UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM 8-K

CURRENT REPORT

Pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934

Date of Report (Date of earliest event reported): October 23, 2024

Nxu, Inc.

(Exact Name of Registrant as Specified in Charter)

| Delaware | 001-41509 | 92-2819012 |
|------------------------------|--------------|---------------------|
| (State or Other Jurisdiction | (Commission | (IRS Employer |
| of Incorporation | File Number) | Identification No.) |

1828 N. Higley Rd. Ste 116, Mesa, AZ 85205

(Address of Principal Executive Offices) (Zip Code)

Registrant's telephone number, including area code: (602) 309-5425

Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions (*see* General Instruction A.2. below):

☑ Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)

□ Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)

□ Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))

□ Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))

Securities registered pursuant to Section 12(b) of the Act:

| Title of each Class | Trading Symbol(s) | Name of each exchange on which registered |
|--|----------------------|--|
| Class A Common Stock, par value \$0.0001 per share | NXU | NASDAQ |

Indicate by checkmark whether the registrant is an emerging growth company as defined in Rule 405 of the Securities Act of 1933 (§230.405 of this chapter) of Rule 12B-2 of the Securities Exchange act of 1934 (§240.12b-2 of this chapter).

Emerging Growth Company \boxtimes

If an emerging growth company, indicate by check mark if the Registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 13(a) of the Exchange Act. \Box

Merger Agreement

On October 23, 2024, Nxu, Inc., a Delaware corporation ("Nxu"), NXU Merger Sub, Inc., a Delaware corporation and a wholly owned subsidiary of Nxu ("Merger Sub I"), NXU Merger Sub, LLC, a Delaware limited liability company ("Merger Sub II") and Verde Bioresins, Inc., a Delaware corporation ("Verde"), entered into an Agreement and Plan of Merger (the "Merger Agreement"), pursuant to which, among other matters, and subject to the satisfaction or waiver of the conditions set forth in the Merger Agreement, Merger Sub I will merge with and into Verde, with Verde continuing as a wholly owned subsidiary of Nxu and the surviving corporation of the first merger (the "First Merger") and promptly following the First Merger, Verde shall merge with and into Merger Sub II (the "Second Merger" and together with the First Merger, the "Merger"), with Merger Sub II continuing as the surviving entity of the Second Merger. The First Merger and the Second Merger are intended to be treated as integrated steps in a single transaction that qualify for federal income tax purposes as a tax-free reorganization under the provisions of Section 368(a) of the Internal Revenue Code of 1986, as amended.

Subject to the terms and conditions of the Merger Agreement, (i) immediately prior to the effective time of the First Merger (the "Effective Time"), all outstanding convertible notes of Verde will be converted into shares of Verde common stock and all outstanding and unexercised Verde warrants will be exercised for shares of Verde common stock, and (ii) at the Effective Time, (a) each then-outstanding share of Verde common stock, other than any cancelled shares and dissenting shares, will be converted into the right to receive a number of shares of Nxu common stock, and (b) each then-outstanding and unexercised Verde option to purchase shares of Verde common stock, whether vested or unvested, will be assumed by Nxu and converted into an option to purchase a number of shares of Nxu common stock. The shares of Nxu common stock that will be issued to Verde stockholders and the number of shares of Nxu common stock underlying options that will be issuable to Verde optionholders will be calculated using a formula in the Merger Agreement based on the enterprise value of each of Verde and Nxu. Verde has been ascribed an aggregate enterprise value of approximately \$306.9 million, and Nxu's aggregate enterprise value will be an amount equal to approximately \$16.2 million less an amount equal to the excess of certain lease payments remaining unpaid at the closing of the Merger Verde stockholders will own approximately 95% of the combined company and pre-Merger Nxu stockholders will own approximately 5% of the combined company, in each case, on a fully-diluted and as-converted basis.

At the Effective Time, (i) each then-outstanding option to acquire Nxu Class A common stock that is out-of-the-money immediately prior to the Effective Time will be cancelled without payment therefor, (ii) each option to acquire Nxu Class A common stock that is in-the-money will remain outstanding and will remain exercisable subject to the terms and conditions of the applicable option award agreement and incentive plan, and (iii) each Nxu restricted stock unit will be settled in accordance with its terms.

In connection with the Merger, following the execution and delivery of the Merger Agreement, Nxu and Verde will prepare and Nxu will file with the U.S. Securities and Exchange Commission (the "SEC") a combined registration statement on Form S-4 (together with all amendments thereto, the "Registration Statement"), registering under the Securities Act of 1933, as amended (the "Securities Act"), the shares of Nxu common stock to be used in connection with the Merger, and a proxy statement with respect to the special meeting of Nxu's stockholders at which Nxu has agreed to seek the approval of Nxu's stockholders with respect to certain actions, including, among other things, (i) the issuance of Nxu common stock pursuant to the terms of the Merger Agreement for purposes of the rules of the Nasdaq Stock Market, LLC ("Nasdaq"), (ii) an amendment and restatement of Nxu's certificate of incorporation to, among other things, (a) eliminate the dual class structure of the Nxu common stock and effectuate a reverse stock split, (b) change the number of authorized shares of Nxu's common stock from 5,000,000,000 shares to such number as subsequently approved by the board of directors of Nxu (the "Nxu Board"), (b) change the number of authorized shares of Nxu's preferred stock from 10,000,000 shares to such number as subsequently approved by the Nxu Board; (c) classify the board of directors of the Nxu Board, (d) require a supermajority stockholder vote in connection with certain actions under the certificate of incorporation, and (h) change the post-combination company's name to "Verde Bioresins, Corp.", and (iii) a new Nxu or its stockholders for monetary damages for breach of fiduciary duties to include Nxu's officers, (g) update provisions governing indemnification and advance provisions in the certificate of incorporation, and (h) change the post-combination company's name to "Verde Bioresins, Corp.", and (iii) a new Nxu equity incentive plan (collectively, the "Nxu Proposals").

Each of Nxu and Verde has agreed to customary representations, warranties and covenants in the Merger Agreement, including, among others, covenants relating to (i) efforts to obtain the requisite approvals of their stockholders, (ii) non-solicitation of alternative acquisition proposals, (iii) the conduct of their respective businesses during the period between the date of signing the Merger Agreement and the closing of the Merger, (iv) Nxu using commercially reasonable efforts to maintain the existing listing of Nxu common stock on Nasdaq and cause the shares of Nxu common stock to be issued in connection with the Merger to be approved for listing on Nasdaq prior to the closing of the Merger, and (v) Nxu filing with the SEC and causing to become effective the Registration Statement to register the shares of Nxu common stock to be issued in connection with the Merger.

Consummation of the Merger, and the other transactions contemplated by the Merger Agreement, is subject to certain closing conditions, including, among other things, (i) approval by the requisite Nxu stockholders of the Nxu Proposals, (ii) approval by the requisite Verde stockholders of the adoption and approval of the Merger Agreement and the transactions contemplated thereby, (iii) Nasdaq's approval of the listing of the shares of Nxu common stock to be issued in connection with the Merger, and (iv) the Registration Statement having become effective in accordance with the Securities Act and not being subject to any stop order or proceeding seeking a stop order. In addition, immediately prior to the Effective Time, (i) except for Jessica Billingsley, each current director of Nxu will resign from the Nxu Board; (ii) each current officer of Nxu will resign from his or her position with Verde. Immediately following the Effective Time, (i) the Nxu Board will consist of seven members, six of whom will be appointed by Verde and one of whom will be appointed by Nxu; and (ii) the officers of Verde (as of immediately prior to the Effective Time) will be appointed to serve as officers of Nxu in identical positions. Each party's obligation to consummate the Merger is also subject to other specified customary conditions, including the accuracy of the representations and warranties of the other party (subject to certain materiality standards) and the performance in all material respects by the other party of its obligations under the Merger Agreement required to be performed on or prior to the date of the closing of the Merger.

The Merger Agreement contains certain termination rights, including, among others, (i) the mutual written consent of the parties, (ii) in the event that there is a breach by a party of any of its representations, warranties, covenants or agreements, which breach is not timely cured or curable, (iii) in the event that the Merger has not been consummated by March 31, 2025, (iv) in the event that the requisite approval of Nxu's stockholders is not obtained upon a vote thereon (triggering Verde's right to terminate), and (v) in the event that any governmental authority shall have taken action to restrain, enjoin or prohibit the consummation of the Merger, which action shall have become final and non-appealable. In addition, upon termination of the Merger Agreement by Verde by written notice to Nxu or by Nxu due to Verde's failure to receive the requisite approval of Verde's stockholders within 10 business days after the effectiveness of the Registration Statement, Verde will be required to pay Nxu a termination fee of \$1.0 million. Further, Nxu may terminate the Merger Agreement prior to receipt of the requisite approval of Nxu's stockholders to enter into a definitive agreement with respect to a Superior Proposal (as such term is defined in the Merger Agreement), provided that Nxu has provided Verde with at least five business days' notice prior to such termination.

As soon as practicable following the Merger, Nxu will be renamed "Verde Bioresins, Corp.", or such other name mutually agreed to by Nxu and Verde and will cause its current ticker symbol "NXU" to be changed to "VRDE" upon Nasdaq's approval of the listing application.

Support Agreements, Lock-Up Agreements and Voting Agreement

Concurrently with the execution of the Merger Agreement, (a) Humanitario Capital LLC, the controlling stockholder of Verde, has entered into a support agreement with Nxu and Verde to vote all of its shares of Verde common stock in favor of the adoption and approval of the Merger Agreement and the transactions contemplated thereby and against any alternative acquisition proposals (the "Verde Support Agreement") and (b) certain directors and officers of Nxu (solely in their respective capacities as Nxu stockholders), holding an aggregate of approximately 19% of the total voting power of Nxu's common stock, have entered into a support agreement with Nxu and Verde to vote all of their shares of Nxu common stock in favor of the Nxu Proposals and against any alternative acquisition proposals (the "Nxu Support Agreement," and, together with the Verde Support Agreements, the "Support Agreements").

Concurrently with the execution of the Merger Agreement, certain executive officers, directors and stockholders of Verde have entered into lock-up agreements (each a "Lock-Up Agreement" and collectively, the "Lock-Up Agreements") pursuant to which, subject to specified exceptions, they have agreed not to transfer any of their shares of Nxu common stock acquired in connection with the Merger for a period of 180 days following the closing of the Merger. Following this initial lock-up period, the transfer of any shares of Nxu common stock representing more than 5% of the aggregate shares in any calendar month is prohibited for a period of 24 months.

Concurrently with the execution of the Merger Agreement, Nxu has entered into a Voting Agreement and Irrevocable Proxy (the "Voting Agreement") with its Chief Executive Officer, Mark Hanchett (the "Holder"), related to which Nxu will issue a share of Series B preferred stock to the Holder, and, in exchange, the Holder has agreed to, subject to specified exceptions, (i) cast all of the votes to which he is entitled by reason of holding the Series B preferred stock "for" or "against" (as applicable) a special action subject to the votes cast by the holders of Class A common stock and Class B common stock of Nxu; and (ii) irrevocably grant each director of Nxu as the Holder's proxy and attorney-in-fact to vote the Series B preferred stock on behalf of the Holder if he fails at any time to vote as required in the Voting Agreement.

The preceding summaries of the Merger Agreement, the Support Agreements, the Lock-Up Agreements and the Voting Agreement and the transactions contemplated thereby do not purport to be complete and are qualified in their entirety by reference to the Merger Agreement, the form of the Nxu Support Agreement, the form of the Verde Support Agreement, the form of the Lock-Up Agreement and the Voting Agreement, which are filed as Exhibits 2.1, 10.1, 10.2, 10.3 and 10.4, respectively, to this Current Report on Form 8-K (this "Report") and which are incorporated herein by reference. The Merger Agreement has been attached as an exhibit to this Report to provide investors and securityholders with information regarding its terms. It is not intended to provide any other factual information about Nxu or Verde or to modify or supplement any factual disclosures about Nxu in its public reports filed with the SEC. The Merger Agreement includes representations, warranties and covenants of Verde, Nxu and Merger Subs made solely for the purpose of the Merger Agreement and solely for the benefit of the parties thereto in connection with the negotiated terms of the Merger Agreement. Investors should not rely on the representations, warranties and covenants in the Merger Agreement or any descriptions thereof as characterizations of the actual state of facts or conditions of Nxu, Verde or any of their respective affiliates. Moreover, certain of those representations and warranties may not be accurate or complete as of any specified date, may be subject to a contractual standard of materiality different from those generally applicable to SEC filings or may have been used for purposes of allocating risk among the parties to the Merger Agreement, rather than establishing matters of fact.

Item 5.02 Departure of Directors or Certain Officers; Election of Directors; Appointment of Certain Officers; Compensatory Arrangements of Certain Officers.

Amended Executive Employment Agreements and Wyant Employment Agreement

Concurrently with the signing of the Merger Agreement, on October 23, 2024, Nxu amended the employment agreements (collectively, the "Employment Agreement Amendments") with each of its Chief Executive Officer, Mark Hanchett, and its President, Annie Pratt (each an "Executive" and collectively, the "Executives"). The Nxu Board also appointed Sarah Wyant (together with the Executives, the "Nxu Executives"), Nxu's Interim Chief Financial Officer, to serve as Nxu's Chief Financial Officer, effective on October 23, 2024. Biographical and other information regarding Ms. Wyant required by Items 401(b) and 401(e) of Regulation S-K is contained in Nxu's proxy statement on Schedule 14A for its 2024 annual meeting of stockholders, filed with the SEC on May 2, 2024, and such information is incorporated by reference into this Report. There are no arrangements or understandings between Ms. Wyant and any other persons pursuant to which she was selected Nxu's Chief Financial Officer. There are no family relationships between Ms. Wyant and any director or executive officer of Nxu, and there are no transactions between Ms. Wyant and Nxu that would be required to be reported under Item 404(a) of Regulation S-K. In connection with Ms. Wyant's appointment, on October 23, 2024, Nxu entered into an employment agreement with Ms. Wyant (the "Wyant Employment Agreement").

Each Employment Agreement Amendment, among other things, (i) provides that the Executive may not hold other employment unless such employment does not interfere or conflict with the Executive's duty under the Executive Agreement, (ii) amends and restates the provisions governing benefits upon an involuntary termination without Cause by Nxu or the Executive's resignation for Good Reason (as such terms are defined in the Employment Agreement Amendment) such that if the Executive is involuntarily terminated by Nxu without Cause or the Executive resigns for Good Reason in each case solely following a termination of the Merger Agreement, provided that the Executive executes a general release of claims, the Executive is entitled to amounts accrued prior to the Executive's termination, a lump sum payment equal to six (6) months of base salary, less applicable taxes and withholding, the settlement of outstanding equity awards in accordance with their terms, and a monthly cash payment equal to the monthly premium costs for COBRA coverage for up to 12 months, (iii) amends the definition of Good Reason to clarify that none of (a) the execution of the Employment Agreement Amendment, (b) changes to Executive's duties, authority or responsibilities or Executive's principal place of employment, in each case, while the Merger Agreement is in effect, nor (c) any of the transactions contemplated by the Merger Agreement, including the Executive's resignation from his or her position as an officer of Nxu or member of the Nxu Board will constitute Good Reason for purposes of the Employment Agreement, (iv) provides for the payment to the Executive of a financing bonus or bonuses not to exceed an aggregate amount equal to six months of the Executive's annual base salary in the event that after the closing of the Merger Nxu raises at least \$5 million during 2025 through the sale of Nxu equity securities to one or more third parties unaffiliated with either Nxu or Verde, provided that the Executive is employed by the Company on the closing of the Merger and provided Executive executes a general release of claims in connection with any termination of employment (the "Financing Bonus"), and (v) amends the outstanding equity awards as described below.

The Wyant Employment Agreement increases Ms. Wyant's annual base salary to \$300,000 from \$275,000, provides that Ms. Wyant is eligible for the Financing Bonus and amends her outstanding equity awards as described below. Ms. Wyant will also be eligible to participate in Nxu's standard employee benefit programs made available by Nxu to its employees generally, and will be eligible to receive paid vacation in accordance with Nxu's vacation policies. If Nxu terminates Ms. Wyant (other than as a result of her death, disability or for Cause) or Ms. Wyant resigns for Good Reason (as such terms are defined in the Wyant Employment Agreement), in either case, following a termination of the Merger Agreement, provided that she executes a general release of claims, Ms. Wyant will be entitled to receive amounts accrued prior to her termination, a lump sum payment of \$150,000, less applicable taxes and withholding, and continued vesting of outstanding equity awards.

Amendments to Equity Awards under the Employment Agreement Amendments and the Wyant Employment Agreement

The Nxu Board and the Nxu Executives previously agreed to extend the vesting dates for all restricted stock units ("RSUs") previously granted to the Executives until January 31, 2025 (the "Delayed Vesting"), as the delivery of the RSUs during the original vesting periods would jeopardize Nxu's ability to continue as a going concern. Pursuant to the Employment Agreement Amendments and the Wyant Employment Agreement, following the signing of the Merger Agreement, Nxu will deliver such RSUs to the Nxu Executives in installments in such amounts as Nxu determines may be delivered to the Nxu Executives without jeopardizing Nxu's ability to continue as a going concern. Such RSUs will be delivered until the earlier of (a) the date all RSUs subject to Delayed Vesting have been delivered to the Nxu Executive or (b) the date that is no later than five business days prior to the closing of the Merger (the "Cut-Off Date"). Any RSUs that have not been delivered to the Nxu Executives as of the Cut-Off Date will be forfeited for no consideration. No RSUs will be settled if such settlement would cause the Nxu Executive to be subject to excise taxes imposed by Section 4999 of the Code, unless at the Company and Nxu Executive's request, Verde may consider permitting such RSUs to be settled upon receipt of satisfactory evidence that the Nxu Executive has remitted to an escrow account the amount necessary to satisfy the Company's tax withholding obligations under Section 4999 upon the closing of the transaction contemplated by the Merger Agreement.

The Nxu Executives hold stock options under the Amended 2023 Omnibus Plan and/or the Stock Option Plan of Atlis Motor Vehicles, Inc. All of the stock options have a per share exercise price in excess of the current fair market value of the Company's stock as of the effective date of the Employment Agreement Amendments or the effective date of the Wyant Employment Agreement, as applicable. The Nxu Executives have agreed to forfeit all outstanding stock options, whether vested or unvested, for no consideration immediately prior to a Change in Control, provided the per share exercise price, as adjusted to reflect any changes in the Company's capitalization between the effective date of the Employment Agreement Amendments or the Wyant Employment Agreement and the date on which a Change in Control occurs, remains in excess of the Company's closing stock price on the trading day immediately preceding the date on which the Change in Control occur.

The preceding summaries of the Employment Agreement Amendments and the Wyant Employment Agreement do not purport to be complete and are qualified in their entirety by reference to the form of the Employment Agreement Amendment and the Wyant Employment Agreement, which are filed as Exhibits 10.5 and 10.6, respectively, to this Report and which are incorporated herein by reference.

Item 7.01 Regulation FD Disclosure.

On October 23, 2024, Nxu and Verde issued a joint press release announcing execution of the Merger Agreement. A copy of that press release is furnished as Exhibit 99.1 to this Report and is incorporated herein by reference.

The information furnished in this Item 7.01, including Exhibit 99.1 attached hereto, shall not be deemed "filed" for purposes of Section 18 of the Securities Exchange Act of 1934, as amended (the "Exchange Act"), or otherwise subject to the liabilities of that section, nor shall it be deemed incorporated by reference into any filing or other document pursuant to the Securities Act or the Exchange Act, regardless of any general incorporation language in such filing or document, except as shall be expressly set forth by specific reference in such a filing or document.

Item 8.01 Other Events.

Concurrently with the signing of the Merger Agreement, on October 23, 2024, Nxu entered into a Board of Directors Agreement (collectively, the "Board of Directors Agreements") with each of Jessica Billingsley and Britt Ide (each a "Non-Employee Director"), pursuant to which each Non-Employee Director will receive a \$15,000 per quarter cash stipend and each Non-Employee Director will be granted 591,715 RSUs under the Nxu, Inc. Amended and Restated 2023 Omnibus Incentive Plan. Such RSUs will vest as follows: (1) 147,928 RSUs are consideration for services rendered for the quarter ending September 30, 2024 and shall be immediately vested, (2) the remaining RSUs shall vest ratably at the end of each calendar quarter from December 31, 2024 until June 30, 2025, provided that the Non-Employee Director continues to serve on the Nxu Board on the applicable vesting date. Vesting of the RSUs granted to the Non-Employee Directors and quarterly Non-Employee Director stipend payments shall cease upon the closing of the Merger. The Board of Directors Agreements also amend outstanding equity awards as described below.

On October 23, 2024, Nxu also amended the restricted stock unit award agreement (the "Nightengale Award Amendment") with Caryn Nightengale, a former member of Nxu's Board, to amend her outstanding equity awards as described below.

Pursuant to the Board of Directors Agreements and the Nightengale Award Amendment, following the signing of the Merger Agreement, Nxu will deliver RSUs to the Non-Employee Directors and Ms. Nightengale that were granted as consideration for their prior service as members of the Nxu Board (the "Director Delayed RSUs"). Such Director Delayed RSUs shall be delivered in installments in such amounts as Nxu determines may be delivered to the Non-Employee Directors and Ms. Nightengale without jeopardizing Nxu's ability to continue as a going concern. Such Director Delayed RSUs will be delivered until the earlier of (a) the date all Director Delayed RSUs have been delivered to the Non-Employee Director and Ms. Nightengale or (b) the Cut-Off Date. Any Director Delayed RSUs that have not been delivered to the Non-Employee Directors and Ms. Nightengale as of the Cut-Off Date will be forfeited for no consideration.

The preceding summaries of the Board of Directors Agreements and the Nightengale Award Amendment do not purport to be complete and are qualified in their entirety by reference to the form of the Board of Directors Agreement and the Nightengale Award Amendment, which are filed as Exhibits 10.7 and 10.8, respectively, to this Report and which are incorporated herein by reference.

Cautionary Note Regarding Forward-Looking Statements

Certain statements contained in this Report and in the exhibits filed or furnished herewith may constitute "forward-looking statements" within the meaning of the Private Securities Litigation Reform Act of 1995. Forward-looking statements can typically be identified by such words as "aim", "anticipate," "believe," "continue," "could," "estimate," "evolve," "expect," "forecast," "intend," "looking ahead," "may," "opinion," "plan," "possible," "potential," "project," "should," "will," and variations of such words and other similar expressions.

These forward-looking statements are only predictions based on current expectations and assumptions and are inherently subject to risks and uncertainties, some of which cannot be predicted or quantified. There are important factors that could cause actual results to differ materially from the results expressed or implied by forward-looking statements, including (i) the risk factors set forth under "Risk Factors" in Part I, Item 1A of Nxu's Annual Report on Form 10-K for the fiscal year ended December 31, 2023 ("Nxu's Form 10-K"), set forth in Nxu's subsequent Quarterly Reports on Form 10-Q or set forth in Nxu's other filings with the SEC and (ii) any of the following:

- the expected timing to consummate the Merger and/or the transactions contemplated by the Merger Agreement;
- the satisfaction (or waiver) of closing conditions to the consummation of the Merger and/or the transactions contemplated by the Merger Agreement, including obtaining stockholder approvals and any required regulatory clearances;

- potential litigation relating to the Merger and/or the transactions contemplated by the Merger Agreement that could be instituted against Nxu, Verde or their respective directors;
- · risks related to Nxu's continued listing on Nasdaq until closing of the Merger;
- risks related to (i) the combined company's satisfaction of the Nasdaq initial listing application qualifications for the listing on the Nasdaq Capital Market and (ii) Nasdaq's approval of the listing of the shares of Nxu common stock to be issued in connection with the Merger;
- risks associated with the possible failure to realize certain anticipated benefits of the Merger, including with respect to future financial and operating results;
- uncertainties regarding unexpected costs, liabilities or delays in connection with the consummation of the Merger and/or the transactions contemplated by the Merger Agreement, and the impact such costs, liabilities or delays would have on the anticipated cash resources of the combined company upon closing;
- operating costs, customer loss and business disruptions arising from the Merger and/or the transactions contemplated by the Merger Agreement and the pendency or consummation thereof (including, without limitation, difficulties in maintaining relationships with employees, customers, clients or suppliers), which may be greater than expected;
- the ability of Nxu and Verde to protect their respective intellectual property rights;
- · competitive responses to the proposed Merger and changes in expected or existing competition;
- Nxu's, Verde's, and the combined company's limited operating history;
- the combined company's ability to manage growth;
- · changes in capital requirements;
- the combined company's ability to obtain additional financing;
- other adverse economic, business, and/or competitive factors; and
- other risks to the consummation of the Merger and the other transactions contemplated by the Merger Agreement, including circumstances that could give rise to the termination of the Merger Agreement and the risk that the transactions contemplated thereby will not be consummated within the expected time period, without undue delay, cost or expense, or at all.

All forward-looking statements are qualified by, and should be considered in conjunction with, these cautionary statements. Readers are cautioned not to place undue reliance on any forward-looking statements, which speak only as of the date on which such statements are made. Except as required by applicable law, neither Nxu nor Verde undertakes any obligation to update forward-looking statements to reflect events or circumstances arising after such date.

Important Additional Information About the Proposed Transaction and Where to Find It

This Report and the exhibits filed or furnished herewith are not substitutes for the Registration Statement or for any other document that Nxu may file with the SEC in connection with the proposed transaction. In connection with the proposed transaction, Nxu intends to file relevant materials with the SEC, including the Registration Statement that will contain a proxy statement of Nxu and that will constitute a prospectus with respect to shares of Nxu's common stock to be issued in connection with the proposed transaction (the "Proxy Statement/Prospectus"). NXU URGES ITS INVESTORS AND STOCKHOLDERS TO READ THE REGISTRATION STATEMENT, PROXY STATEMENT/PROSPECTUS AND ANY OTHER RELEVANT DOCUMENTS THAT MAY BE FILED WITH THE SEC, AS WELL AS ANY AMENDMENTS OR SUPPLEMENTS TO THESE DOCUMENTS, CAREFULLY AND IN THEIR ENTIRETY IF AND WHEN THEY BECOME AVAILABLE BECAUSE THEY WILL CONTAIN IMPORTANT INFORMATION ABOUT NXU, VERDE, THE PROPOSED TRANSACTION AND RELATED MATTERS. Investors and stockholders will be able to obtain free copies of the Registration Statement, Proxy Statement/Prospectus and other documents filed by Nxu with the SEC (when they become available) through the website maintained by the SEC at www.sec.gov. In addition, investors and stockholders should note that Nxu communicates with investors and the public using its website (www.nxuenergy.com) and the investor relations website (investors.nxuenergy.com), where anyone will be able to obtain free copies of the Registration Statement, Proxy Statement/Prospectus and other documents filed by Nxu with the SEC and investors and stockholders should note that Registration Statement, Proxy Statement/Prospectus and other relevant materials when they become available before making any voting or investment decision with respect to the proposed transaction.

No Offer or Solicitation

This Report and the exhibits filed or furnished herewith are not intended to and do not constitute (i) a solicitation of a proxy, consent or approval with respect to any securities or in respect of the proposed transaction or (ii) an offer to sell or the solicitation of an offer to subscribe for or buy or an invitation to purchase or subscribe for any securities pursuant to the proposed transaction or otherwise, nor shall there be any sale, issuance or transfer of securities in any jurisdiction in contravention of applicable law. No offer of securities shall be made except by means of a prospectus meeting the requirements of the Securities Act or an exemption therefrom. Subject to certain exceptions to be approved by the relevant regulators or certain facts to be ascertained, the public offer will not be made directly or indirectly, in or into any jurisdiction where to do so would constitute a violation of the laws of such jurisdiction, or by use of the mails or by any means or instrumentality (including without limitation, facsimile transmission, telephone and the internet) of interstate or foreign commerce, or any facility of a national securities exchange, of any such jurisdiction.

NEITHER THE SEC NOR ANY STATE SECURITIES COMMISSION HAS APPROVED OR DISAPPROVED OF THE SECURITIES OR DETERMINED IF THIS REPORT AND THE EXHIBITS FILED OR FURNISHED HEREWITH ARE TRUTHFUL OR COMPLETE.

Participants in the Solicitation

Nxu, Verde and their respective directors, executive officers and other members of management may be considered participants in the solicitation of proxies in connection with the proposed transaction. Information about Nxu's directors and executive officers, consisting of Mark Hanchett, Annie Pratt, Britt Ide, Jessica Billingsley and Sarah Wyant, including a description of their direct or indirect interest, by security holdings or otherwise, can be found under the captions, "Security Ownership of Certain Beneficial Owners and Management," "Executive Compensation," and "Director Compensation" contained in the definitive proxy statement on Schedule 14A for Nxu's 2024 annual meeting of stockholders, filed with the SEC on May 2, 2024 (the "2024 Nxu Proxy Statement"). To the extent that Nxu's directors and executive officers and their respective affiliates have acquired or disposed of security holdings since the applicable "as of" date disclosed in the 2024 Nxu Proxy Statement, such transactions have been or will be reflected on Statements of Change in Beneficial Ownership on Form 4 filed with the SEC. Additional information regarding the persons who may be deemed participants in the solicitation of proxies, including the information about the directors and executive officers of Verde, will be included in the Proxy Statement/Prospectus and other relevant materials relating to the proposed transaction when it is filed with the SEC. Investors should read the Registration Statement, Proxy Statement/Prospectus and the other relevant materials when they become available before making any voting or investment decision with respect to the proposed transaction. These documents can be obtained free of charge from the sources indicated above.

Item 9.01 Financial Statements and Exhibits.

(d) Exhibits.

| Exhibit No. | Description |
|----------------|---|
| 2.1* | Agreement and Plan of Merger, dated as of October 23, 2024, by and among Nxu, Inc., Nxu Merger Sub, Inc., Nxu Merger Sub, LLC and Verde Bioresins, Inc. |
| 10.1 | Form of Nxu Support Agreement |
| 10.2 | Form of Verde Support Agreement |
| 10.3 | Form of Lock-Up Agreement |
| 10.4 | Voting Agreement and Irrevocable Proxy, dated as of October 23, 2024, by and between Nxu, Inc. and Mark Hanchett |
| 10.5 | Form of Amendment to Employment Agreement, by and between Nxu, Inc. and each of Mark Hanchett and Annie Pratt |
| 10.6 | Executive Employment Agreement for Sarah Wyant, dated as of October 23, 2024, by and between Nxu, Inc. and Sarah Wyant |
| 10.7 | Form of Board of Directors Agreement, by and between Nxu, Inc. and each of Jessica Billingsley and Britt Ide |
| 10.8 | Amendment to Restricted Stock Unit Award Agreement, dated as of October 23, 2024, by and between Nxu, Inc. and Caryn Nightengale |
| 99.1 | Joint Press Release of Nxu and Verde, issued on October 23, 2024 |
| 104 | Cover Page Interactive Data File (embedded within the Inline XBRL document) |

* Exhibits and schedules to this Exhibit have been omitted pursuant to Item 601(a)(5) of Regulation S-K. The Registrant agrees to furnish supplementally a copy of any omitted schedule or exhibit to the SEC upon request; provided, however, that the Registrant may request confidential treatment pursuant to Rule 24b-2 under the Exchange Act for any exhibits or schedules so furnished.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, as amended, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

Date: October 23, 2024

NXU, INC.

By: /s/ Mark Hanchett Mark Hanchett

Chief Executive Officer

AGREEMENT AND PLAN OF MERGER

dated as of

October 23, 2024

by and among

NXU, INC.,

NXU MERGER SUB, INC.,

NXU MERGER SUB, LLC

and

VERDE BIORESINS, INC.

| | I Certain Definitions | 3 |
|---------|--|----|
| 1.01 | Definitions | 3 |
| 1.02 | Construction | 19 |
| Article | II The Merger; Closing | 20 |
| 2.01 | The Merger | 20 |
| 2.02 | Effects of the Merger | 20 |
| 2.03 | Closing | 20 |
| 2.04 | Organizational Documents of the Company | 21 |
| 2.05 | Directors and Officers of the Company and Acquiror | 21 |
| Article | III Effects of the Merger | 22 |
| 3.01 | Effect on Capital Stock | 22 |
| 3.02 | Effect on Company Equity Awards | 23 |
| 3.03 | Acquiror Equity Awards | 23 |
| 3.04 | Equitable Adjustments | 24 |
| 3.05 | Exchange of Company Certificates and Company Book-Entry Shares | 24 |
| 3.06 | Withholding | 25 |
| 3.07 | No Fractional Shares | 25 |
| 3.08 | Dissenting Shares | 26 |
| 3.09 | Company Merger Consideration Schedule | 26 |
| 3.10 | No Liability | 26 |
| Article | IV Representations and Warranties of the Company | 27 |
| 4.01 | Organization, Standing and Corporate Power | 27 |
| 4.02 | Corporate Authority; Approval; Non-Contravention | 27 |
| 4.03 | Governmental Approvals | 28 |
| 4.04 | Capitalization | 28 |
| 4.05 | Subsidiaries | 29 |
| 4.06 | Financial Statements; Internal Controls | 29 |
| 4.07 | Compliance with Laws | 30 |
| 4.08 | Absence of Certain Changes or Events | 31 |
| 4.09 | No Undisclosed Liabilities | 31 |
| 4.10 | Information Supplied | 31 |
| 4.11 | Litigation | 31 |

Page

| 4.12 | Contracts | 32 |
|-----------|--|----|
| 4.13 | Employment Matters | 34 |
| 4.14 | Employee Benefits | 35 |
| 4.15 | Taxes | 37 |
| 4.16 | Intellectual Property | 39 |
| 4.17 | Data Protection | 40 |
| 4.18 | Information Technology and Cybersecurity | 41 |
| 4.19 | Real Property | 42 |
| 4.20 | Corrupt Practices; Sanctions | 43 |
| 4.21 | Competition and Trade Regulation | 44 |
| 4.22 | Environmental Matters | 44 |
| 4.23 | Brokers | 45 |
| 4.24 | Affiliate Agreements | 45 |
| 4.25 | Insurance | 45 |
| 4.26 | COVID-19 | 46 |
| 4.27 | Takeover Statutes and Charter Provisions | 46 |
| 4.28 | Board Approval; Stockholders' Approval; Vote Required | 46 |
| 4.29 | No Other Representations or Warranties | 46 |
| Article V | / Representations and Warranties of Acquiror AND Merger Subs | 47 |
| 5.01 | Organization, Standing and Corporate Power | 47 |
| 5.02 | Corporate Authority; Approval; Non-Contravention | 47 |
| 5.03 | Governmental Approvals | 48 |
| 5.04 | Capitalization | 49 |
| 5.05 | Subsidiaries | 50 |
| 5.06 | Compliance with Laws | 50 |
| 5.07 | Absence of Certain Changes or Events | 51 |
| 5.08 | No Undisclosed Liabilities | 51 |
| 5.09 | Litigation | 51 |
| 5.10 | Contracts | 52 |
| 5.11 | Employment Matters | 54 |
| 5.12 | Employee Benefits | 55 |
| 5.13 | Taxes | 57 |
| 5.14 | Brokers | 59 |
| 5.15 | Acquiror SEC Reports; Financial Statements; Sarbanes-Oxley Act | 59 |

| 5.16 | Registration Statement | 60 |
|-----------|---|----|
| 5.17 | NASDAQ Stock Market Quotation | 61 |
| 5.18 | Intellectual Property. | 61 |
| 5.19 | Data Protection | 62 |
| 5.20 | Information Technology and Cybersecurity | 63 |
| 5.21 | Real Property | 64 |
| 5.22 | Corrupt Practices; Sanctions | 66 |
| 5.23 | Competition and Trade Regulation | 67 |
| 5.24 | Environmental Matters | 67 |
| 5.25 | Affiliate Transactions | 68 |
| 5.26 | Insurance | 68 |
| 5.27 | COVID-19 | 68 |
| 5.28 | Takeover Statutes and Charter Provisions | 68 |
| 5.29 | No Other Representations or Warranties | 69 |
| Article V | /I Covenants of the Company | 69 |
| 6.01 | Conduct of Business | 69 |
| 6.02 | Inspection | 72 |
| 6.03 | Proxy Solicitation; Other Actions | 73 |
| 6.04 | Non-Solicitation; Acquisition Proposals | 73 |
| 6.05 | Company Lock-up | 75 |
| 6.06 | Litigation | 75 |
| 6.07 | Conversion of Company Notes; Exercise of Company Warrants | 75 |
| 6.08 | Company Support Agreement | 75 |
| 6.09 | Company Material Contract Consents | 75 |
| 6.10 | Voting Trust Transfer | 76 |
| 6.11 | Interim Financial Statements | 76 |
| Article V | /II Covenants of Acquiror | 76 |
| 7.01 | Indemnification and Insurance | 76 |
| 7.02 | Conduct of Acquiror During the Interim Period | 77 |
| 7.03 | Inspection | 79 |
| 7.04 | Acquiror NASDAQ Listing | 80 |
| 7.05 | Acquiror SEC Filings | 80 |
| 7.06 | Section 16 Matters | 80 |
| 7.07 | Director and Officer Appointments | 80 |

| 7.08 | Exclusivity | 81 |
|-----------|---|----|
| 7.09 | Acquiror Plan | 82 |
| 7.10 | Stockholder Litigation | 82 |
| 7.11 | Acquiror Certificate of Incorporation Amendment | 83 |
| 7.12 | COBRA Coverage | 83 |
| 7.13 | Acquiror Material Contract Consents | 83 |
| Article Y | VIII Joint Covenants | 83 |
| 8.01 | Support of Transaction | 83 |
| 8.02 | Preparation of Registration Statement; Acquiror Stockholders Meeting; | |
| Solicitat | tion of Company Stockholder Approvals | 84 |
| 8.03 | Tax Matters | 87 |
| 8.04 | Confidentiality; Publicity | 88 |
| 8.05 | Financing Cooperation | 88 |
| 8.06 | Listing Application | 88 |
| Article l | IX Conditions to Obligations | 89 |
| 9.01 | Conditions to Obligations of All Parties | 89 |
| 9.02 | Additional Conditions to Obligations of Acquiror | 90 |
| 9.03 | Additional Conditions to the Obligations of the Company | 91 |
| Article 2 | X Termination/Effectiveness | 92 |
| 10.01 | Termination | 92 |
| 10.02 | Effect of Termination | 93 |
| | XI Miscellaneous | 94 |
| 11.01 | Waiver | 94 |
| 11.02 | Notices | 94 |
| 11.03 | Assignment | 95 |
| 11.04 | Rights of Third Parties | 95 |
| 11.05 | Expenses | 95 |
| 11.06 | Governing Law | 95 |
| 11.07 | Captions; Counterparts | 95 |
| 11.08 | Schedules and Exhibits | 96 |
| 11.09 | Entire Agreement | 96 |
| 11.10 | Amendments | 96 |
| 11.11 | Severability | 96 |
| 11.12 | Jurisdiction; WAIVER OF TRIAL BY JURY | 97 |
| 11.13 | Enforcement | 97 |
| | | |

11.14 Non-Recourse

11.15 Non-survival of Representations, Warranties and Covenants

Exhibits

Exhibit A – Form of Company Support Agreement

Exhibit B – Form of Acquiror Support Agreement Exhibit C – Form of Registration Rights Agreement Exhibit D - Form of Company Lock-Up Agreement

Exhibit E - Form of Amended and Restated Certificate of Incorporation

Exhibit F - Form of FIRPTA Certificate

Exhibit G - Post-Closing Directors and Officers

AGREEMENT AND PLAN OF MERGER

This Agreement and Plan of Merger (this "<u>Agreement</u>"), dated as of $[\bullet]$, 2024, is entered into by and among NXU, Inc. ("<u>Acquiror</u>"), Nxu Merger Sub, Inc., a Delaware corporation ("<u>Merger Sub I</u>"), Nxu Merger Sub, LLC, a Delaware limited liability company ("<u>Merger Sub II</u>", and together with Merger Sub I, the "<u>Merger Subs</u>") and Verde Bioresins, Inc., a Delaware corporation (the "<u>Company</u>"). Except as otherwise indicated, capitalized terms used but not defined herein shall have the meanings set forth in <u>Article I</u>.

RECITALS

WHEREAS, in accordance with the terms and subject to the conditions set forth herein, at the First Merger Effective Time, Merger Sub I will merge with and into the Company (the "First Merger"), with the Company surviving the First Merger as a wholly-owned Subsidiary of Acquiror (the Company, in its capacity as the surviving corporation of the Merger, is referred to as the "Surviving Corporation");

WHEREAS, promptly following the First Merger, the Surviving Corporation shall merge (the "Second Merger" and together with the First Merger, the "Merger") with and into Merger Sub II (which Acquiror shall form prior to the Closing for the sole purpose of the Second Merger) pursuant to the terms of an agreement and plan of merger, which shall be in form and substance reasonably satisfactory to each of the Company and Acquiror, whereupon the separate corporate existence of the Surviving Corporation shall cease, and Merger Sub II shall be the surviving entity of such merger (the "Final Surviving Company");

WHEREAS, each of the Merger Subs is a newly-formed, wholly-owned, direct Subsidiary of Acquiror that was formed for the sole purpose of the Merger;

WHEREAS, the Company Board has (a) determined that this Agreement and the Transactions, including the Merger, and the Ancillary Agreements to which the Company is a party are advisable, fair to, and in the best interests of the Company and its stockholders, (b) approved and declared advisable this Agreement and the Transactions, including the Merger and the Ancillary Agreements to which the Company is a party, and (c) recommended the Company Stockholders approve and adopt this Agreement and approve the Merger and the other Transactions by the requisite affirmative vote in accordance with the DGCL and the Company Organizational Documents;

WHEREAS, each of the Acquiror Board and the board of directors or board of members or managers, as applicable, of each of the Merger Subs has (a) determined that this Agreement and the Transactions, including the Merger, and the Ancillary Agreements to which it is a party are advisable, fair to and in the best interests of each of Acquiror, the Merger Subs and their respective stockholders or members, (b) approved and declared advisable this Agreement and the Transactions, including the Merger, and the Ancillary Agreements to which it is a party, as applicable, (c) directed that the adoption of this Agreement be submitted to a vote of the Acquiror Stockholders, the sole stockholder of Merger Sub I and the sole member of Merger Sub II, as applicable, and (d) recommended that the Acquiror Stockholders, the sole stockholder of Merger Sub I and the sole member of Merger Sub II, as applicable, vote in favor of the adoption of this Agreement; WHEREAS, immediately following the execution and delivery of this agreement, Acquiror, as the sole stockholder of Merger Sub I and the sole member of Merger Sub II, shall adopt this Agreement;

WHEREAS, in connection with the Transactions, Humanitario Capital LLC ("<u>Humanitario</u>"), the controlling stockholder of the Company, will enter into a Support Agreement (the "<u>Company Support Agreement</u>") with Acquiror and the Company, in the form set forth on <u>Exhibit A</u>, pursuant to which, among other things, Humanitario will agree to execute and deliver a consent constituting the Company Stockholder Approvals;

WHEREAS, contemporaneously with the execution and delivery of this Agreement, in connection with the Transactions, certain stockholders of Acquiror, Acquiror and the Company have entered into a Support Agreement, dated as of the date hereof (the "Acquiror Support Agreement" and together with the Company Support Agreements, the "Support Agreements"), in the form set forth on Exhibit B, pursuant to which, among other things, such stockholders agreed to vote in favor of the adoption of this Agreement and the Transactions;

WHEREAS, in connection with the Transactions, within ten (10) Business Days after the Registration Statement is declared effective under the Securities Act, Humanitario shall enter into a voting trust agreement (the "<u>Voting Trust Agreement</u>") with an independent trustee (the "<u>Voting Trustee</u>"), pursuant to which Humanitario will transfer and assign to the Voting Trustee all of the Closing Merger Consideration received by Humanitario pursuant to the terms of this Agreement, in each case, upon the terms and subject to the conditions set forth in the Voting Trust Agreement and the Support Agreement (the "<u>Voting Trust Transfer</u>");

WHEREAS, contemporaneously with the Closing, in connection with the Transactions, Acquiror, the Company, and certain Company Stockholders who will receive Acquiror Common Stock pursuant to <u>Article III</u>, will enter into that certain Registration Rights Agreement (the "<u>Registration Rights Agreement</u>"), substantially in the form set forth on <u>Exhibit C</u>;

WHEREAS, prior to the Closing, Acquiror shall, subject to obtaining the Required Acquiror Stockholder Approvals, adopt an equity incentive plan of the Acquiror (the "Acquiror Equity Incentive Plan"), pursuant to Section 7.09 and in a form prepared by the Company in consultation with the Acquiror;

WHEREAS, Acquiror shall be renamed "Verde Bioresins, Corp.", or such other name mutually agreed to by the Company and Acquiror, and shall trade publicly on the NASDAQ under the new ticker symbol "VRDE"; and

WHEREAS, the parties intend that, for United States federal income tax purposes, (a) the First Merger and the Second Merger be treated as integrated steps in a single transaction that qualifies as a "reorganization" within the meaning of Section 368(a) of the Code and the Treasury Regulations to which each of Acquiror, the Merger Subs and the Company are parties under Section 368(b) of the Code and the Treasury Regulations, and (b) this Agreement is intended to be, and is adopted as, a plan of reorganization for purposes of Sections 354, 361 and 368 of the Code and within the meaning of Treasury Regulations Section 1.368-2(g).

NOW, THEREFORE, in consideration of the foregoing and the respective representations, warranties, covenants and agreements set forth in this Agreement, and intending to be legally bound hereby, Acquiror, the Merger Subs and the Company agree as follows:

ARTICLE I CERTAIN DEFINITIONS

1.01 <u>Definitions</u>. As used herein, the following terms shall have the following meanings:

"Acquiror" has the meaning specified in the preamble.

"<u>Acquiror Benefit Plan</u>" means each Benefit Plan, sponsored, maintained or contributed to (or obligated to be sponsored, maintained or contributed to) by Acquiror or Nxu Tech (a) in which any current or former employee or other service provider to Acquiror or Nxu Tech participates or has any rights or benefits, whether absolute or contingent, or (b) with respect to which Acquiror or Nxu Tech has or could have any Obligation, except to the extent such plan is sponsored or maintained by a Governmental Authority as required by applicable Law.

"Acquiror Board" means the board of directors of Acquiror.

"Acquiror Board Recommendation" has the meaning specified in Section 8.02(d).

"Acquiror Bylaws" means the Bylaws of Acquiror, adopted March 27, 2023.

"<u>Acquiror Certificate of Incorporation</u>" means the Certificate of Incorporation of Acquiror, dated as of March 10, 2023, as amended by the Certificate of Amendment thereto, dated as of December 26, 2023.

"Acquiror Change in Recommendation" has the meaning specified in Section 8.02(d).

"Acquiror Class A Common Stock," means the shares of Class A Common Stock, par value \$0.0001 per share, of Acquiror.

"Acquiror Class B Common Stock" means the shares of Class B Common Stock, par value \$0.0001 per share, of Acquiror.

"Acquiror Common Stock" means, collectively, the Acquiror Class A Common Stock and Acquiror Class B Common Stock.

"Acquiror Cure Period" has the meaning specified in Section 10.01(c).

"Acquiror Enterprise Value" means an amount equal to \$16,154,600 minus the Lease Shortfall Amount.

"Acquiror Environmental Permits" has the meaning specified in Section 5.24(a).

"Acquiror Equity Incentive Plan" has the meaning specified in the recitals.

"Acquiror Equity Plan Proposal" has the meaning specified in Section 8.02(c).

"Acquiror Insurance Policies" has the meaning specified in Section 5.26.

"Acquiror Intellectual Property" means all Owned Acquiror Intellectual Property and all other Intellectual Property used in, held for use or necessary to the business of Acquiror.

"Acquiror Lease Documents" has the meaning specified in Section 5.21(c).

"Acquiror Material Contracts" has the meaning specified in Section 5.10(a).

"<u>Acquiror Option & RSU Plans</u>" means the NXU, Inc. 2023 Omnibus Incentive Plan, as amended from time to time, the Atlis Motor Vehicles, Inc. Employee Stock Option Plan, effective as of April 27, 2022 and amended from time to time (together with any predecessor plan) and the Atlis Motor Vehicles, Inc. Non-Employee Stock Option Plan, effective as of April 27, 2022 and amended from time to time (together with any predecessor plan).

"Acquiror Options" means options to acquire Acquiror Class A Common Stock granted under the Acquiror Option & RSU Plans.

"Acquiror Organizational Documents" means the Acquiror Certificate of Incorporation and the Acquiror Bylaws.

"<u>Acquiror Outstanding Shares</u>" means total number of shares of Acquiror Common Stock outstanding immediately prior to the First Merger Effective Time expressed on a fully-diluted and as converted to Acquiror Common Stock basis, assuming, without limitation or duplication, the issuance of shares of Acquiror Common Stock in respect of all outstanding options, warrants or rights to receive such shares, in each case, outstanding as of immediately prior to the First Merger Effective Time.

"Acquiror Permits" has the meaning specified in Section 5.06(d).

"Acquiror Preferred Stock" means the Preferred Stock, par value \$0.0001 per share, of Acquiror.

"Acquiror Privacy Commitments" has the meaning specified in Section 5.19(a).

"Acquiror Properties" has the meaning specified in Section 5.21(b).

"Acquiror Registered IP" has the meaning specified in Section 5.18(a).

"Acquiror RSUs" means awards of restricted stock units over Acquiror Class A Common Stock granted under the Acquiror Option & RSU Plans.

"Acquiror SEC Reports" has the meaning specified in Section 5.15(a).

"Acquiror Software" means all material Software used in the business of Acquiror.

"Acquiror Stockholder" means a holder of Acquiror Common Stock or Acquiror Preferred Stock.

"Acquiror Stockholders Meeting" has the meaning specified in Section 8.02(d).

"Acquiror Support Agreement" has the meaning specified in the recitals.

"Acquiror Warrants" means the outstanding warrants set forth in Annex 5.04(d) to the Disclosure Schedules.

"Acquisition Proposal" has the meaning specified in Section 6.04(b)(i).

"Action" means any claim, action, suit, assessment, audit, investigation, examination, arbitration or proceeding, in each case that is brought by or before any Governmental Authority.

"Additional Proposal" has the meaning specified in Section 8.02(c).

"Affiliate" means, with respect to any specified Person, any Person that, directly or indirectly, controls, is controlled by, or is under common control with, such specified Person, through one or more intermediaries or otherwise.

"Agreement" has the meaning specified in the preamble.

"Amendment Proposal" has the meaning specified in Section 7.11.

"Ancillary Agreements" means the Support Agreements, the Company Lock-Up Agreements, the Registration Rights Agreement and any other agreement entered into in connection with to the Transactions.

"<u>Anti-Corruption Laws</u>" means any applicable Laws relating to anti-bribery or anti-corruption (governmental or commercial), including political donations or campaign finance Laws, the U.S. Foreign Corrupt Practices Act, all national and international Laws enacted to implement the OECD Convention on Combating Bribery of Foreign Officials in International Business Transactions, and Laws that prohibit the corrupt payment, offer, promise, or authorization of the payment or transfer of anything of value (including gifts or entertainment), directly or indirectly, to any person to obtain or retain business or an improper business advantage.

"Balance Sheet Date" means December 31, 2023.

"Benefit Plan" means any benefit or compensation plan, program, policy, practice, agreement, contract, arrangement or other obligation, whether or not in writing and whether or not funded, including "employee benefit plans" within the meaning of Section 3(3) of ERISA (whether or not subject to ERISA), "voluntary employees' beneficiary associations," under Section 501(c)(9) of the Code, employment, individual consulting, retirement, severance, termination pay, change in control, transaction or retention arrangements, deferred compensation, equity or equity-based compensation, incentive compensation, bonus, supplemental retirement, profit sharing, insurance, medical, welfare, vacation, fringe or other benefits or remuneration plan, program, policy, agreement or arrangement of any kind.

"Business Day" means a day other than a Saturday, Sunday or other day on which commercial banks in New York, New York are authorized or required by Law to close.

"Cancelled Shares" has the meaning specified in Section 3.01(c).

"<u>Capitalized Lease Obligations</u>" means obligations pursuant to a lease that is, or is required in accordance with GAAP to be, classified as a finance lease (specifically excluding any obligations pursuant to operating leases that, prior to adoption of the Financial Accounting Standards Board's Accounting Standard Codification Topic 842, would not have been required to be recognized on a balance sheet prepared in accordance with GAAP).

"CARES Act" means the Coronavirus Aid, Relief, and Economic Security Act.

"CBA" has the meaning specified in Section 4.12(a)(xi).

"Certificates of Merger" has the meaning specified in Section 2.03(b).

"Closing" has the meaning specified in Section 2.03.

"Closing Date" has the meaning specified in Section 2.03.

"Closing Merger Consideration" has the meaning specified in Section 3.01(d).

"COBRA" means the health insurance continuation requirements under Section 4980B of the Code and comparable provisions of ERISA and applicable state Laws.

"Code" means the Internal Revenue Code of 1986, as amended.

"<u>Commercially Available Software</u>" means "off-the-shelf", non-customized or non-modified Software that is (a) made available to or licensed on a non-exclusive basis by any Person solely in executable or object code form, (b) not incorporated into any of the products or services, and (c) made generally commercially available on and actually licensed under standard non-discriminatory terms, for either (i) annual payments of \$10,000 or less or (ii) aggregate payments by of \$10,000 or less.

"Company" has the meaning specified in the preamble.

"Company Audited Financial Statements" has the meaning specified in Section 4.06(a).

"<u>Company Benefit Plan</u>" means each Benefit Plan, sponsored, maintained or contributed to (or obligated to be sponsored, maintained or contributed to) by the Company (a) in which any current or former employee or other service provider to the Company participates or has any rights or benefits, whether absolute or contingent or (b) with respect to which the Company has or could have any Obligation, except to the extent such plan is sponsored or maintained by a Governmental Authority as required by applicable Law.

"Company Board" means the board of directors of the Company.

"Company Board Recommendation" has the meaning specified in Section 8.02(f).

"Company Certificates" has the meaning specified in Section 3.05(a).

"Company Change in Recommendation" has the meaning specified in Section 8.02(f).

"Company Common Stock" means a share of the Company's common stock, par value \$0.0001 per share.

"Company Conversion" has the meaning specified in Section 6.07.

"Company Cure Period" has the meaning specified in Section 10.01(b).

"Company Enterprise Value" means \$306,937,395.

"Company Environmental Permits" has the meaning specified in Section 4.22(a).

"<u>Company Existing Notes</u>" means (a) that certain Secured Convertible Promissory Note, dated February 17, 2023, between the Company and Acuitas (and assigned by Acuitas to Humanitario on June 19, 2023), (b) that certain Secured Convertible Promissory Note, dated June 15, 2023, between the Company and Acuitas (and assigned by Acuitas to Humanitario on June 19, 2023) (c) that certain Senior Secured Convertible Promissory Note issued by the Company to Humanitario, dated March 6, 2024, (d) that certain Senior Secured Convertible Promissory Note issued by the Company to Humanitario, dated April 17, 2024, (e) that certain Senior Secured Convertible Promissory Note issued by the Company to Humanitario, dated June 5, 2024, (f) that certain Senior Secured Convertible Promissory Note issued by the Company to Humanitario, dated June 5, 2024, (f) that certain Senior Secured Convertible Promissory Note issued by the Company to Humanitario, dated June 5, 2024, (g) that certain Senior Secured Convertible Promissory Note issued by the Company to Humanitario, dated June 5, 2024, (g) that certain Senior Secured Convertible Promissory Note issued by the Company to Humanitario, dated June 5, 2024, (g) that certain Senior Secured Convertible Promissory Note issued by the Company to Humanitario, dated June 5, 2024, (g) that certain Senior Secured Convertible Promissory Note issued by the Company to Humanitario, dated June 5, 2024, (g) that certain Senior Secured Convertible Promissory Note issued by the Company to Humanitario, dated June 5, 2024, (g) that certain Senior Secured Convertible Promissory Note issued by the Company to Humanitario, dated June 5, 2024, (g) that certain Senior Secured Convertible Promissory Note issued by the Company to Humanitario, dated June 5, 2024, and (g) that certain Senior Secured Convertible Promissory Note issued by the Company to Humanitario, dated August 30, 2024.

"Company Insurance Policies" has the meaning specified in Section 4.25.

"Company Intellectual Property" means all Owned Company Intellectual Property and all other Intellectual Property used in, held for use or necessary to the business of the Company.

"Company Interim Notes" means convertible notes issued by the Company to Humanitario after the execution of this Agreement but prior to the Closing Date.

"Company Interim Financial Statements" has the meaning specified in Section 6.11.

"Company Lease Documents" has the meaning specified in Section 4.19(c).

"Company Lock-Up Agreement" has the meaning specified in Section 6.05.

"Company Material Contracts" has the meaning specified in Section 4.12(a).

"Company Merger Consideration Schedule" has the meaning specified in Section 3.09.

"Company Notes" means the Company Existing Notes and the Company Interim Notes.

"Company Option" means an option to purchase Company Common Stock issued by the Company pursuant to the Company Stock Plan.

"<u>Company Organizational Documents</u>" means the certificate of incorporation and bylaws of the Company, in each case as may be amended from time to time in accordance with the terms of this Agreement.

"<u>Company Outstanding Shares</u>" means the total number of shares of Company Common Stock outstanding immediately prior to the First Merger Effective Time expressed on a fully-diluted and as-converted to Company Common Stock basis and assuming, without limitation or duplication, the issuance of shares of Company Common Stock issuable (i) upon the conversion of the Company Notes and the exercise of the Company Warrants, and (ii) in respect of all other outstanding options, restricted stock awards, warrants or rights to receive such shares, whether conditional or unconditional and including any outstanding options, warrants or rights triggered by or associated with the consummation of the Merger.

"Company Permits" has the meaning specified in Section 4.07(d).

"Company Privacy Commitments" has the meaning specified in Section 4.17(a).

"Company Properties" has the meaning specified in Section 4.19(b).

"Company Registered IP" has the meaning specified in Section 4.16(a).

"Company Software" means all material Software used in the business of the Company.

"Company Stock" means the Company Common Stock and any other equity securities issued by the Company.

"Company Stock Plan" means the Company's 2020 Stock Incentive Plan.

"Company Stockholder" means each holder of a share of Company Stock.

"<u>Company Stockholder Approvals</u>" means a consent approving and adopting this Agreement, the Merger and, to the extent required by Law, the Transactions in writing or by electronic transmission executed by the Company Stockholders holding sufficient voting power to approve and adopt this Agreement, the Merger and the Transactions.

"Company Stockholder Written Consent Deadline" has the meaning specified in Section 8.02(f).

"Company Support Agreement" has the meaning specified in the recitals.

"Company Warrants" means each outstanding and unexercised warrant to purchase Company Stock whether or not then vested or fully exercisable.

"Competing Proposal" has the meaning specified in Section 7.08(c).

"<u>Confidentiality Agreement</u>" means the Mutual Nondisclosure Agreement (as may be amended or supplement), dated as of April 22, 2024, by and between Acquiror and the Company.

"Contract" means any legally binding contract, agreement, license or sublicense, subcontract, lease, and purchase order.

"<u>COVID-19</u>" means the novel coronavirus, SARS-CoV-2 or COVID-19 or any mutation of the same, including any resulting epidemics, pandemics, disease outbreaks or public health emergencies.

"<u>COVID-19 Measures</u>" means any quarantine, isolation, "shelter in place," "stay at home," workforce reduction, social distancing, shut down, closure, sequester or any other Law, decree, judgment, injunction or other order, directive, guidelines or recommendations by any Governmental Authority or industry group in connection with or in response to COVID-19, including, the CARES Act.

"<u>Data Security Requirements</u>" means, collectively, all of the following: (a) all applicable industry guidelines and self-regulatory programs, including self-regulatory programs related to data security, cyber security, information security or online behavioral advertising; (b) with respect to Acquiror, Acquiror's privacy policies and procedures; and (c) with respect to the Company, the Company's privacy policies and procedures.

"DGCL" means the General Corporation Law of the State of Delaware.

"Dissenting Shares" has the meaning specified in Section 3.08.

"DLLCA" means the Delaware Limited Liability Company Act of the State of Delaware.

"Enforceability Exceptions" has the meaning specified in Section 4.02(a).

"<u>Environmental Laws</u>" means any and all applicable Laws relating to pollution or protection of the environment (including natural, cultural and biological resources), public and worker health and safety (to the extent relating to exposure to Hazardous Materials), or the investigation, cleanup, use, generation, management, manufacture, labeling, registration, import, export, storage, emission, treatment, transportation, disposal or Release of or exposure to Hazardous Materials.

"ERISA" means the Employee Retirement Income Security Act of 1974.

"ERISA Affiliate" means any entity (whether or not incorporated) which, together with the Company or Acquiror, as applicable, currently is, or has been within the last six (6) years treated as a single employer under Section 414(b), (c), (m), or (o) of the Code.

"Exchange Act" means the Securities Exchange Act of 1934, as amended.

"Exchange Ratio" means, subject to Section 3.04, (rounded to four decimal places) the ratio resulting from (a) (i) the quotient of (A) Company Enterprise Value, divided by (B) Acquiror Enterprise Value, multiplied by (ii) Acquiror Outstanding Shares, divided by (b) Company Outstanding Shares.

"<u>Ex-Im Laws</u>" means (a) all applicable trade, export control, import, and antiboycott laws and regulations imposed, administered, or enforced by the U.S. government, including the Arms Export Control Act (22 U.S.C. § 1778), the International Emergency Economic Powers Act (50 U.S.C. §§ 1701–1706), Section 999 of the Internal Revenue Code, the U.S. customs laws at Title 19 of the U.S. Code, the Export Control Reform Act of 2018 (50 U.S.C. §§ 4801-4861), the International Traffic in Arms Regulations (22 C.F.R. Parts 120–130), the Export Administration Regulations (15 C.F.R. Parts 730-774), the U.S. customs regulations at 19 C.F.R. Chapter I, and the Foreign Trade Regulations (15 C.F.R. Part 30), and (b) all applicable trade, export control, import, and antiboycott laws and regulations imposed, administered or enforced by any other country, except to the extent inconsistent with U.S. law.

"<u>Financial Derivative/Hedging Arrangement</u>" means any transaction (including an agreement with respect thereto) that is a rate swap transaction, basis swap, forward rate transaction, commodity swap, commodity option, equity or equity index swap, equity or equity index option, bond option, interest rate option, foreign exchange transaction, cap transaction, floor transaction, collar transaction, currency swap transaction, cross-currency rate swap transaction, currency option or any combination of these transactions.

"Final Surviving Company" has the meaning specified in the recitals.

"First Certificate of Merger" has the meaning specified in Section 2.03(a).

"First Merger" has the meaning specified in the recitals.

"First Merger Effective Time" has the meaning specified in Section 2.03(a).

"Fraud" means actual and intentional fraud under Delaware common law with a specific intent to deceive brought against a party hereto based on a representation or warranty of such party hereto contained in this Agreement; provided, that at the time such representation was made (a) such representation was inaccurate, (b) such party hereto had actual Knowledge (and not imputed or constructive Knowledge) of the inaccuracy of such representation, (c) such party hereto had the specific intent to deceive another party hereto, and (d) the other party hereto acted in reliance on such inaccurate representation and suffered losses as a result of such inaccuracy. For the avoidance of doubt, "Fraud" does not include any claim for equitable fraud, promissory fraud, unfair dealings fraud, or any torts (including a claim for fraud) based on negligence or recklessness.

"GAAP" means generally accepted accounting principles in the United States set forth in the opinions and pronouncements of the Financial Accounting Standards Board or such other principles as may be approved by a significant segment of the accounting profession in the United States that are applicable to the circumstances as of the date of determination, consistently applied.

"Government Official" means any official or employee of any directly or indirectly government-owned or controlled entity, and any officer or employee of a public international organization, as well as any person acting in an official capacity for or on behalf of any such entity or for or on behalf of any such public international organization.

"Governmental Authority" means any federal, state, provincial, municipal, local or foreign government, governmental authority, regulatory or administrative agency, governmental commission, department, board, bureau, agency or instrumentality, arbitrator, court or tribunal.

"Governmental Order" means any order, judgment, injunction, decree, writ, stipulation, determination or award, in each case, entered by or with any Governmental Authority.

"<u>Hazardous Material</u>" means any material, substance or waste that is listed, regulated, or defined as "hazardous," "toxic," or "radioactive," or as a "pollutant" or "contaminant" (or words of similar intent or meaning), or otherwise subject to liability or standards of care under applicable Environmental Laws, including but not limited to petroleum, petroleum by-products, asbestos or asbestos-containing material, polychlorinated biphenyls, per- and polyfluoroalkyl substances, flammable or explosive substances, toxic mold or pesticides.

"Humanitario" has the meaning specified in the recitals.

"Indebtedness" means, with respect to any Person, without duplication, any obligations (whether or not contingent) consisting of (a) the outstanding principal amount of and accrued and unpaid interest on, and other payment obligations for, borrowed money, or payment obligations issued or incurred in substitution or exchange for payment obligations for borrowed money, (b) amounts owing as deferred purchase price for property or services, including "earnout" payments, (c) payment obligations evidenced by any promissory note, bond, debenture, mortgage or other debt instrument or debt security, (d) contingent reimbursement obligations with respect to letters of credit, bankers' acceptance or similar facilities (in each case to the extent drawn), (e) payment obligations of a third party secured by (or for which the holder of such payment obligations has an existing right, contingent or otherwise, to be secured by) any Lien, other than a Permitted Lien, on assets or properties of such Person, whether or not the obligations secured thereby have been assumed, (f) Capitalized Lease Obligations, (g) obligations under any Financial Derivative/Hedging Arrangement (assuming termination thereof), (h) guarantees, make-whole agreements, hold harmless agreements or other similar arrangements with respect to any amounts of a type described in clauses (a) through (h) above, and (i) with respect to each of the foregoing, any unpaid interest, breakage costs, prepayment or redemption penalties or premiums, or other unpaid fees or obligations (including unreimbursed expenses or indemnification obligations for which a claim has been made); provided, however, that Indebtedness shall not include accounts payable to trade creditors that are not past due and accrued expenses arising in the ordinary course of business consistent with past practice.

"Intellectual Property" means all intellectual property rights and other proprietary rights of any kind, as they exist anywhere in the world, whether registered or unregistered, including all: (a) patents and patent applications (including any renewals, extensions, divisions, continuations, continuations-inpart, reissues, reexaminations and interferences thereof, substitutions and foreign counterparts relating to any such patents and patent applications, and industrial property rights); (b) trademarks, service marks, trade dress, trade and brand names and logos and social media designations, and any and all other indicia of source or origin, together with all goodwill associated therewith and any registrations and applications for registrations therefor; (c) rights associated with works of authorship, including exclusive exploitation rights, copyrights, copyrightable works, and any registrations and applications for registrations and applications for registrations and applications for registrations and applications for registration therefor; (d) internet domain names, IP addresses and URLs; (e) trade secrets and other intellectual property rights in know-how, inventions, processes, procedures, technical data, algorithms, formulae, protocols, techniques, results of experimentation and testing, and business information (including financial and marketing plans, supplier lists, and pricing and cost information) and other confidential and proprietary information rights; (f) rights in databases and data collections (including knowledge databases, customer lists and customer databases); (g) intellectual property rights in Software (including in object code and source code); and (h) all past, present and future claims and causes of action arising out of or related to infringement or misappropriation of any of the foregoing.

"Intended Tax Treatment" has the meaning specified in Section 8.03(b).

"Interim Period" has the meaning specified in Section 6.01.

"<u>Intervening Event</u>" means (i) an event, fact, development, circumstance or occurrence, that materially and negatively affects the business, assets, operations or prospects of the Company, and that was not known and was not reasonably foreseeable to Acquiror or the Acquiror Board as of the date hereof (or the consequences of which were not reasonably foreseeable to the Acquiror Board as of the date hereof), and that becomes known to the Acquiror Board after the date of this Agreement (but specifically excluding a Competing Proposal) and/or (ii) a Superior Proposal.

"Intervening Event Notice" has the meaning specified in Section 8.02(d).

"Intervening Event Notice Period" has the meaning specified in Section 8.02(d).

"<u>IT Systems</u>" means all computer hardware (including computer systems, servers, routers, hardware, firmware, peripherals, communication equipment and links, storage media, networking equipment, power supplies and any other components used in conjunction with the foregoing), Proprietary Databases, data Processing systems, Software, and all other information technology equipment and assets, including any outsourced systems and processes, and documentation related to the foregoing, in each case, that are owned, leased, licensed by, used, relied on, held for use or controlled by the Company or Acquiror, as the case may be.

"Knowledge" shall mean the actual knowledge of (a) in the case of the Company, Brian Gordon and Joe Paolucci, and (b) in the case of Acquiror, Mark Hanchett, Annie Pratt, Sarah Wyant and Jordan Christensen.

"Law" means any statute, law (including common law), constitution, act, treaty, convention, code, ordinance, rule, regulation or Governmental Order, in each case, of any Governmental Authority, including, without limitation, the Code, the Treasury Regulations thereunder and other tax rules and regulations.

"Lease Shortfall Amount" means an amount equal the excess of (a) any rental payments on the Tempe Lease and the Mesa Lease remaining unpaid as of the Closing Date (net of the security deposits with respect thereto) over (b) all cash and cash equivalents on the Company's balance sheet as of such time.

"Lien" means any mortgage, deed of trust, pledge, hypothecation, easement, right of way, purchase option, right of first refusal, covenant, restriction, security interest, title defect, encroachment or other survey defect, or other lien or encumbrance of any kind, except for (a) any restrictions arising under any applicable Securities Laws, and (b) immaterial easements, rights of way, covenants, encumbrances or restrictions that do not materially detract from the value of the underlying asset or interfere with the use of, or in the case of real property, occupancy of, the asset.

"Listing Application" has the meaning specified in Section 8.06.

"Material Adverse Effect" means any event, change, circumstance or development that has a material adverse effect on the assets, business, results of operations or financial condition of (x) the Company or (y) Acquiror and Nxu Tech, together, in either case; provided, however, that in no event would any of the following (or any event, change, circumstance or development to the extent primarily arising out of any of the following), alone or in combination, be deemed to constitute, or be taken into account in determining whether there has been or will be, a "Material Adverse Effect": (a) any actual or proposed change or development in applicable Laws (including COVID-19 Measures) or GAAP or any official interpretation thereof; (b) any change or development in interest rates, exchange rates, or economic, political, legislative, regulatory, business, financial, commodity, currency or market conditions generally affecting the economy or the industry in which the Company operates; (c) the announcement or the execution of this Agreement, the pendency or consummation of the Merger or the performance of this Agreement, including the impact thereof on relationships, contractual or otherwise, with customers, suppliers, licensors, distributors, partners, providers, employees, or contractors; (d) any change generally affecting any of the industries or markets in which the Company or Acquiror, as the case may be, operate or the economy as a whole; (e) the compliance with the terms of this Agreement or the taking of any action, or failure to take action, required or contemplated by this Agreement or with the prior written consent of Acquiror; (f) any earthquake, hurricane, tsunami, tornado, flood, mudslide, wild fire or other natural disaster, epidemic, disease outbreak, pandemic (including COVID-19 (or any mutation or variation thereof or related health condition)), weather condition, explosion fire, act of God or other force majeure event; (g) any (x) armed hostilities or the escalation thereof, whether or not pursuant to the declaration of any national emergency or war, or (y) military or terrorist attack, or escalation thereof; and (h) any failure of the Company or Acquiror, as the case may be, to meet any projections, forecasts or budgets; provided, that clause (h) shall not prevent or otherwise affect a determination that any change or effect underlying such failure to meet projections or forecasts has resulted in, or contributed to, or would reasonably be expected to result in or contribute to, a Material Adverse Effect (to the extent such change or effect is not otherwise excluded from this definition of Material Adverse Effect); except in the case of clause (a), (b), (d), (f) and (g) to the extent that such change has a disproportionate impact on the Company or Acquiror as the case may be, as compared to other participants in the industry in which such party operates.

"Merger" has the meaning specified in the recitals.

"Merger Sub I" has the meaning specified in the preamble.

"Merger Subs" has the meaning specified in the preamble.

"Merger Sub II" has the meaning specified in the preamble.

"Mesa Lease" means the Standard Industrial Real Estate Lease dated February 12, 2020, between the Company and Majestic Mesa Partners, LLC.

"Multiemployer Plan" means a multiemployer plan, as defined in Section 3(37) or 4001(a)(3) of ERISA.

"<u>NASDAQ</u>" means the Nasdaq Stock Market, LLC, including the Nasdaq Capital Market or such other Nasdaq market on which shares of Acquiror Class A Common Stock may be listed.

"Nxu Tech" has the meaning given in Section 5.01(b).

"<u>Obligations</u>" means any duties, debts, guarantees, responsibilities, liabilities, assurances, commitments and obligations, costs and expenses of whatever kind and nature, whether legal or equitable, whether vested, absolute or contingent, primary or secondary, direct or indirect, known or unknown, foreseen or unforeseen, asserted or unasserted, accrued or unaccrued, liquidated or unliquidated, due or to become due, patent or latent, and whether based in common law or statute or arising under Contract or by action of any Governmental Authority or otherwise.

"<u>Open Source Software</u>" means Software that is subject to or licensed, provided or distributed under any open source license (including any copyleft license), including any license that satisfies the definition of free software by the Free Software Foundation, satisfies the definition of open source software by the Open Source Initiative, or that is offered under similar types of computer Software licenses.

"Owned Acquiror Intellectual Property" means all Intellectual Property owned by Acquiror or Nxu Tech or that Acquiror or Nxu Tech purports to own, including Acquiror Registered IP.

"Owned Company Intellectual Property" means all Intellectual Property owned by the Company or that the Company purports to own, including Company Registered IP.

"Payor" has the meaning specified in Section 3.06.

"Permitted Liens" means (a) statutory or common law Liens of mechanics, materialmen, warehousemen, landlords, carriers, repairmen, construction contractors and other similar Liens (i) that arise in the ordinary course of business, (ii) relate to amounts not yet delinquent or (iii) that are being contested in good faith through appropriate Actions and either are not material or appropriate reserves for the amount being contested have been established in accordance with GAAP on the Company Audited Financial Statements (in the case of the Company) or on the financial statements included in the Acquiror SEC Reports (in the case of Acquiror); (b) Liens arising under original purchase price conditional sales contracts and equipment leases with third parties entered into in the ordinary course of business; (c) Liens for Taxes (i) not yet delinquent or (ii) that are being contested in good faith through appropriate Actions and for which appropriate reserves have been established in accordance with GAAP; (d) non-monetary Liens, encumbrances and restrictions on real property (including easements, covenants, rights of way and similar restrictions of record) that do not, individually or in the aggregate, materially interfere with the present uses or occupancy of such real property, or which do not materially detract from the value of the underlying asset; (e) non-exclusive licenses of Intellectual Property granted to customers and service providers in the ordinary course of business; (f) Liens that secure obligations that are reflected as liabilities on the balance sheet included (i) with respect to the Company, in the Company Audited Financial Statements (which such Liens are referenced or the existence of which such Liens is referred to in the notes to the balance sheet included in the Company Audited Financial Statements), or (ii) with respect to Acquiror, in the financial statements included in the Acquiror SEC Reports (which such Liens are referenced or the existence of which such Liens is referred to in the Acquiror SEC Reports or the notes to the balance sheet included in the financial statements included in the Acquiror SEC Reports); (g) requirements and restrictions of zoning, building and other applicable Laws and municipal by-laws, and development, site plan, subdivision or other agreements with municipalities, which do not materially interfere with the current use or occupancy of any real property leased by the Company, or which do not materially detract from the value of the underlying asset; and (h) Liens described on Schedule 101(a).

"Person" means any individual, firm, corporation, partnership, limited liability company, incorporated or unincorporated association, joint venture, joint stock company, Governmental Authority or other entity of any kind.

"<u>Personal Information</u>" means any information that, alone or in combination with other information, identifies, relates to, describes, is reasonably capable of being associated with, or could reasonably be linked with a natural Person, including names, addresses, telephone numbers, personal health information, drivers' license numbers and government-issued identification numbers, credit card information, billing and transactional information, contact preferences, financial information or other information that constitutes "personal information", "personally identifiable information" or "personal data" under applicable laws.

"<u>Pre-Closing Financing</u>" means (a) the issuance by the Company or Acquiror of any shares of capital stock or equity-linked securities or rights exercisable for or convertible into shares of capital stock or (b) equity financing facilities or non-redemption pools entered into by the Company or Acquiror, in each case with respect to clauses (a) and (b), the definitive agreements of which will be entered into on or after the date hereof and prior to or simultaneously with the Closing, but the closing of which will be conditioned on the Closing of the Transactions.

"<u>Privacy Laws</u>" means any and all Laws applicable to the Company or Acquiror, as the case may be, relating to data privacy, protection or security or the receipt or the Processing of Personal Information, including all applicable Laws governing cybercrime and breach notification, Section 5 of the Federal Trade Commission Act, California Consumer Privacy Act of 2018, California Privacy Rights Act of 2020, social security number protection Laws, EU General Data Protection Regulation 2016/679, EU Privacy and Electronic Communications Directive 2002/58/EC, applicable Laws concerning requirements for any outbound communications and/or electronic marketing (including e-mail marketing, telemarketing and text messaging), PCI-DSS, HIPAA, and any amendments, successor and/or implementing legislation or regulation to any of the foregoing Laws.

"Processing" means any operation or set of operations performed on data, including adaption or alteration, combination, recording, distribution, import, export, organization, protection, disposal, blocking, erasure, destruction, storing, use, disclosure, collection, transferring or otherwise interacting with such data, including Personal Information.

"Proposals" has the meaning specified in Section 8.02(c).

"Proprietary Database" means any computer database owned by Acquiror or Company, respectively, and used in connection with its business.

"Proxy Statement" means the proxy statement and notice of Acquiror Stockholders Meeting filed by Acquiror as part of the Registration Statement with respect to the Acquiror Stockholders Meeting for the purpose of soliciting proxies from holders of Acquiror Common Stock to approve the Proposals.

"Registration Rights Agreement" has the meaning specified in the recitals.

"Registration Statement" has the meaning specified in Section 8.02(a).

"<u>Related Party</u>" means, with respect to any party, any Subsidiary or Affiliate thereof, or any business, entity or Person that any of the foregoing controls, is controlled by or is under common control with.

"<u>Release</u>" means any actual or threatened releasing, depositing, spilling, leaking, pumping, pouring, placing, emitting, discarding, abandoning, emptying, discharging, migrating, injecting, escaping, leaching, dumping, or disposing into or through the indoor or outdoor environment.

"<u>Representative</u>" means, as to any Person, any of the officers, directors, managers, employees, counsel, accountants, financial advisors, lenders, debt financing sources and consultants of such Person.

"<u>Required Acquiror Stockholder Approvals</u>" means (a) approval of the Transaction Proposal and the Acquiror Equity Plan Proposal by a majority of the total votes cast by Acquiror Stockholders at the Acquiror Stockholders Meeting and (b) approval of the Amendment Proposal at the Acquiror Stockholders Meeting by (i) the holders of a majority of the voting power of the then-outstanding shares of capital stock of Acquiror entitled to vote generally in the election of directors, voting together as a single class and (ii) holders of two-thirds (2/3) of the outstanding shares of Acquiror Class B Common Stock.

"<u>Sanctioned Person</u>" means any Person that is the target of Sanctions, including (a) any Person listed in any Sanctions-related list of designated Persons maintained by the Office of Foreign Assets Control of the U.S. Department of the Treasury, the U.S. Department of Commerce, or the U.S. Department of State, the United Nations Security Council, the European Union, any Member State of the European Union, or the United Kingdom; (b) any Person located, organized, or resident in a Sanctioned Territory; (c) the government of a Sanctioned Territory or the Government of Venezuela; or (d) any Person 50% or more owned or controlled by any such Person or Persons or acting for or on behalf of such Person or Persons.

"<u>Sanctioned Territory</u>" means, at any time, a country or territory that is itself the target of comprehensive Sanctions (as of the date of this Agreement, Cuba, Iran, North Korea, Syria, the Crimea region of Ukraine, the so-called Donetsk People's Republic, the so-called Luhansk People's Republic, and the non-Ukrainian-government controlled areas of Kherson and Zaporizhzhia).

"<u>Sanctions</u>" means economic or financial sanctions, trade controls, or embargoes imposed, administered or enforced from time to time by (a) the U.S. government, including those administered by the Office of Foreign Assets Control of the U.S. Department of the Treasury, the U.S. Department of Commerce, or the U.S. Department of State, or (b) the United Nations Security Council, the European Union, any European Union member state, the United Kingdom, or (c) any other applicable jurisdiction.

"Schedules" means the disclosure schedules of the Company or Acquiror, as applicable.

"SEC" means the United States Securities and Exchange Commission.

"Second Certificate of Merger" has the meaning specified in Section 2.03(b).

"Second Merger" has the meaning specified in the recitals.

"Second Merger Effective Time" has the meaning specified in Section 2.03(b).

"Securities Act" means the Securities Act of 1933, as amended.

"Securities Laws" means the securities laws of any state or federal entity and the rules and regulations promulgated thereunder.

"Software" means any and all (a) computer programs, including any and all software implementation of algorithms, models and methodologies, whether in source code (including any Open Source Software), object code, executable code, human readable form or other form, operating systems, application programs, routines and modules, software tools, firmware and software embedded in equipment, as well as any software-as-a-service and other software-enabled hosted services and platforms, (b) Proprietary Databases and compilations, including any and all data and collections of data, whether machine readable or otherwise, (c) descriptions, models and methodologies, schematics, specification, flow charts and other work products used to design, plan, organize and develop any of the foregoing, screens, user interfaces, report formats, firmware, development tools, templates, menus, buttons and icons and (d) all written or electronic materials and documentation that explain the structure or use of software or that were used in the development of software, including logic diagrams, screens, user interfaces, reports, firmware, development tools, templates, menus, buttons, icons, flow charts, code notes, procedural diagrams, error reports, manuals and training materials used to design, plan, organize and develop any of the foregoing.

"Subsidiary" means, with respect to a Person, any corporation or other organization (including a limited liability company or a partnership), whether incorporated or unincorporated, of which such Person directly or indirectly owns or controls a majority of the securities or other interests having by their terms ordinary voting power to elect a majority of the board of directors or others performing similar functions with respect to such corporation or other organization or any organization of which such Person or any of its Subsidiaries is, directly or indirectly, a general partner or managing member.

"Superior Proposal" means an unsolicited written bona fide offer to consummate any of the following transactions made by a third party that the Acquiror Board determines, in its good faith judgment (after having received the advice of a financial advisor), is capable of doing so (taking into account all financial, regulatory, legal and other aspects of such proposal): (i) a merger, consolidation, share exchange, business combination or other similar transaction involving Acquiror pursuant to which the stockholders of Acquiror immediately preceding such transaction would hold less than 50% of the equity interest in the surviving or resulting entity of such transaction; or (ii) the acquisition by any person or group (including by means of a tender offer or an exchange offer or a two-step transaction involving a tender offer followed with reasonable promptness by a cash-out merger involving Acquiror), directly or indirectly, of ownership of 100% of the then outstanding shares of stock of Acquiror, in each case on terms (including conditions to consummation of the contemplated transaction) that the Acquiror Board determines, in its good faith judgment (after having received the advice of a financial advisor), to be more favorable, from a financial point of view, to Acquiror's stockholders than the Transactions (including any proposal by the Company to amend the terms of the Transactions) and for which financing, to the extent required, is then committed.

"Support Agreements" has the meaning specified in the recitals.

"Surviving Corporation" has the meaning specified in the recitals hereto.

"Surviving Provisions" has the meaning specified in Section 10.02.

"Tax" means any federal, state, provincial, territorial, local, foreign and other net income, alternative or add-on minimum, franchise, gross income, adjusted gross income or gross receipts, employment, unemployment, compensation, utility, social security (or similar), withholding, payroll, ad valorem, transfer, windfall profits, franchise, license, branch, excise, severance, production, stamp, occupation, premium, personal property, real property, capital stock, profits, disability, registration, value added, capital gains, goods and services, estimated, sales, use, or other tax, escheat or unclaimed property obligation, governmental fee or other like assessment, together with any interest, penalty, fine, levy, impost, duty, charge, addition to tax or additional amount imposed with respect thereto by a Governmental Authority.

"Tax Authority" means any Governmental Authority with jurisdiction or authority to impose, administer, levy, assess or collect Tax.

"<u>Tax Return</u>" means any return, report, statement, refund, claim, election, disclosure, declaration, information report or return, statement, estimate or other document filed or required to be filed with a Tax Authority with respect to Taxes, including any schedule or attachment thereto and including any amendments thereof.

"Tempe Lease" means the Sublease Agreement, dated as of June 9, 2023, by and between Diffusion Group, L.L.C. and Nxu Technologies, LLC.

"Terminating Acquiror Breach" has the meaning specified in Section 10.01(c).

"Terminating Company Breach" has the meaning specified in Section 10.01(b).

"Termination Date" has the meaning specified in Section 10.01(b).

"Transaction Proposal" has the meaning specified in Section 8.02(c).

"Transactions" means the transactions contemplated by this Agreement, including the Merger.

"Transfer Taxes" has the meaning specified in Section 8.03(a).

"Treasury Regulations" means the regulations promulgated under the Code.

"Voting Trust Transfer" has the meaning specified in the recitals.

"Voting Trustee" has the meaning specified in the recitals.

"WARN Act" has the meaning specified in Section 4.13(e).

"<u>Willful Breach</u>" means, with respect to any agreement, a party's knowing and intentional material breach of its covenants or other agreements set forth in such agreement, which material breach constitutes, or is a consequence of, a purposeful act or failure to act by such party with the knowledge that the taking of such act or failure to take such act would cause a material breach of such agreement.

1.02 <u>Construction</u>.

(a) Unless the context of this Agreement otherwise requires, (i) words of any gender include each other gender, (ii) words using the singular or plural number also include the plural or singular number, respectively, (iii) the terms "hereof," "herein," "hereby," "hereto" and derivative or similar words refer to this entire Agreement, (iv) the terms "Article", "Section", "Schedule", "Exhibit" and "Annex" refer to the specified Article, Section, Schedule, Exhibit or Annex of or to this Agreement unless otherwise specified, (v) the word "including" shall mean "including without limitation", (vi) the word "or" shall be disjunctive but not exclusive and (vii) any reference to a Law shall mean such Law as amended through the relevant date.

(b) Unless the context of this Agreement otherwise requires, references to agreements and other documents shall be deemed to include all subsequent amendments and other modifications thereto.

(c) Unless the context of this Agreement otherwise requires, references to statutes shall include all regulations promulgated thereunder and references to statutes or regulations shall be construed as including all statutory and regulatory provisions consolidating, amending or replacing the statute or regulation.

(d) The language used in this Agreement shall be deemed to be the language chosen by the parties to express their mutual intent and no rule of strict construction shall be applied against any party.

(e) Whenever this Agreement refers to a number of days, such number shall refer to calendar days unless Business Days are specified. If any action is to be taken or given on or by a particular calendar day, and such calendar day is not a Business Day, then such action may be deferred until the next Business Day.

(f) All accounting terms used herein and not expressly defined herein shall have the meanings given to them under GAAP.

(g) The phrases "delivered," "provided to," "furnished to," "made available" and phrases of similar import when used herein, unless the context otherwise requires, means that a copy of the information or material referred to has been provided no later than the Business Day prior to the date of this Agreement to the party to which such information or material is to be provided or furnished (i) in the virtual "data room" set up by the Company or Acquiror, as the case may be, in connection with this Agreement or (ii) by delivery to such party or its legal counsel via electronic mail or hard copy form.

ARTICLE II THE MERGER; CLOSING

2.01 The Merger.

(a) Upon the terms and subject to the conditions set forth in this Agreement, at the First Merger Effective Time, Merger Sub I shall be merged with and into the Company, with the Company continuing as the Surviving Corporation. Following the First Merger, the separate corporate existence of Merger Sub I shall cease. The First Merger shall be consummated in accordance with this Agreement (including Section 2.03) and the DGCL.

(b) Promptly following the First Merger, the Surviving Corporation shall merge with and into Merger Sub II, whereupon the separate corporate existence of the Surviving Corporation shall cease. The Second Merger shall be consummated in accordance with this Agreement (including Section 2.03), the DGCL and the DLLCA.

2.02 Effects of the Merger.

(a) The First Merger shall have the effects set forth in this Agreement and the DGCL. Without limiting the generality of the foregoing and subject thereto, by virtue of the First Merger and without further act or deed, at the First Merger Effective Time, all of the property, rights, privileges, powers and franchises of the Company and Merger Sub I shall vest in the Surviving Corporation and all of the debts, liabilities and duties of the Company and Merger Sub I shall become the debts, liabilities and duties of the Surviving Corporation.

(b) The Second Merger shall have the effects set forth in this Agreement, the DGCL and the DLLCA. Without limiting the generality of the foregoing and subject thereto, by virtue of the Second Merger and without further act or deed, at the Second Merger Effective Time, all of the property, rights, privileges, powers and franchises of the Surviving Corporation and Merger Sub II shall vest in the Final Surviving Company and all of the debts, liabilities and duties of the Surviving Company.

2.03 <u>Closing</u>. Subject to the terms and conditions of this Agreement, the closing of the Transactions (the "<u>Closing</u>") shall take place electronically through the exchange of documents via e-mail or facsimile beginning on the date which is three (3) Business Days after the date on which all conditions set forth in <u>Article IX</u> shall have been satisfied or waived (other than those conditions that by their terms are to be satisfied at the Closing, but subject to the satisfaction or waiver thereof) or such other time and place as Acquiror and the Company may mutually agree in writing. The date on which the Closing actually occurs is referred to in this Agreement as the "<u>Closing Date</u>." Subject to the satisfaction or waiver of all of the conditions set forth in <u>Article IX</u> of this Agreement, and provided this Agreement has not theretofore been terminated pursuant to its terms, at the Closing:

(a) the Company shall cause a certificate of merger in form and substance reasonably satisfactory to Acquiror and the Company (the "<u>First Certificate of Merger</u>"), together with all other necessary documentation, to be executed, acknowledged and filed with the Secretary of State of the State of Delaware as provided in Sections 251 and 103 of the DGCL, and the First Certificate of Merger shall specify that the Merger shall become effective upon filing of the First Certificate of Merger with the Secretary of State of the State of Delaware or at such later time as may be agreed by Acquiror and the Company in writing and specified in the Certificate of Merger (the "<u>First Merger Effective Time</u>");

(b) Promptly following the First Merger Effective Time, the Company shall cause a certificate of merger in form and substance reasonably satisfactory to Acquiror and the Company (the "Second Certificate of Merger", and together with the First Certificate of Merger, the "Certificates of Merger"), together with all other necessary documentation, to be executed, acknowledged and filed with the Secretary of State of the State of Delaware as provided in Sections 264 of the DGCL and Section 18-209 of the DLLCA, and the Second Certificate of Merger shall specify that the Second Merger shall become effective upon filing of the Second Certificate of Merger with the Secretary of State of Delaware or at such later time as may be agreed by Acquiror and the Company in writing and specified in the Certificate of Merger (the "Second Merger Effective Time"); and

(c) the Merger will become effective, and Acquiror shall have been approved to trade publicly on the NASDAQ under the new ticker symbol, "VRDE".

2.04 Organizational Documents of the Company.

(a) At the First Merger Effective Time, the certificate of incorporation and bylaws of the Merger Sub I as in effect immediately prior to the First Merger Effective Time shall be the certificate of incorporation and bylaws of the Surviving Corporation, until thereafter amended in accordance with their terms and the DGCL.

(b) At the Second Merger Effective Time, the limited liability company agreement of Merger Sub II, as in effect immediately prior to the Second Merger Effective Time, shall be the limited liability company agreement of the Final Surviving Company at the Second Merger Effective Time, until thereafter amended in accordance with its terms and the DLLCA; provided, however, in each case, that the name of the company set forth therein shall be changed to the name of the Company.

2.05 Directors and Officers of the Company and Acquiror.

(a) Persons constituting the officers of the Company prior to the First Merger Effective Time shall be appointed as the officers of the Final Surviving Company until the earlier of their death, resignation or removal or until their respective successors are duly elected or appointed.

Except as otherwise agreed in writing by the Company and Acquiror prior to the Closing, and conditioned upon the occurrence of (b) the Closing, subject to any limitation imposed under applicable Laws and NASDAQ listing requirements, Acquiror shall take all necessary action prior to the First Merger Effective Time such that (i) except for Jessica Billingsley, each director of Acquiror in office immediately prior to the First Merger Effective Time shall cease to be a director immediately following the First Merger Effective Time (including by causing each such director to tender an irrevocable resignation as a director, effective as of the First Merger Effective Time), (ii) immediately after the First Merger Effective Time, the Acquiror Board shall consist of seven (7) members, with one (1) member of the Acquiror Board to be appointed by Acquiror (the "Acquiror Director") (who initially shall be Jessica Billingsley) and (iii) the initial members of the Acquiror Board shall be the individuals set forth on Exhibit G, or other individuals determined by the Company after signing, each to serve in the applicable class of directors identified by the Company after signing. The initial members of the audit committee, the compensation committee and the corporate governance committee of the Acquiror Board immediately after the First Merger Effective Time shall be such individuals determined by the Company prior to the Closing. If necessary to effect the foregoing, the Acquiror Board shall adopt resolutions prior to the First Merger Effective Time that expand or decrease the size of the Acquiror Board and appoint such persons to the vacancies resulting from the incumbent directors' respective resignations or, if applicable, the newly created directorships upon any expansion of the size of the Acquiror Board. Each person appointed as a director of Acquiror pursuant to this Section 2.05(b) shall remain in office as a director of Acquiror until his or her successor is elected and qualified or until his or her earlier death, resignation or removal. If any of the directors set forth on Exhibit G shall be unable or unwilling to serve at the Closing, the party that appointed such individual shall promptly designate a replacement director and provide any relevant information about such appointee as the other party may reasonably request.

(c) Acquiror shall take all necessary action prior to the First Merger Effective Time such that (i) each officer of Acquiror in office immediately prior to the First Merger Effective Time shall cease to be an officer immediately following the First Merger Effective Time and (ii) the persons constituting the officers of the Company prior to the First Merger Effective Time shall, as of immediately following the First Merger Effective Time, be appointed the officers of Acquiror in identical positions until the earlier of their death, resignation or removal or until their respective successors are duly elected or appointed.

ARTICLE III EFFECTS OF THE MERGER

3.01 <u>Effect on Capital Stock</u>. Upon the terms and subject to the conditions of this Agreement, at the First Merger Effective Time, by virtue of the Merger and without any action on the part of the Company, Acquiror, the Merger Subs or the holder of any Company Stock:

(a) Treatment of Merger Sub I Common Stock. Each share of common stock of Merger Sub I, par value \$0.0001 per share, issued and outstanding immediately prior to the First Merger Effective Time shall be automatically converted at the First Merger Effective Time into one validly issued, fully paid and nonassessable share of common stock of the Surviving Corporation, par value \$0.0001 per share, and such shares will constitute the only outstanding equity interests of the Surviving Corporation following the Merger.

(b) Treatment of Merger Sub II Membership Interests. Each share of common stock of the Surviving Corporation, par value \$0.0001 per share issued and outstanding immediately prior to the Second Merger Effective Time shall be cancelled and extinguished without any conversion thereof. At the Second Merger Effective Time, each membership interest of Merger Sub II that is issued and outstanding immediately prior to the Second Merger Effective Time will constitute one membership interest of the Final Surviving Company. Such membership interest shall be the only membership interest of the Final Surviving Company that is issued and outstanding immediately after the Second Merger Effective Time.

(c) *Cancellation of Certain Company Stock.* Each share of Company Stock issued and outstanding immediately prior to the First Merger Effective Time that is held by the Company in treasury or owned by Acquiror or by the Merger Subs shall no longer be outstanding and shall be automatically cancelled and shall cease to exist (the "<u>Cancelled Shares</u>"), and no consideration shall be delivered in exchange therefor.

(d) Conversion of All Other Company Stock. Each share of Company Common Stock issued and outstanding immediately prior to the First Merger Effective Time, other than any Cancelled Shares and Dissenting Shares, shall be automatically converted into the right to receive the number of shares of duly authorized, validly issued, fully paid and nonassessable shares of Acquiror Common Stock equal to the Exchange Ratio (the "<u>Closing Merger Consideration</u>").

3.02 <u>Effect on Company Equity Awards</u>. At the First Merger Effective Time, each Company Option, whether vested or unvested, that is then outstanding and unexercised (each such Company Option, an "<u>Assumed Option</u>") shall cease to represent a right to acquire shares of Company Common Stock and shall be assumed and converted, as of the First Merger Effective Time, into an option to purchase a number of shares of Acquiror Common Stock equal to the Exchange Ratio, multiplied by the number of shares of Company Common Stock subject to such Company Option as of immediately prior to the First Merger Effective Time rounded down to the nearest whole share. Each Assumed Option shall continue to be governed by the Company Stock Plan, which shall be assumed by Acquiror solely to govern and administer the Assumed Options, and shall be subject to the same terms and conditions (including with respect to the exercise price and any vesting or forfeiture and exercise provisions) that are otherwise the same as were applicable to such Company Option as of immediately prior to the First Effective Time.

3.03 Acquiror Equity Awards. Each Acquiror Option that has an exercise price that is greater than the closing price of the Acquiror Class A Common Stock on the trading day immediately preceding the Closing Date will be surrendered or cancelled for no consideration in accordance with the terms of the Acquiror Option & RSU Plans immediately prior to the First Merger Effective Time. Each other Acquiror Option shall remain outstanding and shall remain exercisable subject to the terms and conditions of the Acquiror Option award and the Acquiror Option & RSU Plans. Each Acquiror RSU will be settled in accordance with its terms, as amended, such that there shall be no outstanding Acquiror RSU awards (or obligations to deliver shares of Acquiror Common Stock with respect to previously outstanding and vested Acquiror RSU awards) as of the First Merger Effective Time. Effective as of immediately prior to the First Merger Effective Time, the Acquiror Option & RSU Plans will be terminated, except with respect to Acquiror Options that remain outstanding following the First Merger Effective Time and any shares that remain available for issuance under the Acquiror Option & RSU Plans immediately prior to the First Merger Effective Time shall become available for issuance under the Acquiror Equity Incentive Plan in accordance with Section 7.09 hereof.

3.04 Equitable Adjustments. If, between the date of this Agreement and the Closing, the outstanding shares of Company Stock or Acquiror Common Stock shall have been changed into a different number of shares or a different class or series, by reason of any stock dividend, subdivision, reclassification, recapitalization, split, change, combination or exchange of shares, or any similar event shall have occurred, then any number, value (including the Exchange Ratio) or amount contained herein which is based upon the number of shares of Company Stock or Acquiror Common Stock will be appropriately adjusted to provide to the holders of Company Stock and the holders of Acquiror Common Stock the same economic effect as contemplated by this Agreement; provided, however, that this Section 3.04 shall not be construed to permit Acquiror, the Company or Merger Subs to take any action with respect to their respective securities that is prohibited by the terms and conditions of this Agreement.

3.05 Exchange of Company Certificates and Company Book-Entry Shares.

(a) *Exchange Procedures.* At the Closing, the Company Stockholders holding shares of Company Common Stock shall deliver the certificates or book entry statements evidencing their right to shares of Company Common Stock (collectively, the "<u>Company Certificates</u>"), other than Cancelled Shares and Dissenting Shares, to Acquiror for cancellation, or in the case of a lost, stolen or destroyed Company Certificate, will deliver to Acquiror an affidavit (including an indemnity) in the manner provided in <u>Section 3.05(e)</u>, and, in exchange therefor, Acquiror shall instruct its transfer agent to issue the Closing Merger Consideration in book-entry form promptly thereafter (but in no event earlier than the First Merger Effective Time), and Company Certificates shall forthwith be cancelled. Until so surrendered, outstanding Company Certificates will be deemed, from and after the First Merger Effective Time, to evidence only the right to receive the applicable portion of the Closing Merger Consideration.

(b) Distributions With Respect to Unexchanged Shares. No dividends or other distributions declared or made after the date hereof with respect to Acquiror Common Stock with a record date after the First Merger Effective Time will be paid to the holders of any unsurrendered Company Certificates with respect to the Closing Merger Consideration to be issued upon surrender thereof until the holders of record of such Company Certificates shall surrender such certificates. Subject to applicable Law, following surrender of any such Company Certificates, Acquiror shall promptly deliver to the record holders thereof, without interest, the Closing Merger Consideration issued in exchange therefor in book-entry form and the amount of any such dividends or other distributions with a record date after the First Merger Effective Time theretofore paid with respect thereto.

(c) Transfers of Ownership. If Closing Merger Consideration is to be issued in book-entry form in a name other than that in which the Company Certificates surrendered in exchange therefor are registered, it will be a condition of the issuance thereof that the Company Certificates so surrendered will be properly endorsed and otherwise in proper form for transfer and that the persons requesting such exchange will have paid to Acquiror or any agent designated by it any transfer or other taxes required by reason of the issuance of the Closing Merger Consideration in any name other than that of the registered holder of the Company Certificates surrendered, or established to the reasonable satisfaction of Acquiror or any agent designated by it that such tax has been paid or is not payable.

(d) No Further Ownership Rights in Company Common Stock. All shares of Acquiror Common Stock issued in accordance with the terms hereof shall be deemed to have been issued in full satisfaction of all rights pertaining to the Company Common Stock and there shall be no further registration of transfers on the records of the Surviving Corporation of the shares of Company Common Stock that were outstanding immediately prior to the First Merger Effective Time. If, after the First Merger Effective Time, Company Certificates are presented to the Surviving Corporation for any reason, they shall be cancelled and exchanged as provided in this Section 3.05.

(e) Lost, Stolen or Destroyed Certificates. In the event that any Company Certificates shall have been lost, stolen or destroyed, Acquiror shall issue in exchange for such lost, stolen or destroyed Company Certificates, upon the making of an affidavit of that fact by the holder thereof, the aggregate Closing Merger Consideration into which the shares of Company Common Stock formerly represented by such Company Certificates was converted into and any dividends or distributions payable pursuant to <u>Section 3.05(b)</u>; provided, however, that, as a condition precedent to the delivery of such Closing Merger Consideration, the owner of such lost, stolen or destroyed Company Certificates shall indemnify Acquiror against any claim that may be made against Acquiror or the Surviving Corporation with respect to the Company Certificates alleged to have been lost, stolen or destroyed.

3.06 <u>Withholding</u>. Each of Acquiror, Merger Sub I, Merger Sub II, the Company, the Surviving Corporation, the Final Surviving Company and their respective Affiliates and agents (each, a "<u>Payor</u>") shall be entitled to deduct and withhold from any consideration or amounts otherwise deliverable or payable under this Agreement such amounts that any such Persons are required to deduct and withhold with respect to any of the deliveries and payments contemplated by this Agreement under the Code or any other applicable Tax Law. To the extent that a Payor withholds or deducts such amounts with respect to any Person, such withheld or deducted amounts shall be treated as having been paid to or on behalf of such Person in respect of which such withholding or deduction was made for all purposes.

3.07 <u>No Fractional Shares</u>. Notwithstanding anything to the contrary contained herein, no certificates or scrip representing fractional shares of Acquiror Common Stock shall be issued upon the conversion of Company Common Stock pursuant to <u>Section 3.01</u>, and such fractional share interests shall not entitle the owner thereof to vote or to any other rights of a holder of Acquiror Common Stock. If the aggregate Closing Merger Consideration payable to a holder of Company Common Stock is not a whole number, such holder of Company Common Stock shall receive (in lieu of a fractional share) such number of shares of Acquiror Common Stock equal to the aggregate Closing Merger Consideration to which such holder is entitled rounded (up or down) to the nearest whole share.

3.08 Dissenting Shares. Notwithstanding anything in this Agreement to the contrary, shares of Company Common Stock issued and outstanding immediately prior to the First Merger Effective Time and held by a Company Stockholder entitled to vote in respect of such shares of Company Common Stock who has not voted in favor of the Merger or consented thereto in writing pursuant to Section 228 of the DGCL and has properly demanded appraisal for such shares in accordance with, and who complies in all respects with, Section 262 of the DGCL (such shares, "Dissenting Shares"), shall not be converted into the right to receive the Closing Merger Consideration and shall instead be entitled to receive payment of the appraised value of such Dissenting Shares in accordance with the provisions of Section 262 of the DGCL. At the First Merger Effective Time, (a) all Dissenting Shares shall be cancelled, extinguished and cease to exist and (b) the holders of Dissenting Shares shall be entitled to only such rights as may be granted to him, her or it under the DGCL. If any holder of Dissenting Shares fails to perfect or otherwise fails to comply with the provision of Section 262 of the DGCL or otherwise effectively waives, withdraws or loses such holder's right to appraisal under Section 262 of the DGCL or a court of competent jurisdiction shall determine such holder is not entitled to the relief provided by Section 262 of the DGCL, then the right of such holder to be paid the appraised value of such Dissenting Shares under Section 262 of the DGCL shall cease and such Dissenting Shares shall be deemed to have been converted, as of the First Merger Effective Time, into, and to have become exchangeable for the right to receive the Closing Merger Consideration upon the surrender of such shares in accordance with this Article III and thereafter shall not be deemed to be Dissenting Shares. The Company shall give Acquiror reasonably prompt notice of any demands received by the Company for appraisal of shares of Company Stock, attempted withdrawals of such demands and any other instruments served pursuant to the DGCL and received by the Company relating to rights to be paid the appraised value of Dissenting Shares, and Acquiror shall have the right to participate in and direct all negotiations and proceedings with respect to such demands. Prior to the First Merger Effective Time, the Company shall not, except with the prior written consent of Acquiror (such consent not to be unreasonably withheld, conditioned or delayed), make any payment with respect to, or settle or compromise or offer to settle or compromise, any such demands or waive any failure to timely deliver a written demand for appraisal or otherwise comply with the provisions under Section 262 of the DGCL, or agree or commit to do any of the foregoing.

3.09 <u>Company Merger Consideration Schedule</u>. The Company shall deliver to Acquiror, no later than five Business Days prior to the Closing Date, a schedule (the "<u>Company Merger Consideration Schedule</u>") reflecting the calculation of the Closing Merger Consideration and the allocation of the Closing Merger Consideration among the Company Stockholders. Acquiror and its Representatives shall have a reasonable opportunity to review and to discuss with the Company and its Representatives the documentation provided pursuant to this <u>Section 3.09</u> and any relevant books and records of the Company. The Company and its Representatives shall reasonably assist Acquiror and its Representatives in their review of the documentation and shall consider in good faith Acquiror's comments to the Company Merger Consideration Schedule, and if any adjustments are so made to the Company Merger Consideration Schedule prior to the Closing, such adjusted Company Merger Consideration Schedule shall thereafter become the Company Merger Consideration Schedule for all purposes of this Agreement. The Company Merger Consideration Schedule and the determinations contained therein shall be prepared in accordance with the applicable definitions contained in this Agreement. Acquiror will be entitled to rely in all respects upon the Company Merger Consideration Schedule.

3.10 <u>No Liability</u>. Neither Acquiror's transfer agent nor Acquiror shall be liable to any holder of Company Stock or Acquiror Common Stock for any such Company Stock or Acquiror Common Stock, as applicable (or dividends or distributions with respect thereto), or cash delivered to a public official pursuant to any abandoned property, escheat or similar Law.

ARTICLE IV REPRESENTATIONS AND WARRANTIES OF THE COMPANY

Except as set forth in the Schedules to this Agreement (each of which qualifies (a) the correspondingly numbered representation, warranty or covenant if specified therein and (b) such other representations, warranties or covenants where its relevance as an exception to (or disclosure for purposes of) such other representation, warranty or covenant is reasonably apparent on its face), the Company represents and warrants to Acquiror and the Merger Subs as of the date hereof and as of the Closing Date (or if a specific date is indicated in any such statement, as of such specified date) as follows:

4.01 <u>Organization, Standing and Corporate Power</u>. The Company is an entity validly existing and in good standing under the DGCL and has all requisite corporate power and authority to carry on its business as now being conducted in all material respects. The Company is duly qualified or licensed to do business and is in good standing in each jurisdiction in which the nature of its business or the ownership, leasing or operation of its properties, assets or rights makes such qualification or licensing necessary, except as would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect on the Company. The Company Organizational Documents that have been made available to Acquiror are true, correct and complete and are in effect as of the date of this Agreement and the Company is not in default under or in violation of any provision thereunder.

4.02 Corporate Authority; Approval; Non-Contravention.

(a) Except for the Company Stockholder Approvals, the Company has all requisite corporate power and authority and has taken all corporate or other legal entity action necessary in order to execute, deliver and perform its obligations under this Agreement and the Ancillary Agreements to which it is a party and, subject to satisfaction of the conditions to Closing contemplated hereby, to consummate the Transactions. The execution, delivery and performance by the Company of this Agreement and the Ancillary Agreements to which it is a party, and the consummation by it of the Transactions, have been duly and validly authorized by all necessary corporate consent and authorizations on the part of the Company, and no other corporate actions on the part of the Company is necessary to authorize the execution and delivery by the Company of this Agreement, the Ancillary Agreements to which it is a party and the consummation by it of the Transactions, in each case, subject to receipt of the Company Stockholder Approvals. This Agreement has been duly executed and delivered by the Company and, assuming due authorization, execution and delivery hereof by the other parties, is a legal, valid and binding obligation of the Company, enforceable against the Company in accordance with its terms (subject to applicable bankruptcy, solvency, fraudulent transfer, reorganization, moratorium and other Laws affecting creditors' rights generally from time to time in effect and by general principles of equity (the "Enforceability Exceptions")).

(b) Except as set forth on <u>Schedule 4.02(b)</u>, the execution, delivery and, subject to receipt of the Company Stockholder Approvals, performance of this Agreement and the Ancillary Agreements to which the Company is a party, and the consummation of the Transactions, do not, and will not, constitute or result in (i) a breach or violation of, or a default under, the Company Organizational Documents, (ii) to the Knowledge of the Company, conflict with or violate any Law applicable to the Company or by which any property or asset of the Company is bound or affected and (iii) with or without notice, lapse of time or both, a breach or violation of, a termination (or right of termination) of or default or change of control under, the creation or acceleration of any obligations under or the creation of a Lien (other than a Permitted Lien) on any of the assets of the Company pursuant to, any Company Material Contract or, assuming (solely with respect to performance of this Agreement and consummation of the Transactions) compliance with the matters referred to in <u>Section 4.02(a)</u>, under any Law to which the Company or any of its Affiliates is subject, except (in the case of clause (ii) above) for such violations of Laws, breaches, or defaults that would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect on the Company.

(c) The Company Support Agreement will have been duly executed and delivered by the Company and the Company Stockholder party thereto, and will constitute a legal, valid and binding obligation of the Company, enforceable against the parties thereto in accordance with their terms (subject to the Enforceability Exceptions). The Company Stockholder that will be party to the Company Support Agreement holds Company Stock representing the voting power sufficient as of the date of such Company Support Agreement to obtain the Company Stockholder Approvals.

4.03 <u>Governmental Approvals</u>. No consent of, or registration, declaration, notice or filing with, any Governmental Authority is required by or with respect to the Company in connection with the execution and delivery by the Company of this Agreement or the consummation by the Company of the Transactions, except for (a) such other consents, registrations, declarations, notices and filings which, if not obtained or made, would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect on the Company, and (b) the filing of the Certificate of Merger with the Secretary of State of the State of Delaware.

4.04 <u>Capitalization</u>.

(a) As of the date hereof, the authorized capital stock of the Company consisted of 58,687,483,705 shares of Company Common Stock, of which 2,648,364,984 shares of Company Common Stock were outstanding. All of the issued Company Stock have been duly authorized and are validly issued, fully paid and nonassessable. Set forth on <u>Schedule 4.04</u> is a true, correct and complete list of each holder of issued and outstanding capital stock or other equity securities (including any securities convertible into equity securities, including the Company Existing Notes and Company Warrants) of the Company and the number of shares or other equity interests held by each such holder as of the date hereof. Each of the outstanding shares of capital stock of the Company (i) was issued in compliance in all material respects with applicable Laws and in compliance with the Company Organizational Documents, (ii) was not issued in breach or violation of any preemptive rights or Contract to which the Company is a party, and (iii) is owned free and clear of any Lien imposed by or resulting from any Contract to which the Company is a party (other than the Company Organizational Documents and Contracts that have been provided to Acquiror that set forth the Company Stockholders' obligations to the Company).

(b) Except as set forth on <u>Schedule 4.04</u>, there are no other outstanding equity interest in the Company, including any warrants, options, calls, pre-emptive rights, subscriptions, "phantom" stock rights or other rights, agreements, arrangements, convertible or exchangeable securities or other commitments (other than this Agreement) pursuant to which the Company is obligated to issue, transfer, sell, purchase, return or redeem or cause to be issued, transferred, sold, purchased, returned or redeemed any equity securities of the Company. As of the Closing, after giving effect to the Company Conversion and the assumption of the Company Options, except for Company Common Stock, there will be no other outstanding equity interest in the Company, including any warrants, options, calls, pre-emptive rights, subscriptions, "phantom" stock rights or other rights, agreements, arrangements, convertible or exchangeable securities or other commitments (other than this Agreement) pursuant to which the Company Stock, there will be no other rights, agreements, arrangements, convertible or exchangeable securities or other commitments (other than this Agreement) pursuant to which the Company is obligated to issue, transfer, sell, purchase, return or redeem or cause to be issued, transferred, sold, purchased, returned or redeemed any equity securities of the Company is obligated to issue, transfer, sell, purchase, return or redeem or cause to be issued, transferred, sold, purchased, returned or redeemed any equity securities of the Company.

(c) <u>Schedule 4.04(c)</u> sets forth a true and correct list, as of the date hereof, of (i) each award of Company Options, (ii) the name of each holder of Company Options, (iii) the number of shares of Company Common Stock underlying each award of Company Options, (iv) the date on which the award of Company Options was granted, (v) the vesting schedule of each award of Company Options and (vi) the exercise price and expiration date of each award of Company Option. Other than set forth on <u>Schedule 4.05</u>, the Company has no equity interest in, nor has it agreed to acquire, any share capital or other equity security of any other company (wherever incorporated).

4.05 <u>Subsidiaries</u>. The Company has no Subsidiaries or other ownership, or right to acquire, capital stock, security, partnership interest or other equity interest of any kind in any Person.

4.06 Financial Statements; Internal Controls.

(a) The audited statements of total assets, total liabilities and shareholders' equity and total comprehensive income for the fiscal years ended December 31, 2021, 2022 and 2023 (collectively, the "<u>Company Audited Financial Statements</u>") were prepared in accordance with the standards, principles and practices specified therein and, subject thereto, in accordance with GAAP, the standards of the American Institute of Certified Public Accountants and applicable Law throughout and among the periods involved, except as otherwise noted therein. Prior to the date hereof, true, complete and correct copies of the Company Audited Financial Statements and the accompanying independent auditors' reports, if and as applicable have been made available to Acquiror.

(b) The Company Audited Financial Statements are true, complete and correct in all material respects and were derived from the books and records of the Company and prepared in accordance with GAAP (except as may be indicated in the notes thereto). The Company Interim Financial Statements to be delivered pursuant to Section 6.11 hereof, when delivered, will be true, complete and correct in all material respects and derived from the books and records of the Company and prepared in accordance with GAAP (except as may be indicated in the notes thereto), subject to normal and recurring year-end adjustments and the absence of notes, none of which will be material individually or in the aggregate. The Company Audited Financial Statements fairly present, and the Company Interim Financial Statements when delivered will fairly present, in all material respects the assets, liabilities, cash flow and financial condition and results of operations of the Company as of the times and for the periods referred to therein. Since the Balance Sheet Date, the Company has not made any material change in the accounting practices or policies applied in the preparation of the Company Audited Financial Statements, except as required by applicable Law or GAAP.

(c) The Company maintains a standard system of accounting established and administered in accordance with GAAP applied on a consistent basis throughout the periods indicated.

(d) Except as set forth in <u>Schedule 4.06(d)</u>, the Company maintains a system of internal accounting controls sufficient to provide reasonable assurance that (i) transactions are executed in accordance with management's general or specific authorizations, (ii) transactions are recorded as necessary to permit preparation of financial statements in conformity with GAAP and to maintain asset accountability, (iii) access to assets is permitted only in accordance with management's general or specific authorization, and (iv) the recorded accountability for assets is compared with the existing assets at reasonable intervals and appropriate action is taken with respect to any differences.

4.07 <u>Compliance with Laws</u>. Except as would not reasonably be expected to have a Material Adverse Effect on the Company:

(a) The Company is conducting and has, since March 4, 2020, conducted its business in compliance with all Laws applicable to it and the Company's business, properties, rights or other assets.

(b) There is no, and since March 4, 2020 there has been no, Action by or against the Company pending or threatened in writing, nor has any Governmental Authority indicated in writing to the Company an intention to initiate or conduct the same.

(c) Since March 4, 2020, the Company has not received any written notice from any Governmental Authority (i) with respect to an alleged, actual or potential violation and/or failure to comply with any such applicable Law or (ii) requiring the Company to take or omit any action to ensure compliance with any such applicable Law.

(d) The Company possesses all governmental permits, approvals, orders, authorizations, consents, licenses, certificates, franchises, exemptions or waivers of, or filings, notices, or registrations with, or issued by, any Governmental Authority necessary for the ownership and use of the assets of the Company and the operation of the Company's business as currently conducted, (the "Company Permits"). All such Company Permits are valid and in full force and effect, and there are no lawsuits or other proceedings pending or threatened before any Governmental Authority that seek the revocation, cancellation, suspension or adverse modification thereof. The Company is not in default in any respect of, and no condition exists that with notice or lapse of time or both would constitute a material default under, the Company Permits.

4.08 <u>Absence of Certain Changes or Events</u>. (a) Since the Balance Sheet Date and except as expressly required or permitted by this Agreement, (i) the Company has conducted its business in all material respects in the ordinary course of business and (ii) except as disclosed on <u>Schedule 4.08</u>, no action has been taken by the Company that would require consent under <u>Section 6.01</u> if such action were taken after signing of this Agreement and prior to Closing (other than for any such actions for which such consent has been received in accordance with <u>Section 6.01</u>) and (b) as of the date hereof, there has not been any change, effect, event, circumstance, occurrence or state of facts that would, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect on the Company.

4.09 <u>No Undisclosed Liabilities</u>. Except (a) as disclosed, reflected or reserved against in the balance sheet dated as of December 31, 2023 included within the Company Audited Financial Statements, (b) for liabilities incurred in the ordinary course of business since the Balance Sheet Date, (c) as expressly permitted or contemplated by this Agreement or otherwise incurred in connection with the Transactions, (d) as disclosed on <u>Schedule 4.09</u>, (e) contingent liabilities under executory contracts and (f) for liabilities that have been discharged or paid in full in the ordinary course of business, as of the date hereof, the Company does not have any liabilities of the type required by GAAP to be disclosed or reserved for on a balance sheet of the Company.

4.10 <u>Information Supplied</u>. The information supplied in writing by the Company for inclusion in the Registration Statement and the Proxy Statement will not (a) in the case of the Registration Statement, at the time the Registration Statement is declared effective under the Securities Act and (b) in the case of the Proxy Statement, as of the date the Proxy Statement is first mailed to the Acquiror Stockholders and at the time of any meeting of the Acquiror Stockholders to be held in connection with the Transactions, contain any untrue statement of a material fact, or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not false or misleading. Notwithstanding the foregoing sentence, the Company makes no representation or warranty or covenant with respect to: (i) statements made or incorporated by reference therein in any of the foregoing documents based on information supplied by Acquiror for inclusion therein or (ii) any projections or forecasts or forward looking statements included in the Registration Statement or Proxy Statement.

4.11 Litigation.

(a) The Company is not and, to the Knowledge of the Company, none of its officers, directors, agents or employees, in their capacities as such, is the subject of or engaged in any material Action or other dispute resolution process before a third party, whether as claimant, defendant or otherwise, and no such Action or dispute resolution process is pending or threatened in writing on the date hereof. As of the date hereof, the Company is not and, to the Knowledge of the Company, none of its officers, directors, agents or employees, in their capacities as such, is subject to any settlement agreements or arrangements, whether written or oral, or is in discussions for a settlement or arrangement, regarding any material Actions.

(b) The Company is not a party to or subject to the provisions of any outstanding material Governmental Order (except if generally applicable without the Company being named therein).

4.12 <u>Contracts</u>.

(a) <u>Schedule 4.12(a)</u> sets forth a true and complete list as of the date hereof, of the following Contracts that are effective as of the date hereof and to which the Company is a party or is bound (all such Contracts set forth on <u>Schedule 4.12(a)</u>, or which are required to be so disclosed, the "<u>Company Material Contracts</u>"):

(i) all such Contracts with third party manufacturers and suppliers for the manufacture and supply of products providing for minimum order quantities, minimum purchase requirements, exclusive supply, or manufacturing or purchase requirements with a total annual payment or financial commitment exceeding \$500,000 on an annual basis;

(ii) all such Contracts with (or with obligations of the Company to) a Related Party;

(iii) all such Contracts that contain any covenant materially limiting or prohibiting the right of the Company (A) to engage in any line of business or conduct business in any geographic area, (B) to distribute or offer any products or services, or (C) to compete with any other person in any line of business or in any geographic area or levying a fine, charge or other payment for doing any of the foregoing;

(iv) all such Contracts in which the aggregate outstanding expenditure or payment obligations of the Company exceeds \$1,000,000, excluding obligations that are contingent liabilities in respect of a breach or indemnification obligation or similar contingent obligation as a result of a breach or default;

\$250,000;

(v)

any Contract for the use by Company of any tangible property where the annual lease payments are greater than

(vi) any partnership, joint venture or other similar agreement or arrangement providing for the formation, creation, operation, management or control of any partnership or joint venture with a third party to which the Company is a party, other than *bona fide* customer-supplier relationships or a trade association;

(vii) all such Contracts providing for the acquisition or disposition of any business, equity interests or material assets (whether by merger, sale of stock, sale of assets or otherwise) pursuant to which the Company has any ongoing obligation (including for deferred purchase price obligations, earn-out obligations, indemnification obligations and other contingent liabilities (including payment obligations in respect of the future utilization of any net operating losses));

(viii) all such Contracts that obligate the Company to make any loans, advances or capital contributions to, or investments in, any Person (other than advances to employees for business expenses in the ordinary course of business consistent with past practice);

(ix) any note, mortgage, indenture or other obligation or agreement or other instrument for or relating to indebtedness for borrowed money in excess of \$250,000, or any guarantee of third party obligations in excess of \$250,000, or any letters of credit, performance bonds or other credit support for the Company;

(x) any Contract that creates guarantees or Liens of any nature on the Company's assets not in the ordinary course of business and in an amount equal to or greater than \$250,000;

(xi) any collective bargaining agreement or other Contract with any labor or trade union, works council, or other labor organization (each a "<u>CBA</u>");

(xii) all such Contracts or agreements with any employee, officer, director or other individual service provider that (A) provide for annual compensation in excess of \$200,000, (B) are not terminable by the Company on no more than thirty (30) days' notice and without liability or financial obligation to the Company other than accrued compensation and other payments required by Law or (C) provide for sale, change in control, severance, retention, transaction or similar bonuses, or any other compensation or benefits triggered or accelerated in connection with or otherwise related to the consummation of the transaction contemplated hereby, whether alone or in connection with any other event, that are (or may become) payable to any current or former employee or other service provider of the Company;

(xiii) all Contracts entered into to settle or resolve any material Intellectual Property-related dispute, including settlement agreements, covenants not to sue, consent agreements, and co-existence agreements;

(xiv) all Contracts providing for the development of any material Intellectual Property, independently or jointly, by or for the Company, other than Contracts entered into pursuant to a form employee or independent contractor invention assignment agreement between the Company and an employee or independent contractor of the Company regarding the development of Intellectual Property by such employee or independent contractor; and

(xv) all Contracts pursuant to which the Company (A) grants or (B) is granted a license (whether or not such license is currently exercisable) to, or other rights under, any Intellectual Property, excluding, in the case of (A), Permitted Liens, and in the case of (B), any Commercially Available Software.

(b) The Company (i) is not, nor has it received written notice that any other party to any Material Contract is, in material violation or material breach of or material default (immediately or upon notice or lapse of time) under or (ii) has not waived or failed to enforce any material rights or material benefits under any Material Contract to which it is a party or any of its properties, rights or other assets is subject. No Material Contract is the subject of a notice to terminate, except for any expiration of the term of a Material Contract following the date of this Agreement in accordance with its terms. Each Material Contract is in full force and effect and, subject to the Enforceability Exceptions, is legal, valid and binding on the Company, and, to the Knowledge of the Company, each other party thereto, except as would not reasonably be expected to have a Material Adverse Effect on the Company. There is no default under any such Material Contracts by the Company, or, to the Knowledge of the Company, or, to the Knowledge of the a default thereunder by the Company, or, to the Knowledge of the Company.

4.13 Employment Matters.

(a) Except as set forth on <u>Schedule 4.13(a)</u>: (i) the Company is not a party to or bound by any CBA (including agreements with works councils and trade unions and side letters), nor is it negotiating or required to negotiate a CBA; (ii) no employees of the Company are represented by any labor or trade union, works council, or other labor organization with respect to their employment; (iii) in the past three (3) years, no labor or trade union, works council, other labor organization, or group of employees of the Company has made a demand for recognition or certification, and there are no representation or certification proceedings presently pending or threatened to be brought or filed with the National Labor Relations Board or any other labor relations tribunal or authority; (iv) in the past three (3) years, there have been no actual or threatened organizing activities with respect to any employees of the Company and no such activities are currently pending or threatened; (v) in the past three (3) years, there has been no actual or threatened strike, lockout, work stoppage, slowdown, picketing, hand billing, unfair labor practice charge, material labor grievance, material labor arbitration or other material labor dispute against or affecting the Company, and no such dispute is currently pending or threatened; and (vi) with respect to the Transactions, the Company has satisfied all notice, bargaining, consent, consultation or other obligations to its employees and employees' Representatives under applicable Law and any CBA.

(b) Over the past three (3) years (i) no employee of the Company has made allegations of sexual harassment, sexual misconduct or other harassment against any officer, director or key employee of the Company, and (ii) the Company has not entered into any settlement agreement related to sexual harassment, sexual misconduct or other harassment by or against any current or former employee, independent contractor, director or officer of the Company.

(c) The Company is not a party to nor has it received an active written threat or, an oral threat, of any Action brought by on or on behalf of any applicant, any former or current employee, any former or current individual or sole proprietor independent contractor relating to any labor or employment matters of the Company.

(d) The Company is, and for the past three (3) years has been, in compliance with all Laws relating to labor and/or employment, including Laws regarding terms and conditions of employment, hiring, background checks, trainings, notices, immigration, authorization to work, health and safety, wages, hours, employee classification (for overtime purposes or as employee versus independent contractor), overtime (including the proper determination of regular pay and the treatment of bonuses), meal and rest periods, harassment, discrimination, retaliation, accommodations, disability rights or benefits, plant closings and mass layoffs, workers' compensation, labor relations, leaves of absences, time off, COVID-19, affirmative action, unemployment insurance and/or termination of employment other than any failure to so comply not reasonably likely to result in material liability to the Company. The Company has no actual or contingent liability with respect to (i) any misclassification of any person as an independent contractor rather than as an employee, as an employee rather than as an independent contractor, or as a non-employee when in fact employed, (ii) any employee or contractor leased from or staffed by another employer, or (iii) any person currently or formerly classified as exempt from, or otherwise not paid where required, overtime and minimum or other wages.

(e) The Company has not, in the past three (3) years, experienced or implemented a "plant closing" or "mass layoff" as defined in the Worker Adjustment and Retraining Notification Act or any similar foreign, state or local Law (each a "<u>WARN Act</u>") or other event requiring notice under a WARN Act. In the six (6) month period immediately prior to the date hereof, the Company has not carried out any temporary layoffs, furloughs, or material reductions in hours of work that, if continued for six (6) months, alone or in the aggregate with any other "employment loss" (as defined under any WARN Act), could reasonably be expected to constitute a "plant closing" or "mass layoff" under any WARN Act.

4.14 Employee Benefits.

(a) <u>Schedule 4.14(a)</u> contains a true, correct and complete list of each Company Benefit Plan. The Company has provided to Acquiror with respect to each Company Benefit Plan, as applicable, true, correct and complete copies of: (i) such Company Benefit Plan, or with respect to any unwritten Company Benefit Plan, a written description of all material terms thereof, and form of any award documentation thereunder; (ii) the most recent determination, opinion or advisory letter received from the Internal Revenue Service, if any; (iii) the Form 5500 annual report and the financial statements and actuarial summary or other valuation reports prepared with respect thereto in the past three (3) years; (iv) each trust, insurance, annuity, stop-loss or other funding Contract or arrangement related thereto; (v) the most recent summary plan description and any summaries of material modification with respect thereto; (vi) discrimination or similar testing (and evidence of corrective action related thereto) for the past three (3) years; and (vii) all non-routine correspondence or communications relating thereto with any Governmental Authority or participant within the past three (3) years.

(b) With respect to each Company Benefit Plan: (i) such Company Benefit Plan has been maintained, operated and administered in compliance in all material respects with its terms and with the requirements of applicable Law, including the Code and ERISA; (ii) all contributions or payments due to date have been made timely and in compliance with the terms of such Company Benefit Plan and applicable Law, or if not yet due, properly accrued (to the extent required or appropriate to be accrued in accordance with GAAP or otherwise in the ordinary course of business); (iii) no failure to pay premiums due or payable with respect to insurance policies relating to such Company Benefit Plan has resulted in default under any such policies; and (iv) there are no proceedings pending or, to the Knowledge of the Company, threatened against or involving such Company Benefit Plan brought by or on behalf of any current or former employee or other service provider of the Company (or dependent or beneficiary thereof) or Governmental Authority (other than routine claims for benefits made in the ordinary course of business), in each case, except as could not reasonably be expected to result in material liability to the Company. Each Company Benefit Plan that is intended to be qualified under Section 401(a) of the Code is so qualified and has received a favorable determination letter from the Internal Revenue Service or, with respect to a prototype or volume submitter plan, can rely on an opinion or advisory letter from the Internal Revenue Service to the prototype plan or volume submitter plan sponsor, as to its qualification and to the effect that the plan's related trust is exempt from federal income taxes under Section 501(a) of the Code, and nothing has occurred that could reasonably be expected to result in the revocation of such favorable determination.

(c) Neither the Company nor any ERISA Affiliate maintains, sponsors, contributes to or been obligated to contribute to (or has ever maintained, sponsored or contributed to or been obligated to contribute to) or has any obligation with respect to, (i) any employee benefit plan that is subject to Title IV of ERISA, (ii) any Multiemployer Plan, (iii) any multiple employer plan (as defined in Section 413(c) of the Code), (iv) any voluntary employees' beneficiary association described under Section 501(c)(9) of the Code or any "funded welfare plan" within the meaning of Section 419 of the Code or (v) any multiple employer welfare arrangement (as defined under Section 3(40) of ERISA).

(d) The Company has no actual or contingent liability under or with respect to COBRA for failure to comply with its notice or coverage requirements or for failure to comply with the Patient Protection and Affordable Care Act, as amended by the Health Care and Education Reconciliation Act ("PPACA"), and any other comparable domestic or foreign Laws, including all requirements relating to eligibility waiting periods and the offer of or provision of minimum essential coverage that is compliant with Section 36B(c)(2)(C) of the Code. No excise Tax or penalty under PPACA, including Sections 4980D and 4980H of the Code is outstanding, has accrued, or has arisen with respect to any period prior to the Closing, with respect to any Company Benefit Plan.

(e) Each Company Benefit Plan that provides deferred compensation subject to Section 409A of the Code satisfies and has satisfied, in form and operation, the requirements of Section 409A of the Code and the guidance thereunder. No current or former employee or other service provider of the Company is entitled to receive any gross-up or additional payment in connection with any Tax (including those imposed under Section 409A or Section 4999 of the Code).

(f) No Company Benefit Plan, fiduciary of such plan or administrator of such plan has taken any action, or failed to take any action, which action or failure could subject Acquiror or any of its Affiliates, or any current or former employee or other service provider of the Company, to any material Tax, fine, Lien, penalty or other obligation imposed by ERISA, the Code or other applicable Laws, with respect to or in connection with any Company Benefit Plan.

(g) The Company does not provide (and has not at any time provided) or have any obligation to provide post-employment health, life or other welfare benefits, other than as required under Section 4980B of the Code (the full cost of which is borne by the applicable recipient of such benefits).

(h) Neither the execution of this Agreement nor the consummation of the Transactions, either alone or in combination with the occurrence of any other event, would be reasonably likely to result in: (i) any material payment or benefit becoming due to, or a material increase in, or acceleration of the timing of payment, funding or vesting of, the compensation or benefits of, any current or former employee, officer, director or other individual service provider of the Company; (ii) the receipt (whether in cash, property or the vesting of property) by any "disqualified individual" of any "parachute payment" (as such terms are defined in Section 280G of the Code); or (iii) any breach or violation of or default under or, except pursuant to applicable Law, limitation of the rights of the Company, Acquiror or their respective Affiliates to amend, modify or terminate any Company Benefit Plan.

4.15 <u>Taxes</u>.

(a) The Company has filed with the appropriate Tax Authority, or has caused to be filed on its behalf (taking into account any valid extension of time within which to file), all material Tax Returns required to be filed by it, and all such Tax Returns were and are true, correct and complete in all material respects. The Company has paid all Taxes due and payable (whether or not shown on any Tax Return). No claim has ever been made in writing by a Tax Authority in a jurisdiction where the Company does not file a Tax Return that such entity is or may be subject to taxation by that jurisdiction in respect of Taxes that would be covered by or the subject of such Tax Return.

(b) The Company has (i) withheld all Taxes required to have been withheld by it in connection with amounts paid to any employee, independent contractor, creditor, stockholder or any other third party, and (ii) remitted such amounts required to have been remitted to the appropriate Tax Authority, except where such failure would not reasonably be expected to result in a material liability.

(c) No claim, assessment, deficiency or proposed adjustment for any amount of Tax has been asserted or assessed in writing by any Tax Authority against the Company that remains unresolved or unpaid. There is no Tax audit or other examination of the Company presently in progress, and there are no waivers, extensions or requests for any waivers or extensions of any statute of limitations currently in effect with respect to any Taxes or Tax Returns of the Company, nor has any request been made in writing for any such extension or waiver.

(d) The Company is not and has not been (i) a party to any Tax sharing, indemnification, allocation or similar agreement or arrangement (excluding any commercial Contract entered into in the ordinary course of business and not primarily related to Taxes), (ii) a member of an affiliated, consolidated, combined, unitary or similar Tax group for income tax purposes (other than any such Tax group the common parent of which was the Company), or (iii) a party to any "listed transaction" under Treasury Regulations Section 1.6011-4(b)(2) (or any similar or corresponding provision of state, local or foreign Law).

(e) The Company has no liability for Taxes of any other Person (other than any such Tax group the common parent of which is the Company) as a result of Treasury Regulations Section 1.1502-6 (or any similar provision of state, local or foreign Law), as a transferee or successor, by operation of Law, by Contract or otherwise (excluding, in each case, any commercial Contract entered into in the ordinary course of business and not primarily related to Taxes).

(f) The Company has not distributed stock of another Person, or had its stock distributed by another Person, in a transaction that was purported or intended to be governed in whole or in part by Section 355 of the Code (or so much of Section 356 of the Code as relates to Section 355 of the Code).

The Company is in compliance in all material respects with all applicable transfer pricing Laws and regulations. All intercompany (g) agreements have been adequately documented, and such documents have been duly executed in a timely manner. The prices and terms for the provision of any property or services by or to the Company are arm's length for purposes of the relevant transfer pricing Laws, and all related documentation required by such Laws has been timely prepared or obtained and, if necessary, retained.

(h) The Company is not a partner for Tax purposes with respect to any joint venture, partnership or other arrangement or Contract which is treated as a partnership for Tax purposes.

The Company has not (i) been a United States real property holding corporation within the meaning of Section 897(c)(2) of the (i) Code during the past five years; (ii) been a "personal holding company" as defined in Section 542 of the Code (or any similar provision of state, local or foreign Law); or (iii) become subject to Tax in a country other than the country of its formation by reason of being engaged in a trade or business, having a permanent establishment (within the meaning of an applicable Tax treaty), or having employees in such country.

There are no Liens with respect to Taxes on any of the assets of the Company, other than Liens for Taxes which are not yet due and

payable.

(k) The Company will not be required to include any item of income in, or exclude any material item of deduction from, taxable income for any period (or any portion thereof) ending after the Closing Date as a result of any installment sale, open transaction or other transaction on or prior to the Closing Date, any accounting method change, improper use of an accounting method or agreement with any Tax Authority filed or made on or prior to the Closing Date or any material deferred revenue or prepaid amount received on or prior to the Closing.

> (1)The Company has complied in all material respects with all escheat and unclaimed property Laws.

(m) The Company has not taken or agreed to take any action (nor to the Knowledge of the Company is there any fact or circumstance) that would prevent the Merger from qualifying as a "reorganization" within the meaning of Section 368(a) of the Code.

4.16 Intellectual Property.

(j)

Schedule 4.16(a) contains a complete and accurate list of all (i) issued patents and pending patent applications, (ii) trademark and (a) service mark registrations and pending applications, (iii) copyright registrations and pending applications and (iv) internet domain names, in each case that are owned by the Company or that the Company purports to own (collectively, "Company Registered IP"), indicating for each item, as applicable, the registration or application number, the applicable filing jurisdiction and the date of filing or issuance and, with respect to domain names, the applicable domain name registrar. Each item of Company Registered IP is subsisting and unexpired and to the Knowledge of the Company, valid and enforceable. The Company solely and exclusively owns, free and clear of all Liens (other than Permitted Liens), all right, title and interest in and to the Owned Company Intellectual Property.

(b) (i) The operation of the business of the Company as currently conducted does not infringe, misappropriate, dilute or otherwise violate, and since March 4, 2020, has not infringed, misappropriated, diluted or otherwise violated, any third-party Intellectual Property (provided that the foregoing representation 4.16(b)(i) is being made to the Knowledge of the Company in respect of patents and patent applications) and (ii) to the Knowledge of the Company: no third party is infringing, misappropriating, diluting or otherwise violating, and no third party has infringed, misappropriated, diluted or otherwise violated since March 4, 2020, any Owned Company Intellectual Property. The Company has not received any written charge, complaint, claim, demand or notice alleging that the conduct of the business infringes, misappropriates, dilutes or otherwise violates any Intellectual Property of any other Person.

(c) There is no, and since March 4, 2020 there has not been any, Action pending or threatened in writing, or, to the Knowledge of the Company, orally (including "cease and desist" letters or invitations to take a license) (i) against the Company (A) challenging the ownership, validity, registrability, patentability, or enforceability of the Owned Company Intellectual Property (excluding ordinary course office actions and similar ex-parte proceedings in connection with the prosecution of applications for the registration or issuance of any Intellectual Property) or (B) asserting that the Company has infringed, misappropriated, diluted or otherwise violated any third-party Intellectual Property of any third party (excluding ordinary course office actions and similar ex-parte proceedings in connection with the prosecution with the prosecution of applications for the registration or issuance of any third party (excluding ordinary course office actions and similar ex-parte proceedings in connection with the prosecution of applications for the registration or issuance of any third party (excluding ordinary course office actions and similar ex-parte proceedings in connection with the prosecution of applications for the registration or issuance of any Intellectual Property or (B) asserting that any third party has infringed, misappropriated, diluted or otherwise violated any Owned Company Intellectual Property.

(d) The Company has taken commercially reasonable measures to protect, enforce and maintain (i) the confidentiality of all material proprietary information and trade secrets included in the Owned Company Intellectual Property or of third parties provided to the Company under obligations of confidentiality, and (ii) its ownership of, and rights in, all Company Intellectual Property. Without limiting the foregoing, the Company has not made any of its trade secrets or other material confidential or proprietary information that it intended to maintain as confidential information (including source code of any Company Software) available to any other Person except pursuant to written binding agreements requiring such Person to maintain the confidentiality of such confidential information or trade secrets.

4.17 <u>Data Protection</u>. Except as would not reasonably be expected to have a Material Adverse Effect on the Company:

(a) The Company is, and at all times since March 4, 2020 has been, (i) in material compliance with all Privacy Laws, Data Security Requirements, and the Company's contractual requirements relating to the Company's Processing of Personal Information (collectively, the "<u>Company</u> <u>Privacy Commitments</u>"), and (ii) has not been subject to any Actions, regulatory audits or investigations by any Person or Governmental Authority relating to Company Privacy Commitments. The Company has appointed a data protection officer if required to do so under Privacy Laws, and has complied at all times with and made privacy notices (complete and accurate copies of which have been made available to the Acquiror) available to all individuals about whom the Company Processes or directs the Processing of Personal Information, in conformance with Company Privacy Commitments, including with respect to the collection, processing, receipt, security, storage, transmission, transfer (including cross-border transfers), disclosure, destruction and use of Personal Information. The Company has taken commercially reasonable steps to ensure that all Personal Information is protected against loss or unauthorized Processing, including by maintaining and enforcing policies, procedures, and rules regarding data privacy, protection, and security to protect the security, integrity and privacy of Personal Information as required by all Company Privacy Commitments.

(b) Since March 4, 2020, (i) to the Company's knowledge, there has not been any breach, security incident, loss, theft, modification, destruction, alteration, disclosure or other misuse of, or third party claims, complaints, fines, practice recommendation, request, or proceeding of any nature alleging, unauthorized access or unauthorized use of any Personal Information or confidential business information maintained by any Company Data Related Vendor on behalf of the Company; and (ii) the Company has not received any written requests, complaints or objections to its collection or Processing of Personal Information from any Governmental Authority or other third party (including data subjects). No Person has threatened in writing to bring or been awarded compensation from the Company under any Company Privacy Commitments, there is no basis for any such claim to be brought that may give rise or has given rise to any liability, and no written claim for such compensation is outstanding. The Company has complied with all data subject requests, including any requests for access to Personal Information, the cessation of specified processing activities or the rectification or erasure of any Personal Information, in each case in accordance with the requirements of applicable Privacy Laws.

(c) The Company does not sell, rent or otherwise makes available to any Personal Information, except in a manner that complies with all applicable Privacy Laws. The execution, delivery and performance of this Agreement and the transaction contemplated herein comply and will comply with all Company Privacy Commitments.

(d) The Company has entered into written contracts with each third-party service provider, vendor and business partner that has access (including storage) to Personal Information which complies with all applicable requirements of Company Privacy Commitments ("<u>Company Data Related Vendors</u>"), including payment processors, advertising and marketing agencies, cloud storage vendors and outsourced technology or human resource functions, containing all appropriate, sufficient and legally required contractual clauses regarding privacy and information security. The Company has made available to Acquiror true and accurate copies of all contracts with any Company Data Related Vendor.

(e) To the extent that the Company collects Personal Information of Persons who reside outside of the United States, the EU, and/or the UK, and/or Canada, respectively, the Company has implemented sufficient mechanisms to ensure that the transfers of such Personal Information from such Persons' home country to any other country complies with Privacy Laws restricting the transfer of such Personal Information.

4.18 <u>Information Technology and Cybersecurity</u>. Except as would not reasonably be expected to have a Material Adverse Effect on the Company:

(a) The IT Systems used in the conduct of the Company's business: (i) operate and perform in accordance with their documentation and functional specifications; (ii) are adequate for, and operate and perform in all material respects as required in connection with, the needs and operation of Company's business as currently conducted and are fulfilling the purposes for which they were acquired, licensed or established in an efficient manner without material downtime or errors; and (iii) are free from any bugs, computer virus, unauthorized disabling or erasing mechanism, worm, unauthorized software lock, drop dead device, Trojan horse, back door, trap door or time bomb or other software routines or hardware components that permit unauthorized access or the unauthorized disablement or erasure of the IT Systems. There has been no failure or other material substandard performance of any IT Systems which has caused any disruption to the business of the Company.

(b) The Company has implemented with respect to the IT Systems used in its business commercially reasonable (i) steps to provide for backup, security and disaster recovery technology consistent with industry practices and (ii) actions (including implementing and monitoring compliance with administrative, technical and physical safeguards, policies, procedures and security measures that conform with all applicable contractual obligations and Privacy Laws) to protect the operation, confidentiality, integrity and security of its IT Systems and the information (including trade secrets and Personal Information) and data stored thereon or transmitted thereby, including from unauthorized use or access by third parties and from viruses and contaminants. The Company has carried out and documented reasonable security and technology audits, investigations, data mapping exercises, impact assessments, and other activities relating to the Company's information security practices, and no such audits, investigations, assessments and/or reports have identified material risks and/or vulnerabilities in the Company IT Systems.

(c) There has been no security breach of, or unauthorized use or access to, the Company IT Systems (including ransomware attacks), including any breach, which resulted in the unauthorized use, misappropriation, modification, encryption, corruption, disclosure or transfer of any information or data contained therein or transmitted thereby (including trade secrets and Personal Information).

(d) The Open Source Software used in the Company Software and Proprietary Databases does not have license or other usage terms that require as a condition of how the Company currently uses or has used the Company Software and Proprietary Databases, that any Software or databases or data incorporated into, derived from or distributed with the Company Software and Proprietary Databases be (i) disclosed or, in the case of Software, distributed in source code form; (ii) licensed for the purpose of making derivative works; (iii) redistributable at no charge; or (iv) subject to any other material limitation, restriction or condition on the right or ability of the Company to use or distribute the same. The Company has not used any Company Software or Proprietary Database disseminated by an entity on behalf of the Company that is installed on consumers' or partners' device(s) and used by any Person on behalf of the Company to monitor, record or transmit information about activities occurring on the device(s) on which it is installed, or about data that is stored or created on, transmitted from or transmitted to the computers on which it is installed in a manner that violates any applicable Company Privacy Commitments. The Company has not entered in any escrow or other agreements with any Person granting access to any Company Software and Proprietary Databases.

4.19 <u>Real Property</u>.

(a) The Company owns no interest in real property.

(b) <u>Schedule 4.19(b)</u> contains a complete and accurate list by property, city, state and country, of all land and real property leasehold or subleasehold estates and other rights to use or occupy any interest in land or real property held by the Company as of the date of this Agreement (the "<u>Company Properties</u>"). The Company Properties are the only properties used by the Company in, or otherwise related to, the Company's business as of the date of this Agreement, and subject to any permitted action pursuant to <u>Section 6.01</u>, as of the Closing Date. The Company is the sole legal and beneficial owner of (or is solely legally and beneficially entitled to) a leasehold interest in, or a right to use or occupy, the Company Properties. The Company has not leased or otherwise granted to any Person the right to use or occupy any Company Properties or any portion thereof.

(c) <u>Schedule 4.19(d)</u> contains a complete and accurate list and description of all leases, subleases, licenses, concessions, and other Contracts, agreements and leasehold or land use arrangements and all related supplemental or ancillary documents pursuant to which the Company leases, licenses, subleases or otherwise occupies any Company Property on the date hereof, except for any leases or licenses which arrange for a temporary occupancy arrangement of less than six months (collectively, the "<u>Company Lease Documents</u>"). The Company has delivered to Acquiror a true and complete copy of each such Company Lease Document. The Company is not and, to the Knowledge of the Company, no other party to any Company Lease Document is in material breach or material default under such Company Lease Document, nor has any event occurred which with notice or the passage of time or both would constitute a breach or default under any Company Lease Document.

(d) To the Knowledge of the Company, each Company Lease Document is a written agreement in full force and effect, is valid, binding and enforceable, subject to proper authorization and execution of each Company Lease Document by the other parties thereto. The Company has paid the rent and all other sums that are due and payable under such Company Lease Documents and there are no material arrears nor any sums which have been waived, deferred or accelerated, and no rent reviews are outstanding, in progress nor have been deferred. All consents, permits and approvals required for the grant of each Company Lease Document have been obtained and complied with in all material respects.

(e) To the Knowledge of the Company, except as would not reasonably be expected to have a Material Adverse Effect on the Company, there exists no restrictions, covenants or encumbrances that prevent any of the Company Properties from being used now or in the future for their current use and at materially the same cost as at present or would prevent or require consent from a third party as a result of the transactions contemplated by this Agreement.

(f) There are no outstanding options, rights of first offer or rights of first refusal to purchase any Company Properties or any portion thereof or interest therein (which are binding on or in favor of the Company). There are no Contracts relating to the right to receive any portion of the income or profits from the sale, operation or development of any Company Properties or any portion thereof or interest therein. The Company is not a party to any agreement or option to purchase any land or real property or interest therein, nor is in the process of negotiating any such agreement or option to purchase as at the date of this Agreement.

(g) Except as would not reasonably be expected to have a Material Adverse Effect on the Company, as of the date hereof, each Company Property and any structures built on them comply with all applicable Laws, the current use of each Company Property is the lawful use, no Company Property is subject to any restrictions relating to flood zoning limiting its use in any material respect, and there are no material outstanding or threatened disputes, actions, claims, demands, adverse notices or complaints to which the Company has received notice or is a party in respect of any of the Company Properties.

(h) As of the date hereof, there are no pending, or, to the Knowledge of the Company, threatened, material appropriation, condemnation, eminent domain, compulsory purchase or like proceedings relating to the whole or any part of any Company Properties.

(i) The Company has no material actual or contingent liability in respect of previously owned, leased, licensed, used or occupied land or buildings.

4.20 <u>Corrupt Practices; Sanctions</u>. Except as would not reasonably be expected to have a Material Adverse Effect on the Company:

(a) Neither the Company, any directors, officers or employees of the Company, nor, to the Knowledge of the Company, any of its other respective Representatives have violated, conspired to violate or aided and abetted the violation of any Anti-Corruption Laws. The Company (i) has instituted policies and procedures reasonably designed to ensure compliance with Anti-Corruption Laws and anti-money laundering Laws and (ii) has maintained such policies and procedures in force. No Government Official nor any of his or her immediate family members is an officer or director or owns any securities of the Company.

(b) Neither the Company, any directors, officers or employees of the Company nor, to the Knowledge of the Company, any of their other respective Representatives, is engaging and has not engaging, directly or indirectly, in any activities (including sales, reselling, licensing or sublicensing arrangements, funding, making payments, procuring, insurance or otherwise providing assistance or support in connection with operations, business or any other activity) with or involving any Sanctioned Person or Sanctioned Territory, nor otherwise violated any applicable Sanctions or Ex-Im Laws.

(c) Neither the Company nor any of its directors, officers, employees, nor to the Knowledge of the Company, any other Persons acting for or on behalf of any of the foregoing is, or has been, a Sanctioned Person.

(d) Neither the Company nor, to the Knowledge of the Company, any Representatives of the Company is currently or has in the past been a subject, target, or party of any current investigation, allegation (whether raised internally or externally), request for information, voluntary selfdisclosure, or any other inquiry by or involving any Governmental Authority regarding the actual or possible violation of Sanctions, and the Company has not received any notice that there is any investigation, allegation, request for information, or any other inquiry by any Governmental Authority regarding an actual or possible violation of Sanctions.

4.21 <u>Competition and Trade Regulation</u>. Except as would not reasonably be expected to have a Material Adverse Effect on the Company:

(a) The Company and its Representatives have been and currently are in compliance with Sanctions and Ex-Im Laws. The Company also maintains policies and procedures reasonably designed to ensure compliance with Sanctions and Ex-Im Laws and are following such policies and procedures.

(b) The Company is in compliance with all applicable antitrust Laws in all material respects. The Company is not nor has been a party to or is or has been concerned in any agreement or arrangement with a Governmental Authority under any anti-trust, competition or similar legislation in any jurisdiction in which the Company has assets or carries or intends to carry on business or where its activities may have an effect.

(c) The Company does not currently, or expect in the future to, produce, design, test, manufacture, fabricate, or develop any critical technologies, as defined under 31 C.F.R. § 800.215.

4.22 <u>Environmental Matters</u>. Except as would not reasonably be expected to have a Material Adverse Effect on the Company:

(a) the Company is and has been, in compliance in all material respects with all Environmental Laws, which material compliance includes possessing and complying with all Company Permits required under Environmental Laws in connection with the operation of the Company's business or the Company Properties ("<u>Company Environmental Permits</u>");

(b) the Company has not received any notice alleging any violation of or noncompliance by the Company with respect to any Environmental Law or Company Environmental Permit;

(c) there are no Actions pending, or to the Knowledge of the Company, threatened, against the Company, and the Company has not received any written notification of, nor, to the Knowledge of the Company, is the Company otherwise responsible for any violation of, noncompliance with or material liability under, Environmental Laws, including for the contamination by or manufacture, management, labeling, registration, import, export, generation, storage, treatment, use, transportation, disposal, Release or threatened Release at any location of, or exposure of any Person to, any Hazardous Material;

(d) the Company (i) is not a party to or subject to the provisions of any Governmental Order pursuant to Environmental Law or (ii) has not retained or assumed by Contract any material liabilities or obligations pursuant to Environmental Laws;

(e) there have been no Releases of any Hazardous Materials at, on, under, or from the current or former Company Properties, or to the Knowledge of the Company, any third-party property that would result in a requirement for notification to or investigation by a Governmental Authority, or the need for investigation, remediation, removal, cleanup or monitoring of Hazardous Materials pursuant to Environmental Laws; and

(f) the Company has furnished to Acquiror, or has otherwise made available for inspection by the Acquiror, all material Phase I or Phase II environmental reports and similar material environmental documents in the possession of the Company.

4.23 <u>Brokers</u>. No broker, investment banker, financial advisor or other Person, other than those set out in <u>Schedule 4.23</u>, the fees and expenses of which will be paid by the Company pursuant to an engagement letter entered into therewith, is entitled to any broker's, finder's, financial advisor's or other similar fee or commission in connection with the Transactions based upon arrangements made by or on behalf of the Company.

4.24 <u>Affiliate Agreements</u>. The Company is not a party to any transaction, agreement, arrangement or understanding with any (a) present or former executive officer or director of the Company, (b) beneficial owner (within the meaning of Section 13(d) of the Exchange Act) of 5% or more of the capital stock or equity interests of Acquiror, the Merger Subs or the Company or (c) Affiliate, "associate" or member of the "immediate family" (as such terms are respectively defined in Rules 12b-2 and 16a-1 of the Exchange Act) of any of the foregoing.

4.25 Insurance. The Company maintains such insurance policies or binders of fire, liability, umbrella liability, real and personal property, workers' compensation, vehicular, directors' and officers' liability, fiduciary liability and other casualty and property insurance as commercially reasonable and appropriate (the "<u>Company Insurance Policies</u>"). To the Knowledge of the Company, there are no events, circumstances or other liabilities that give rise to a material claim under the Company Insurance Policies. Except as has not had or would not reasonably be expected to have a Material Adverse Effect on the Company, the Company Insurance Policies are in full force and effect as of the date of this Agreement with respect to the Company, and the limits thereunder have not been impaired, exhausted or materially diminished. As of the date hereof, the Company has not received any written notice of cancellation of, of a material premium increase (relative to others in the industry in which the Company operates) with respect to, or of a material alteration of coverage under, any Company Insurance Policies (i) are valid and binding in accordance with their terms, subject to Enforceability Exceptions and (ii) have not been subject to any lapse in coverage. There are no material claims related to the Company or its assets, business, operations, employees, officers and directors pending under any such Company Insurance Policies as to which coverage has been denied or disputed or in respect of which there is an outstanding reservation of rights.

4.26 <u>COVID-19</u>. The Company has not participated in the federal Paycheck Protection Program or sought material benefits or relief thereunder.

4.27 <u>Takeover Statutes and Charter Provisions</u>. As of the date hereof and through the First Merger Effective Time, no "fair price," "moratorium," "control share acquisition" or other anti-takeover statute or similar domestic or foreign Law applies with respect to the Company in connection with this Agreement, the Merger, the issuance of the Closing Merger Consideration or any of the other transactions contemplated hereby. As of the date hereof and through the First Merger Effective Time, there is no stockholder rights plan, "poison pill" or similar anti-takeover agreement or plan in effect to which the Company is subject, a party or otherwise bound.

4.28 <u>Board Approval; Stockholders' Approval; Vote Required</u>. The Company Board, by resolutions duly adopted by unanimous vote of those voting at a meeting duly called and held and not subsequently rescinded or modified in any way, has duly (i) determined that this Agreement, the Merger and other Transactions are fair to and in the best interests of the Company and its stockholders, (ii) approved this Agreement, the Merger and other Transactions and declared their advisability, (iii) recommend that the stockholders of the Company approve and adopt this Agreement and approve the Merger and other Transactions and directed that this Agreement and the transactions contemplated hereby be submitted for consideration by the Company's stockholders. The only vote or written consent of the holders of any class or series of capital stock of the Company necessary to approve this Agreement, the Merger and the other Transactions is the affirmative vote or written consent of the holders of a majority of the outstanding shares of Company Common Stock in favor of the approval and adoption of this Agreement and approval of the Merger and other Transactions.

4.29 <u>No Other Representations or Warranties</u>. The representations and warranties made by the Company in this <u>Article IV</u> are the exclusive representations and warranties made by the Company, its Affiliates and their respective Representatives. Except for the representations and warranties contained in this <u>Article IV</u>, neither the Company nor any other Person has made or makes any other express or implied representation or warranty, either written or oral, on behalf of the Company, to the accuracy or completeness of any information regarding the Company available to the other parties or their respective Representatives and expressly disclaims any such other representations or warranties. In particular, without limiting the foregoing, neither the Company nor any other Person makes or has made any representation or warranty to the other parties hereto, and shall have no liability in respect of, (a) any financial projection, forecast, estimate, budget or prospect information relating to the Company or (b) any oral or, except for the representations and warranties expressly made by the Company in this <u>Article IV</u>, written information made available to the other parties hereto in the course of their evaluation of the Company and the negotiation of this Agreement or in the course of the Transactions.

ARTICLE V REPRESENTATIONS AND WARRANTIES OF ACQUIROR AND MERGER SUBS

Except as set forth in the Schedules to this Agreement (each of which qualifies (a) the correspondingly numbered representation, warranty or covenant if specified therein and (b) such other representations, warranties or covenants where its relevance as an exception to (or disclosure for purposes of) such other representation, warranty or covenant is reasonably apparent on its face) or in the Acquiror SEC Reports filed or furnished by Acquiror on or prior to the date hereof, each of Acquiror and the Merger Subs represents and warrants to the Company as of the date hereof and as of the Closing Date (or if a specific date is indicated in any such statement, as of such specified date) as follows:

5.01 Organization, Standing and Corporate Power.

(a) Acquiror is an entity validly existing and in good standing under the DGCL, and has all requisite corporate power and authority to carry on its business as now being conducted in all material respects. Acquiror is duly qualified or licensed to do business and is in good standing in each jurisdiction in which the nature of its business or the ownership, leasing or operation of its properties, assets or rights makes such qualification or licensing necessary, except as would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect on Acquiror or Nxu Tech. The Acquiror Organizational Documents that have been made available to the Company are true, correct and complete and are in effect as of the date of this Agreement and Acquiror is not in default under or in violation of any provision thereunder.

(b) Each of the Merger Subs is an entity duly organized, validly existing and in good standing under the Laws of Delaware, with full corporate power and authority to enter into this Agreement and perform its obligations hereunder. Other than the Merger Subs and NXU Technologies, LLC, a Delaware limited liability company and a wholly owned Subsidiary of Acquiror ("<u>NXU Tech</u>"), Acquiror has no other Subsidiaries or any equity or other interests in any other Person.

5.02 Corporate Authority; Approval; Non-Contravention.

(a) Except for the Required Acquiror Stockholder Approvals, each of Acquiror, Merger Sub I and Merger Sub II has all requisite corporate power and authority and has taken all corporate or other legal entity action necessary in order to execute, deliver and perform its obligations under this Agreement and the Ancillary Agreements to which it is a party and, subject to satisfaction of the conditions to Closing contemplated hereby, to consummate the Transactions. The execution, delivery and performance by Acquiror, Merger Sub I and Merger Sub II of this Agreement and the Ancillary Agreements to which it is a party, and the consummation by it of the Transactions, have been duly and validly authorized by all necessary corporate consent and authorizations on the part of Acquiror, Merger Sub I and Merger Sub II or Merger Sub II is necessary to authorize the execution and delivery by Acquiror, Merger Sub I or Merger Sub II of this Agreement, the Ancillary Agreements to which it is a party and the consummation by it of the Transactions, in each case, subject to receipt of the Required Acquiror Stockholder Approvals. This Agreement has been duly executed and delivered by each of Acquiror, Merger Sub I and Merger Sub II and, assuming due authorization, execution and delivery hereof by the Company, is a legal, valid and binding obligation of each of Acquiror, Merger Sub I and Merger Sub II and Merger Sub II in accordance with its terms (subject to the Enforceability Exceptions).

(b) The execution, delivery, and performance of this Agreement and the Ancillary Agreements to which Acquiror, Merger Sub I or the Merger Sub II are a party, and the consummation of the Transactions, and (in the case of Acquiror) subject to receipt of the Required Acquiror Stockholder Approvals, do not, and will not, constitute or result in (i) a breach or violation of, or a default under, the Acquiror Organizational Documents or the applicable organizational documents of the Merger Sub I or Meger Sub II or (ii) with or without notice, lapse of time or both, a breach or violation of, a termination (or right of termination) of or default under, the creation or acceleration of any obligations under or the creation of a Lien (other than a Permitted Lien) on any of the assets of Acquiror, Merger Sub I, Merger Sub II or Nxu Tech pursuant to, any Acquiror Material Contract or, assuming (solely with respect to performance of this Agreement and consummation of the Transactions) compliance with the matters referred to in Section 5.02(a), under any Law to which Acquiror, Merger Sub I and Merger Sub II or any of their Affiliates is subject, except (in the case of clause (ii) above) for such violations, breaches or defaults which (A) are set forth on Schedule 5.02(b), or (B) has not had or would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect on Acquiror or Nxu Tech.

(c) The Acquiror Support Agreement will have been duly executed by Acquiror and the stockholders of Acquiror parties thereto and, assuming due authorization, execution and delivery thereof by the other parties thereto, will constitute a legal, valid and binding obligation of Acquiror, and, to the Knowledge of Acquiror, will be enforceable against the other parties thereto (other than the Company) in accordance with its terms (subject to the Enforceability Exceptions).

5.03 <u>Governmental Approvals</u>. No consent of, or registration, declaration, notice or filing with, any Governmental Authority is required by or with respect to Acquiror, Merger Sub I or Merger Sub II in connection with the execution and delivery by Acquiror, Merger Sub I or Merger Sub II of this Agreement or the consummation by Acquiror, Merger Sub I or Merger Sub II of the Transactions, except for (a) such other consents, registrations, declarations, notices and filings which, if not obtained or made, would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect on Acquiror or Nxu Tech, (b) the filing of the Certificates of Merger and amendment to Acquiror's certificate of incorporation with the Secretary of State of the State of Delaware, (c) applicable requirements, if any, of the Exchange Act, state securities or "blue sky" laws, (d) the NASDAQ's approval of Listing Application and (e) the SEC's declaration of the Registration Statement effective under the Securities Act.

5.04 Capitalization.

(a) The authorized capital stock of Acquiror consists of (i) 4,000,000,0000 shares of Acquiror Class A Common Stock, (ii) 1,000,000,000 shares of Acquiror Class B Common Stock, and (iii) 10,000,000 shares of Acquiror Preferred Stock, of which 5,000 shares have been designated as Series A Convertible Preferred Stock and 1 share has been designated as Series B Preferred Stock. At the close of business on the immediately preceding Business Day prior to the date hereof, there were (i) 11,934,072 shares of Acquiror Class A Common Stock outstanding, (ii) 273,503 shares of Acquiror Class B Common Stock outstanding, (iii) no shares of Series A Convertible Preferred Stock outstanding, (iv) zero shares of Series B Preferred Stock outstanding (provided that the Acquiror's board approved the issuance of one share of Series B Preferred Stock on the date hereof), (v) an aggregate of 49,354,991 reserved for future issuance under the NXU, Inc. 2023 Omnibus Incentive Plan, as amended from time to time (the "<u>Acquiror Omnibus Plan</u>"), (vi) no shares are reserved for the future grant of awards under any Acquiror Option & RSU Plan other than the Acquiror Omnibus Plan, (vii) an aggregate of 997,698 were subject to outstanding Acquiror Options and an aggregate of 7,293,934 were subject to outstanding Acquiror RSUs and (vii) an aggregate of 109,513 reserved for future issuance under Acquiror Warrants. Acquiror does not hold any shares of its capital stock in its treasury.

(b) All of the issued and outstanding shares of capital stock of Acquiror have been, and all shares that may be issued pursuant to the Acquiror Option & RSU Plans will be, when issued in accordance with the terms thereof, duly authorized and validly issued, fully paid and nonassessable. None of the issued and outstanding shares of capital stock of Acquiror are entitled or subject to any preemptive right, right of participation, right of maintenance or any similar right and none of the issued and outstanding shares capital stock of Acquiror is a party relating to the voting or registration of, or restricting any Person from purchasing, selling, pledging or otherwise disposing of (or granting any option or similar right with respect to), any shares of capital stock of Acquiror. Acquiror is not under any obligation, nor is it bound by any Contract pursuant to which it may become obligated, to repurchase, redeem or otherwise acquire any issued outstanding shares of capital stock of Acquiror or other securities.

(c) Schedule 5.04(c) sets forth a true and correct list, as of the close of business on the immediately preceding Business Day prior to the date hereof, of (i) each award of Acquiror Options and Acquiror RSUs and the Acquiror Option and RSU Plan under which the award was granted, (ii) the name of each holder of Acquiror Options and the name of each holder of Acquiror RSUs, (iii) the number of shares of Acquiror Class A Common Stock underlying each award of Acquiror Options and Acquiror RSUs, (iv) the date on which the award of Acquiror Options or Acquiror RSUs was granted, (v) the vesting schedule of each award of Acquiror Options and Acquiror RSUs and Acquiror RSUs and (vi) the exercise price and expiration date of each award of Acquiror Options. Each Acquiror Option has an exercise price that is at least equal to the fair market value (within the meaning of Section 409A of the Code) of the underlying shares on the date of grant.

(d) Except as set forth on <u>Schedule 5.04(d)</u>, as of the close of business on the immediately preceding Business Day prior to the date hereof, there are no other outstanding equity interest in Acquiror, including any warrants, options, calls, pre-emptive rights, subscriptions, "phantom" stock rights or other rights, agreements, arrangements, convertible or exchangeable securities or other commitments (other than this Agreement) pursuant to which Acquiror is obligated to issue, transfer, sell, purchase, return or redeem or cause to be issued, transferred, sold, purchased, returned or redeemed any equity securities of Acquiror.

(e) Other than NXU Tech, Merger Sub I or Merger Sub II or as set forth on <u>Schedule 5.04(e)</u>, Acquiror has no equity interest in, nor has it agreed to acquire, any share capital or other equity security of any other company (wherever incorporated).

5.05 <u>Subsidiaries</u>. Other than the Merger Subs, which have been formed solely in connection with the Transaction and has undertaken no other operations or activities, NXU Tech, and as set forth on <u>Schedule 5.04(e)</u>, Acquiror has no Subsidiaries or other ownership, or right to acquire, capital stock, security, partnership interest or other equity interest of any kind in any Person.

5.06 <u>Compliance with Laws</u>. Except as would not reasonably be expected to have a Material Adverse Effect on Acquiror or Nxu Tech:

(a) Each of Nxu Tech and Acquiror is conducting and has, since January 1, 2020, conducted its business in compliance with all Laws applicable to it and Acquiror's business, properties, rights or other assets.

(b) There is no, and since January 1, 2020 has been no, Action by or against Acquiror or Nxu Tech pending or threatened in writing, nor has any Governmental Authority indicated in writing to Acquiror or Nxu Tech an intention to initiate or conduct the same.

(c) Since January 1, 2020, neither Nxu Tech nor Acquiror has received any written notice from any Governmental Authority (i) with respect to an alleged, actual or potential violation and/or failure to comply with any such applicable Law or (ii) requiring Acquiror to take or omit any action to ensure compliance with any such applicable Law.

(d) Acquiror possesses all governmental permits, approvals, orders, authorizations, consents, licenses, certificates, franchises, exemptions or waivers of, or filings, notices, or registrations with, or issued by, any Governmental Authority necessary for the ownership and use of the assets of Acquiror and Nxu Tech and the operation of Acquiror's and Nxu Tech's business as currently conducted, (the "Acquiror Permits"). All such Acquiror Permits are valid and in full force and effect, and there are no lawsuits or other proceedings pending or threatened before any Governmental Authority that seek the revocation, cancellation, suspension or adverse modification thereof. Acquiror Permits.

5.07 <u>Absence of Certain Changes or Events</u>. Except as set forth on <u>Schedule 5.07</u>: (a) since the Balance Sheet Date and except as expressly required or permitted by this Agreement each of Nxu Tech and Acquiror has conducted its business in all material respects in the ordinary course of business, and (b) since the Balance Sheet Date, there has not been any change, effect, event, circumstance, occurrence or state of facts that would, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect on Acquiror or Nxu Tech.

5.08 <u>No Undisclosed Liabilities</u>. Except (a) as disclosed, reflected or reserved against in the most recent balance sheet of Acquiror included within the Acquiror SEC Reports, (b) for liabilities incurred in the ordinary course of business since the Balance Sheet Date, (c) as expressly permitted or contemplated by this Agreement or otherwise incurred in connection with the Transactions, (d) as disclosed on <u>Schedule 5.08</u>, (e) contingent liabilities under executory contracts and (f) for liabilities that have been discharged or paid in full in the ordinary course of business, as of the date hereof, Acquiror does not have any liabilities of the type required by GAAP to be disclosed or reserved for on a balance sheet of Acquiror.

5.09 Litigation.

(a) Except as set forth in <u>Schedule 5.09(a)(i)</u>, neither Nxu Tech, Acquiror nor, to the Knowledge of Acquiror, any of the officers of Nxu Tech or Acquiror, in their capacities as such, is the subject of or engaged in any material Action before a Governmental Authority, arbitration or other dispute resolution process before a third party unrelated to the dispute, whether as claimant, defendant or otherwise, and no such litigation, arbitration or dispute resolution process is pending or threatened in writing on the date hereof, in each case, that would, individually or in the aggregate, reasonably be expected to have a material adverse effect on the ability of Acquiror, Merger Sub I or Merger Sub II to enter into, perform its obligations under this Agreement and consummate the Transactions. Except as set forth in <u>Schedule 5.09(a)(i)</u>, neither of Nxu Tech nor Acquiror is, nor to the Knowledge of Acquiror are any of the officers of Nxu Tech or Acquiror, in their capacities as such, subject to any settlement agreements or arrangements, whether written or oral, or is in discussions for a settlement or arrangement, regarding any disputes or claims, that would, individually or in the aggregate, reasonably be expected to have a material adverse effect on the ability of Acquiror, Merger Sub I or Merger Sub II to enter into, perform its obligations under this Agreement and consummate the Transactions.

(b) Neither Acquiror, Nxu Tech, Merger Sub I nor Merger Sub II is a party to or subject to the provisions of any outstanding judgment, order, writ, injunction, decree or award of any Governmental Authority (except if generally applicable without Acquiror, Nxu Tech, Merger Sub I or Merger Sub II being named therein) that would, individually or in the aggregate, reasonably be expected to have a material adverse effect on the ability of Acquiror, Merger Sub I or Merger Sub I o

5.10 Contracts.

(a) <u>Schedule 5.10(a)</u> sets forth a true and complete list as of the date hereof, of the following Contracts that are effective as of the date hereof and to which Acquiror or Nxu Tech is a party or is bound (all such Contracts set forth on <u>Schedule 5.10(a)</u>, or which are required to be so disclosed, the "<u>Acquiror Material Contracts</u>"):

(i) all such Contracts with third party manufacturers and suppliers for the manufacture and supply of products;

(ii) all such Contracts with (or with obligations of Acquiror to) a Related Party;

(iii) all such Contracts that contain any covenant materially limiting or prohibiting the right of Acquiror (A) to engage in any line of business or conduct business in any geographic area, (B) to distribute or offer any products or services, or (C) to compete with any other person in any line of business or in any geographic area or levying a fine, charge or other payment for doing any of the foregoing;

(iv) all such Contracts in which the aggregate outstanding expenditure or payment obligations of Acquiror or Nxu Tech exceeds \$250,000, excluding obligations that are contingent liabilities in respect of a breach or indemnification obligation or similar contingent obligation as a result of a breach or default;

(v) any Contract for the use by Acquiror or Nxu Tech of any tangible property;

(vi) any partnership, joint venture or other similar agreement or arrangement providing for the formation, creation, operation, management or control of any partnership or joint venture with a third party to which Acquiror or Nxu Tech is a party, other than *bona fide* customer-supplier relationships or a trade association;

(vii) all such Contracts providing for the acquisition or disposition of any business, equity interests or material assets (whether by merger, sale of stock, sale of assets or otherwise) pursuant to which Acquiror or Nxu Tech has any ongoing obligation (including for deferred purchase price obligations, earn-out obligations, indemnification obligations and other contingent liabilities (including payment obligations in respect of the future utilization of any net operating losses));

(viii) all such Contracts that obligate Acquiror or Nxu Tech to make any loans, advances or capital contributions to, or investments in, any Person (other than advances to employees for business expenses in the ordinary course of business consistent with past practice);

(ix) any note, mortgage, indenture or other obligation or agreement or other instrument for or relating to indebtedness for borrowed money, or any guarantee of third-party obligations, or any letters of credit, performance bonds or other credit support for Acquiror or Nxu Tech;

any Contract that creates guarantees or Liens of any nature on Acquiror's or Nxu Tech's assets not in the ordinary course

of business;

(xi) any CBA;

(x)

(xii) all such Contracts or agreements with any employee, officer, director or other individual service provider that (A) provide for annual compensation in excess of \$100,000, (B) are not terminable by Acquiror or Nxu Tech on no more than thirty (30) days' notice and without liability or financial obligation to Acquiror other than accrued compensation and other payments required by Law or (C) provide for sale, change in control, severance, retention, transaction or similar bonuses, or any other compensation or benefits triggered or accelerated in connection with or otherwise related to the consummation of the transaction contemplated hereby, whether alone or in connection with any other event, that are (or may become) payable to any current or former employee of or other service provider to Acquiror or Nxu Tech;

(xiii) all Contracts entered into to settle or resolve any material Intellectual Property-related dispute, including settlement agreements, covenants not to sue, consent agreements, and co-existence agreements;

(xiv) all Contracts providing for the development of any material Intellectual Property, independently or jointly, by or for Acquiror or Nxu Tech, other than Contracts entered into pursuant to a form employee or independent contractor invention assignment agreement between Acquiror or Nxu Tech and an employee or independent contractor of Acquiror or Nxu Tech regarding the development of Intellectual Property by such employee or independent contractor; and

(xv) all Contracts pursuant to which Acquiror or Nxu Tech (A) grants or (B) is granted a license (whether or not such license is currently exercisable) to, or other rights under, any Intellectual Property, excluding, in the case of (A), Permitted Liens, and in the case of (B), any Commercially Available Software.

(b) Except as set forth in <u>Schedule 5.10(b)</u>, neither of Acquiror nor Nxu Tech (i) is, nor has it received written notice that any other party to any Acquiror Material Contract is, in material violation or material breach of or material default (immediately or upon notice or lapse of time) under or (ii) has waived or failed to enforce any material rights or material benefits under any Acquiror Material Contract to which it is a party or any of its properties, rights or other assets is subject. Except as set forth in <u>Schedule 5.10(b)</u>, no Material Contract is the subject of a notice to terminate, except for any expiration of the term of an Acquiror Material Contract following the date of this Agreement in accordance with its terms. Except as set forth in <u>Schedule 5.10(b)</u>, each Acquiror Material Contract is in full force and effect and, subject to the Enforceability Exceptions, is legal, valid and binding on the Acquiror or Nxu Tech, as applicable, and, to the Knowledge of Acquiror, each other party thereto, except as would not reasonably be expected to have a Material Adverse Effect on Acquiror or Nxu Tech. Except as set forth in <u>Schedule 5.10(b)</u>, there is no default under any such Material Contracts by Acquiror or Nxu Tech, as applicable, or, to the Knowledge of Acquiror, any other party thereto, and no event has occurred that with the lapse of time or the giving of notice or both would constitute a default thereunder by Acquiror or Nxu Tech.

5.11 Employment Matters.

(a) Except as set forth on <u>Schedule 5.11(a)</u>, (i) neither of Acquiror nor Nxu Tech is a party to or bound by any CBA (including agreements with works councils and trade unions and side letters), or is negotiating or required to negotiate a CBA; (ii) no employees of Acquiror or Nxu Tech are represented by any labor or trade union, works council, or other labor organization with respect to their employment; (iii) in the past three (3) years, no labor or trade union, works council, other labor organization, or group of employees of Acquiror or Nxu Tech has made a demand for recognition or certification, and there are no representation or certification proceedings presently pending or threatened to be brought or filed with the National Labor Relations Board or any other labor relations tribunal or authority; (iv) in the past three (3) years, there have been no actual or threatened organizing activities with respect to any employees of Acquiror or Nxu Tech and no such activities are currently pending or threatened; (v) in the past three (3) years, there has been no actual or threatened strike, lockout, work stoppage, slowdown, picketing, hand billing, unfair labor practice charge, material labor grievance, material labor arbitration or other material labor dispute against or affecting Acquiror or Nxu Tech, and no such dispute is currently pending or threatened; (vi) with respect to the Transactions, each of Acquiror and Nxu Tech has satisfied all notice, bargaining, consent, consultation or other obligations to its employees and employees' Representatives under applicable Law and any CBA.

(b) Over the past three (3) years (i) no employee of Acquiror or Nxu Tech has made allegations of sexual harassment, sexual misconduct or other harassment against any officer, director or key employee of Acquiror or Nxu Tech, as applicable, and (ii) neither Acquiror nor Nxu Tech has entered into any settlement agreement related to sexual harassment, sexual misconduct or other harassment by or against any current or former employee, independent contractor, director or officer of Acquiror or Nxu Tech, as applicable.

(c) Neither of Acquiror or Nxu Tech is a party to or has received an active written threat or, an oral threat, of any Action brought by on or on behalf of any applicant, any former or current employee, any former or current individual or sole proprietor independent contractor relating to any labor or employment matters of Acquiror or Nxu Tech, as applicable.

(d) Except as set forth in <u>Schedule 5.11(d)</u>, each of Acquiror and Nxu Tech is, and for the past three (3) years has been, in compliance with all Laws relating to labor or employment, including Laws regarding terms and conditions of employment, hiring, background checks, trainings, notices, immigration, authorization to work, health and safety, wages, hours, employee classification (for overtime purposes or as employee versus independent contractor), overtime (including the proper determination of regular pay and the treatment of bonuses), meal and rest periods, harassment, discrimination, retaliation, accommodations, disability rights or benefits, plant closings and mass layoffs, workers' compensation, labor relations, leaves of absences, time off, COVID-19, affirmative action, unemployment insurance and/or termination of employment other than any failure to so comply not reasonably likely to result in material liability to Acquiror or Nxu Tech, as applicable. Neither of Acquiror nor Nxu Tech has any actual or contingent liability with respect to (i) any misclassification of any person as an independent contractor rather than as an employee, as an employee when in fact employed, (ii) any employee or contractor leased from or staffed by another employer, or (iii) any person currently or formerly classified as exempt from, or otherwise not paid where required, overtime and minimum or other wages.

(e) Neither Acquiror nor Nxu Tech has, in the past three (3) years, experienced or implemented a "plant closing" or "mass layoff" as defined in the WARN Act or other event requiring notice under a WARN Act. In the six (6) month period immediately prior to the date hereof, neither Acquiror nor Nxu Tech has carried out any temporary layoffs, furloughs, or material reductions in hours of work that, if continued for six (6) months, alone or in the aggregate with any other "employment loss" (as defined under any WARN Act), could reasonably be expected to constitute a "plant closing" or "mass layoff" under any WARN Act.

5.12 Employee Benefits.

(a) <u>Schedule 5.12(a)</u> contains a true, correct and complete list of each Acquiror Benefit Plan. Acquiror has provided to the Company with respect to each Acquiror Benefit Plan, as applicable, true, correct and complete copies of: (i) such Acquiror Benefit Plan, or with respect to any unwritten Acquiror Benefit Plan, a written description of all material terms thereof, and form of any award documentation thereunder; (ii) the most recent determination, opinion or advisory letter received from the Internal Revenue Service, if any; (iii) the Form 5500 annual report and the financial statements and actuarial summary or other valuation reports prepared with respect thereto in the past three (3) years; (iv) each trust, insurance, annuity, stop-loss or other funding Contract or arrangement related thereto; (v) the most recent summary plan description and any summaries of material modification with respect thereto; (vi) discrimination or similar testing (and evidence of corrective action related thereto) for the past three (3) years; and (vii) all non-routine correspondence or communications relating thereto with any Governmental Authority or participant within the past three (3) years.

(b) With respect to each Acquiror Benefit Plan: (i) such Acquiror Benefit Plan has been maintained, operated and administered in compliance in all material respects with its terms and with the requirements of applicable Law, including the Code and ERISA; (ii) all contributions or payments due to date have been made timely and in compliance with the terms of such Acquiror Benefit Plan and applicable Law, or if not yet due, properly accrued (to the extent required or appropriate to be accrued in accordance with GAAP or otherwise in the ordinary course of business); (iii) no failure to pay premiums due or payable with respect to insurance policies relating to such Acquiror Benefit Plan has resulted in default under any such policies; and (iv) there are no proceedings pending or, to the Knowledge of Acquiror, threatened against or involving such Acquiror Benefit Plan brought by or on behalf of any current or former employee or other service provider of Acquiror (or dependent or beneficiary thereof) or Governmental Authority (other than routine claims for benefits made in the ordinary course of business), in each case, except as could not reasonably be expected to result in material liability to Acquiror. Each Acquiror Benefit Plan that is intended to be qualified under Section 401(a) of the Code is so qualified and has received a favorable determination letter from the Internal Revenue Service or, with respect to a prototype or volume submitter plan, can rely on an opinion or advisory letter from the Internal Revenue Service to the prototype plan or volume submitter plan sponsor, as to its qualification and to the effect that the plan's related trust is exempt from federal income taxes under Section 501(a) of the Code, and nothing has occurred that could reasonably be expected to result in the revocation of such favorable determination.

(c) Neither Acquiror, Nxu Tech nor any ERISA Affiliate maintains, sponsors, contributes to or been obligated to contribute to (or has ever maintained, sponsored or contributed to or been obligated to contribute to) or has any obligation with respect to, (i) any employee benefit plan that is subject to Title IV of ERISA, (ii) any Multiemployer Plan, (iii) any multiple employer plan (as defined in Section 413(c) of the Code), (iv) any voluntary employees' beneficiary association described under Section 501(c)(9) of the Code or any "funded welfare plan" within the meaning of Section 419 of the Code or (v) any multiple employer welfare arrangement (as defined under Section 3(40) of ERISA).

(d) Neither Acquiror nor Nxu Tech has no actual or contingent liability under or with respect to COBRA for failure to comply with its notice or coverage requirements or for failure to comply with PPACA, and any other comparable domestic or foreign Laws, including all requirements relating to eligibility waiting periods and the offer of or provision of minimum essential coverage that is compliant with Section 36B(c)(2)(C) of the Code. No excise Tax or penalty under PPACA, including Sections 4980D and 4980H of the Code is outstanding, has accrued, or has arisen with respect to any period prior to the Closing, with respect to any Acquiror Benefit Plan.

(e) Each Acquiror Benefit Plan that provides deferred compensation subject to Section 409A of the Code satisfies and has satisfied, in form and operation, the requirements of Section 409A of the Code and the guidance thereunder. No current or former employee or other service provider of Acquiror or Nxu Tech is entitled to receive any gross-up or additional payment in connection with any Tax (including those imposed under Section 409A or Section 4999 of the Code).

(f) No Acquiror Benefit Plan, fiduciary of such plan or administrator of such plan has taken any action, or failed to take any action, which action or failure could subject Acquiror, Nxu Tech or any of their Affiliates, or any current or former employee or other service provider of Acquiror or Nxu Tech, to any material Tax, fine, Lien, penalty or other obligation imposed by ERISA, the Code or other applicable Laws, with respect to or in connection with any Company Benefit Plan.

(g) Neither of Acquiror nor Nxu Tech provide (or have at any time provide) or have any obligation to provide post-employment health, life or other welfare benefits, other than as required under Section 4980B of the Code (the full cost of which is borne by the applicable recipient of such benefits).

(h) Except as set forth in <u>Schedule 5.12(h)</u>, neither the execution of this Agreement nor the consummation of the Transactions, either alone or in combination with the occurrence of any other event, would be reasonably likely to result in: (i) any material payment or benefit becoming due to, or a material increase in, or acceleration of the timing of payment, funding or vesting of, the compensation or benefits of, any current or former employee, officer, director or other individual service provider of Acquiror or Nxu Tech; (ii) the receipt (whether in cash, property or the vesting of property) by any "disqualified individual" of any "parachute payment" (as such terms are defined in Section 280G of the Code); or (iii) any breach or violation of or default under or, except pursuant to applicable Law, limitation of the rights of Acquiror, Nxu Tech or their respective Affiliates to amend, modify or terminate any Acquiror Benefit Plan.

5.13 <u>Taxes</u>.

(a) Each of Acquiror and Nxu Tech has filed with the appropriate Tax Authority, or has caused to be filed on its behalf (taking into account any valid extension of time within which to file), all material Tax Returns required to be filed by it, and all such Tax Returns were and are true, correct and complete in all material respects. Each of Acquiror and Nxu Tech has paid all Taxes due and payable (whether or not shown on any Tax Return). No claim has ever been made in writing by a Tax Authority in a jurisdiction where Acquiror or Nxu Tech does not file a Tax Return that such entity is or may be subject to taxation by that jurisdiction in respect of Taxes that would be covered by or the subject of such Tax Return.

(b) Except as set forth in <u>Schedule 5.13(b)</u>, each of Acquiror and Nxu Tech has (i) withheld all Taxes required to have been withheld by it in connection with amounts paid to any employee, independent contractor, creditor, stockholder or any other third party, and (ii) remitted such amounts required to have been remitted to the appropriate Tax Authority, except where such failure would not reasonably be expected to result in a material liability.

(c) No claim, assessment, deficiency or proposed adjustment for any amount of Tax has been asserted or assessed in writing by any Tax Authority against Acquiror or Nxu Tech that remains unresolved or unpaid. There is no Tax audit or other examination of Acquiror or Nxu Tech presently in progress, and there are no waivers, extensions or requests for any waivers or extensions of any statute of limitations currently in effect with respect to any Taxes or Tax Returns of Acquiror, nor has any request been made in writing for any such extension or waiver.

(d) Neither of Acquiror nor Nxu Tech is or has been (i) a party to any Tax sharing, indemnification, allocation or similar agreement or arrangement (excluding any commercial Contract entered into in the ordinary course of business and not primarily related to Taxes), (ii) a member of an affiliated, consolidated, combined, unitary or similar Tax group for income tax purposes (other than any such Tax group the common parent of which was Acquiror), or (iii) a party to any "listed transaction" under Treasury Regulations Section 1.6011-4(b)(2) (or any similar or corresponding provision of state, local or foreign Law).

(e) Neither of Acquiror nor Nxu Tech has any liability for Taxes of any other Person (other than any such Tax group the common parent of which is Acquiror) as a result of Treasury Regulations Section 1.1502-6 (or any similar provision of state, local or foreign Law), as a transferee or successor, by operation of Law, by Contract or otherwise (excluding, in each case, any commercial Contract entered into in the ordinary course of business and not primarily related to Taxes).

(f) Neither of Acquiror nor Nxu Tech has distributed stock of another Person, or had its stock distributed by another Person, in a transaction that was purported or intended to be governed in whole or in part by Section 355 of the Code (or so much of Section 356 of the Code as relates to Section 355 of the Code).

(g) Each of Acquiror and Nxu Tech are in compliance in all material respects with all applicable transfer pricing Laws and regulations. All intercompany agreements have been adequately documented, and such documents have been duly executed in a timely manner. The prices and terms for the provision of any property or services by or to Acquiror or Nxu Tech are arm's length for purposes of the relevant transfer pricing Laws, and all related documentation required by such Laws has been timely prepared or obtained and, if necessary, retained.

(h) Neither Acquiror nor Nxu Tech is a partner for Tax purposes with respect to any joint venture, partnership or other arrangement or Contract which is treated as a partnership for Tax purposes.

(i) Neither Acquiror nor Nxu Tech has (i) been a United States real property holding corporation within the meaning of Section 897(c) (2) of the Code during the past five years, (ii) been a "personal holding company" as defined in Section 542 of the Code (or any similar provision of state, local or foreign Law) or (iii) become subject to Tax in a country other than the country of its formation by reason of being engaged in a trade or business, having a permanent establishment (within the meaning of an applicable Tax treaty), or having employees in such country.

(j) There are no Liens with respect to Taxes on any of the assets of Acquiror or Nxu Tech, other than Liens for Taxes which are not yet due and payable.

(k) Neither of Acquiror nor Nxu Tech will be required to include any item of income in, or exclude any material item of deduction from, taxable income for any period (or any portion thereof) ending after the Closing Date as a result of any installment sale, open transaction or other transaction on or prior to the Closing Date, any accounting method change, improper use of an accounting method or agreement with any Tax Authority filed or made on or prior to the Closing Date or any material deferred revenue or prepaid amount received on or prior to the Closing.

(l) Each of Acquiror and Nxu Tech has complied in all material respects with all escheat and unclaimed property Laws.

(m) Neither Acquiror nor Nxu Tech has taken or agreed to take any action (nor to the Knowledge of Acquiror is there any fact or circumstance) that would prevent the Merger from qualifying as a "reorganization" within the meaning of Section 368(a) of the Code.

(n) For U.S. federal income Tax purposes, Merger Sub II is, and immediately following the Merger will be, a "disregarded entity" within the meaning of Treasury Regulations Section 1.368-2(b)(1)(i)(A) whose separate existence as an equity is disregarded from that of its sole owner, which is Acquiror.

5.14 <u>Brokers</u>. No broker, investment banker, financial advisor or other Person, other than those set out in <u>Schedule 5.14</u>, the fees and expenses of which will be paid by Acquiror, Nxu Tech, Merger Sub I or Merger Sub II pursuant to an engagement letter entered into therewith, is entitled to any broker's, financial advisor's or other similar fee or commission in connection with the Transactions based upon arrangements made by or on behalf of Acquiror, Nxu Tech, Merger Sub I or Merger Sub II.

5.15 Acquiror SEC Reports; Financial Statements; Sarbanes-Oxley Act.

(a) To the Knowledge of Acquiror, except as disclosed in <u>Schedule 5.15(a)</u>. Acquiror has filed all required registration statements, reports, schedules, forms, statements and other documents required to be filed by it with the SEC (collectively, as they have been amended since the time of their filing and including all exhibits thereto, the "<u>Acquiror SEC Reports</u>"). Except as set forth on <u>Schedule 5.15(a)</u>, to the Knowledge of Acquiror, none of the Acquiror SEC Reports, as of their respective dates, contained any untrue statement of a material fact or omitted to state a material fact required to be stated therein or necessary in order to make the statements made therein, in light of the circumstances under which they were made, not misleading. Except as set forth on <u>Schedule 5.15(a)</u>, to the Knowledge of Acquiror, the audited financial statements and unaudited interim financial statements (including, in each case, the notes and schedules thereto) included in the Acquiror SEC Reports complied as to form, as of their respective dates of filing with the SEC, in all material respects with the published rules and regulations of the SEC with respect thereto, were prepared in all material respects in accordance with GAAP applied on a consistent basis during the periods involved (except as may be indicated therein or in the notes thereto and except with respect to unaudited statements as permitted by Form 10-Q of the SEC), and fairly present (subject, in the case of the unaudited interim financial statements included therein, to normal year-end adjustments and the absence of complete footnotes) in all material respects the financial position of Acquiror and Nxu Tech as of the respective dates thereof and the results of its operations and cash flows for the respective periods then ended.

(b) Acquiror has established and maintains disclosure controls and procedures (as defined in Rule 13a-15 under the Exchange Act). Such disclosure controls and procedures are designed to ensure that material information relating to Acquiror and other material information required to be disclosed by Acquiror in the reports and other documents that it files or furnishes under the Exchange Act is recorded, processed, summarized and reported within the time periods specified in the rules and forms of the SEC, and that all such material information is accumulated and communicated to Acquiror's principal executive officer and its principal financial officer as appropriate to allow timely decisions regarding required disclosure and to make the certifications required pursuant to Sections 302 and 906 of the Sarbanes-Oxley Act. As of March 31, 2024, to the Knowledge of Acquiror, such disclosure controls and procedures were effective in timely alerting Acquiror's principal executive officer and principal financial officer to material information required to be included in Acquiror's periodic reports required under the Exchange Act.

(c) Acquiror has established and maintained a system of internal control over financial reporting (as defined in Rule 13a-15 under the Exchange Act). Acquiror's management assessed the effectiveness of the Company's internal control over financial reporting for the fiscal year ended December 31, 2023 based on the criteria for effective internal control over financial reporting established in "Internal Control-Integrated Framework," issued by the Committee of Sponsoring Organizations of the Treadway Commission in 2013, and based on such assessment, determined that the Company maintained effective internal control over financial reporting as of December 31, 2023. Since December 31, 2023, there have been no changes in Acquiror's internal control over financial reporting that have materially affected, or are reasonably likely to materially affect, Acquiror's internal control over financial reporting.

(d) There are no outstanding loans or other extensions of credit made by Acquiror or Nxu Tech to any executive officer (as defined in Rule 3b-7 under the Exchange Act) or director of Acquiror or Nxu Tech. Neither Acquiror nor Nxu Tech has taken any action prohibited by Section 402 of the Sarbanes-Oxley Act.

(e) Neither Acquiror nor Nxu Tech (including any employee thereof) nor, to the Knowledge of Acquiror, Acquiror's independent registered public accounting firm has identified or been made aware of (i) any significant deficiency or material weakness (each as defined in Rule 13a-15(f) of the Exchange Act) in the system of internal controls over financial reporting utilized by Acquiror, (ii) any Fraud, whether or not material, that involves Acquiror's management or other employees who have a role in the preparation of financial statements or the internal controls over financial reporting utilized by Acquiror or (iii) any claim or allegation regarding any of the foregoing.

(f) To the Knowledge of Acquiror, as of the date hereof, there are no outstanding SEC comments from the SEC with respect to the Acquiror SEC Reports. Except as set forth in <u>Schedule 5.15(f)</u>, to the Knowledge of Acquiror, none of the Acquiror SEC Reports filed on or prior to the date hereof is subject to ongoing SEC review or investigation as of the date hereof.

(g) To the Knowledge of Acquiror, each director and executive officer of Acquiror and Nxu Tech has filed with the SEC all statements required by Section 16(a) of the Exchange Act and the rules and regulations promulgated thereunder.

5.16 <u>Registration Statement</u>. As of the time the Registration Statement becomes effective under the Securities Act, the Registration Statement (together with any amendments or supplements thereto) will not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements therein, in the light of the circumstances under which they were made, not misleading; <u>provided</u>, <u>however</u>, that Acquiror makes no representations or warranties as to the information contained in or omitted from the Registration Statement in reliance upon and in conformity with information furnished in writing to Acquiror by or on behalf of the Company specifically for inclusion in the Registration Statement or as to any projections or forecasts or forward looking statements included in the Registration Statement.

5.17 <u>NASDAQ Stock Market Quotation</u>. The issued and outstanding shares of Acquiror Class A Common Stock are registered pursuant to Section 12(b) of the Exchange Act and are listed for trading on the NASDAQ under the symbol "NXU". Except as disclosed in the Acquiror SEC Reports, Acquiror is in compliance in all material respects with the rules of the NASDAQ and there is no action or proceeding pending or, to the Knowledge of Acquiror, threatened against Acquiror by the NASDAQ, the Financial Industry Regulatory Authority or the SEC with respect to any intention by such entity to deregister the Acquiror Class A Common Stock or terminate the listing of Acquiror Class A Common Stock on the NASDAQ. None of Acquiror or its Affiliates has taken any action in an attempt to terminate the registration of the Acquiror Class A Common Stock under the Exchange Act.

5.18 Intellectual Property.

(a) <u>Schedule 5.18(a)</u> contains a complete and accurate list of all (i) issued patents and pending patent applications, (ii) trademark and service mark registrations and pending applications, (iii) copyright registrations and pending applications and (iv) internet domain names, in each case that are owned by Acquiror or Nxu Tech or that Acquiror or Nxu Tech purport to own (collectively, "<u>Acquiror Registered IP</u>"), indicating for each item, as applicable, the registration or application number, the applicable filing jurisdiction and the date of filing or issuance and, with respect to domain names, the applicable domain name registrar. Each item of Acquiror Registered IP is subsisting and unexpired and to the Knowledge of Acquiror, valid and enforceable. Acquiror solely and exclusively owns, free and clear of all Liens (other than Permitted Liens), all right, title and interest in and to the Owned Acquiror Intellectual Property.

(b) (i) The operation of the business of Acquiror and Nxu Tech as currently conducted does not infringe, misappropriate, dilute or otherwise violate, and since January 1, 2020, has not infringed, misappropriated, diluted or otherwise violated, any third-party Intellectual Property (provided that the foregoing representation 5.18(b)(i) is being made to the Knowledge of the Acquiror in respect of patents and patent applications) and (ii) to the Knowledge of Acquiror: no third party is infringing, misappropriating, diluting or otherwise violating, and no third party has infringed, misappropriated, diluted or otherwise violated since December 31, 2016, any Owned Acquiror Intellectual Property. Neither Acquiror nor Nxu Tech has received any written charge, complaint, claim, demand or notice alleging that the conduct of the business infringes, misappropriates, dilutes or otherwise violates any Intellectual Property of any other Person.

(c) There is no, and since January 1, 2020 there has not been any, Action pending or threatened in writing, or, to the Knowledge of Acquiror, orally (including "cease and desist" letters or invitations to take a license) (i) against Acquiror or Nxu Tech (A) challenging the ownership, validity, registrability, patentability, or enforceability of the Owned Acquiror Intellectual Property (excluding ordinary course office actions and similar exparte proceedings in connection with the prosecution of applications for the registration or issuance of any Intellectual Property) or (B) asserting that Acquiror or Nxu Tech has infringed, misappropriated, diluted or otherwise violated any third-party Intellectual Property, or (ii) brought by Acquiror or Nxu Tech (A) challenging the ownership, validity, registrability, patentability, or enforceability of the Intellectual Property of any third party (excluding ordinary course office actions and similar exparte proceedings in connection with the prosecution of applications for the registration or issuance of any Intellectual Property (excluding ordinary course office actions and similar exparte proceedings in connection with the prosecution of applications for the registration or issuance of any Intellectual Property of any third party (excluding ordinary course office actions and similar exparte proceedings in connection with the prosecution of applications for the registration or issuance of any Intellectual Property) or (B) asserting that any third party has infringed, misappropriated, diluted or otherwise violated any Owned Acquiror Intellectual Property.

(d) Each of Acquiror and Nxu Tech has taken commercially reasonable measures to protect, enforce and maintain (i) the confidentiality of all material proprietary information and trade secrets included in the Owned Acquiror Intellectual Property or of third parties provided to Acquiror or Nxu Tech, as applicable, under obligations of confidentiality, and (ii) its ownership of, and rights in, all Owned Acquiror Intellectual Property. Without limiting the foregoing, neither Acquiror nor Nxu Tech has made any of its trade secrets or other material confidential or proprietary information that it intended to maintain as confidential information (including source code of any Acquiror Software) available to any other Person except pursuant to written binding agreements requiring such Person to maintain the confidentiality of such confidential information or trade secrets.

5.19 Data Protection. Except as would not reasonably be expected to have a Material Adverse Effect on Acquiror or Nxu Tech:

(a) Each of Acquiror and Nxu Tech is, and at all times since January 1, 2020 has been, (i) in material compliance with all Privacy Laws, Data Security Requirements, and Acquiror's or Nxu Tech's, as applicable, contractual requirements relating to the Acquiror's or Nxu Tech's, as applicable, Processing of Personal Information (collectively, the "<u>Acquiror Privacy Commitments</u>"), and (ii) has not been subject to any Actions, regulatory audits or investigations by any Person or Governmental Authority relating to the Acquiror Privacy Commitments. Each of Acquiror and Nxu Tech has appointed a data protection officer if required to do so under Privacy Laws, and complied at all times with and made privacy notices (complete and accurate copies of which have been made available to the Company) available to all individuals about whom the Acquiror or Nxu Tech, as applicable, Processes or directs the Processing of Personal Information, in conformance with Acquiror Privacy Commitments, including with respect to the collection, processing, receipt, security, storage, transmission, transfer (including cross-border transfers), disclosure, destruction and use of Personal Information. Each of Acquiror and Nxu Tech has taken commercially reasonable steps to ensure that all Personal Information is protected against loss or unauthorized Processing, including by maintaining and enforcing policies, procedures, and rules regarding data privacy, protection, and security to protect the security, integrity and privacy of Personal Information as required by all Acquiror Privacy Commitments.

(b) Since January 1, 2020, (i) to Acquiror's knowledge, there has not been any breach, security incident, loss, theft, modification, destruction, alteration, disclosure or other misuse of, or third party claims, complaints, fines, practice recommendation, request, or proceeding of any nature alleging, unauthorized access or unauthorized use of any Personal Information or confidential business information maintained by any Acquiror Data Related Vendor on behalf of the Acquiror or Nxu Tech and (ii) neither Acquiror nor Nxu Tech has received any written requests, complaints or objections to its collection or use of Personal Information from any Governmental Authority or other third party (including data subjects). No Person has threatened to bring or been awarded compensation from Acquiror or Nxu Tech under any Acquiror Privacy Commitments and there is no basis for any such claim to be brought that may give rise or has given rise to any liability, and no written claim for such compensation is outstanding. Each of Acquiror and Nxu Tech has complied with all data subject requests, including any requests for access to Personal Information, the cessation of specified processing activities or the rectification or erasure of any Personal Information, in each case in accordance with the requirements of applicable Privacy Laws.

(c) Neither Acquiror nor Nxu Tech sell, rent or otherwise make available to any Personal Information, except in a manner that complies with all applicable Privacy Laws. The execution, delivery and performance of this Agreement and the transaction contemplated herein comply and will comply with all Acquiror Privacy Commitments.

(d) Each of Acquiror and Nxu Tech, as applicable, has entered into written contracts with each third-party service provider, vendor and business partner that has access (including storage) to Personal Information which complies with all applicable requirements of Acquiror Privacy Commitments ("<u>Acquiror Data Related Vendors</u>"), including payment processors, advertising and marketing agencies, cloud storage vendors and outsourced technology or human resource functions, containing all appropriate, sufficient and legally required contractual clauses regarding privacy and information security. The Acquiror has made available to the Company true and accurate copies of all contracts with any Acquiror Data Related Vendor.

(e) To the extent that either of Acquiror or Nxu Tech collect Personal Information of Persons who reside outside of the United States, the EU, and/or the UK, and/or Canada, respectively, Acquiror or Nxu Tech, as applicable, has implemented sufficient mechanisms to ensure that the transfers of such Personal Information from such Persons' home country to any other country complies with Privacy Laws restricting the transfer of such Personal Information.

5.20 <u>Information Technology and Cybersecurity</u>. Except as would not reasonably be expected to have a Material Adverse Effect on Acquiror or Nxu Tech:

(a) The IT Systems used in the conduct of Acquiror's and Nxu Tech's business (i) operate and perform in accordance with their documentation and functional specifications; (ii) are adequate for, and operate and perform in all material respects as required in connection with, the needs and operation of Acquiror's and Nxu Tech's business as currently conducted and are fulfilling the purposes for which they were acquired, licensed or established in an efficient manner without material downtime or errors; and (iii) are free from any bugs, computer virus, unauthorized disabling or erasing mechanism, worm, unauthorized software lock, drop dead device, Trojan horse, back door, trap door or time bomb or other software routines or hardware components that permit unauthorized access or the unauthorized disablement or erasure of the IT Systems. There has been no failure or other material substandard performance of any IT Systems which has caused any disruption to the business of Acquiror or Nxu Tech.

(b) Each of Acquiror and Nxu Tech has implemented with respect to the IT Systems used in its business commercially reasonable (i) steps to provide for backup, security and disaster recovery technology consistent with industry practices and (ii) actions (including implementing and monitoring compliance with administrative, technical and physical safeguards, policies, procedures and security measures that conform with all applicable contractual obligations and Privacy Laws) to protect the operation, confidentiality, integrity and security of its IT Systems and the information (including trade secrets and Personal Information) and data stored thereon or transmitted thereby, including from unauthorized use or access by third parties and from viruses and contaminants. Each of Acquiror and Nxu Tech has carried out and documented reasonable security and technology audits, investigations, data mapping exercises, impact assessments, and other activities relating to Acquiror's and Nxu Tech's information security practices, and no such audits, investigations, assessments and/or reports have identified material risks and/or vulnerabilities in Acquiror's or Nxu Tech's IT Systems.

(c) There has been no security breach of, or unauthorized use or access to, the Acquiror or Nxu's IT Systems (including ransomware attacks), including any breach which resulted in the unauthorized use, misappropriation, modification, encryption, corruption, disclosure or transfer of any information or data contained therein or transmitted thereby (including trade secrets and Personal Information).

(d) The Open Source Software used in the Acquiror and Nxu Tech Software and Proprietary Databases does not have license or other usage terms that require as a condition of how Acquiror or Nxu Tech, as applicable, currently uses or has used the Acquiror and Nxu Tech Software and Proprietary Databases, that any Software or databases or data incorporated into, derived from or distributed with the Acquiror and Nxu Tech Software and Proprietary Databases be (i) disclosed or, in the case of Software, distributed in source code form; (ii) licensed for the purpose of making derivative works; (iii) redistributable at no charge; or (iv) subject to any other material limitation, restriction or condition on the right or ability of Acquiror or Nxu Tech, as applicable, has used any Acquiror or Nxu Tech Software or Proprietary Database disseminated by an entity on behalf of Acquiror or Nxu Tech, as applicable, that is installed on consumers' or partners' device(s) and used by any Person on behalf of Acquiror or Nxu Tech, as applicable, that is installed on consumers' or partners' device(s) on which it is installed, or about data that is stored or created on, transmitted from or transmitted to the computers on which it is installed in a manner that violates any applicable Acquiror Privacy Commitments. Neither Acquiror nor Nxu Tech has entered in any escrow or other agreements with any Person granting access to any Acquiror and Nxu Tech Software and Proprietary Databases.

5.21 <u>Real Property</u>.

(a) Neither of Acquiror nor Nxu Tech own any fee simple interest in real property.

(b) <u>Schedule 5.21(b)</u> contains a complete and accurate list by property, city, state and country, of all land and real property leasehold or subleasehold estates and other rights to use or occupy any interest in land or real property held by Acquiror or Nxu Tech as of the date of this Agreement (the "<u>Acquiror Properties</u>"). The Acquiror Properties are the only properties used by Acquiror or Nxu Tech in, or otherwise related to, Acquiror's business as of the date of this Agreement, and subject to any permitted action pursuant to <u>Section 7.02</u>, as of the Closing Date. Acquiror is the sole legal and beneficial owner of (or is solely legally and beneficially entitled to) a leasehold interest in, or a right to use or occupy, the Acquiror Properties. Neither Acquiror nor Nxu Tech has leased or otherwise granted to any Person the right to use or occupy any Acquiror Properties or any portion thereof.

(c) <u>Schedule 5.21(c)</u> contains a complete and accurate list and description of all leases, subleases, licenses, concessions, and other Contracts, agreements and leasehold or land use arrangements and all related supplemental or ancillary documents pursuant to which Acquiror or Nxu Tech leases, licenses, subleases or otherwise occupies any Acquiror Property on the date hereof, except for any leases or licenses which arrange for a temporary occupancy arrangement of less than six months (collectively, the "<u>Acquiror Lease Documents</u>"). Acquiror has delivered to the Company a true and complete copy of each such Acquiror Lease Document. Neither Acquiror nor Nxu Tech is and, to the Knowledge of Acquiror, no other party to any Lease Document is in material breach or material default under such Acquiror Lease Document, nor has any event occurred which with notice or the passage of time or both would constitute a breach or default under any Acquiror Lease Document.

(d) To the Knowledge of Acquiror, each Acquiror Lease Document is a written agreement in full force and effect, is valid, binding and enforceable, subject to proper authorization and execution of each Acquiror Lease Document by the other parties thereto. Acquiror or Nxu Tech has paid the rent and all other sums that are due and payable under such Acquiror Lease Documents and there are no material arrears nor any sums which have been waived, deferred or accelerated, and no rent reviews are outstanding, in progress nor have been deferred. All consents, permits and approvals required for the grant of each Acquiror Lease Document have been obtained and complied with in all material respects.

(e) To the Knowledge of Acquiror, except as would not reasonably be expected to have a Material Adverse Effect on Acquiror or Nxu Tech, there exists no restrictions, covenants or encumbrances that prevent any of the Acquiror Properties from being used now or in the future for their current use and at materially the same cost as at present or would prevent or require consent from a third party as a result of the transactions contemplated by this Agreement.

(f) There are no outstanding options, rights of first offer or rights of first refusal to purchase any Acquiror Properties or any portion thereof or interest therein (which are binding on or in favor of Acquiror). There are no Contracts relating to the right to receive any portion of the income or profits from the sale, operation or development of any Acquiror Properties or any portion thereof or interest therein. Neither Acquiror nor Nxu Tech is a party to any agreement or option to purchase any land or real property or interest therein, nor is in the process of negotiating any such agreement or option to purchase as at the date of this Agreement.

(g) Except as would not reasonably be expected to have a Material Adverse Effect on Acquiror or Nxu Tech, as of the date hereof, each Acquiror Property and any structures built on them comply with all applicable Laws, the current use of each Acquiror Property is the lawful use, no Acquiror Property is subject to any restrictions relating to flood zoning limiting its use in any material respect, and there are no material outstanding or threatened disputes, actions, claims, demands, adverse notices or complaints to which Acquiror has received notice or is a party in respect of any of the Acquiror Properties.

(h) As of the date hereof, there are no pending, or, to the Knowledge of Acquiror, threatened, material appropriation, condemnation, eminent domain, compulsory purchase or like proceedings relating to the whole or any part of any Acquiror Properties.

(i) Neither Acquiror nor Nxu Tech has any material actual or contingent liability in respect of previously owned, leased, licensed, used or occupied land or buildings.

5.22 <u>Corrupt Practices; Sanctions</u>. Except as would not reasonably be expected to have a Material Adverse Effect on Acquiror or Nxu Tech:

(a) Neither Acquiror, Nxu Tech, any directors, officers or employees of Acquiror or Nxu Tech, nor, to the Knowledge of Acquiror, any of their other Representatives have violated, conspired to violate or aided and abetted the violation of any Anti-Corruption Laws. Acquiror (i) has instituted policies and procedures reasonably designed to ensure compliance with Anti-Corruption Laws and anti-money laundering Laws and (ii) has maintained such policies and procedures in force. No Government Official nor any of his or her immediate family members is an officer or director or owns any securities of Acquiror or Nxu Tech.

(b) Neither Acquiror, Nxu Tech, any directors, officers or employees of Acquiror or Nxu Tech nor, to the Knowledge of Acquiror, any of their other Representatives, is engaging and has not engaged, directly or indirectly, in any activities (including sales, reselling, licensing or sub-licensing arrangements, funding, making payments, procuring, insurance or otherwise providing assistance or support in connection with operations, business or any other activity) with or involving any Sanctioned Person or Sanctioned Territory, nor otherwise violated any applicable Sanctions or Ex-Im Laws.

(c) Neither Acquiror nor Nxu Tech, nor any of their directors, officers, employees, nor to the Knowledge of Acquiror, any other Persons acting for or on behalf of any of the foregoing is, or has been, a Sanctioned Person.

(d) Neither Acquiror nor Nxu Tech, nor, to the Knowledge of Acquiror, any Representatives of Acquiror or Nxu Tech, is currently or has in the past been a subject, target, or party of any current investigation, allegation (whether raised internally or externally), request for information, voluntary self-disclosure, or any other inquiry by or involving any Governmental Authority regarding the actual or possible violation of Sanctions, and neither Acquiror nor Nxu Tech has received any notice that there is any investigation, allegation, request for information, or any other inquiry by any Governmental Authority regarding an actual or possible violation of Sanctions.

5.23 <u>Competition and Trade Regulation</u>. Except as would not reasonably be expected to have a Material Adverse Effect on Acquiror or Nxu Tech:

(a) Acquiror, Nxu Tech and their Representatives have been and currently are in compliance with Sanctions and Ex-Im Laws. Acquiror also maintains policies and procedures reasonably designed to ensure compliance with Sanctions and Ex-Im Laws and are following such policies and procedures.

(b) Each of Acquiror and Nxu Tech is in compliance with all applicable antitrust Laws in all material respects. Neither Acquiror nor Nxu Tech is nor has been a party to or is or has been concerned in any agreement or arrangement with a Governmental Authority under any anti-trust, competition or similar legislation in any jurisdiction in which Acquiror or Nxu Tech has assets or carries or intends to carry on business or where its activities may have an effect.

(c) Neither Acquiror nor Nxu Tech currently, nor expect in the future to, produce, design, test, manufacture, fabricate, or develop any critical technologies, as defined under 31 C.F.R. § 800.215.

5.24 <u>Environmental Matters</u>. Except as set forth in <u>Schedule 5.24</u> or as would not reasonably be expected to have a Material Adverse Effect on Acquiror or Nxu Tech:

(a) Each of Acquiror and Nxu Tech is and has been, in compliance in all material respects with all Environmental Laws, which material compliance includes possessing and complying with all Acquiror Permits required under Environmental Laws in connection with the operation of Acquiror's business or the Acquiror Properties ("Acquiror Environmental Permits");

(b) Neither Acquiror nor Nxu Tech has received any notice alleging any violation of or noncompliance by Acquiror or Nxu Tech with respect to any Environmental Law or Acquiror Environmental Permit;

(c) there are no Actions pending, or to the Knowledge of Acquiror, threatened, against Acquiror or Nxu Tech, and neither Acquiror nor Nxu Tech has received any written notification of, nor, to the Knowledge of Acquiror, is Acquiror or Nxu Tech otherwise responsible for any violation of, noncompliance with or material liability under, Environmental Laws, including for the contamination by or manufacture, management, labeling, registration, import, export, generation, storage, treatment, use, transportation, disposal, Release or threatened Release at any location of, or exposure of any Person to, any Hazardous Material;

(d) Neither Acquiror nor Nxu Tech (i) is a party to or subject to the provisions of any Governmental Order pursuant to Environmental Law or (ii) has retained or assumed by Contract any material liabilities or obligations pursuant to Environmental Laws;

(e) there have been no Releases of any Hazardous Materials at, on, under, or from the current or former Acquiror Properties, or to the Knowledge of Acquiror, any third-party property that would result in a requirement for notification to or investigation by a Governmental Authority, or the need for investigation, remediation, removal, cleanup or monitoring of Hazardous Materials pursuant to Environmental Laws; and

(f) Acquiror has furnished to the Company, or has otherwise made available for inspection by the Company, all material Phase I or Phase II environmental reports and similar material environmental documents in the possession of Acquiror and Nxu Tech.

5.25 <u>Affiliate Transactions</u>. Since January 1, 2022, all transactions required to be disclosed by Acquiror and Nxu Tech pursuant to Item 404 of Regulation S-K have been disclosed in the Acquiror SEC Reports. No transaction, arrangement or relationship, or series of similar transactions, arrangements or relationships, is currently proposed by Acquiror or Nxu Tech or, to the Knowledge of Acquiror, by any other Person to which Acquiror or Nxu Tech would be a participant that would be required to be disclosed under Item 404 of Regulation S-K if consummated.

5.26 Insurance. Acquiror and Nxu Tech maintain such insurance policies or binders of fire, liability, umbrella liability, real and personal property, workers' compensation, vehicular, directors' and officers' liability, fiduciary liability and other casualty and property insurance as commercially reasonable and appropriate (the "<u>Acquiror Insurance Policies</u>"). To the Knowledge of Acquiror, there are no events, circumstances or other liabilities that give rise to a material claim under the Acquiror Insurance Policies. Except as has not had or would not reasonably be expected to have a Material Adverse Effect on Acquiror or Nxu Tech, the Acquiror Insurance Policies are in full force and effect as of the date of this Agreement with respect to Acquiror and Nxu Tech, and the limits thereunder have not been impaired, exhausted or materially diminished. As of the date hereof, Acquiror has not received any written notice of cancellation of, of a material premium increase (relative to others in the industry in which Acquiror operates) with respect to, or of a material alteration of coverage under, any Acquiror Insurance Policies (i) are valid and binding in accordance with their terms, subject to Enforceability Exceptions and (ii) have not been subject to any lapse in coverage. There are no material claims related to Acquiror or Nxu Tech or their assets, business, operations, employees, officers and directors pending under any such Acquiror Insurance Policies as to which coverage has been denied or disputed or in respect of which there is an outstanding reservation of rights.

5.27 <u>COVID-19</u>. Acquiror has disclosed its or Nxu Tech's participation in the federal Paycheck Protection Program in the Acquiror SEC Reports in all material respects.

5.28 <u>Takeover Statutes and Charter Provisions</u>. As of the date hereof and through the Second Merger Effective Time, no "fair price," "moratorium," "control share acquisition" or other anti-takeover statute or similar domestic or foreign Law applies with respect to Acquiror, Nxu Tech, Merger Sub I or Merger Sub II in connection with this Agreement, the Merger, the issuance of the Closing Merger Consideration or any of the other Transactions. As of the date hereof and through the Second Merger Effective Time, there is no stockholder rights plan, "poison pill" or similar antitakeover agreement or plan in effect to which Acquiror, Merger Sub I or Merger Sub II are subject, a party or otherwise bound.

5.29 <u>No Other Representations or Warranties</u>. The representations and warranties made by Acquiror, Merger Sub I and Merger Sub II in this <u>Article V</u> are the exclusive representations and warranties made by Acquiror, Merger Sub I, Merger Sub I nor Merger Sub II, nor any other Person, has made or makes any other express or implied representation or warranty, either written or oral, on behalf of Acquiror, Merger Sub I or Merger Sub I nor Merger Sub I or Merger Sub II, to the accuracy or completeness of any information regarding Acquiror, Merger Sub I or Merger Sub II available to the other parties or their respective Representatives and expressly disclaims any such other representations or warranties. Without limiting the foregoing, neither Acquiror, Merger Sub I nor Merger Sub I nor Merger Sub II, nor any other Person, makes or has made any representation or warranty to the other parties hereto with respect to, and shall have no liability in respect of, (a) any financial projection, forecast, estimate, budget or prospect information relating to Acquiror, Merger Sub I or Merger Sub I in this Article V, written information made available to the other parties hereto in the course of their evaluation of Acquiror, Merger Sub I and Merger Sub II and the negotiation of this Agreement or in the course of the Transactions.

ARTICLE VI COVENANTS OF THE COMPANY

6.01 Conduct of Business. From the date of this Agreement until the earlier of the Closing Date or the termination of this Agreement in accordance with its terms (the "Interim Period"), the Company shall, except as set forth on Schedule 6.01, as expressly contemplated by this Agreement or as consented to by Acquiror in writing (which consent shall not be unreasonably conditioned, withheld or delayed), or as may be required by Law (including COVID-19 Measures), (i) conduct and operate its business in the ordinary course consistent with past practice in all material respects, (ii) use reasonable best efforts to preserve intact the current business organization and ongoing businesses of the Company and maintain the existing relations and goodwill of the Company with customers, suppliers, distributors and creditors of the Company and (iii) use reasonable best efforts to keep available the services of its present officers; provided that in the case of each of the preceding clauses (i)-(iii), during any period of full or partial suspension of operations related to COVID-19, or any similar national or international health concern, the Company may, in connection with therewith, take such actions in good faith as are reasonably necessary (A) to protect the health and safety of the Company's employees and other individuals having business dealings with the Company or (B) to respond to third-party supply or service disruptions caused thereby, including, but not limited to COVID-19 Measures, and any such actions taken (or not taken) as a result of, in response to, or otherwise related to such health concerns shall be deemed to be taken in the "ordinary course of business" for all purposes of this Section 6.01 and not be considered a breach of this Section 6.01; provided, further, that following any such suspension, to the extent that the Company took any actions pursuant to the immediately preceding proviso that caused deviations from its business being conducted in the ordinary course of business consistent with past practice, to resume conducting its business in the ordinary course of business consistent with past practice in all material respects as soon as reasonably practicable. Without limiting the generality of the foregoing, except as set forth on Schedule 6.01, as expressly contemplated by this Agreement or as consented to by Acquiror in writing (which consent shall not be unreasonably conditioned, withheld or delayed), or as may be required by Law, the Company shall not, during the Interim Period:

(a) change or amend organizational documents of the Company;

(b) declare, make or pay any dividend or other distribution (whether in cash, equity or property) to stockholders of the Company or repurchase or redeem any Company Stock;

(c) create, allot, issue, redeem or repurchase or agree to create, allot, issue, redeem or repurchase any shares or other securities of whatsoever nature convertible into shares (or any option to subscribe for the same) of the Company (excluding repurchases in connection with termination of employment pursuant to an existing repurchase right);

(d) enter into, or amend or modify any material term of, terminate, or waive or release any material rights, claim or benefits under any Company Material Contract (or any Contract, that if existing on the date hereof, would be a Material Contract), other than entry into, amendments of, modifications of, terminations of, or waivers or releases under, such Company Material Contracts in the ordinary course of business;

(e) enter into, or amend or modify any material term of, terminate, or waive or release any material rights, claim or benefits under any Contract, Lease Document or any document governing the occupation of real property, or other arrangement to which the Company, on one hand, and a Company Stockholder or its Affiliate, on the other hand, are parties or by which they are bound or which is for the benefit of a Company Stockholder or its Affiliates, other than entry into, amendments of, modifications of, terminations of, or waivers or releases under, such Contracts or arrangements either (i) in the ordinary course of business consistent with past practice or (ii) involving an annual aggregate payment of less than \$1,000,000;

(f) sell, transfer, lease, abandon, cancel, fail to maintain, let lapse or convey, surrender or dispose of any assets, properties or business of the Company (other than Company Intellectual Property and Company Software), except in the ordinary course of business and in an amount not in excess of \$250,000 in the aggregate;

(g) (i) sell, assign, license, transfer, pledge, grant or suffer to exist any Lien on, convey or otherwise dispose of any Owned Company Intellectual Property, other than non-exclusive licenses of Intellectual Property granted to customers and service providers in the ordinary course of business or (ii) fail to continue to prosecute or defend, allow to enter the public domain, abandon, cancel, fail to renew or maintain or otherwise allow to lapse any material Owned Company Intellectual Property;

(h) fail to continue to protect and maintain the confidentiality of any trade secrets or material proprietary information included in the Owned Company Intellectual Property;

(i) except as otherwise required pursuant to applicable Law, (i) adopt, enter into, amend, modify, or terminate any Company Benefit Plan or any collective bargaining or similar agreement to which the Company is a party or by which it is bound, (ii) grant or provide any equity or equitybased awards (or obligations in respect thereto), severance or termination payments, deferred compensation, or transaction, retention or change in control payments or benefits to any current or former director, employee, officer or other individual service provider of the Company, except in connection with the promotion, hiring or firing (in each case, to the extent permitted by clause (v)) of any employee in the ordinary course of business and consistent with past practice or which is less than \$200,000 in aggregate value, (iv) recognize or certify any labor union, works council, other labor organization or group of employees as the bargaining Representative for any employees of the Company, except to the extent already in existence or as required by Law, (v) implement or announce any employee layoffs, plant closings, reductions-in-force, furloughs, temporary layoffs, reduction in terms and conditions of employment, or other actions that could implicate any WARN Act, or (vi) waive or release any noncompetition, nonsolicitation, nondisclosure, noninterference, nondisparagement or other restrictive covenant obligation of any current or former employee or independent contractor;

(j) (i) fail to maintain its existence or acquire by merger or consolidation with, or merge or consolidate with, or purchase a material portion of the assets or equity of, any corporation, partnership, limited liability company, association, joint venture or other business organization or division thereof, other than such acquisitions and purchases that would not require financial statements of the acquired business to be included in the Registration Statement pursuant to Rule 3-05 of Regulation S-X under the Securities Act, or (ii) adopt or enter into a plan of complete or partial liquidation, dissolution, merger, consolidation, restructuring, recapitalization or other reorganization of the Company (other than the Transactions);

(k) make any capital expenditures (or commitment to make any capital expenditures) that in the aggregate exceed \$2,500,000, other than any capital expenditure (or series of related capital expenditures) (i) consistent in all material respects with historical practices or (ii) reasonably forecasted by the Company and included in the Company's capital expenditure projections as of the date of this Agreement;

(1) make any loans, advances or capital contributions to, or investments in, any other Person (including to any of its officers, directors, agents or consultants), or enter into any agreement in respect of the acquisition of real property, make any material change in its existing borrowing or lending arrangements for or on behalf of such Persons, or enter into any "keep well" or similar agreement to maintain the financial condition of any other Person, except (i) advances to employees or officers of the Company in the ordinary course of business consistent with past practice, and (ii) extended payment terms for customers in the ordinary course of business;

(m) make, revoke or change any material Tax election, change any Tax accounting method or period, file any amendment to a material Tax Return, enter into any agreement with a Governmental Authority with respect to Taxes, settle or compromise any examination, audit or other Action with a Governmental Authority of or relating to Taxes, consent to any extension or waiver of the statutory period of limitations applicable to any claim or assessment in respect of Taxes, incur any liability for Taxes outside the ordinary course of business, or enter into any Tax sharing, indemnification, allocation or similar agreement or arrangement (excluding pursuant to customary provisions in any commercial Contract entered into in the ordinary course of business; movided, for the avoidance of doubt the Company may pay Taxes in the ordinary course of business;

(n) waive, release, compromise, settle or satisfy any pending or threatened material claim (which shall include, but not be limited to, any pending or threatened Action) or compromise or settle any related liability, other than in the ordinary course of business consistent with past practice or where such waiver, release, compromise, settlement or satisfaction involves monetary damages not to exceed \$250,000 in the aggregate;

(o) incur, issue, assume, guarantee or otherwise become liable for any Indebtedness, or in any material respect, modify any Indebtedness, other than Indebtedness in an amount not to exceed \$250,000.

(p) enter into any material new line of business outside of the business currently conducted by the Company as of the date of this Agreement other than natural extensions of existing lines of business;

(q) make any material change in financial accounting methods, principles or practices, except insofar as may have been required by the SEC, by a change in GAAP (including pursuant to standards, guidelines and interpretations of the Financial Accounting Standards Board or any similar organization), or by applicable Law;

(r) voluntarily fail to maintain, cancel or change coverage under, in a manner materially detrimental to the Company, any Company Insurance Policy;

(s) liquidate, dissolve, reorganize or otherwise wind up the business or operations of the Company; and

(t) enter into any agreement or undertaking to do any action prohibited under this <u>Section 6.01</u>.

Notwithstanding anything to the contrary herein, the Company may issue one or more Company Interim Notes during the Interim Period.

6.02 <u>Inspection</u>. The Company shall afford to Acquiror and its Representatives reasonable access during the Interim Period, during normal business hours and with reasonable advance written notice, in such manner as to not interfere with the normal operation of the Company, to all of their respective properties, books, projections, plans, systems, Contracts, commitments, Tax Returns, records, commitments, analyses and appropriate officers and employees of the Company, and shall furnish such Representatives with all financial and operating data and other information concerning the affairs of the Company and that are in the possession of the Company as such Representatives may reasonably request. All information obtained by Acquiror and its Representatives under this Agreement shall be subject to the Confidentiality Agreement. Notwithstanding the foregoing, the Company shall not be required to provide to Acquiror or any of its Representatives any information (a) if and to the extent doing so would (i) violate any Law to which Company is subject, (ii) result in the disclosure of any trade secrets of third parties in breach of any Contract with such third party, (iii) violate any legally-binding obligation of the Company with respect to confidentially, non-disclosure or privacy, if Company shall have used commercially reasonable efforts (without payment of any consideration, fees or expenses) to obtain the consent of such third party to such inspection or disclosure or (iv) jeopardize protections afforded to Company under the attorney-client privilege or the attorney work product doctrine; provided that, in case of each of clauses (i) through (iv), the parties shall use commercially reasonable efforts to make alternative arrangements for such disclosure in a manner that does not result in the events set out in clauses (i) through (iv).

6.03 Proxy Solicitation; Other Actions.

(a) The Company shall be available to, and the Company shall use reasonable best efforts to make its officers and employees available to, in each case, during normal business hours and upon reasonable advanced notice, Acquiror and its counsel in connection with the drafting of the Registration Statement and responding in a timely manner to comments on the Registration Statement from the SEC. The Company agrees to provide Acquiror as promptly as practicable following the date hereof, (i) unaudited financial statements, including condensed balance sheets and condensed statements of operations, stockholders' equity and cash flows, of the Company for each fiscal quarter beginning with the fiscal quarter ending June 30, 2024 and ending at least 45 days prior to the date on which the Registration Statement is effective, in each case, prepared in accordance with GAAP and Regulation S-X and (ii) auditor's reports and consents to use such financial statements and reports and the Company Audited Financial Statements in the Registration Statement. Without limiting the generality of the foregoing, the Company shall use reasonable best efforts to cooperate with Acquiror in connection with Acquiror's preparation for inclusion in the Registration Statement of pro forma financial statements that comply with the requirements of Regulation S-X under the rules and regulations of the SEC (as interpreted by the staff of the SEC) to the extent such pro forma financial statements are required by Form S-4.

(b) From and after the date on which the Registration Statement becomes effective under the Securities Act until the Closing Date, the Company will give Acquiror prompt written notice of any action taken or not taken by the Company or of any development regarding the Company, in any such case which is known by the Company, that would cause the Registration Statement to contain an untrue statement of a material fact or omit to state a material fact necessary in order to make the statements, in light of the circumstances under which they were made, not misleading; provided that, if any such action shall be taken or fail to be taken or such development shall otherwise occur, Acquiror and the Company shall cooperate fully to cause an amendment or supplement to be made promptly to the Registration Statement, such that the Registration Statement no longer contains an untrue statement of a material fact or omit to state a material fact necessary in order to make the statements, in light of the circumstances under which they were made, not misleading; provided, further, however, that no information received by Acquiror pursuant to this Section 6.03 shall operate as a waiver or otherwise affect any representation, warranty or agreement given or made by the party who disclosed such information, and no such information shall be deemed to change, supplement or amend the Schedules.

6.04 Non-Solicitation; Acquisition Proposals.

(a) From the date of this Agreement until the First Merger Effective Time or, if earlier, the valid termination of this Agreement in accordance with <u>Section 10.01</u>, the Company shall not and shall use its reasonable best efforts to cause their respective Representatives not to, directly or indirectly:

(i) initiate, solicit or knowingly encourage or knowingly facilitate any inquiries or requests for information with respect to, or the making of, any inquiry regarding, or any proposal or offer that constitutes, or which is otherwise intended or is reasonably likely to result in or lead to, an Acquisition Proposal;

(ii) engage in, continue or otherwise participate in any negotiations or discussions concerning, or provide access to its properties, books and records or any confidential information or data to, any Person relating to any proposal, offer, inquiry or request for information that constitutes, or which is otherwise intended or is reasonably likely to result in or lead to, an Acquisition Proposal;

(iii) approve, endorse or recommend, or propose publicly to approve, endorse or recommend, any Acquisition Proposal;

(iv) execute or enter into, any letter of intent, memorandum of understanding, agreement in principle, confidentiality agreement, merger agreement, acquisition agreement, exchange agreement, joint venture agreement, partnership agreement, option agreement or other similar agreement for or relating to any Acquisition Proposal; or

(v) resolve or agree to do any of the foregoing.

The Company also agrees that immediately following the execution of this Agreement it shall use its reasonable best efforts to cause its Representatives to, cease any solicitations, discussions or negotiations with any Person (other than Acquiror, the Merger Subs and their respective Representatives) conducted heretofore in connection with an Acquisition Proposal or any inquiry or request for information that could reasonably be expected to lead to, or result in, an Acquisition Proposal. The Company also agrees that within three (3) Business Days of the execution of this Agreement, the Company shall request each Person (other than the parties hereto and their respective Representatives) that has prior to the date hereof executed a confidentiality agreement in connection with its consideration of acquiring the Company (and with whom the Company has had contact in twelve (12) months prior to the date of this Agreement regarding the acquisition of the Company) to return or destroy all confidential information furnished to such Person by or on behalf of it prior to the date hereof and terminate access to any physical or electronic data room maintained by or on behalf of the Company. The Company shall promptly (and in any event within one (1) Business Day) notify, in writing, Acquiror of the receipt of any inquiry, proposal, offer or request for information received after the date hereof that constitutes, or would reasonably be expected to result in or lead to, any Acquisition Proposal. Without limiting the foregoing, it is understood that any violation of the restrictions contained in this <u>Section 6.04</u> by any of the Company's or Representatives acting on the Company's behalf, shall be deemed to be a breach of this <u>Section 6.04</u> by the Company.

(b) For purposes of this Agreement, the following terms shall have the meanings assigned below:

(i) "Acquisition Proposal" means any proposal or offer from any Person or "group" (as defined in the Exchange Act) (other than Acquiror, the Merger Subs or their respective Affiliates) relating to, in a single transaction or series of related transactions, (A) any direct or indirect acquisition or purchase of a business that constitutes 20% or more of the net revenues, net income or assets of the Company, (B) any direct or indirect acquisition of 20% or more of the assets of the Company (based on the fair market value thereof, as determined in good faith by the Company Board), (C) acquisition of beneficial ownership, or the right to acquire beneficial ownership, of 20% or more of the total voting power of the equity securities of the Company to the extent such Person does not beneficially own 20% or more of the total voting power of the equity securities of the Company prior to such tender offer or exchange offer, or any merger, reorganization, consolidation, share exchange, business combination, recapitalization, liquidation, dissolution or similar transaction involving the Company, (D) any issuance or sale or other disposition (including by way of merger, reorganization, division, consolidation, share exchange, business combination, recapitalization or other similar transaction) of 20% or more of the total voting power of the total voting by way of merger, reorganization, division, consolidation, share exchange, business combination, recapitalization or other similar transaction) of 20% or more of the total voting power of the company or (E) any other transaction that is similar in nature to the Transactions or otherwise materially impedes or conflicts with the Transactions.

6.05 <u>Company Lock-up</u>. Prior to the Closing, the Company shall use commercially reasonable efforts to procure each Company Stockholder with greater than five percent ownership in the Company to enter into a Lock-Up Agreement with Acquiror in substantially the form attached as <u>Exhibit D</u> (the "<u>Company Lock-Up Agreement</u>").

6.06 <u>Litigation</u>. In the event that the Company is subject to any material litigation or material Action or other dispute resolution process before a third party unrelated to the dispute, whether as claimant, defendant or otherwise, or such Action or dispute resolution process is pending or threatened in writing, the Company shall promptly notify the Acquiror of any such litigation or Action and keep the Acquiror reasonably informed with respect to the status thereof.

6.07 <u>Conversion of Company Notes; Exercise of Company Warrants</u>. The Company shall procure that, prior to the Closing, all Company Notes shall be converted into Company Common Stock (and any securities under such Company Notes shall be released), and all Company Warrants shall be exercised for Company Common Stock, in each case, in accordance with their terms (the "Company Conversion").

6.08 <u>Company Support Agreement</u>. Concurrently with the execution of this Agreement, the Company shall deliver to Acquiror counterparts to the Company Support Agreement duly executed by Humanitario and the Company.

6.09 <u>Company Material Contract Consents</u>. The Company shall use commercially reasonable efforts to obtain such Company Material Contract Consents prior to the Closing to the extent the failure to do so would reasonably be expected to cause a Material Adverse Effect on the Company and shall use commercially reasonable efforts to obtain the other Company Material Contract Consents. "<u>Company Material Contract Consent</u>" shall mean any consent or waiver pursuant to any Company Material Contract that is required by the execution, delivery or performance of this Agreement and consummation of the Transactions.

6.10 <u>Voting Trust Transfer</u>. The Company shall cooperate with and assist Humanitario with entering into the Voting Trust Agreement, and, after the First Merger Effective Time, the Acquiror shall cooperate with Humanitario with completing the Voting Trust Transfer, in each case, in the manner and on the terms and conditions described in the Company Support Agreement.

6.11 <u>Interim Financial Statements</u>. Prior to the Closing, the Company shall prepare and deliver to Acquiror (i) an unaudited balance sheet as of the Company as of September 30, 2024 and the related unaudited statements of operations, stockholders' equity and cash flows of the Company for the 9-month period ending on such date, and (ii) an unaudited balance sheet of the Company as of September 30, 2023 and the related unaudited statement of operations, stockholders' equity and cash flows of the Company for the 9-month period ending on such date (collectively, the "Company Interim Financial Statements").

ARTICLE VII COVENANTS OF ACQUIROR

7.01 Indemnification and Insurance.

From and after the First Merger Effective Time, Acquiror and the Final Surviving Company agree that they shall indemnify and (a) hold harmless each present and former director and officer of the Company against any costs or expenses (including reasonable attorneys' fees), judgments, fines, losses, claims, damages or liabilities incurred in connection with any Action, whether civil, criminal, administrative or investigative, arising out of or pertaining to matters existing or occurring at or prior to the First Merger Effective Time, whether asserted or claimed prior to, at or after the First Merger Effective Time, to the fullest extent that the Company would have been permitted under applicable Law and its certificate of incorporation, bylaws and indemnification agreements in effect on the date of this Agreement to indemnify such Person (including the advancing of expenses as incurred to the fullest extent permitted under applicable Law). Without limiting the foregoing, Acquiror shall, and shall cause the Final Surviving Company and its Subsidiaries to, for a period of not less than six (6) years from the First Merger Effective Time (i) maintain provisions in its certificate of incorporation, bylaws, and indemnification agreements, to the extent applicable, concerning the indemnification and exculpation (including provisions relating to expense advancement) of officers and directors that are no less favorable to those Persons than the provisions of its certificate of incorporation, bylaws, and indemnification agreements, to the extent applicable, as of the date of this Agreement and (ii) not amend, repeal or otherwise modify such provisions in any respect that would adversely affect the rights of those Persons thereunder, in each case, except as required by Law. Acquiror shall assume, and be liable for, and shall cause the Final Surviving Company to honor, each of the covenants in this Section 7.01. Prior to the Closing, Acquiror shall obtain directors' and officers' liability insurance that shall be (A) effective as of the Closing and that provides coverage for Acquiror and its directors and officers both (1) on an ongoing basis, but no less than six (6) years from the Closing, and (2) with respect to claims arising from factors or events that occurred on or prior to the Closing, in each case in an amount and scope reasonably comparable to the existing policy of Acquiror, (B) at a cost that is reasonable and customary for insurance policies with Acquiror's existing directors' and officers' liability policy insurer or an insurer with a comparable insurer financial strength rating. On or prior to the Closing Date, Acquiror shall enter into customary indemnification agreements reasonably satisfactory to the Company with each of the post-Closing directors and officers of Acquiror, which indemnification agreements shall continue to be effective following the Closing.

(b) Notwithstanding anything contained in this Agreement to the contrary, this <u>Section 7.01</u> shall survive the consummation of the Merger indefinitely and shall be binding, jointly and severally, on Acquiror and the Final Surviving Company and all successors and assigns of Acquiror and the Final Surviving Company. In the event that Acquiror, the Final Surviving Company or any of their respective successors or assigns consolidates with or merges into any other Person and shall not be the continuing or surviving corporation or entity of such consolidation or merger or transfers or conveys all or substantially all of its properties and assets to any Person, then, and in each such case, Acquiror and the Final Surviving Company, as the case may be, shall succeed to the obligations set forth in this <u>Section 7.01</u>. The obligations of Acquiror and the Final Surviving Company under this <u>Section 7.01</u> shall not be terminated or modified in such a manner as to materially and adversely affect any present and former director and officer of the Company without the consent of the affected Person.

7.02 Conduct of Acquiror During the Interim Period.

(a) During the Interim Period, Acquiror and each of the Merger Subs shall, except as set forth on <u>Schedule 7.02(a)</u>, as expressly contemplated by this Agreement (including as contemplated by any Pre-Closing Financing), as consented to by the Company in writing (which consent shall not be unreasonably conditioned, withheld or delayed), or as may be required by Law, carry on their business in the manner conducted immediately prior to the execution and delivery of this Agreement and in accordance with applicable Law. During the Interim Period, except as set forth on <u>Schedule 7.02(a)</u>, or as expressly contemplated by this Agreement (including as contemplated by any Pre-Closing Financing) or as consented to by the Company in writing (which consent shall not be unreasonably conditioned, withheld or delayed), or as may be required by Law, Acquiror shall not and shall not permit the Merger Subs to:

Tech;

(i) change, modify or amend the Acquiror Organizational Documents or the organizational documents of Merger Sub or Nxu

(i) (A) make, declare, set aside or pay any dividends on, or make any other distribution (whether in cash, stock or property) in respect of any of its outstanding share capital, capital stock or other equity interests; (B) split, sub-divide, combine, reclassify or otherwise change any of its share capital, capital stock or other equity interests; or (C) repurchase, redeem or otherwise acquire, or offer to repurchase, redeem or otherwise acquire, any share capital, capital stock of, or other equity interests in, Acquiror;

(iii) enter into, renew or amend in any material respect, any agreement with any affiliate;

(iv) enter into, or amend or modify any material term of (in a manner adverse to Acquiror or the Merger Subs (including the Company)), terminate (excluding any expiration in accordance with its terms), or waive or release any material rights, claims or benefits under, any Acquiror Material Contracts (or any Contract, that if existing on the date hereof, would have been an Acquiror Material Contract) to which Acquiror or Merger Subs is a party or by which it is bound;

(v) waive, release, compromise, settle or satisfy any pending or threatened claim (which shall include, but not be limited to, any pending or threatened Action relating to this Agreement or otherwise) or compromise or settle any liability, other than (A) in the ordinary course of business or (B) to pay any amounts payable to landlord of Acquiror in connection with a real property lease or to pay the liabilities, fees and costs incurred in connection with the Transactions;

any Indebtedness;

(vi) incur, create, assume, refinance, guarantee or otherwise become liable for (whether directly, contingently or otherwise)

(vii) except with respect to the issuance of a single share of Series B Preferred Stock, offer, issue, deliver, grant or sell, or authorize or propose to offer, issue, deliver, grant or sell, any share capital or capital stock of, or other equity interests in, Acquiror or Merger Subs or any securities convertible into, or any rights, warrants or options to acquire, any such capital stock or equity interests;

(viii) (A) adopt any Benefit Plan other than as set forth in the Proposals, or enter into any employment Contract or any CBA, (B) hire any employee or any other individual independent contractor (not including advisors in connection with the Transactions) to provide services to Acquiror (including the Company) following Closing, (C) terminate any officer, employee or independent contractor (other than for cause), (D) make or grant any severance, bonus, including any change-in-control, retention, transaction, or similar compensation or benefits, or any increase in base salary or wages to any director, executive, or other current or former employee or service provider, or (E) take any action to accelerate the vesting of, or payment of, any compensation or benefit to any current or former employee or other service provider under any Company Benefit Plan or Benefit Plan maintained by the Acquiror;

(ix) (A) fail to maintain its existence or acquire by merger or consolidation with, or merge or consolidate with, or purchase a material portion of the assets or equity of, any corporation, partnership, limited liability company, association, joint venture or other business organization or division thereof; or (B) adopt or enter into a plan of complete or partial liquidation, dissolution, merger, consolidation, restructuring, recapitalization or other reorganization of Acquiror or the Merger Subs (other than the Transactions);

(x) make any capital expenditures, other than costs incurred in connection with (i) the termination of a real property lease, or (ii) the sale of any equipment;

(xi) make any loans, advances or capital contributions to, or investments in, any other Person (including to any of its officers, directors, agents or consultants), make any change in its existing borrowing or lending arrangements for or on behalf of such Persons, or enter into any "keep well" or similar agreement to maintain the financial condition of any other Person;

(xii) make, revoke or change any material Tax election, change any Tax accounting method or period, file any amendment to a material Tax Return, enter into any agreement with a Governmental Authority with respect to Taxes, settle or compromise any examination, audit or other Action with a Governmental Authority of or relating to Taxes, consent to any extension or waiver of the statutory period of limitations applicable to any claim or assessment in respect of Taxes, incur any liability for Taxes outside the ordinary course of business, or enter into any Tax sharing, indemnification, allocation or similar agreement or arrangement (excluding pursuant to customary provisions in any commercial Contract entered into in the ordinary course of business and not primarily related to Taxes); provided for the avoidance of doubt Acquiror and the Merger Subs may pay Taxes in the ordinary course of business;

Agreement;

(xiii) enter into any new line of business outside of the business currently conducted by Acquiror as of the date of this

(xiv) make any change in financial accounting methods, principles or practices, except insofar as may have been required by a change in GAAP (including pursuant to standards, guidelines and interpretations of the Financial Accounting Standards Board or any similar organization) or applicable Law;

(xv) voluntarily fail to maintain, cancel or materially change coverage under any insurance policy in form and amount equivalent in all material respects to the insurance coverage currently maintained with respect to Acquiror and its assets and properties;

(xvi) enter into any agreement with any broker, investment banker, or financial advisor pursuant to which such party shall be entitled to any broker's, financial advisor's or other similar fee or commission; or

(xvii) enter into any agreement or undertaking to do any action prohibited under this Section 7.02.

(b) Except as set forth in <u>Schedule 7.02(b)</u>, during the Interim Period, Acquiror shall, and shall cause the Merger Subs to comply with, and continue performing under, as applicable, the Acquiror Organizational Documents, and all other agreements or Contracts to which Acquiror or the Merger Subs may be a party.

7.03 Inspection. Acquiror shall afford to the Company, its Affiliates and their respective Representatives reasonable access during the Interim Period, during normal business hours and with reasonable advance notice, to all of their respective properties, books, projections, plans, systems, Contracts, commitments, Tax Returns, records, commitments, analyses and appropriate officers and employees of Acquiror, and shall furnish such Representatives with all financial and operating data and other information concerning the affairs of Acquiror that are in the possession of Acquiror as such Representatives may reasonably request. All information obtained by the Company, its Affiliates and their respective Representatives under this Agreement shall be subject to the Confidentiality Agreement prior to the First Merger Effective Time. Notwithstanding the foregoing, Acquiror shall not be required to provide to the Company or any of its Representatives any information (a) if and to the extent doing so would (i) violate any Law to which Acquiror or Merger Subs are subject, (ii) result in the disclosure of any trade secrets of third parties in breach of any Contract with such third party, (iii) violate any legally-binding obligation of Acquiror or Merger Subs under the attorney-client privilege or the attorney work product doctrine; provided that, in case of each of clauses (i) through (iv), the parties shall use commercially reasonable efforts to make alternative arrangements for such disclosure in a manner that does not result in the events set out in clauses (i) through (iv).

7.04 <u>Acquiror NASDAQ Listing</u>. From the date hereof through the Closing, Acquiror shall use reasonable efforts to ensure Acquiror remains listed as a public company on, and for Acquiror Common Stock to be listed on, the NASDAQ.

7.05 <u>Acquiror SEC Filings</u>. From the date hereof through the Closing, Acquiror will keep current and timely file all reports required to be filed or furnished with the SEC and otherwise comply in all material respects with its reporting obligations under applicable Securities Laws; provided that, notwithstanding anything to the contrary contained herein, late filings, if any, of any reports that are required pursuant to Items 1.01, 1.02, 1.04, 1.05, 2.03, 2.04, 2.05, 2.06, 4.02(a) or 5.02(e) of Form 8-K shall not be deemed a breach of this Agreement.

7.06 <u>Section 16 Matters</u>. Prior to the Closing, the board of directors of Acquiror, or an appropriate committee of "non-employee directors" (as defined in Rule 16b-3 of the Exchange Act) thereof, shall adopt a resolution consistent with the interpretive guidance of the SEC so that the acquisition of Acquiror Common Stock pursuant to this Agreement and the other agreements contemplated hereby, by any person owning securities of the Company who is expected to become a director or officer (as defined under Rule 16a-1(f) under the Exchange Act) of Acquiror following the Closing shall be an exempt transaction for purposes of Section 16(b) of the Exchange Act pursuant to Rule 16b-3 thereunder.

7.07 <u>Director and Officer Appointments</u>. Except as otherwise agreed in writing by the Company and Acquiror prior to the Closing, and conditioned upon the occurrence of the Closing, subject to any limitation imposed under applicable Laws and NASDAQ listing requirements, Acquiror shall take all actions necessary or appropriate to cause (a) the individuals designated in accordance with <u>Section 2.05(b)</u> to be elected as members of the Acquiror Board, effective as of the Closing and (b) the individuals designated in accordance with <u>Section 2.05(c)</u> to be the executive officers of Acquiror effective as of the Closing.

7.08 Exclusivity.

(a) From the date of this Agreement until the First Merger Effective Time or, if earlier, the valid termination of this Agreement in accordance with Section 10.01, Acquiror shall not, and shall use its reasonable efforts to cause their respective Representatives not to, directly or indirectly:

(i) initiate, solicit or knowingly encourage or knowingly facilitate any inquiries or requests for information with respect to, or the making of, any inquiry regarding, or any proposal or offer that constitutes any Competing Proposal;

(ii) engage in, continue or otherwise participate in any negotiations or discussions concerning, or provide access to its properties, books and records or any confidential information or data to, any Person relating to any proposal, offer, inquiry or request for information that constitutes any Competing Proposal;

(iii) approve, endorse or recommend, or propose publicly to approve, endorse or recommend, any Competing Proposal;

(iv) execute or enter into, any letter of intent, memorandum of understanding, agreement in principle, confidentiality agreement, merger agreement, acquisition agreement, exchange agreement, joint venture agreement, partnership agreement, option agreement or other similar agreement for or relating to any Competing Proposal; or

(v) resolve or agree to do any of the foregoing.

(b) Acquiror also agrees that immediately following the execution of this Agreement it shall, and shall use its reasonable efforts to cause its Representatives to, cease any solicitations, discussions or negotiations with any Person (other than the parties hereto and their respective Representatives) conducted heretofore in connection with a Competing Proposal. Acquiror shall promptly (and in any event within one (1) Business Day) notify, in writing, the Company of the receipt of any inquiry, proposal, offer or request for information received after the date hereof that constitutes any Competing Proposal. For the avoidance of doubt, nothing contained in this Agreement shall prohibit Acquiror from taking and disclosing to its stockholders a position contemplated by Rule 14e-2 or Rule 14d-9 promulgated under the Exchange Act or from making any disclosure to Acquiror's stockholders if, in the good faith judgment of the Acquiror Board, after consultation with its financial advisors and outside counsel, failure so to disclose would be inconsistent with applicable Law; provided, however, that neither Acquiror nor Acquiror Board nor any committee thereof shall, except as specifically permitted by Section 8.02(d), withdraw or modify, or propose to withdraw or modify, in a manner adverse to the Company, the Acquiror Board Recommendation, or propose to approve or recommend, a Competing Proposal. Notwithstanding anything to the contrary contained in this Agreement or any Ancillary Agreements, the Acquiror Board may furnish information to, and enter into discussions with, a person who has made a Competing Proposal, and the Acquiror Board has (i) determined, in its good faith judgment (after having received the advice of a financial advisor and independent legal counsel, who may be Acquiror's regularly engaged independent legal counsel), that such Competing Proposal constitutes a Superior Proposal, or is reasonably likely to result in a Superior Proposal, (ii) determined, in its good faith judgment after consultation with independent legal counsel (who may be Acquiror's regularly engaged independent legal counsel), that, in light of such Competing Proposal, the furnishing of such information or entering into discussions is required to comply with its fiduciary obligations to Acquiror and its stockholders under applicable Law, (iii) provided written notice to the Company of its intent to furnish information or enter into discussions with such person at least three (3) business days prior to taking any such action, and (iv) obtained from such person an executed confidentiality agreement on terms no less favorable to Acquiror than those contained in the Confidentiality Agreement (it being understood that such confidentiality agreement and any related agreements shall not include any provision calling for any exclusive right to negotiate with such party or having the effect of prohibiting Acquiror from satisfying its obligations under this Agreement); provided further that Acquiror shall concurrently make available to the Company and its Representatives any information concerning the Acquiror and its Subsidiaries that is provided to any such Person and that was not previously made available to the Company.

(c) For purposes of this Agreement, "<u>Competing Proposal</u>" means (other than the Transactions): (i) any merger, consolidation, share exchange, business combination, recapitalization, liquidation, dissolution or other similar transaction involving Acquiror; (ii) any sale, lease, exchange, transfer or other disposition of all or a substantial part of the assets of the Acquiror; (iii) any sale, exchange, transfer or other disposition of 15% or more of any class of equity securities of Acquiror; (iv) any tender offer or exchange offer that, if consummated, would result in any person beneficially owning 15% or more of any class of equity securities of Acquiror; or (v) any solicitation in opposition to approval and adoption of this Agreement by Acquiror's stockholders.

7.09 Acquiror Plan. Acquiror shall adopt the Acquiror Equity Incentive Plan, in the form to be prepared by the Company in consultation with the Acquiror. The number of shares of Acquiror Common Stock available for issuance under the Acquiror Equity Incentive Plan shall be a number of shares of Acquiror Common Stock that is no less than the sum of (i) 10% of the total number of shares of Acquiror Common Stock that are issued and outstanding as of the Closing and (ii) such additional number of shares of Acquiror Common Stock reserved for issuance under the Acquiror Option & RSU Plans that remain available for grant under the Acquiror Option & RSU Plans immediately prior to the Closing (after giving effect to the forfeiture of any Acquiror Options or Acquiror Option & RSU Plans, and the number of shares of Common Stock subject to the Assumed Options that are outstanding as of the Closing Date and which awards expire, terminate or are otherwise surrendered, cancelled, forfeited or repurchased by the Company at their original issuance price pursuant to a contractual repurchase right, and shall be increased by an annual "evergreen" increase to take effect on each January 1 during the ten-year term of the Acquiror Equity Incentive Plan of at least 3% of the number of outstanding shares of Acquiror Common Stock as of December 31 immediately prior to the date of such increase, or such lesser increase as may be determined by the post-closing Acquiror Board.

7.10 <u>Stockholder Litigation</u>. In the event that any litigation or Action related to this Agreement, any Ancillary Agreement or the transactions contemplated hereby or thereby is brought, or, to the Knowledge of Acquiror, threatened in writing, against Acquiror or the Acquiror Board by any of Acquiror's stockholders prior to the Closing, Acquiror shall promptly notify the Company in writing of any such Action and keep the Company reasonably informed on a current basis with respect to the status thereof. Acquiror shall provide the Company (a) the opportunity to participate in the defense of any such Action and (b) the right to review and comment on all material filings or responses to be made by the Acquiror in connection with any such Action (and the Acquiror shall in good faith take such comments and other advice into consideration). The parties agree to cooperate in the defense and settlement of any such litigation, and Acquiror shall not settle any such Action without the prior written consent of the Company (not to be unreasonably withheld, conditioned or delayed), except that the Company will not be obligated to consent to any settlement that does not include a full release of the Company and its Affiliates or that imposes an injunction or other equitable relief upon the Company or any of its Affiliates.

7.11 <u>Acquiror Certificate of Incorporation Amendment</u>. Acquiror shall submit to Acquiror Stockholders at the Acquiror Stockholder Meeting a proposal to adopt amending and restating the Acquiror Certificate of Incorporation in its entirely to read as set forth on <u>Exhibit E</u> (the "<u>Amendment</u> <u>Proposal</u>").

7.12 <u>COBRA Coverage</u>. Acquiror or the Company shall comply with any applicable Laws requiring it to provide COBRA coverage, including any obligations with respect to current or former employees of Acquiror and the Company.

7.13 <u>Acquiror Material Contract Consents</u>. Acquiror shall use commercially reasonable efforts to obtain such Acquiror Material Contract Consents prior to the Closing to the extent the failure to do so would reasonably be expected to cause a Material Adverse Effect on Acquiror or Nxu Tech and shall use commercially reasonable efforts to obtain the other Acquiror Material Contract Consents. "<u>Acquiror Material Contract Consent</u>" shall mean any consent or waiver pursuant to any Acquiror Material Contract that is required by the execution, delivery or performance of this Agreement and consummation of the Transactions.

ARTICLE VIII JOINT COVENANTS

8.01 <u>Support of Transaction</u>. Without limiting any covenant contained in <u>Article VI</u> or <u>Article VII</u>, which obligations shall control to the extent of any conflict with the succeeding provisions of this <u>Section 8.01</u>, Acquiror and the Company shall each: (a) use reasonable efforts to assemble, prepare and file any information (and, as needed, to supplement such information) as may be reasonably necessary to obtain as promptly as practicable all governmental and regulatory consents required to be obtained in connection with the Transactions, (b) use reasonable efforts to obtain all material consents and approvals of third parties that any of Acquiror, the Company, or their respective Affiliates are required to obtain in order to consummate the Transactions, including any required approvals of parties to Material Contracts, and (c) subject to any Intervening Event, take such other action or refrain from taking such actions as may reasonably be necessary or as another party may reasonably request to satisfy the conditions of <u>Article IX</u> or otherwise to comply with this Agreement and to consummate the Transactions as soon as practicable.

8.02 Preparation of Registration Statement; Acquiror Stockholders Meeting; Solicitation of Company Stockholder Approvals.

(a) As promptly as practicable following the execution and delivery of this Agreement, Acquiror and the Company shall jointly prepare, and Acquiror shall cause to be filed with the SEC, a mutually-acceptable registration statement on Form S-4 (as amended or supplemented from time to time, and including the Proxy Statement contained therein and any exhibits thereto, the "<u>Registration Statement</u>") in connection with the registration under the Securities Act of the issuance of Acquiror Common Stock under this Agreement, which Registration Statement will also contain the Proxy Statement. Each of Acquiror and the Company shall use its reasonable efforts to cause the Registration Statement and the Proxy Statement to comply with the rules and regulations promulgated by the SEC, to have the Registration Statement declared effective under the Securities Act as promptly as practicable after such filing and to keep the Registration Statement effective as long as is necessary to consummate the Merger and the issuance of Acquiror Common Stock in connection therewith. Each of Acquiror and the Company shall furnish all information concerning itself and its officers, directors, and holders of equity securities as may reasonably be requested by the other Party in connection with such actions and the preparation of the Registration Statement.

(b) Each of Acquiror and the Company shall cooperate and mutually agree upon (such agreement not to be unreasonably withheld or delayed), any response to comments of the SEC or its staff with respect to the Registration Statement and any amendment to the Registration Statement filed in response thereto. If Acquiror or the Company becomes aware that any information contained in the Registration Statement shall have become false or misleading in any material respect or that the Registration Statement is required to be amended in order to comply with applicable Law, then (i) such party shall promptly inform the other party and (ii) Acquiror, on the one hand, and the Company, on the other hand, shall cooperate and mutually agree upon (such agreement not to be unreasonably withheld or delayed) an amendment or supplement to the Registration Statement. Acquiror and the Company shall use reasonable efforts to cause the Registration Statement as so amended or supplemented, to be filed with the SEC and to be disseminated to the holders of Acquiror Common Stock, as applicable, in each case pursuant to applicable Law and subject to the terms and conditions of this Agreement and the Acquiror Organizational Documents. Each of the Company and Acquiror shall provide the other parties with copies of any written comments, and shall inform such other parties of any oral comments, that Acquiror receives from the SEC or its staff with respect to the Registration Statement promptly after the receipt of such comments and shall give the other parties a reasonable opportunity to review and comment on any proposed written or oral responses to such comments prior to responding to the SEC or its staff.

(c) Acquiror agrees to include provisions in the Proxy Statement and to take reasonable action related thereto, with respect to (i) the issuance of Acquiror Common Stock pursuant to the terms of this Agreement for purposes of NASDAQ Marketplace Rule 5635(a) and the change of control of Acquiror resulting from the Transaction for purposes of NASDAQ Marketplace Rule 5635(b) (the "<u>Transaction Proposal</u>"), (ii) approval of the Amendment Proposal, (iii) the approval and adoption of the Acquiror Equity Incentive Plan (the "<u>Acquiror Equity Plan Proposal</u>"), (iv) adjournment of the Acquiror Stockholders Meeting, if necessary, to permit further solicitation of proxies because there are not sufficient votes to approve and adopt any of the foregoing proposals (the "<u>Adjournment Proposal</u>") and (v) approval of any other proposals reasonably agreed by Acquiror and the Company to be necessary or appropriate in connection with the Transactions (the "<u>Additional Proposal</u>" and together with the Transaction Proposal, the Amendment Proposal, the Acquiror Equity Plan Proposal and the Adjournment Proposal, the "Proposals"). Without the prior written consent of the Company, the Proposals shall be the only matters (other than procedural matters) which Acquiror shall propose to be acted on by holders of Acquiror Common Stock at the Acquiror Stockholders Meeting.

(d) Except as set forth in this Section 8.02(d), Acquiror shall use reasonable efforts to, as promptly as practicable after the Registration Statement is declared effective under the Securities Act, (i) establish the record date (which record date shall be mutually agreed with the Company) for, duly call, give notice of, convene and hold the a stockholders meeting in accordance with the Acquiror Organizational Documents and DGCL (the "Acquiror Stockholders Meeting"), (ii) cause the Proxy Statement to be disseminated to Acquiror Stockholders in compliance with applicable Law and (iii) solicit proxies from the holders of Acquiror Common Stock to vote in favor of each of the Proposals. Acquiror shall, through the Acquiror Board, recommend to the holders of Acquiror Common Stock that they approve the Proposals (the "Acquiror Board Recommendation") and shall include the Acquiror Board Recommendation in the Proxy Statement. The Acquiror Board shall not (and no committee or subgroup thereof shall) change, withdraw, withhold, qualify or modify, or publicly propose to change, withdraw, withhold, qualify or modify, the Acquiror Board Recommendation (an "Acquiror Change in Recommendation"); provided that if, at any time prior to obtaining the Required Acquiror Stockholder Approvals, the Acquiror Board determines in good faith, after consultation with its outside legal counsel, that in response to an Intervening Event, the failure to make an Acquiror Change in Recommendation would be result in a breach of its fiduciary duties under applicable Law, the Acquiror Board may, prior to obtaining the Required Acquiror Stockholder Approvals, make an Acquiror Change in Recommendation; provided, however, that Acquiror shall not be entitled to make, or agree or resolve to make, an Acquiror Change in Recommendation unless (A) Acquiror delivers to the Company a written notice (an "Intervening Event Notice") advising the Company that the Acquiror Board proposes to take such action and containing the material facts underlying the Acquiror Board's determination that an Intervening Event has occurred, and (B) at or after 5:00 p.m., New York City time, on the fourth (4th) Business Day immediately following the day on which Acquiror delivered the Intervening Event Notice (such period from the time the Intervening Event Notice is provided until 5:00 p.m., New York City time, on the fourth (4th) Business Day immediately following the day on which Acquiror delivered the Intervening Event Notice (it being understood that any material development with respect to an Intervening Event shall require a new notice but with an additional three Business Day (instead of four Business Day) period from the date of such notice, the "Intervening Event Notice Period")), the Acquiror Board reaffirms in good faith (after consultation with its outside legal counsel and financial advisor) that the failure to make an Acquiror Change in Recommendation would result in a breach of its fiduciary duties under applicable Law. If requested by the Company, Acquiror will, and will use its reasonable efforts to cause its Representatives to, during the Intervening Event Notice Period, engage in good faith negotiations with the Company and its Representatives to make such adjustments in the terms and conditions of this Agreement so as to obviate the need for an Acquiror Change in Recommendation.

(e) Notwithstanding the foregoing provisions of this <u>Section 8.02(e)</u>, if on a date for which the Acquiror Stockholders Meeting is scheduled, Acquiror has not received proxies representing a sufficient number of Acquiror Common Stock to obtain the Required Acquiror Stockholder Approvals, as applicable, whether or not a quorum is present, Acquiror shall have the right to make one or more successive postponements or adjournments of the Acquiror Stockholders Meeting; <u>provided</u> that if a quorum is present, that the proposal for the adjournment of the Acquiror Stockholders Meeting shall have been approved by a holders of sufficient number of Acquiror Common Stock; <u>provided</u>, <u>further</u>, that the Acquiror Stockholders Meeting, without the prior written consent of the Company, (i) may not be adjourned to a date that is more than fifteen (15) Business Days after the date for which the Acquiror Stockholders Meeting was originally scheduled or the most recently adjourned Acquiror Stockholders Meeting (excluding any adjournments required by applicable Law) and (ii) is held no later than four (4) Business Days prior to the Termination Date.

As promptly as practicable after the effectiveness of the Registration Statement, the Company shall solicit a consent (in form and (f) substance reasonably satisfactory to Acquiror) in writing or by electronic transmission from the Company Stockholders entitled to vote approving and adopting this Agreement, the Merger and, to the extent required by Law, the Transactions (when executed by the Company Stockholders holding sufficient voting power to approve and adopt this Agreement, the Merger and the Transactions, the "Company Stockholder Approvals"). In connection therewith, the Company shall use reasonable best efforts to, as promptly as practicable, (i) establish the record date (which record date shall be mutually agreed with Acquiror) for determining the Company Stockholders entitled to provide such written consent and (ii) solicit written consents from such Company Stockholders to give the Company Stockholder Approvals. The Company shall, through the Company Board, recommend to the Company Stockholders entitled to vote that they adopt this Agreement (the "Company Board Recommendation"). The Company Board shall not (and no committee or subgroup thereof shall) change, withdraw, withhold, qualify or modify, or publicly propose to change, withdraw, withhold, qualify or modify, the Company Board Recommendation unless a failure to do so would result in a breach of the Company Board's fiduciary duties under applicable Law (such action, a "Company Change in Recommendation"). Within ten (10) Business Days after the effectiveness of the Registration Statement (the "Company Stockholder" Written Consent Deadline"), the Company will provide Acquiror with copies of all stockholder consents it receives evidencing that the Company Stockholder Approvals have been obtained. Promptly following the receipt of the Company Stockholder Approvals, the Company will prepare and deliver to its stockholders who have not consented the notice required by Sections 228(e) and 262 of the DGCL. Unless this Agreement has been terminated in accordance with its terms, the Company's obligation to solicit written consents from the Company Stockholders entitled to vote to give the Company Stockholder Approvals in accordance with this Section 8.02(f) shall not be limited or otherwise affected by the making, commencement, disclosure, announcement or submission of any Acquisition Proposal or by any Company Change in Recommendation.

8.03 Tax Matters.

(a) *Transfer Taxes*. Notwithstanding anything to the contrary contained herein, any transfer, documentary, sales, use, stamp, registration, value added or other similar Taxes incurred in connection with the Closing of the Merger ("<u>Transfer Taxes</u>") shall be borne by the Company. The Company and Acquiror further agree to reasonably cooperate to reduce or eliminate the amount of any such Transfer Taxes.

(b) Intended Tax Treatment. The parties intend that, for United States federal income tax purposes, the First Merger and Second Merger be treated as integrated steps in a single transaction that qualifies as a "reorganization" within the meaning of Section 368(a) of the Code and the Treasury Regulations to which each of Acquiror, the Merger Subs and the Company are parties under Section 368(b) of the Code and the Treasury Regulations (the "Intended Tax Treatment"). This Agreement is intended to be, and is adopted as, a plan of reorganization for purposes of Sections 354, 361 and the 368 of the Code and within the meaning of Treasury Regulations Section 1.368-2(g). The Merger shall be reported by the parties for all applicable Tax purposes in accordance with the foregoing, unless otherwise required by a Tax Authority as a result of a "determination" within the meaning of Section 1313(a) of the Code (or any similar or corresponding provision of applicable Law). The parties hereto shall, and shall cause their Affiliates to, cooperate with each other and their respective counsel to document and support the Tax treatment of the Merger as a "reorganization" within the meaning of Section 368(a) of the Code, provided, however, that each of the Company and Acquiror is relying on its own advisors in connection with the tax treatment of the Merger and the other transactions contemplated by this Agreement.

(c) None of Acquiror, the Merger Subs or the Company shall take or cause to be taken, or knowingly fail to take or knowingly fail to cause to be taken any action, which action or failure to act prevents or impedes, or could reasonably be expected to prevent, impair or impede or otherwise create a significant risk with respect to, the Intended Tax Treatment.

(d) On the Closing Date, the Company shall deliver to Acquiror (i) a certification from the Company pursuant to Treasury Regulations Section 1.1445-2(c)(3) and (ii) a notice of such certification to the Internal Revenue Service pursuant to Treasury Regulations Section 1.897-2(h)(2), which notice shall be delivered by Acquiror to the Internal Revenue Service following the Closing, in each case, in the form set forth on Exhibit F, dated as of the Closing Date and duly signed by a responsible corporate officer of the Company.

(e) Each party shall promptly notify the other parties hereto in writing if, before the Closing Date, such party knows or has reason to believe that the Merger may not qualify for the Intended Tax Treatment (and whether the terms of this Agreement could be reasonably amended in order to facilitate the Merger qualifying for the Intended Tax Treatment). If the SEC requires that a Tax opinion be prepared and submitted with respect to the Tax treatment of the Merger, and if such a Tax opinion is being provided by a Tax counsel, each of the Company, Acquiror, the Merger Subs and their Affiliates agrees to deliver to such Tax counsel customary Tax representation letters reasonably satisfactory to such Tax opinion; provided that notwithstanding anything herein to the contrary in this Agreement, if and to the extent a Tax opinion with respect to the treatment of the Merger is being requested or required, each of the Company and the Acquiror shall use their reasonable efforts to have their respective counsel (upon receipt of customary Tax representation letters satisfactory to such counsel) provide a Tax opinion on the treatment of the Merger. For the avoidance of doubt, the covenants in this paragraph (e) shall apply regardless of the party to which any SEC inquiry or request is made.

(f) Notwithstanding anything to the contrary in this Agreement and for avoidance of doubt, none of the opinions with respect to any Tax matters relating to the Transactions discussed in <u>Section 8.03(e)</u> shall be a condition for the Closing.

8.04 Confidentiality; Publicity.

(a) Acquiror acknowledges that the information being provided to it in connection with this Agreement and the consummation of the transactions contemplated hereby is subject to the terms of the Confidentiality Agreement, the terms of which are incorporated herein by reference.

(b) The parties agree that the initial press release to be issued with respect to the Transactions shall be in the form previously agreed by the parties. None of Acquiror, the Merger Subs, the Company or any of their respective Affiliates shall make any public announcement or issue any public communication regarding this Agreement or the transactions contemplated hereby, or any matter related to the foregoing, without first obtaining the prior consent of the Company or Acquiror, as applicable (which consent shall not be unreasonably withheld, conditioned or delayed), except if such announcement or other communication is required by applicable Law or legal process (including pursuant to the Securities Law or the rules of any national securities exchange), in which case Acquiror or the Company, as applicable, shall use their commercially reasonable efforts to coordinate such announcement or communication with the other party, prior to announcement or issuance and allow the other party a reasonable opportunity to comment thereon (which shall be considered by Acquiror or the Company, as applicable, in good faith); provided, however, that notwithstanding anything contained in this Agreement to the contrary, (i) each party and its Affiliates may make announcements and may provide information regarding this Agreement and the transactions contemplated hereby to their respective owners, their Affiliates, and its and their respective directors, officers, employees, managers, advisors, direct and indirect investors and prospective investors without the consent of any other party hereto and (ii) the Company may exercise its rights and communicate with third parties as contemplated by Section 6.04; and provided, further, that the foregoing shall not prohibit any party hereto from communicating with third parties to the extent necessary for the purpose of seeking any third party consent in connection with the Transactions.

Financing Cooperation. From the date hereof until the earlier of (a) the First Merger Effective Time or (b) the valid termination of this 8.05 Agreement, each of the Acquiror and the Merger Subs agree to use reasonable efforts to cooperate with the Company in connection with the arrangement of any Pre-Closing Financing, including by (i) reasonably cooperating with the Company's preparation of definitive financing documentation and the Schedules and exhibits thereto, in each case, customarily required to be delivered under such definitive financing documentation, (ii) providing to the Company and its prospective sources of Pre-Closing Financing all reasonably necessary documentation and other information required by regulatory authorities under applicable "know your customer" and anti-money laundering rules and regulations, including the PATRIOT Act and (iii) cooperating in satisfying the conditions precedent set forth in the definitive Pre-Closing Financing documentation to the extent the satisfaction of such condition requires the cooperation of, or is within the control of, the Company, the Acquiror, the Merger Subs, as applicable; provided, however, that neither party shall be obligated to repay any Pre-Closing Financing of the other party without such first party's express written consent; and provided, further, that any Pre-Closing Financing shall comply with the limitations set forth herein. Notwithstanding the foregoing, (A) such requested cooperation shall not (1) unreasonably interfere with the business or operations of Acquiror, (2) cause significant competitive harm to the Acquiror if Transactions are not consummated or (3) result in the material contravention of, or that could reasonably be expected to result in, a material violation or breach of, or a default under, any Laws or under any Material Contract, (B) nothing in this Section 8.05 shall require cooperation to the extent that it would (1) cause any condition to the Closing set forth in Article IX to not be satisfied or (2) cause any breach of this Agreement, (C) Acquiror shall not be required to (1) pay any commitment or other similar fee prior to the Closing, (2) deliver or obtain auditor comfort letters or opinions of internal or external counsel, (3) provide access to or disclose information where Acquiror determines that such access or disclosure could jeopardize the attorney-client privilege or contravene any Law or Contract, (4) deliver any audited financial statements to the extent not already available, or (5) waive or amend any terms of this Agreement or any other Contract to which Acquiror is a party, and (D) none of Acquiror or its directors, officers or employees shall be required to execute, deliver or enter into, or perform any agreement, document or instrument, including any definitive agreements with respect to the Pre-Closing Financing that is not contingent upon the Closing or that would be effective prior to the First Merger Effective Time and the directors and officers of Acquiror shall not be required to adopt resolutions approving the agreements, documents and instruments pursuant to which the Pre-Closing Financing is obtained, in each case which are effective prior to the Closing.

8.06 Listing Application.

Acquiror and the Company shall use reasonable efforts to cause the Acquiror Common Stock to be issued in connection with the (a) Transactions, or otherwise reserved for issuance to be approved for listing on the NASDAQ as promptly as practicable following the issuance thereof, subject to official notice of issuance, on or prior to the Closing Date, including by submitting, subject to the Company's corporation, prior to the Closing a listing application (the "Listing Application") with the NASDAQ with respect to such Acquiror Common Stock. Each of the Company and Acquiror shall promptly furnish all information concerning itself and its Affiliates as may be reasonably requested by the other such party and shall otherwise reasonably assist and cooperate with the other such party in connection with the preparation and filing of the Listing Application. Acquiror and the Company will use reasonable efforts to (i) cause the Listing Application, when filed, to comply in all material respects with all requirements applicable thereto, (ii) respond as promptly as reasonably practicable to and resolve all comments received from the NASDAQ or its staff concerning the Listing Application and (iii) have the Listing Application approved by the NASDAQ, as promptly as practicable after such filing. Acquiror shall not submit the Listing Application or any supplement or amendment thereto, or respond to comments received from the NASDAQ with respect thereto, without the Company's prior consent (which shall not be unreasonably withheld, conditioned or delayed) and without providing the Company a reasonable opportunity to review and comment thereon. Acquiror shall promptly notify the Company upon the receipt of any comments from the NASDAO, or any request from the NASDAO for amendments or supplements to the Listing Application and shall provide the Company with copies of all material correspondence between Acquiror or any of its Representatives, on the one hand, and the NASDAQ, on the other hand, and all written comments with respect to the Listing Application received from the NASDAQ, and advise the Company of any oral comments with respect to the Listing Application received from the NASDAQ. Promptly after receiving notice thereof, Acquiror shall advise the Company of the time of the approval of the Listing Application and the approval for listing on the NASDAQ of the Acquiror Common Stock to be issued in connection with the transactions contemplated hereby.

(b) In furtherance of and without limiting the foregoing, if approval for listing on the NASDAQ of the Acquiror Common Stock to be issued in connection with the transactions contemplated is not approved by NASDAQ as a result of a failure to satisfy NASDAQ Rule 5505(b)(1)(B), each of Acquiror and the Company shall cooperate in good faith and use commercially reasonable efforts to take such actions as may be required to cause such rule to be satisfied, including through the issuance and sale of securities of either Acquiror or the Company, or through requesting Humanitario to sell or otherwise transfer equity securities of the Company owned by it, to unaffiliated third parties prior to the Closing (it being understood and agreed that there cannot be any assurance that either Acquiror or the Company will be successful in arranging or completing any such issuance or sale or that Humanitario be successful in negotiating commercially reasonable terms with respect to such sale or other transfer); provided, that neither Acquiror or the Company shall be required to take any action that, based on advice of counsel, could be inconsistent with Law or NASDAQ rules.

ARTICLE IX CONDITIONS TO OBLIGATIONS

9.01 <u>Conditions to Obligations of All Parties</u>. The obligations of the parties hereto to consummate, or cause to be consummated, the Closing are subject to the satisfaction of the following conditions, any one or more of which may be waived (if legally permitted) in writing by all of such parties:

(a) *No Prohibition.* No Governmental Authority shall have enacted, issued, promulgated, enforced or entered any Law, judgment, decree, executive order or award, which is then in effect and has the effect of making the Transactions illegal or otherwise prohibiting or enjoining consummation of the Transactions; <u>provided</u> that the Governmental Authority enacting, issuing, promulgating, enforcing or entering such Law, judgment, decree, executive order or award has jurisdiction over the parties hereto with respect to the transactions contemplated hereby.

(b) *Registration Statement*. The Registration Statement shall have become effective under the Securities Act and no stop order suspending the effectiveness of the Registration Statement shall have been issued and no proceedings for that purpose shall have been initiated or threatened by the SEC and not withdrawn.

(c) *NASDAQ*. The Acquiror Common Stock shall have been approved for listing on NASDAQ, subject only to official notice of issuance thereof.

Section 2.05.

(d) *Directors and Officers*. The Directors and Officers of Acquiror upon the First Merger Effective Time shall be in accordance with

(e) Acquiror Stockholder Approvals. The Required Acquiror Stockholder Approvals shall have been obtained.

(f) Company Stockholder Approvals. The Company Stockholder Approvals shall have been obtained.

9.02 <u>Additional Conditions to Obligations of Acquiror</u>. The obligations of Acquiror to consummate, or cause to be consummated, the Closing are subject to the satisfaction of the following additional conditions, any one or more of which may be waived in writing by Acquiror:

(a) Representations and Warranties. The representations and warranties of the Company contained in Section 4.01 (Organization, Standing and Corporate Power), Section 4.02(a) (Corporate Authority; Approval; Non-Contravention) and Section 4.23 (Brokers) shall each be true and correct in all material respects as of the Closing Date as though made on the Closing Date, except to the extent that any such representation and warranty expressly speaks as of an earlier date, in which case such representation), shall be true and correct in all respects other than *de minimis* inaccuracies as of the Closing Date as though made on the extent that any such representation and warranty expressly speaks as of an earlier date, in which case such representation), shall be true and correct in all respects other than *de minimis* inaccuracies as of the Closing Date as though made on the Closing Date, except to the extent that any such representation and warranty expressly speaks as of an earlier date, in which case such representation and correct as of such earlier date. All other representations and warranties of the Company contained in this Agreement shall be true and correct (without giving any effect to any limitation as to "materiality" or "Material Adverse Effect" or any similar limitation set forth therein) as of the Closing Date, as though made on and as of the Closing Date, except (i) to the extent that any such representation and warranty shall be true and correct as of such earlier date and (ii) where the failure of such representations and warranties to be true and correct (whether as of the Closing Date or such earlier date), individually or in the aggregate, does not result in a Material Adverse Effect on the Company.

(b) Agreements and Covenants. Each of the covenants of the Company to be performed or complied with as of or prior to the Closing shall have been performed or complied with in all material respects.

(c) Officer's Certificate. The Company shall have delivered to Acquiror a certificate signed by an officer of the Company, dated the Closing Date, certifying that, to the knowledge and belief of such officer, the conditions specified in Section 9.02(a), Section 9.02(b) and Section 9.02(f) have been fulfilled.

(d) *Ancillary Agreements*. The Company shall have delivered to Acquiror executed counterparts to all of the Ancillary Agreements to which the Company, or any Company Stockholder, is a party.

(e) *No Material Adverse Effect.* Since the date of this Agreement, there shall not have occurred and be continuing any Material Adverse Effect on the Company.

(f) Voting Trust Transfer. The Voting Trust Transfer shall have been completed in accordance with the Company Support Agreement.

9.03 <u>Additional Conditions to the Obligations of the Company</u>. The obligation of the Company to consummate the Closing is subject to the satisfaction of the following additional conditions, any one or more of which may be waived in writing by the Company:

(a) Representations and Warranties. The representations and warranties of Acquiror and the Merger Subs contained in Section 5.01 (Organization, Standing and Corporate Power), Section 5.02(a) (Corporate Authority; Approval; Non-Contravention) and Section 5.14 (Brokers) shall each be true and correct in all material respects as of the Closing Date as though made on the Closing Date, except to the extent that any such representation and warranty expressly speaks as of an earlier date, in which case such representation and warranty shall be true and correct in all respects other than *de minimis* inaccuracies as of the Closing Date as though made on the Closing Date, except to the extent that any such representations and warranty expressly speaks as of an earlier date, in which case such representation <u>Section 5.04</u> (Capitalization), shall be true and correct in all respects other than *de minimis* inaccuracies as of the Closing Date as though made on the Closing Date, except to the extent that any such representations and warranty expressly speaks as of an earlier date, in which case such representation and warranty shall be true and correct as of such earlier date. All other representations and warranties of Acquiror and the Merger Subs contained in this Agreement shall be true and correct (without giving any effect to any limitation as to "material adverse effect" or any similar limitation set forth therein) as of the Closing Date, as though made on and as of the Closing Date, except (i) to the extent that any such representation and warranty expressly speaks as of an earlier date and (ii) where the failure of such representations and warranties to be true and correct (whether as of the Closing Date or such earlier date), taken as a whole, does not result in a Material Adverse Effect on Acquiror or Nxu Tech.

(b) Agreements and Covenants. Each of the covenants of Acquiror to be performed or complied with as of or prior to the Closing shall have been performed or complied with in all material respects.

(c) *Officer's Certificate*. Acquiror shall have delivered to the Company a certificate signed by an officer of Acquiror, dated the Closing Date, certifying that, to the knowledge and belief of such officer, the conditions specified in <u>Section 9.03(a)</u> and <u>Section 9.03(b)</u> have been fulfilled.

(d) Ancillary Agreements. Acquiror shall have delivered to the Company executed counterparts to all of the Ancillary Agreements to which Acquiror is a party.

(e) *Acquiror Officers and Directors.* Acquiror shall have delivered all resignations of its officers and directors, as applicable, such that, immediately following the First Merger Effective Time, the officers and directors of the Acquiror shall be as contemplated by <u>Section 2.05</u>.

(f) No Material Unforeseen Extraordinary Liability. Since the date of this Agreement, neither Acquiror nor Nxu Tech have suffered or incurred any unforeseen extraordinary liability that first arises or the consequences of which first become apparent after the date of this Agreement and is not otherwise disclosed to the Company, or the consequences of which are not reasonably foreseeable as of the date of this agreement, if such liability would reasonably be expected to cause a Material Adverse Effect on Acquiror and Nxu Tech, taken as a whole.

ARTICLE X TERMINATION/EFFECTIVENESS

10.01 <u>Termination</u>. This Agreement may be terminated, and the transactions contemplated hereby abandoned:

(a) by mutual written consent of the Company and Acquiror;

prior to the Closing, by written notice to the Company from Acquiror if (i) there is any breach of any representation, warranty, (b)covenant or agreement on the part of the Company set forth in this Agreement, such that any condition specified in Section 9.02(a) or Section 9.02(b) would not be satisfied at the Closing (a "Terminating Company Breach"), except that, if any such Terminating Company Breach is curable by the Company through the exercise of its commercially reasonable efforts, then, for a period of up to 30 days (or any shorter period of the time that remains between the date Acquiror provides written notice of such violation or breach and the Termination Date) after receipt by the Company of notice from Acquiror of such breach, but only as long as the Company continues to use its commercially reasonable efforts to cure such Terminating Company Breach (the "Company Cure Period"), such termination shall not be effective, and such termination shall become effective only if the Terminating Company Breach is not cured within the Company Cure Period, (ii) the Closing has not occurred on or before March 31, 2025 (the "Termination Date"), (iii) at any time prior to receipt of the Required Acquiror Stockholder Approvals, to enter into a definitive agreement with respect to a Superior Proposal, provided that Acquiror has provided the Company with not less than five (5) business days' notice prior to such termination, (iv) at any time after the Company Stockholder Written Consent Deadline if the Company has not previously received the Company Stockholder Approvals (provided, that upon the Company receiving the Company Stockholder Approvals, Acquiror shall no longer have any right to terminate this Agreement under this clause (iv)), or (v) the consummation of the Merger is permanently enjoined or prohibited by the terms of a final, non-appealable Governmental Order or other Law; provided, that the right to terminate this Agreement under Section 10.01(b)(ii) shall not be available if Acquiror's failure to fulfill any obligation under this Agreement has been the primary cause of, or primarily resulted in, the failure of the Closing to occur on or before such date; provided, further, that the right to terminate this Agreement under Section 10.01(b)(ii) shall not be available if Acquiror is in breach of this Agreement on such date, which breach could give rise to a right of the Company to terminate this Agreement;

(c) prior to the Closing, by written notice to Acquiror from the Company if (i) there is any breach of any representation, warranty, covenant or agreement on the part of Acquiror set forth in this Agreement, such that any condition specified in Section 9.03(a) or Section 9.03(b) would not be satisfied at the Closing (a "Terminating Acquiror Breach"), except that, if any such Terminating Acquiror Breach is curable by Acquiror through the exercise of its commercially reasonable efforts, then, for a period of up to 30 days (or any shorter period of the time that remains between the date the Company provides written notice of such violation or breach and the Termination Date) after receipt by Acquiror of notice from the Company of such breach, but only as long as Acquiror continues to use its commercially reasonable efforts to cure such Terminating Acquiror Breach is not cured within the Acquiror Cure Period, (ii) the Closing has not occurred on or before the Termination Date, or (iii) the consummation of the Merger is permanently enjoined or prohibited by the terms of a final, non-appealable Governmental Order or other Law, or (iv) there has been any Acquiror Change in Recommendation; provided, that the right to terminate this Agreement under Section 10.01(c)(ii) shall not be available if the Company's failure to fulfill any obligation under this Agreement has been the primary cause of, or primarily resulted in, the failure of the Closing to occur on or before such date; provided, further, that the right to terminate this Agreement under Section 10.01(c)(ii) shall not be available if the Company is in breach of this Agreement on such date, which breach could give rise to a right of Acquiror to terminate this Agreement;

(d) by written notice from the Company to Acquiror prior to obtaining the Required Acquiror Stockholder Approvals if the Acquiror Board shall have (i) made an Acquiror Change in Recommendation or (ii) failed to include the Acquiror Board Recommendation in the Proxy Statement;

(e) by written notice from the Company to Acquiror if the Required Acquiror Stockholder Approvals are not obtained at the Acquiror Stockholders Meeting (subject to any adjournment or recess of the meeting in accordance with <u>Section 8.02(e)</u>);

(f) by written notice from Acquiror to the Company if the Company Support Agreement has not been delivered concurrently with the execution of this Agreement pursuant to <u>Section 6.09</u>; or

(g) by written notice from the Company to Acquiror.

10.02 Effect of Termination. Except as otherwise set forth in this Section 10.02, in the event of the termination of this Agreement pursuant to Section 10.01, this Agreement shall forthwith become void and have no effect, without any liability on the part of any party hereto or its respective Affiliates, officers, directors, employees or stockholders, other than liability of any party hereto for any Fraud or Willful Breach of this Agreement by such party occurring prior to such termination; provided, however, that if the Company terminates this Agreement pursuant to Section 10.01(g) or Acquiror terminates this Agreement pursuant to Section 10.01(b)(iv), the Company shall pay to Acquiror a termination fee of \$1,000,000, via wire transfer of immediately available funds to such account as may be designated in writing by Acquiror. The provisions of Sections 6.03, 8.04, 10.02 and Article XI (collectively, the "Surviving Provisions") and the Confidentiality Agreement, and any other Section or Article of this Agreement referenced in the Surviving Provisions, to the extent required to survive in order to give appropriate effect to the Surviving Provisions, shall in each case survive any termination of this Agreement.

ARTICLE XI MISCELLANEOUS

11.01 <u>Waiver</u>. Any party to this Agreement may, at any time prior to the Closing, by action taken by its board of directors, or officers thereunto duly authorized, waive any of the terms or conditions of this Agreement, or agree to an amendment or modification to this Agreement in the manner contemplated by <u>Section 11.10</u> and by an agreement in writing executed in the same manner (but not necessarily by the same Persons) as this Agreement.

11.02 <u>Notices</u>. All notices and other communications among the parties shall be in writing and shall be deemed to have been duly given (i) when delivered in person, (ii) when delivered after posting in the United States mail having been sent registered or certified mail return receipt requested, postage prepaid, (iii) when delivered by FedEx or other nationally recognized overnight delivery service or (iv) when e-mailed during normal business hours (and otherwise as of the immediately following Business Day), addressed as follows:

(a) If to Acquiror or Merger Subs, to:

Attn: Mark Hanchett E-mail: mark@nxuenergy.com

with a copy (which shall not constitute notice) to:

Snell & Wilmer L.L.P. Attn: Mike Donahey 1 E Washington Street, Suite 2700 Phoenix, AZ 85004 E-mail: mdonahey@swlaw.com

(b) If to the Company to:

Verde Bioresins, Inc 1431 E. Orangethorpe Ave. Fullerton, CA 92831 Attn: Brian Gordon E-mail: brian@verdebioresins.com

And

Wilmer Cutler Pickering Hale and Dorr LLP 2600 El Camino Real, Suite 400 Palo Alto, CA 94306 Attn: Daniel Zimmermann E-mail: Daniel.Zimmermann@wilmerhale.com

And

Wilmer Cutler Pickering Hale and Dorr LLP 7 World Trade Center New York, New York 10007 Attn: Michael E. Gilligan Glenn R. Pollner E-mail: Michael.Gilligan@wilmerhale.com Glenn.Pollner@wilmerhale.com

or to such other address or addresses as the parties may from time to time designate in writing.

11.03 <u>Assignment</u>. No party hereto shall assign this Agreement or any part hereof without the prior written consent of the other parties. Subject to the foregoing, this Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective permitted successors and assigns. Any attempted assignment in violation of the terms of this <u>Section 11.03</u> shall be null and void, *ab initio*.

11.04 <u>Rights of Third Parties</u>. Nothing expressed or implied in this Agreement is intended or shall be construed to confer upon or give any Person, other than the parties hereto, any right or remedies under or by reason of this Agreement; <u>provided</u>, <u>however</u>, that notwithstanding the foregoing (a) in the event the Closing occurs, the present and former officers and directors of the Company and Acquiror (and their successors and Representatives) are intended third-party beneficiaries of, and may enforce, <u>Section 7.01</u> and (b) the past, present and future directors, officers, employees, incorporators, members, partners, stockholders, Affiliates, agents, attorneys, advisors and Representatives of the parties, and any Affiliate of any of the foregoing (and their successors, heirs and Representatives), are intended third-party beneficiaries of, and may enforce, <u>Sections 7.12</u>, <u>11.14</u>, and <u>11.15</u>.

11.05 <u>Expenses</u>. Unless otherwise specified herein, each party hereto shall bear its own expenses incurred in connection with this Agreement and the transactions herein contemplated whether or not such transactions shall be consummated, including all fees of its legal counsel, financial advisers and accountants.

11.06 <u>Governing Law</u>. This Agreement, and all claims or causes of action based upon, arising out of, or related to this Agreement or the transactions contemplated hereby, shall be governed by, and construed in accordance with, the Laws of the State of Delaware, without giving effect to principles or rules of conflict of laws to the extent such principles or rules would require or permit the application of Laws of another jurisdiction.

11.07 <u>Captions; Counterparts</u>. The captions in this Agreement are for convenience only and shall not be considered a part of or affect the construction or interpretation of any provision of this Agreement. This Agreement may be executed in two (2) or more counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument. The words "execution," "signed," "signature," and words of like import in this Agreement or in any other certificate, agreement or document related to this Agreement or the other Ancillary Agreements shall include images of manually executed signatures transmitted by facsimile or other electronic format (including, "pdf," "tif" or "jpg") and other electronic signatures (including, DocuSign and AdobeSign). The use of electronic signatures and electronic records (including, any contract or other record created, generated, sent, communicated, received, or stored by electronic means) shall be of the same legal effect, validity and enforceability as a manually executed signature or use of a paper-based record-keeping system to the fullest extent permitted by applicable Law, including the Federal Electronic Signatures in Global and National Commerce Act, the Delaware Uniform Electronic Transactions Act and any other applicable Law. Minor variations in the form of the signature page, including footers from earlier versions of this Agreement or any such other document, shall be disregarded in determining the party's intent or the effectiveness of such signature.

11.08 <u>Schedules and Exhibits</u>. The Schedules and Exhibits referenced herein are a part of this Agreement as if fully set forth herein. All references herein to Schedules and Exhibits shall be deemed references to such parts of this Agreement, unless the context shall otherwise require. Any disclosure made by a party in the Schedules with reference to any section or schedule of this Agreement shall be deemed to be a disclosure with respect to all other sections or schedules to which such disclosure may apply solely to the extent the relevance of such disclosure is reasonably apparent on the face of the disclosure in such Schedule. Certain information set forth in the Schedules is included solely for informational purposes.

11.09 Entire Agreement. This Agreement (together with the Schedules and Exhibits to this Agreement), the Ancillary Agreements and the Confidentiality Agreement constitute the entire agreement among the parties relating to the transactions contemplated hereby and supersede any other agreements, whether written or oral, that may have been made or entered into by or among any of the parties relating to the Transactions. No representations, warranties, covenants, understandings, agreements, oral or otherwise, relating to the transactions contemplated by this Agreement exist between the Parties except as expressly set forth or referenced in this Agreement, the Ancillary Agreements and the Confidentiality Agreement.

11.10 <u>Amendments</u>. This Agreement may be amended or modified in whole or in part, only by a duly authorized agreement in writing executed in the same manner as this Agreement (but not necessarily by the same natural persons who executed this Agreement) and which makes reference to this Agreement. The approval of this Agreement by the shareholders or stockholders of any of the parties shall not restrict the ability of the board of directors of any of the parties to terminate this Agreement in accordance with <u>Section 10.01</u> or to cause such party to enter into an amendment to this Agreement pursuant to this <u>Section 11.10</u>.

11.11 Severability. If any provision of this Agreement is held invalid or unenforceable by any court of competent jurisdiction, the other provisions of this Agreement shall remain in full force and effect. The parties further agree that if any provision contained herein is, to any extent, held invalid or unenforceable in any respect under the Laws governing this Agreement, they shall take any actions necessary to render the remaining provisions of this Agreement valid and enforceable to the fullest extent permitted by Law and shall amend or otherwise modify this Agreement to replace any provision contained herein that is held invalid or unenforceable with a valid and enforceable provision giving effect to the intent of the parties.

Jurisdiction; WAIVER OF TRIAL BY JURY. Any Action based upon, arising out of or related to this Agreement, or the transactions 11 12 contemplated hereby, shall be brought in the Court of Chancery of the State of Delaware or, if such court lacks jurisdiction, the state or federal courts in the State of Delaware, and each of the parties irrevocably submits to the exclusive jurisdiction of each such court in any such Action, waives any objection it may now or hereafter have to personal jurisdiction, venue or to convenience of forum, agrees that all claims in respect of the Action shall be heard and determined only in any such court, and agrees not to bring any Action arising out of or relating to this Agreement or the transactions contemplated hereby in any other court. Nothing herein contained shall be deemed to affect the right of any party to serve process in any manner permitted by Law, or to commence legal proceedings or otherwise proceed against any other party in any other jurisdiction, in each case, to enforce judgments obtained in any Action brought pursuant to this Section 11.12. EACH OF THE PARTIES HERETO HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVES TO THE FULLEST EXTENT PERMITTED BY LAW ANY AND ALL RIGHT TO TRIAL BY JURY IN ANY ACTION BASED UPON, ARISING OUT OF OR RELATED TO THIS AGREEMENT OR THE TRANSACTIONS CONTEMPLATED HEREBY. EACH OF THE PARTIES HERETO CERTIFIES AND ACKNOWLEDGES THAT (A) NO REPRESENTATIVE OF ANY OTHER PARTY HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PARTY WOULD NOT, IN THE EVENT OF LITIGATION, SEEK TO ENFORCE THE FOREGOING WAIVER, (B) EACH SUCH PARTY UNDERSTANDS AND HAS CONSIDERED THE IMPLICATIONS OF THIS WAIVER, (C) EACH SUCH PARTY MAKES THIS WAIVER VOLUNTARILY, AND (D) EACH SUCH PARTY HAS BEEN INDUCED TO ENTER INTO THIS AGREEMENT AND THE TRANSACTIONS BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS IN THIS SECTION 11.12.

11.13 Enforcement. The parties agree that irreparable damage for which monetary damages, even if available, would not be an adequate remedy, would occur in the event that the parties do not perform their obligations under the provisions of this Agreement (including failing to take such actions as are required of them hereunder to consummate this Agreement) in accordance with its specified terms or otherwise breach such provisions. The parties acknowledge and agree that (a) the parties shall be entitled to an injunction, specific performance, or other equitable relief, to prevent breaches of this Agreement and to enforce specifically the terms and provisions hereof, without proof of damages, prior to the valid termination of this Agreement in accordance with <u>Section 10.01</u>, this being in addition to any other remedy to which they are entitled under this Agreement and (b) the right of specific enforcement is an integral part of the transactions contemplated by this Agreement and without that right, none of the parties would have entered into this Agreement. Each party agrees that it will not oppose the granting of specific performance and other equitable relief on the basis that the other parties have an adequate remedy at Law or that an award of specific performance is not an appropriate remedy for any reason at Law or equity. The parties acknowledge and agree that any party seeking an injunction to prevent breaches of this Agreement and to enforce specifically the terms and provisions of this Agreement in accordance with this <u>Section 11.13</u> shall not be required to provide any bond or other security in connection with any such injunction.

11.14 <u>Non-Recourse</u>. This Agreement may only be enforced against, and any claim or cause of action based upon, arising out of, or related to this Agreement or the transactions contemplated hereby may only be brought against, the entities that are expressly named as parties hereto, and then only with respect to the specific obligations set forth herein with respect to such party. For the avoidance of doubt, Company Stockholders and Acquiror Stockholders are not named parties. Except to the extent a named party to this Agreement (and then only to the extent of the specific obligations undertaken by such named party in this Agreement), (a) no past, present or future director, officer, employee, incorporator, member, partner, stockholder, Affiliate, agent, attorney, advisor or Representative or Affiliate of any named party to this Agreement and (b) no past, present or future director, officer, employee, incorporator, member, partner, stockholder, Affiliate, agent, attorney, advisor or Representative or Affiliate, agent, attorney, advisor or Representative or other wise) for any one or more of the representations, warranties, covenants, agreements or other obligations or liabilities of any one or more of the Company, Acquiror, or Merger Subs under this Agreement of or for any claim based on, arising out of, or related to this Agreement or the transactions contemplated hereby.

11.15 <u>Non-survival of Representations, Warranties and Covenants</u>. None of the representations, warranties, covenants, obligations or other agreements in this Agreement or in any certificate, statement or instrument delivered pursuant to this Agreement, including any rights arising out of any breach of such representations, warranties, covenants, obligations, agreements and other provisions, shall survive the Closing, and each shall terminate and expire upon the occurrence of the First Merger Effective Time (and there shall be no liability after the Closing in respect thereof), except for (a) those covenants and agreements contained herein that by their terms expressly apply in whole or in part after the Closing and then only with respect to any breaches occurring after the Closing and (b) this Article XI.

[Remainder of Page Intentionally Left Blank]

IN WITNESS WHEREOF, Acquiror, the Merger Subs and the Company have caused this Agreement to be executed and delivered as of the date first written above by their respective officers thereunto duly authorized.

NXU, INC.

By: /s/ Mark Hanchett

Name:Mark HanchettTitle:Chief Executive Officer

[Signature Page to the Agreement and Plan of Merger]

IN WITNESS WHEREOF, Acquiror, the Merger Subs and the Company have caused this Agreement to be executed and delivered as of the date first written above by their respective officers thereunto duly authorized.

Nxu Merger Sub, Inc.

By: NXU, Inc. Title: Sole Member

By: /s/ Annie Pratt

Name:Annie PrattTitle:President/Chief Executive Officer

Nxu Merger Sub, LLC

By: NXU, Inc. Title: Sole Member

> By: /s/ <u>Mark Hanchett</u> Name: <u>Mark Hanchett</u> Title: Chief Executive Officer

[Signature Page to the Agreement and Plan of Merger]

IN WITNESS WHEREOF, Acquiror, the Merger Subs and the Company have caused this Agreement to be executed and delivered as of the date first written above by their respective officers thereunto duly authorized.

VERDE BIORESINS, INC.

By: /s/ Brian Gordon

Name: Brian Gordon Title: President

[Signature Page to the Agreement and Plan of Merger]

Exhibit A Form of Company Support Agreement

COMPANY SUPPORT AGREEMENT

This Company Support Agreement (this "Support Agreement") is dated as of October $[\bullet]$, 2024, by and among Humanitario Capital LLC, a Puerto Rico limited liability company (the "Stockholder"), NXU, Inc., a Delaware corporation prior ("Acquiror"), and Verde Bioresins, Inc., a Delaware corporation (the "Company"). Capitalized terms used but not defined herein shall have the meanings ascribed to such terms in the Merger Agreement (as defined below).

RECITALS

WHEREAS, as of the date hereof, the Stockholder is the holder of record and/or the "beneficial owner" (within the meaning of Rule 13d-3 under the Exchange Act) of (i) 2,648,364,984 shares of the Company's Common Stock (the "<u>Stockholder Shares</u>"), (ii) (a) that certain Common Stock Purchase Warrant, dated as of February 17, 2023, (b) that certain Common Stock Purchase Warrant, dated as of February 8, 2021, and (c) that certain Common Stock Purchase Warrant, dated as of June 15, 2023 (together, the "<u>Stockholder Warrants</u>"), and (iii) (a) that certain Secured Convertible Promissory Note issued by the Company to Stockholder, dated as of February 17, 2023, (b) that certain Secured Convertible Promissory Note issued by the Company to Stockholder, dated June 15, 2023, (c) that certain Senior Secured Convertible Promissory Note issued by the Company to Stockholder, dated March 6, 2024, (d) that certain Senior Secured Convertible Promissory Note issued by the Company to Stockholder, dated June 17, 2024, (e) that certain Senior Secured Convertible Promissory Note issued by the Company to Stockholder, dated April 17, 2024, (e) that certain Senior Secured Convertible Promissory Note issued by the Company to Stockholder, dated June 5, 2024, (f) that certain Senior Secured Convertible Promissory Note issued by the Company to Stockholder, dated July 26, 2024 and (g) that certain Senior Secured Convertible Promissory Note issued by the Company to Stockholder, dated August 30, 2024 (together, the "<u>Stockholder Notes</u>");

WHEREAS, prior to the execution and delivery of this Support Agreement, Acquiror, Nxu Merger Sub, Inc., a Delaware corporation and a direct wholly owned subsidiary of Acquiror ("<u>Merger Sub I</u>"), Nxu Merger Sub, LLC a Delaware limited liability company and a wholly owned subsidiary of Acquiror ("<u>Merger Sub II</u>") and the Company have entered into an Agreement and Plan of Merger (as amended or modified from time to time, the "<u>Merger Agreement</u>"), dated as of October [•], 2024, pursuant to which, among other transactions, Merger Sub I is to merge with and into the Company, with the Company continuing on as the surviving company and a wholly owned subsidiary of Acquiror (the "<u>First Merger</u>") and, promptly thereafter, such surviving company shall merge with and into Merger Sub II with Merger Sub II continuing as the surviving entity of such merger as a wholly owned subsidiary of Acquiror, on the terms and conditions set forth therein (the "<u>Second Merger</u>" and together with the First Merger, the "<u>Merger</u>");

WHEREAS, the Merger Agreement contemplates that the Stockholder will receive, pursuant to the terms of the Merger Agreement, certain shares of Acquiror Common Stock at the Closing (such shares, together with any securities paid as dividends or distributions with respect to such securities or into which such securities are exchanged or converted, the "<u>Closing Shares</u>"); and

WHEREAS, as an inducement to Acquiror and the Company to enter into the Merger Agreement and to consummate the transactions contemplated therein, the parties hereto desire to agree to certain matters as set forth herein.

AGREEMENT

NOW, THEREFORE, in consideration of the foregoing and the mutual agreements contained herein, and intending to be legally bound hereby, the parties hereto hereby agree as follows:

ARTICLE I <u>SUPPORT AGREEMENT; COVENANTS; RELEASE; VOTING TRUST AGREEMENT</u>

11.16 <u>Binding Effect of Merger Agreement</u>. The Stockholder hereby acknowledges that it has read the Merger Agreement and this Support Agreement and has had the opportunity to consult with its tax and legal advisors. During the term commencing on the date hereof and ending at the Expiration Time, the Stockholder shall be bound by and comply with Section 8.04 (*Confidentiality; Publicity*) of the Merger Agreement (and any relevant definitions contained in any such Sections) as if (a) the Stockholder was an original signatory to the Merger Agreement with respect to such provision, and (b) each reference to the "Stockholder" contained in Section 8.04 of the Merger Agreement also referred to the Stockholder.

11.17 No Transfer. During the period commencing on the date hereof and ending on the earliest to occur of (a) the Effective Time and (b) such date and time as the Merger Agreement shall be terminated in accordance with its terms (the earlier of clauses (a) and (b), the "Expiration Time"), the Stockholder shall not, except in each case pursuant to the Merger Agreement, Transfer any of the Stockholder Shares, the Stockholder Warrants or the Stockholder Notes; provided, however, that the foregoing shall not prohibit Transfers between the Stockholder and any Affiliate of the Stockholder, so long as, prior to and as a condition to the effectiveness of any such Transfer, such Affiliate executes and delivers to Acquiror and the Company a joinder to this Support Agreement in substantially the form attached hereto as Annex A; provided, further, that any Transfer permitted under this Section 1.2 shall not relieve such Stockholder of its obligations under this Support Agreement. Any Transfer in violation of this Section 1.2 shall be null and void. Notwithstanding anything to the contrary above, the Stockholder shall not be prohibited from exercising or converting the Stockholder Warrants or the Stockholder Notes into Company Common Stock, and such Company Common Stock issued upon conversion of the Stockholder Warrants or the Stockholder Notes shall be deemed to be "Stockholder Shares" for purposes of this Support Agreement. "Transfer" shall mean the (a) sale or assignment of, offer to sell, contract or agreement to sell, hypothecate, pledge, grant of any option to purchase or otherwise dispose of or agreement to dispose of, directly or indirectly, or establishment or increase of a put equivalent position or liquidation with respect to or decrease of a call equivalent position within the meaning of Section 16 of the Securities Exchange Act of 1934, as amended, and the rules and regulations of the Securities and Exchange Commission promulgated thereunder with respect to, any security, (b) entry into any swap or other arrangement that transfers to another, in whole or in part, any of the economic consequences of ownership of any security, whether any such transaction is to be settled by delivery of such securities, in cash or otherwise, or (c) public announcement of any intention to effect any transaction specified in clause (a) or (b).

11.18 <u>New Shares</u>. In the event that after the date hereof (a) any shares of Company Common Stock are issued to the Stockholder pursuant to any stock dividend, stock split, recapitalization, reclassification, combination or exchange of the Stockholder Shares, on or affecting the Stockholder Shares owned by the Stockholder or otherwise, (b) the Stockholder purchases or otherwise acquires beneficial ownership of any Company Common Stock or other equity securities of the Company (such Company Common Stock or other equity securities of the Company (such Company Common Stock or other equity securities of the Company (such Company Common Stock or other equity securities of the Company (such Company Common Stock or other equity securities of the Stockholder acquired or purchased by the Stockholder shall be subject to the terms of this Support Agreement to the same extent as if they constituted the Stockholder Shares owned by the Stockholder as of the date hereof.

11.19 <u>Closing Date Deliverables</u>. Upon the Closing, the Stockholder shall deliver to Acquiror and the Company a duly executed copy of the Company Lock-Up Agreement, by and among Acquiror and the Stockholder.

11.20 <u>Conversion of Stockholder Notes</u>. In connection with the Closing and immediately prior to the Effective Time, the Stockholder and the Company each shall take all actions for the Stockholder Notes to be converted into shares of the Company Common Stock pursuant to the terms of the applicable Stockholder Note.

11.21 <u>Conversion of Stockholder Warrants</u>. In connection with the Closing and immediately prior to the Effective Time, the Stockholder and the Company each shall take all actions for the Stockholder Warrants to be converted into shares of the Company Common Stock pursuant to the terms of that certain Convertible Note and Warrant Purchase Agreement, dated as of June 15, 2023, between the Company and the Stockholder.

11.22 <u>Voting and Support Agreement</u>.

(a) The Stockholder irrevocably and unconditionally agrees that, prior to the Expiration Time, at any meeting of the stockholders of the Company, and in any action by written consent of the stockholders of the Company distributed by the Board of Directors of the Company or otherwise undertaken in respect of or as contemplated by the Merger Agreement or the transactions contemplated thereby in a form reasonably acceptable to Acquiror, the Stockholder shall, if a meeting is held, appear at the meeting, in person or by proxy, or otherwise cause its Stockholder Shares (to the extent such Stockholder Shares are entitled to vote on or provide consent with respect to such matter):

(i) to approve and adopt the Merger Agreement and the transactions contemplated thereby, including the Merger;

(ii) in any other circumstances upon which a consent, waiver or other approval is required under the Company Organizational Documents or under any agreements between the Company and its stockholders or otherwise sought with respect to the Merger Agreement or the transactions contemplated thereby, to vote, consent, waive or approve (or cause to be voted, consented, waived or approved) all of the Stockholder's Stockholder Shares held at such time in favor thereof (to the extent such Stockholder Shares are entitled to vote on or provide consent, waiver or approval with respect to such matter);

(iii) against any merger agreement or merger, consolidation, combination, sale of substantial assets, reorganization, recapitalization, dissolution, liquidation or winding up of or by the Company (other than the Merger Agreement and the transactions contemplated thereby, including the Merger);

(iv) against any change in the business, management or Board of Directors of the Company that would or would reasonably be expected to adversely affect the ability of the Company to consummate the transactions contemplated by the Merger Agreement, including the Merger; and

(v) against any proposal, action or agreement that would (A) impede, frustrate, prevent or nullify any provision of this Support Agreement, the Merger Agreement or the transactions contemplated thereby, including the Merger, (B) result in a breach in any respect of any covenant, representation, warranty or any other obligation or agreement of the Company under the Merger Agreement, (C) result in any of the conditions set forth in Article IX of the Merger Agreement not being fulfilled or (D) change in any manner the dividend policy or capitalization of, including the voting rights of any class of capital stock or securities convertible into capital stock of, the Company; and

(vi) the Stockholder hereby agrees that it shall not commit or agree to take any action inconsistent with the foregoing.

11.23 <u>Further Assurances</u>. The Stockholder shall take, or cause to be taken, all such further actions and do, or cause to be done, all things reasonably necessary (including under applicable Law) to consummate the Merger and the other transactions contemplated by the Merger Agreement and this Support Agreement, in each case, on the terms and subject to the conditions set forth therein and herein, as applicable.

11.24 <u>No Inconsistent Agreement</u>. The Stockholder hereby represents and covenants that such Stockholder has not entered into, and shall not enter into, any agreement that would restrict, limit or interfere with the performance of the Stockholder's obligations hereunder.

11.25 <u>Voting Trust Agreement</u>. The Stockholder hereby covenants and agrees that, within ten (10) Business Days after the Registration Statement is declared effective under the Securities Act, the Stockholder will enter into a voting trust agreement, on terms and conditions set forth in <u>Exhibit A</u> hereto and otherwise on terms and conditions to the reasonable satisfaction of the Stockholder, the Company and the Acquiror (the "<u>Voting Trust</u> <u>Agreement</u>"), with an independent third-party trustee appointed by the Stockholder (which trustee shall not be an Affiliate of the Stockholder or a member of the immediate family of any Affiliate of the Stockholder) (the "<u>Voting Trustee</u>"), pursuant to which the Stockholder will transfer and assign to the Voting Trustee all of the Closing Shares received by the Stockholder pursuant to the terms of the Merger Agreement, effective as of the Effective Time.

11.26 <u>Waiver and Release of Claims</u>. The Stockholder hereby covenants and agrees as follows:

(a) Subject to and conditioned upon the Effective Time, effective as of the Effective Time (and subject to the limitations set forth in paragraph (c) below), the Stockholder, on behalf of itself and its Affiliates and its and their respective successors, assigns, representatives, administrators, executors and agents, and any other person or entity claiming by, through, or under any of the foregoing (each a "<u>Releasing Party</u>" and, collectively, the "<u>Releasing Parties</u>," provided, for the avoidance of doubt, that the Acquiror, shall not be deemed a Releasing Party hereunder), does hereby unconditionally and irrevocably release, waive and forever discharge the Acquiror, the Company, and each of their past and present directors, officers, employees, agents, predecessors, successors, assigns, and Subsidiaries, from any and all past or present claims, demands, damages, judgments, causes of action and liabilities of any nature whatsoever, whether or not known, suspected or claimed, arising directly or indirectly from any act, omission, event or transaction occurring (or any circumstances existing) at or prior to the Effective Time (each a "<u>Claim</u>" and, collectively, the "<u>Claims</u>").

(b) The Stockholder acknowledges that it may hereafter discover facts in addition to or different from those which it now knows or believes to be true with respect to the subject matter of this Support Agreement, and that it may hereafter come to have a different understanding of the law that may apply to potential claims which it is releasing hereunder, but it affirms that, except as is otherwise specifically provided herein, it is its intention to fully, finally and forever settle and release any and all Claims. In furtherance of this intention, the Stockholder acknowledges that the releases contained herein shall be and remain in effect as full and complete general releases notwithstanding the discovery or existence of any such additional facts or different understandings of law. The Stockholder knowingly and voluntarily waives and releases any and all rights and benefits it may now have, or in the future may have, under Section 1542 of the California Civil Code (or any analogous law of any other state), which reads as follows:

"A GENERAL RELEASE DOES NOT EXTEND TO CLAIMS THAT THE CREDITOR OR RELEASING PARTY DOES NOT KNOW OR SUSPECT TO EXIST IN HIS OR HER FAVOR AT THE TIME OF EXECUTING THE RELEASE AND THAT, IF KNOWN BY HIM OR HER, WOULD HAVE MATERIALLY AFFECTED HIS OR HER SETTLEMENT WITH THE DEBTOR OR RELEASED PARTY."

The Stockholder understands that Section 1542, or a comparable statute, rule, regulation or order of another jurisdiction, gives the Stockholder the right not to release existing Claims of which the Stockholder is not aware, unless the Stockholder voluntarily chooses to waive this right. Having been so apprised, the Stockholder nevertheless hereby voluntarily elects to and does waive any rights it may have that are described in Section 1542, or such other comparable statute, rule, regulation or order, and elects to assume all risks for Claims that exist, existed or may hereafter exist in its favor, known or unknown, suspected or unsuspected, arising out of or related to claims or other matters purported to be released pursuant to this <u>Section 1.11</u>, in each case, effective as of the Effective Time. The Stockholder acknowledges and agrees that the foregoing waiver is an essential and material term of the release provided pursuant to this <u>Section 1.11</u> and that, without such waiver, the Acquiror and the Company would not have agreed to the terms of this Support Agreement.

(c) Notwithstanding the foregoing provisions of this <u>Section 1.11</u> or anything to the contrary set forth herein, the Releasing Party expressly does not release or discharge: (i) any Claims that arise under or are based upon the terms of (A) this Support Agreement, the Merger Agreement, any of the Ancillary Agreements or any other document, certificate or Contract executed or delivered in connection with the Merger Agreement; (B) the Registration Rights Agreement; or (C) any underwriting agreement, business combination marketing agreement, financial advisory agreement, engagement letter or any similar agreement in respect of the transactions contemplated by the Merger Agreement to which a Releasing Party may be a party and that (in each case) is expressly disclosed in the Schedules of the Company, as each such agreement or instrument described in this clause (i) may be amended in accordance with its terms and the terms set forth in (x) the Merger Agreement or (y) this Support Agreement or the other Ancillary Agreements (if and to the extent applicable), (ii) any rights with respect to the capital stock or warrants of the Company owned by such Releasing Party or the promissory notes issued by the Company to such Releasing Party, or (iii) any Claims for indemnification, contribution, set-off, reimbursement or similar rights pursuant to any certificate of incorporation or bylaws of the Acquiror or any of its Subsidiaries or any indemnity or similar agreements by the Acquiror or any of its Subsidiaries with or for the benefit of a Releasing Party solely to the extent (in each case) set forth in the Schedules of the Company or as contemplated by Section 7.01 of the Merger Agreement.

(d) Notwithstanding the foregoing provisions of this <u>Section 1.11</u>, nothing contained in this Support Agreement shall be construed as an admission by any party hereto of any liability of any kind to any other party hereto.

ARTICLE II REPRESENTATIONS AND WARRANTIES

12.01 <u>No Inconsistent Agreement</u>. The Stockholder hereby represents and covenants that the Stockholder has not entered into, and shall not enter into, any agreement that would restrict, limit or interfere with the performance of the Stockholder's obligations hereunder.

12.02 <u>Representations and Warranties of the Stockholder</u>. The Stockholder hereby represents and warrants as of the date hereof to Acquiror and the Company as follows:

(a) <u>Organization; Due Authorization</u>. The Stockholder is duly organized, validly existing and in good standing under the laws of the jurisdiction in which it is incorporated, formed, organized or constituted, and the execution, delivery and performance of this Support Agreement and the consummation of the transactions contemplated hereby are within the Stockholder's corporate, limited liability company or organizational powers and have been duly authorized by all necessary corporate, limited liability company or organizational actions on the part of the Stockholder. This Support Agreement has been duly executed and delivered by the Stockholder, and, assuming due authorization, execution and delivery by the other parties to this Support Agreement, this Support Agreement constitutes a legally valid and binding obligation of the Stockholder, enforceable against the Stockholder in accordance with the terms hereof (except as enforceability may be limited by bankruptcy laws, other similar laws affecting creditors' rights and general principles of equity affecting the availability of specific performance and other equitable remedies).

(b) <u>Ownership</u>. The Stockholder is the record and/or beneficial owner (within the meaning of Rule 13d-3 under the Exchange Act) of, and have good title to, all of the Stockholder Shares, which represent 81.66% of the voting power of the outstanding Company Stock entitled to vote, voting together as a single class, and there exist no Liens or any other limitation or restriction (including any restriction on the right to vote, sell or otherwise dispose) affecting any Stockholder Shares, other than pursuant to (i) this Support Agreement, (ii) the Company Organizational Documents, or (iii) any applicable securities laws. Other than the Stockholder Shares, the Stockholder Warrants and the Stockholder Notes, the Stockholder does not hold or own any rights to acquire (directly or indirectly) any equity securities of the Company or outstanding options, warrants, rights, convertible or exchangeable securities, "phantom" stock rights, stock appreciation rights, stock-based performance units or commitments for shares of the Company.

(c) <u>No Conflicts</u>. The execution and delivery of this Support Agreement by the Stockholder does not, and the performance by the Stockholder of its obligations hereunder will not, (i) conflict with or result in a violation of the organizational documents of the Stockholder or (ii) require any consent or approval that has not been given or other action that has not been taken by any Person (including under any contract binding upon the Stockholder or the Stockholder of its obligations under this Support Agreement. The Stockholder has full right and power to enter into this Support Agreement. For the avoidance of doubt, pursuant to the Company Organizational Documents, other than the consent of the Stockholder, no other consent of any other stockholder of the Company is required to obtain the Company Stockholder Approvals.

(d) <u>Litigation</u>. Except as set forth in Schedule 2.2(d) attached hereto, (i) there are no Actions pending against the Stockholder or, to the knowledge of the Stockholder, threatened against the Stockholder, before (or, in the case of any threatened Actions, that would be before) any Governmental Entity, that in any manner challenge or seek to prevent, enjoin or materially delay the performance by the Stockholder of its obligations under this Support Agreement, (ii) the Stockholder has not been suspended or expelled from membership in any securities or commodities exchange or association or had a securities or commodities license or registration denied, suspended or revoked and (iii) the Stockholder (A) is not subject to or a respondent in any legal action for any injunction, cease and desist order or order or stipulation to desist or refrain from any act or practice relating to the offering of securities in any jurisdiction, (B) has not been convicted of, or pleaded guilty to, any crime involving fraud relating to any financial transaction or handling of funds of another person, or pertaining to any dealings in any securities and (C) is not currently a defendant in any such criminal proceeding.

(e) <u>Acknowledgment</u>. The Stockholder understands and acknowledges that each of the Company and the Acquiror is entering into the Merger Agreement in reliance upon the Stockholder's execution, delivery and performance of this Support Agreement.

ARTICLE III MISCELLANEOUS

13.01 <u>Termination</u>. This Support Agreement and all of its provisions shall automatically terminate and be of no further force and effect upon the earliest to occur of (i) the Effective Time, and (ii) such date and time as the Merger Agreement shall be terminated in accordance with its terms. Upon such termination of this Support Agreement, all obligations of the parties under this Support Agreement will terminate, without any liability or other obligation on the part of any party hereto to any Person in respect hereof or the transactions contemplated hereby, and no party hereto shall have any rights against such party), whether under contract, tort or otherwise, with respect to the subject matter hereof; provided, however, that the termination of this Support Agreement shall not relieve any party hereto from liability arising in respect of any breach of this Support Agreement prior to such termination. This <u>ARTICLE III</u> shall survive the termination of this Support Agreement.

13.02 <u>Governing Law</u>. Sections 11.06 and 11.12 of the Merger Agreement are incorporated herein by reference, *mutatis mutandis*.

13.03 <u>Assignment</u>. This Support Agreement and all of the provisions hereof will be binding upon and inure to the benefit of the parties hereto and their respective heirs, successors and permitted assigns. Neither this Support Agreement nor any of the rights, interests or obligations hereunder will be assigned (including by operation of law) without the prior written consent of all of the other parties hereto.

13.04 <u>Specific Performance</u>. The parties hereto agree that irreparable damage may occur in the event that any of the provisions of this Support Agreement were not performed in accordance with their specific terms or were otherwise breached. It is accordingly agreed that the parties hereto shall be entitled to seek an injunction or injunctions to prevent breaches of this Support Agreement and to enforce specifically the terms and provisions of this Support Agreement in the Court of Chancery of the State of Delaware (or, to the extent such court does not have subject matter jurisdiction, the Superior Court of the State of Delaware or the United States District Court for the District of Delaware), this being in addition to any other remedy to which such party is entitled at law or in equity. In the event that any Action shall be brought in equity to enforce the provisions of this Support Agreement, no party shall allege, and each party hereby waives the defense, that there is an adequate remedy at law, and each party agrees to waive any requirement for the securing or posting of any bond in connection therewith.

13.05 <u>Amendment; Waiver</u>. This Support Agreement or any provision hereof may not be amended, changed, supplemented, waived or otherwise modified or terminated, except upon the execution and delivery of a written agreement executed by Acquiror, the Company and the Stockholder.

13.06 <u>Severability</u>. If any provision of this Support Agreement is held invalid or unenforceable by any court of competent jurisdiction, the other provisions of this Support Agreement will remain in full force and effect. Any provision of this Support Agreement held invalid or unenforceable only in part or degree will remain in full force and effect to the extent not held invalid or unenforceable.

13.07 <u>Notices</u>. All notices, consents, waivers and other communications under this Support Agreement must be in writing and will be deemed to have been duly given (a) if personally delivered, on the date of delivery; (b) if delivered by express courier service of national standing for next day delivery (with charges prepaid), on the Business Day following the date of delivery to such courier service; (c) if delivered by electronic mail, on the date of transmission if on a Business Day before 5:00 p.m. local time of the business address of the recipient party (otherwise on the next succeeding Business Day); and (d) if deposited in the United States mail, first-class postage prepaid, on the date of delivery, in each case to the appropriate addresses or electronic mail addresses as a party may designate by notice to the other parties in accordance with this <u>Section 3.7</u>):

If to Acquiror (before the Effective Time):

Attn: Mark Hanchett

E-mail: mark@nxuenergy.com

with a copy (which shall not constitute notice) to:

Snell & Wilmer L.L.P. Attn: Mike Donahey 1 E Washington Street, Suite 2700 Phoenix, AZ 85004 E-mail: mdonahey@swlaw.com

If to the Company:

Verde Bioresins, Inc. 1431 Orangethorpe Ave. Fullerton, CA 92831 Attn: Brian Gordon E-mail: brian@verdebioresins.com

with a copy (which shall not constitute notice) to:

Wilmer Cutler Pickering Hale and Dorr LLP 2600 El Camino Real, Suite 400 Palo Alto, CA 94306 Attn: Daniel Zimmermann E-mail: Daniel.Zimmermann@wilmerhale.com

If to the Stockholder:

Humanitario Capital LLC Attn: Terren S. Peizer E-mail: terren@acuitasgh.com

13.08 <u>Counterparts</u>. This Support Agreement may be executed in two or more counterparts (any of which may be delivered by electronic transmission), each of which shall constitute an original, and all of which taken together shall constitute one and the same instrument.

13.09 Entire Agreement. This Support Agreement and the agreements referenced herein constitute the entire agreement and understanding of the parties hereto in respect of the subject matter hereof and supersede all prior understandings, agreements or representations by or among the parties hereto to the extent they relate in any way to the subject matter hereof.

13.10 <u>Rights of Third Parties</u>. Nothing expressed or implied in this Support Agreement is intended or shall be construed to confer upon or give any Person, other than the parties hereto, any right or remedies under or by reason of this Support Agreement.

13.11 <u>Stockholder Transaction Expenses</u>. If the Closing occurs, at the Closing, Acquiror shall pay, or cause to be paid, to the Stockholder (or as directed by the Stockholder), by wire transfer of immediately available funds, the Stockholder Transaction Expenses (as defined below). If the Merger Agreement is terminated in accordance with its terms, promptly upon such termination, the Company shall reimburse the Stockholder for all Stockholder Transaction Expenses. As used herein, "<u>Stockholder Transaction Expenses</u>" means all reasonable and documented fees and disbursements of the Stockholder and its Affiliates payable to unaffiliated third parties (including, without limitation, outside counsel, agents, advisors, consultants, experts and financial advisors employed by or on behalf of the Stockholder or any of its Affiliates) incurred in connection with the Transactions, in each case, to the extent payable in cash; provided that such fees and disbursements of the Stockholder and its Affiliates shall not exceed \$200,000.00 in the aggregate.

[THE REMAINDER OF THIS PAGE IS INTENTIONALLY BLANK]

IN WITNESS WHEREOF, the Stockholder, Acquiror and the Company have each caused this Support Agreement to be duly executed as of the date first written above.

NXU, Inc.

By:

Name:Mark HanchettTitle:Chief Executive Officer

[Signature Page to Company Support Agreement]

IN WITNESS WHEREOF, the Stockholder, Acquiror and the Company have each caused this Support Agreement to be duly executed as of the date first written above.

Humanitario Capital LLC

By:

Name:Terren S. PeizerTitle:Authorized Signatory

[Signature Page to Company Support Agreement]

IN WITNESS WHEREOF, the Stockholder, Acquiror and the Company have each caused this Support Agreement to be duly executed as of the date first written above.

Verde Bioresins, Inc.

By:

Name: Brian Gordon Title: President

[Signature Page to Company Support Agreement]

Annex A

Form of Joinder Agreement

This Joinder Agreement (this "Joinder Agreement") is made as of the date written below by the undersigned (the "Joining Party") in accordance with the Company Support Agreement, dated as of [___], 2024 (as amended, supplemented or otherwise modified from time to time, the "Support Agreement"), by and among Humanitario Capital LLC, a Puerto Rico limited liability company (the "Stockholder"), NXU, Inc., a Delaware corporation prior ("Acquiror"), and Verde Bioresins, Inc., a Delaware corporation (the "Company"). Capitalized terms used herein and not otherwise defined shall have the meaning ascribed to them in the Support Agreement.

The Joining Party hereby acknowledges, agrees and confirms that, by its execution of this Joinder Agreement, the Joining Party shall be deemed to be a party to, and a "Stockholder" under, the Support Agreement as of the date hereof and shall have all of the rights and obligations of the Stockholder as if it had executed the Support Agreement. The Joining Party hereby ratifies, as of the date hereof, and agrees to be bound by, all of the terms, provisions and conditions contained in the Support Agreement.

IN WITNESS WHEREOF, the undersigned has duly executed this Joinder Agreement as of the date written below.

Name:

By:______ Title: Date:

Notice Address:

[•]

[•]

[•]

Exhibit A

Terms of Voting Trust Agreement

In the event that, at any time during the term of the Voting Trust Agreement, (a) any shares of Acquiror Common Stock are issued to the Stockholder pursuant to any stock dividend, stock split, recapitalization, reclassification, combination or exchange of any of the Closing Shares, on or affecting any of the Closing Shares owned by the Stockholder or otherwise, (b) the Stockholder purchases or otherwise acquires beneficial ownership of any shares of Acquiror Common Stock or (c) the Stockholder acquires the right to vote or share in the voting of any shares of Acquiror Common Stock, then all such shares of Acquiror Common Stock shall be subject to the Voting Trust Agreement.

During the term of the Voting Trust Agreement, the Voting Trustee shall vote all shares of Acquiror Common Stock subject to the Voting Trust Agreement in favor of any vote, consent or other action in the same proportion as voted by public stockholders of the Company that are not Affiliates of the Company with respect to such matter.

The Voting Trust Agreement shall remain in effect until the earliest to occur of the following: (a) Mr. Peizer is sentenced to a non-custodial sentence, or finishes service of any custodial part of a sentence, on the charges described in Schedule 2.2(d) attached to the Support Agreement, or the conviction on such charges is set aside or reversed, (b) the fifth anniversary of the Closing Date, and (c) such time as the Stockholder and its Affiliates collectively beneficially own shares of Acquiror Common Stock representing in the aggregate less than 10% of the total voting power of all shares of Acquiror Common Stock then outstanding.

ACQUIROR SUPPORT AGREEMENT

This Acquiror Support Agreement (this "<u>Support Agreement</u>") is dated as of October [•], 2024, by and among, Mark Hanchett ("<u>Hanchett</u>"), Annie Pratt ("<u>Pratt</u>", and together with Hanchett, the "<u>Stockholders</u>"), NXU, Inc., a Delaware corporation ("<u>NXU</u>"), and Verde Bioresins, Inc., a Delaware corporation (the "<u>Company</u>"). Capitalized terms used but not defined herein shall have the meanings ascribed to such terms in the Merger Agreement.

RECITALS

WHEREAS, as of the date hereof, each Stockholder is the holder of record and/or the "beneficial owner" (within the meaning of Rule 13d-3 under the Exchange Act) of that number of (a) shares of Class A Common Stock, par value \$0.0001, of NXU (the "<u>Class A Common Stock</u>"), (b) Class B Common Stock, par value \$0.0001, of NXU (the "<u>Class B Common Stock</u>", and together with the Class A Common Stock, the "<u>Common Stock</u>") and (c) options set forth on the signature page of such Stockholder (all such securities held by a Stockholder, the "<u>Stockholder Securities</u>");

WHEREAS, contemporaneously with the execution and delivery of this Support Agreement, NXU, Nxu Merger Sub, Inc., a Delaware corporation and a direct wholly owned subsidiary of NXU ("<u>Merger Sub I</u>"), Nxu Merger Sub, LLC, a Delaware limited liability company ("<u>Merger Sub II</u>", and together with Merger Sub I, the "<u>Merger Subs</u>"), and the Company are entering into an Agreement and Plan of Merger (as amended or modified from time to time, the "<u>Merger Agreement</u>"), dated as of the date hereof, pursuant to which, among other transactions, Merger Sub I is to merge with and into the Company, with the Company continuing on as the surviving company and a wholly owned subsidiary of NXU and then merging with and into Merger Sub II, with Merger Sub II becoming the surviving entity, on the terms and conditions set forth therein (the "<u>Merger</u>"); and

WHEREAS, as an inducement to NXU and the Company to enter into the Merger Agreement and to consummate the transactions contemplated therein, the parties hereto desire to agree to certain matters as set forth herein.

AGREEMENT

NOW, THEREFORE, in consideration of the foregoing and the mutual agreements contained herein, and intending to be legally bound hereby, the parties hereto hereby agree as follows:

Article I <u>SUPPORT AGREEMENT; COVENANTS; RELEASE</u>

I.1 <u>Binding Effect of Merger Agreement</u>. Each Stockholder hereby acknowledges that he or she has read the Merger Agreement and this Support Agreement and has had the opportunity to consult with his or her tax and legal advisors. During the period commencing on the date hereof and ending at the Expiration Time (as defined below), each Stockholder shall be bound by and comply with Sections 7.08 (*Exclusivity*) and 8.04 (*Confidentiality; Publicity*) of the Merger Agreement (and any relevant definitions contained in any such Sections) as if (a) such Stockholder was an original signatory to the Merger Agreement with respect to such provisions, and (b) each reference to the "Acquiror" contained in Sections 7.08 and 8.04 also referred to such Stockholder.

12 No Transfer. During the period commencing on the date hereof and ending on the earliest to occur of (a) the Effective Time (and, in such case, for the avoidance of doubt, not following the Effective Time) and (b) such date and time as the Merger Agreement shall be terminated in accordance with its terms (the earlier of clauses (a) and (b), the "Expiration Time"), neither Stockholder shall Transfer any Stockholder Securities held by such Stockholder as of the date of this Support Agreement; provided, however, either Stockholder may Transfer shares of Stockholder Securities in connection with bona fide estate planning purposes, either during such Stockholder's lifetime or on death by will or intestacy to such Stockholder's spouse, child (natural or adopted), or any other direct lineal descendant, or any custodian or trustee of any trust, partnership, limited liability company or other corporate entity for the benefit of, or the ownership interests of which are owned wholly by such Stockholder or any such family members, provided that such permitted transferees execute and deliver to NXU and the Company a joinder to this Support Agreement in substantially the form attached hereto as Annex A; provided, further, that any Transfer permitted under this Section 1.2 shall not relieve such Stockholder of his or her obligations under this Support Agreement. Any Transfer in violation of this Section 1.2 shall be null and void. "Transfer" shall mean the (i) sale or assignment of, offer to sell, contract or agreement to sell, hypothecate, pledge, grant of any option to purchase or otherwise dispose of or agreement to dispose of, directly or indirectly, or establishment or increase of a put equivalent position or liquidation with respect to or decrease of a call equivalent position within the meaning of Section 16 of the Securities Exchange Act of 1934, as amended, and the rules and regulations of the Securities and Exchange Commission promulgated thereunder with respect to, any security, (ii) entry into any swap or other arrangement that transfers to another, in whole or in part, any of the economic consequences of ownership of any security, whether any such transaction is to be settled by delivery of such securities, in cash or otherwise, or (iii) public announcement of any intention to effect any transaction specified in clause (i) or (ii).

I.3 <u>New Shares</u>. In the event that after the date hereof but prior to the Expiration Time (a) either Stockholder acquires any shares of Common Stock or other equity securities of NXU or any rights to acquire such shares of Common Stock or other equity securities of NXU pursuant to any stock dividend, stock split, recapitalization, reclassification, combination or exchange of shares of Common Stock, on or affecting the shares of Common Stock or other equity securities of NXU, or (c) either Stockholder purchases or otherwise acquires beneficial ownership of any shares of Common Stock or other equity securities of NXU, or (c) either Stockholder acquires the right to vote or share in the voting of any shares of Common Stock or other equity securities of NXU, or (c) either Stockholder acquires the right to vote or share in the voting of any shares of Common Stock or other equity securities of NXU, or (c) either Stockholder acquires the right to vote or share in the voting of any shares of Common Stock or other equity securities of NXU, collectively, the "<u>New Securities</u>"), then such New Securities acquired or purchased by either such Stockholder shall be subject to the terms of this Support Agreement to the same extent as if they constituted the shares of Common Stock owned by such Stockholder as of the date hereof.

I.4 Voting and Support Agreement.

a. Each Stockholder irrevocably and unconditionally agrees that, prior to the Expiration Time, at any meeting of the stockholders of NXU, however called, or at any adjournment thereof, or in any other circumstance in which the vote, consent or other approval of the stockholders of NXU is sought, each Stockholder shall (i) appear at each such meeting or otherwise cause all of his or her Common Stock to be counted as present thereat for purposes of calculating a quorum and (ii) vote (or cause to be voted), or execute and deliver a written consent (or cause a written consent to be executed and delivered) covering, all of his or her Common Stock:

i. in favor of the approval and adoption of the Proposals;

ii. against any Competing Proposal;

iii. against any change in the business, management or Board of Directors of NXU (other than in connection with the Transaction Proposals) that would or would reasonably be expected to adversely affect the ability of NXU to consummate the transactions contemplated by the Merger Agreement, including the Merger; and

iv. against any proposal, action or agreement that would (A) impede, frustrate, prevent or nullify any provision of this Support Agreement, the Merger Agreement or the Transactions, including the Merger, (B) result in a breach in any respect of any covenant, representation, warranty or any other obligation or agreement of NXU or Merger Sub under the Merger Agreement, (C) result in any of the conditions set forth in Article IX of the Merger Agreement not being fulfilled or (D) change in any manner the dividend policy or capitalization of, including the voting rights of any class of capital stock of, NXU.

Each Stockholder hereby agrees that he or she shall not commit or agree to take any action inconsistent with the foregoing.

b. During the period commencing on the date hereof and ending at the Expiration Time, such Stockholder shall not modify or amend any agreement set forth on <u>Schedule I</u>.

I.5 <u>Further Assurances</u>. Each Stockholder shall take, or cause to be taken, all such further actions and do, or cause to be done, all things reasonably necessary (including under applicable Law) to consummate the Merger and the other transactions contemplated by the Merger Agreement and this Support Agreement, in each case, on the terms and subject to the conditions set forth therein and herein, as applicable.

I.6 <u>No Inconsistent Agreement</u>. Each Stockholder hereby represents and covenants that such Stockholder has not entered into, and shall not enter into, any agreement that would restrict, limit or interfere with the performance of his or her obligations hereunder.

I.7 <u>Waiver and Release of Claims</u>. Each Stockholder hereby covenants and agrees as follows:

(a) Subject to and conditioned upon the Effective Time, effective as of the Effective Time (and subject to the limitations set forth in paragraph (c) below), such Stockholder, on behalf of himself, herself and his or her Affiliates and his or her and their respective successors, assigns, representatives, administrators, executors and agents, and any other person or entity claiming by, through, or under any of the foregoing (each a "<u>Releasing Party</u>" and, collectively, the "<u>Releasing Parties</u>," provided, for the avoidance of doubt, that NXU, shall not be deemed a Releasing Party hereunder), does hereby unconditionally and irrevocably release, waive and forever discharge the NXU, the Company, and each of their past and present directors, officers, employees, agents, predecessors, successors, assigns, and Subsidiaries, from any and all past or present claims, demands, damages, judgments, causes of action and liabilities of any nature whatsoever, whether or not known, suspected or claimed, arising directly or indirectly from any act, omission, event or transaction occurring (or any circumstances existing) at or prior to the Effective Time (each a "<u>Claim</u>" and, collectively, the "<u>Claims</u>").

(b) Such Stockholder acknowledges that he or she may hereafter discover facts in addition to or different from those which he or she now knows or believes to be true with respect to the subject matter of this Agreement, and that he or she may hereafter come to have a different understanding of the law that may apply to potential claims which it is releasing hereunder, but he or she affirms that, except as is otherwise specifically provided herein, it is his or her intention to fully, finally and forever settle and release any and all Claims. In furtherance of this intention, such Stockholder acknowledges that the releases contained herein shall be and remain in effect as full and complete general releases notwithstanding the discovery or existence of any such additional facts or different understandings of law. Such Stockholder knowingly and voluntarily waives and releases any and all rights and benefits he or she may now have, or in the future may have, under Section 1542 of the California Civil Code (or any analogous law of any other state), which reads as follows:

"A GENERAL RELEASE DOES NOT EXTEND TO CLAIMS THAT THE CREDITOR OR RELEASING PARTY DOES NOT KNOW OR SUSPECT TO EXIST IN HIS OR HER FAVOR AT THE TIME OF EXECUTING THE RELEASE AND THAT, IF KNOWN BY HIM OR HER, WOULD HAVE MATERIALLY AFFECTED HIS OR HER SETTLEMENT WITH THE DEBTOR OR RELEASED PARTY."

Such Stockholder understands that Section 1542, or a comparable statute, rule, regulation or order of another jurisdiction, gives such Stockholder the right not to release existing Claims of which such Stockholder is not aware, unless such Stockholder voluntarily chooses to waive this right. Having been so apprised, such Stockholder nevertheless hereby voluntarily elects to and does waive any rights he or she may have that are described in Section 1542, or such other comparable statute, rule, regulation or order, and elects to assume all risks for Claims that exist, existed or may hereafter exist in his or her favor, known or unknown, suspected or unsuspected, arising out of or related to claims or other matters purported to be released pursuant to this Section 1.7, in each case, effective as of the Effective Time. Such Stockholder acknowledges and agrees that the foregoing waiver is an essential and material term of the release provided pursuant to this Section 1.7 and that, without such waiver, NXU and the Company would not have agreed to the terms of this Agreement.

(c) Notwithstanding the foregoing provisions of this Section 1.7 or anything to the contrary set forth herein, the Releasing Parties do not release or discharge, and each Releasing Party expressly does not release or discharge any Claims (i) that arise under or are based upon the terms of this Agreement, the Merger Agreement, any of the Ancillary Agreements or any other document, certificate or Contract executed or delivered in connection with the Merger Agreement or any similar agreement in respect of the transactions contemplated by the Merger Agreement to which a Releasing Party may be a party and that (in each case) is expressly disclosed in the Schedules of NXU, as each such agreement or instrument described above may be amended in accordance with its terms and the terms set forth in (A) the Merger Agreement or (B) this Agreement or the other Ancillary Agreements (if and to the extent applicable), (ii) any rights with respect to the Common Stock owned by such Releasing Party, or (iii) any Claims for indemnification, contribution, set-off, reimbursement or similar rights pursuant to any certificate of incorporation or bylaws of NXU or any of its Subsidiaries or any indemnity or similar agreements by the NXU with or for the benefit of a Releasing Party solely to the extent (in each case) set forth in the Schedules of Acquiror or as contemplated by Section 7.01 of the Merger Agreement.

(d) Notwithstanding the foregoing provisions of this Section 1.7, nothing contained in this Agreement shall be construed as an admission by any party hereto of any liability of any kind to any other party hereto.

Article II <u>REPRESENTATIONS AND WARRANTIES</u>

II.1 <u>No Inconsistent Agreement</u>. Except as contemplated or permitted under the Merger Agreement, each Stockholder hereby represents and covenants that such Stockholder has not entered into, and shall not enter into, any agreement that would restrict, limit or interfere with the performance of such Stockholder's obligations hereunder.

II.2 <u>Representations and Warranties of the Stockholders</u>. Each Stockholder hereby represents and warrants (severally as to such Stockholder and not jointly) as of the date hereof to NXU and the Company as follows:

a. <u>Organization; Due Authorization</u>. This Agreement has been duly executed and delivered by such Stockholder, and, assuming due authorization, execution and delivery by the other parties to this Agreement, this Agreement constitutes a legally valid and binding obligation of such Stockholder, enforceable against such Stockholder in accordance with the terms hereof (except as enforceability may be limited by bankruptcy laws, other similar laws affecting creditors' rights and general principles of equity affecting the availability of specific performance and other equitable remedies).

b. <u>Ownership</u>. Such Stockholder is the record and/or beneficial owner (within the meaning of Rule 13d-3 under the Exchange Act) of, and has good title to, all of his or her Stockholder Securities, and there exist no Liens or any other limitation or restriction (including any restriction on the right to vote, sell or otherwise dispose) affecting any such Stockholder Securities, other than pursuant to (i) this Agreement, (ii) the organizational documents of NXU, (iii) the Merger Agreement or (iv) any applicable securities laws. Other than the Stockholder Securities described on the signature page of such Stockholder hereto, such Stockholder does not hold or own any rights to acquire (directly or indirectly) any equity securities of NXU or outstanding options, warrants, rights, convertible or exchangeable securities, "phantom" stock rights, stock appreciation rights, stock-based performance units or commitments for shares of NXU.

c. <u>No Conflicts</u>. The execution and delivery of this Agreement by such Stockholder does not, and the performance by such Stockholder of his or her obligations hereunder will not, require any consent or approval that has not been given or other action that has not been taken by any Person (including under any contract binding upon such Stockholder or the Stockholder Shares), in each case to the extent such consent, approval or other action would prevent, enjoin or materially delay the performance by such Stockholder of his or her obligations under this Agreement. Such Stockholder has full right and power to enter into this Agreement.

d. <u>Litigation</u>. There are no Actions pending against such Stockholder, or to the knowledge of such Stockholder threatened against such Stockholder, before (or, in the case of threatened Legal Proceedings, that would be before) any Governmental Entity, that in any manner challenges or seeks to prevent, enjoin or materially delay the performance by such Stockholder of his or her obligations under this Agreement. Such Stockholder (i) is not subject to or a respondent in any legal action for any injunction, cease and desist order or order or stipulation to desist or refrain from any act or practice relating to the offering of securities in any jurisdiction, (ii) has never been convicted of, or pleaded guilty to, any crime involving fraud relating to any financial transaction or handling of funds of another person, or pertaining to any dealings in any securities and (iii) is not currently a defendant in any such criminal proceeding.

e. <u>Acknowledgment</u>. Such Stockholder understands and acknowledges that the Company is entering into the Merger Agreement in reliance upon such Stockholder's execution, delivery and performance of this Support Agreement.

Article III <u>MISCELLANEOUS</u>

III.1 <u>Termination</u>. This Support Agreement and all of its provisions shall automatically terminate and be of no further force and effect upon the earliest to occur of (i) the Effective Time, and (ii) such date and time as the Merger Agreement shall be terminated in accordance with its terms. Upon such termination of this Support Agreement, all obligations of the parties under this Support Agreement will terminate, without any liability or other obligation on the part of any party hereto to any Person in respect hereof or the transactions contemplated hereby, and no party hereto shall have any rights against such party), whether under contract, tort or otherwise, with respect to the subject matter hereof; provided, however, that the termination of this Support Agreement shall not relieve any party hereto from liability arising in respect of any breach of this Support Agreement prior to such termination. This ARTICLE III shall survive the termination of this Support Agreement.

III.2 <u>Governing Law</u>. Sections 11.06 and 11.12 of the Merger Agreement are incorporated herein by reference, *mutatis mutandis*.

III.3 <u>Assignment</u>. This Support Agreement and all of the provisions hereof will be binding upon and inure to the benefit of the parties hereto and their respective heirs, successors and permitted assigns. Neither this Support Agreement nor any of the rights, interests or obligations hereunder will be assigned (including by operation of law) without the prior written consent of all of the other parties hereto.

III.4 <u>Specific Performance</u>. The parties hereto agree that irreparable damage may occur in the event that any of the provisions of this Support Agreement were not performed in accordance with their specific terms or were otherwise breached. It is accordingly agreed that the parties hereto shall be entitled to seek an injunction or injunctions to prevent breaches of this Support Agreement and to enforce specifically the terms and provisions of this Support Agreement in the Court of Chancery of the State of Delaware (or, to the extent such court does not have subject matter jurisdiction, the Superior Court of the State of Delaware or the United States District Court for the District of Delaware), this being in addition to any other remedy to which such party is entitled at law or in equity. In the event that any Action shall be brought in equity to enforce the provisions of this Support Agreement, no party shall allege, and each party hereby waives the defense, that there is an adequate remedy at law, and each party agrees to waive any requirement for the securing or posting of any bond in connection therewith.

III.5 <u>Amendment; Waiver</u>. This Support Agreement or any provision hereof may not be amended, changed, supplemented, waived or otherwise modified or terminated, except upon the execution and delivery of a written agreement executed by NXU, the Company and each Stockholder.

III.6 <u>Severability</u>. If any provision of this Support Agreement is held invalid or unenforceable by any court of competent jurisdiction, the other provisions of this Support Agreement will remain in full force and effect. Any provision of this Support Agreement held invalid or unenforceable only in part or degree will remain in full force and effect to the extent not held invalid or unenforceable.

III.7 <u>Notices</u>. All notices, consents, waivers and other communications under this Agreement must be in writing and will be deemed to have been duly given (a) if personally delivered, on the date of delivery; (b) if delivered by express courier service of national standing for next day delivery (with charges prepaid), on the Business Day following the date of delivery to such courier service; (c) if delivered by electronic mail, on the date of transmission if on a Business Day before 5:00 p.m. local time of the business address of the recipient party (otherwise on the next succeeding Business Day); and (d) if deposited in the United States mail, first-class postage prepaid, on the date of delivery, in each case to the appropriate addresses or electronic mail addresses as a party may designate by notice to the other parties in accordance with this <u>Section 3.7</u>):

If to NXU:

Attn: Mark Hanchett E-mail: mark@nxuenergy.com

with a copy (which shall not constitute notice) to:Attn: Mike Donahey1 E Washington Street, Suite 2700

Phoenix, AZ 85004 E-mail: mdonahey@swlaw.com If to the Company:

Verde Bioresins, Inc. 1431 E. Orangethorpe Ave. Fullerton, CA 92831 Attn: Brian Gordon E-mail: brian@verdebioresins.com

with a copy (which shall not constitute notice) to: Wilmer Cutler Pickering Hale and Dorr LLP 2600 El Camino Real, Suite 400 Palo Alto, CA 94306 Attn: Daniel Zimmermann E-mail: Daniel.Zimmerman@wilmerhale.com

III.8 <u>Counterparts</u>. This Support Agreement may be executed in two or more counterparts (any of which may be delivered by electronic transmission), each of which shall constitute an original, and all of which taken together shall constitute one and the same instrument.

III.9 <u>Entire Agreement</u>. This Support Agreement and the agreements referenced herein constitute the entire agreement and understanding of the parties hereto in respect of the subject matter hereof and supersede all prior understandings, agreements or representations by or among the parties hereto to the extent they relate in any way to the subject matter hereof.

[THE REMAINDER OF THIS PAGE IS INTENTIONALLY BLANK]

IN WITNESS WHEREOF, each Stockholder, NXU and the Company have each caused this Support Agreement to be duly executed as of the date first written above.

NXU, Inc.

By:

Name:Mark HanchettTitle:Chief Executive Officer

[Signature Page to Acquiror Support Agreement]

IN WITNESS WHEREOF, each Stockholder, NXU and the Company have each caused this Support Agreement to be duly executed as of the date first written above.

Verde Bioresins, Inc.

By: Name: Brian Gordon Title: President

[Signature Page to Acquiror Support Agreement]

IN WITNESS WHEREOF, each Stockholder, NXU and the Company have each caused this Support Agreement to be duly executed as of the date first written above.

Mark Hanchett

Notice Address:

Stockholder Securities: 196,692 Options (underwater, none exercised) 3,826.3 shares of Class A Common Stock 26,503,675 shares of Class B Common Stock

[Signature Page to Acquiror Support Agreement]

IN WITNESS WHEREOF, each Stockholder, NXU and the Company have each caused this Support Agreement to be duly executed as of the date first written above.

Annie Pratt

Notice Address:

Stockholder Securities: 76,812 Options (underwater, none exercised) 3,186 shares of Class A Common Stock 10,021,695 shares of Class B Common Stock

[Signature Page to Acquiror Support Agreement]

<u>Annex A</u>

Form of Joinder Agreement

This Joinder Agreement (this "Joinder Agreement") is made as of the date written below by the undersigned (the "Joining Party") in accordance with the Acquiror Support Agreement, dated as of October [_], 2024 (as amended, supplemented or otherwise modified from time to time, the "Support Agreement"), by and among Mark Hanchett, Annie Pratt, Nxu, Inc., a Delaware corporation, and Verde Bioresins, Inc., a Delaware corporation. Capitalized terms used herein and not otherwise defined shall have the meaning ascribed to them in the Support Agreement.

The Joining Party hereby acknowledges, agrees and confirms that, by its execution of this Joinder Agreement, the Joining Party shall be deemed to be a party to, and a "Stockholder" under, the Support Agreement as of the date hereof and shall have all of the rights and obligations of the Stockholders as if it had executed the Support Agreement. The Joining Party hereby ratifies, as of the date hereof, and agrees to be bound by, all of the terms, provisions and conditions contained in the Support Agreement.

IN WITNESS WHEREOF, the undersigned has duly executed this Joinder Agreement as of the date written below.

Name:

By:_____ Title: Date:

Notice Address:

Exhibit C Form of Registration Rights Agreement

REGISTRATION RIGHTS AGREEMENT

THIS REGISTRATION RIGHTS AGREEMENT (this "<u>Agreement</u>"), dated as of October [\bullet], 2024, is made and entered into by and among [Verde Bioresins, Corp (f/k/a NXU, Inc.)], a Delaware corporation (the "<u>Company</u>") and the undersigned parties listed under Holder on the signature page hereto (each such party, and any person or entity who hereafter becomes a party to this Agreement pursuant to <u>Section 5.2</u> or <u>Section 5.11</u> of this Agreement, a "<u>Holder</u>" and collectively the "<u>Holders</u>"). Except as otherwise stated, capitalized terms used but not otherwise defined herein shall have the meanings provided in the Merger Agreement (as defined below).

RECITALS

WHEREAS, pursuant to the Merger Agreement, dated as of October [•], 2024 (the "<u>Merger Agreement</u>"), by and among the Company, Nxu Merger Sub, Inc., a Delaware corporation and wholly-owned subsidiary of the Company ("<u>Merger Sub I</u>"), Nxu Merger Sub, LLC, a Delaware limited liability company and wholly-owned subsidiary of the Company ("<u>Merger Sub II</u>"), and Verde Bioresins, Inc., a Delaware corporation ("<u>Verde</u>"), Verde will become a wholly-owned subsidiary of the Company (the "<u>Merger</u>");

WHEREAS, after the closing of the Merger (the "<u>Closing</u>"), the Holders will own shares of Common Stock, par value \$0.0001 per share, of the Company (the "<u>Common Stock</u>"); and

WHEREAS, the Company and the Holders desire to enter into this Registration Rights Agreement in order to provide the Holders certain registration rights with respect to the Registrable Securities (as defined below) on the terms set forth herein.

NOW, **THEREFORE**, in consideration of the representations, covenants and agreements contained herein, and certain other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto, intending to be legally bound, hereby agree as follows:

ARTICLE I DEFINITIONS

1.1 **Definitions**. The terms defined in this <u>Article I</u> shall, for all purposes of this Agreement, have the respective meanings set forth below:

"Additional Holder" shall have the meaning given in Section 5.11.

"Additional Holder Common Stock" shall have the meaning given in Section 5.11.

"<u>Adverse Disclosure</u>" shall mean any public disclosure of material non-public information, which disclosure, in the good faith judgment of the Chief Executive Officer or Chief Financial Officer of the Company, after consultation with counsel to the Company, (a) would be required to be made in (i) any Registration Statement in order for the applicable Registration Statement not to contain any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary to make the statements therein not misleading or (ii) any Prospectus in order for the applicable Prospectus not to include any untrue statement of a material fact or omit to state a material fact necessary in order to make the statements therein, in the light of the circumstances under which they were made, not misleading, (b) would not be required to be made at such time if the Registration Statement were not being filed, and (c) the Company has a bona fide business purpose for not making such information public.

"Agreement" shall have the meaning given in the Preamble.

"Block Trade" shall have the meaning given to it in subsection 2.4.1.

"Board" shall mean the board of directors of the Company.

"<u>Closing</u>" shall have the meaning given in the Recitals hereto.

"Commission" shall mean the Securities and Exchange Commission.

"Common Stock" shall have the meaning given in the Recitals hereto.

"<u>Company</u>" shall have the meaning given in the Preamble.

"Demanding Holder" shall have the meaning given in subsection 2.1.5.

"Effectiveness Period" shall have the meaning given in subsection 3.1.1 of this Agreement.

"Exchange Act" shall mean the Securities Exchange Act of 1934, as it may be amended from time to time.

"Existing Holders" shall have the meaning given in the Recitals hereto.

"Financial Counterparty" shall have the meaning given in subsection 2.4.1 of this Agreement.

"Holder Indemnified Persons" shall have the meaning given in subsection 4.1.1 of this Agreement.

"Holder Information" shall have the meaning given in subsection 4.1.2.

"Holders" shall have the meaning given in the Preamble.

"Joinder" shall have the meaning given in Section 5.11.

"Maximum Number of Securities" shall have the meaning given in subsection 2.1.6 of this Agreement.

"<u>Merger</u>" has the meaning given in the Recitals hereto.

"Merger Sub" shall have the meaning given in the Recitals hereto.

"Minimum Underwritten Offering Threshold" shall have the meaning given in subsection 2.1.5.

"<u>Misstatement</u>" shall mean, in the case of a Registration Statement, an untrue statement of a material fact or an omission to state a material fact required to be stated therein, or necessary to make the statements therein not misleading, and in the case of a Prospectus, an untrue statement of a material fact or an omission to state a material fact necessary in order to make the statements therein, in the light of the circumstances under which they were made, not misleading.

"Other Coordinated Offering" shall have the meaning given to it in subsection 2.4.1.

"Permitted Transferees" shall mean any person or entity to whom a Holder is permitted to transfer Registrable Securities.

"Piggyback Registration" shall have the meaning given in subsection 2.2.1 of this Agreement.

"Pro Rata" shall have the meaning given in subsection 2.1.6 of this Agreement.

"Prospectus" shall mean the prospectus included in any Registration Statement, as supplemented by any and all prospectus supplements and as amended by any and all post- effective amendments and including all material incorporated by reference in such prospectus.

"Registrable Securities" shall mean (a) any outstanding shares of Common Stock or any other equity security (including the shares of Common Stock issued or issuable upon the exercise of any other equity security) of the Company held by a Holder as of the date of this Agreement, (b) any Additional Holder Common Stock, (c) any shares of the Company issued or to be issued to any Holders in connection with the Merger and (d) any other equity security of the Company issued or issuable with respect to any such shares of Common Stock by way of a share capitalization or share sub-division or in connection with a combination of shares, recapitalization, merger, consolidation or reorganization; provided, however, that, as to any particular Registrable Securities, such securities shall cease to be Registrable Securities when: (A) a Registration Statement with respect to the sale of such securities shall have been effective under the Securities Act and such securities shall have been sold, transferred, disposed of or exchanged in accordance with such Registration Statement; (B) such securities shall have been otherwise transferred, new certificates for such securities not bearing a legend restricting further transfer shall have been delivered by the Company and subsequent public distribution of such securities shall not require registration under the Securities Act; (C) such securities shall have ceased to be outstanding; or (D) such securities may be sold without registration pursuant to Rule 144 and Rule 145 (as applicable) promulgated under the Securities Act (or any successor rule promulgated thereafter by the Commission) (but with no current public information requirement or volume, manner of sale or other restrictions or limitations) with respect to securities that were at no time held by any Affiliate or the Company), as reasonably determined by the Company, upon the advice of counsel to the Company.

"<u>Registration</u>" shall mean a registration effected by preparing and filing a registration statement or similar document in compliance with the requirements of the Securities Act, and the applicable rules and regulations promulgated thereunder, and any such registration statement having been declared effective by, or become effective pursuant to the rules promulgated by, the Commission.

"Registration Expenses" shall mean the out-of-pocket expenses of a Registration, including, without limitation, the following:

a. all registration and filing fees (including fees with respect to filings required to be made with the Financial Industry Regulatory Authority, Inc. and any national securities exchange on which the shares of Common Stock is then listed);

b. fees and expenses of compliance with securities or blue sky laws (including reasonable fees and disbursements of counsel for the Underwriters in connection with blue sky qualifications of Registrable Securities);

c. reasonable printing, messenger, telephone and delivery expenses;

d. reasonable fees and disbursements of counsel for the Company;

e. reasonable fees and disbursements of all independent registered public accountants of the Company incurred specifically in connection with such Registration or Underwritten Offering;

f. the fees and expenses incurred in connection with the listing of any Registrable Securities on each national securities exchange on which the shares of Common Stock is then listed; and

g. the fees and expenses incurred by the Company in connection with any Underwritten Offerings or other offering involving an Underwriter.

"<u>Registration Statement</u>" shall mean any registration statement under the Securities Act that covers the Registrable Securities pursuant to the provisions of this Agreement, including the Prospectus included in such registration statement, amendments (including post-effective amendments) and supplements to such registration statement and all exhibits to and all material incorporated by reference in such registration statement.

"Requesting Holder" shall have the meaning given in <u>subsection 2.1.5</u> of this Agreement.

"Securities Act of 1933, as amended from time to time.

"Shelf Registration" shall have the meaning given in subsection 2.1.1 of this Agreement.

"Subsequent Shelf Registration Statement" shall have the meaning given in subsection 2.1.3.

"Transfer" shall mean the (a) sale or assignment of, offer to sell, contract or agreement to sell, hypothecate, pledge, grant of any option to purchase or otherwise dispose of or agreement to dispose of, directly or indirectly, or establishment or increase of a put equivalent position or liquidation with respect to or decrease of a call equivalent position within the meaning of Section 16 of the Exchange Act with respect to, any security, (b) entry into any swap or other arrangement that transfers to another, in whole or in part, any of the economic consequences of ownership of any security, whether any such transaction is to be settled by delivery of such securities, in cash or otherwise, or (c) public announcement of any intention to effect any transaction specified in clause (a) or (b).

"<u>Underwriter</u>" shall mean a securities dealer who purchases any Registrable Securities as principal or as broker, placement agent or sales agent pursuant to a Registration and not as part of such dealer's market-making activities.

"Underwritten Demand" shall have the meaning given in subsection 2.1.5 of this Agreement.

"<u>Underwritten Offering</u>" or "<u>Underwritten Registration</u>" shall mean a Registration in which shares of Common Stock of the Company are sold to an Underwriter in a firm commitment underwriting for distribution to the public. For the avoidance of doubt, the terms "Underwritten Offering" and "Underwritten Registration" do not include firm commitment underwritten offerings of convertible debt securities.

"Verde Holders" shall mean the parties listed under Verde Holders on Schedule A hereto.

"Withdrawal Notice" shall have the meaning given in subsection 2.1.7.

ARTICLE II REGISTRATIONS

2.1 Registration.

2.1.1 <u>Shelf Registration</u>. The Company agrees that, within forty-five (45) calendar days after the consummation of the Merger, the Company will use its commercially reasonable efforts to file with the Commission (at the Company's sole cost and expense) a Registration Statement registering the resale or other disposition of the Registrable Securities (a "<u>Shelf Registration</u>"), which Shelf Registration may include shares of Common Stock that may be issuable upon exercise of outstanding warrants, or shares of Common Stock issued or issuable in respect of securities that may have been purchased in any private placement that was consummated at or prior to the Effective Time. Such Shelf Registration shall provide for the resale of the Registrable Securities included therein pursuant to any method or combination of methods legally available (the "<u>Plan of Distribution</u>") to, and requested by, any Holder named therein.

2.1.2 Effective Registration. The Company shall use its commercially reasonable efforts to cause such Registration Statement to be declared effective by the Commission as soon as reasonably practicable after the initial filing of the Registration Statement. Subject to the limitations contained in this Agreement, the Company shall effect any Shelf Registration on such appropriate registration form of the Commission (a) as shall be selected by the Company and (b) as shall permit the resale or other disposition of the Registrable Securities by the Holders. The Company shall use its commercially reasonable efforts to have the Registration Statement declared effective as soon as practicable after the filing thereof, but no later than the earlier of (i) the 90th calendar day (or 120th calendar day if the Commission notifies the Company that it will "review" the Registration Statement) following the Effective Time and (ii) the 10th business day after the date the Company is notified (orally or in writing, whichever is earlier) by the Commission that the Registration Statement will not be "reviewed" or will not be subject to further review (such earlier date, the "Effective Date"). If at any time a Registration Statement filed with the Commission pursuant to <u>subsection 2.1.1</u> is effective and a Holder provides written notice to the Company that it intends to effect an offering of all or part of the Registration Statement as may be necessary in order to enable such offering to take place in accordance with the terms of this Agreement.

Subsequent Shelf Registration. If any Registration Statement ceases to be effective under the Securities Act for any reason at 213 any time while Registrable Securities are still outstanding, the Company shall, subject to Section 3.4, use its commercially reasonable efforts to as promptly as is reasonably practicable cause such Registration Statement to again become effective under the Securities Act (including using its commercially reasonable efforts to obtain the prompt withdrawal of any order suspending the effectiveness of such Registration Statement), and shall use its commercially reasonable efforts to as promptly as is reasonably practicable amend such Registration Statement in a manner reasonably expected to result in the withdrawal of any order suspending the effectiveness of such Registration Statement or file an additional Registration Statement as a Shelf Registration (a "Subsequent Shelf Registration Statement") registering the resale of all Registrable Securities (determined as of two (2) business days prior to such filing), and pursuant to the Plan of Distribution, and requested by, any Holder named therein. If a Subsequent Shelf Registration Statement is filed, the Company shall use its commercially reasonable efforts to (i) cause such Subsequent Shelf Registration Statement to become effective under the Securities Act as promptly as is reasonably practicable after the filing thereof (it being agreed that the Subsequent Shelf Registration Statement shall be an automatic shelf registration statement (as defined in Rule 405 promulgated under the Securities Act) if the Company is a well-known seasoned issuer (as defined in Rule 405 promulgated under the Securities Act) at the most recent applicable eligibility determination date), but no later than the earlier of (i) the 60th calendar day (or 90th calendar day if the Commission notifies the Company that it will "review" the Subsequent Shelf Registration Statement) following the date on which the Subsequent Shelf Registration Statement is filed and (ii) keep such Subsequent Shelf Registration Statement continuously effective, available for use to permit the Holders named therein to sell their Registrable Securities included therein and in compliance with the provisions of the Securities Act until such time as there are no longer any Registrable Securities outstanding. Any such Subsequent Shelf Registration Statement shall be a Registration Statement on Form S-3 to the extent that the Company is eligible to use such form. Otherwise, such Subsequent Shelf Registration Statement shall be on another appropriate form. The Company's obligation under this subsection 2.1.3, shall, for the avoidance of doubt, be subject to Section 3.4.

2.1.4 Additional Registrable Securities. Subject to Section 3.4, in the event that any Holder holds Registrable Securities that are not or that are unable to be registered for resale on a delayed or continuous basis, the Company, upon written request of a Holder, shall promptly use its commercially reasonable efforts to cause the resale of such Registrable Securities to be covered, at the Company's option, by any then available Registration Statement (including by means of a post-effective amendment) or by filing a Subsequent Shelf Registration Statement and cause the same to become effective as soon as practicable after such filing, but no later than the earlier of (i) the 60th calendar day (or 90th calendar day if the Commission notifies the Company that it will "review" the Subsequent Shelf Registration Statement is filed, and such Registration Statement or Subsequent Shelf Registration Statement shall be subject to the terms hereof; provided, however, that the Company shall only be required to cause such Registrable Securities to be so covered twice per calendar year for the Holders.

Underwritten Offering. Subject to the provisions of subsection 2.1.6 and Section 2.5 of this Agreement, a Holder or group of 215 Holders (a Holder or group of Holders being in such case, a "Demanding Holder") may make a written demand for an Underwritten Offering pursuant to a Registration Statement filed with the Commission in accordance with subsection 2.1.1 of this Agreement (an "Underwritten Demand"); provided, that the Company shall only be obligated to effect an Underwritten Offering if such offering shall include Registrable Securities proposed to be sold by the Demanding Holder, either individually or together with other Demanding Holders, with a total offering price reasonably expected to exceed, in the aggregate, \$25.0 million (the "Minimum Underwritten Offering Threshold"). The Demanding Holder shall have the responsibility to engage an underwriter(s) (which shall consist of one (1) or more reputable nationally or regionally recognized investment banks); provided that such selection shall be subject to the consent of the Company. The Company shall have no responsibility for engaging any underwriter(s) for an Underwritten Offering. The Company shall, within five (5) business days of the Company's receipt of the Underwritten Demand, notify, in writing, all other Holders of such demand, and each Holder who thereafter requests to include all or a portion of such Holder's Registrable Securities in such Underwritten Offering pursuant to such Underwritten Demand (each such Holder, a "Requesting Holder") shall so notify the Company, in writing, within two (2) business days (one (1) business day if such offering is an overnight or bought Underwritten Offering) after the receipt by such Holder of the notice from the Company. Upon receipt by the Company of any such written notification from a Requesting Holder(s), such Requesting Holder(s) shall be entitled to have their Registrable Securities included in such Underwritten Offering pursuant to such Underwritten Demand. In such event, the right of any Holder or Requesting Holder to registration pursuant to this subsection 2.1.5, shall be conditioned upon such Holder's or Requesting Holder's participation in such underwriting and the inclusion of such Holder's or Requesting Holder's Registrable Securities in the underwriting to the extent provided herein. All such Holders or Requesting Holders proposing to distribute their Registrable Securities through such Underwritten Offering under this subsection 2.1.5 shall enter into an underwriting agreement in customary form with the Underwriter(s) selected for such Underwritten Offering by the Demanding Holders initiating such Underwritten Offering. Notwithstanding the foregoing, the Company is not obligated to effect more than an aggregate of five (5) Underwritten Offerings demanded by the Verde Holders pursuant to this subsection 2.1.5 and is not obligated to effect an Underwritten Offering pursuant to this subsection 2.1.5 within one hundred and eight (180) days after the closing of an Underwritten Offering.

216 Reduction of Underwritten Offering. If the managing Underwriter or Underwriters in an Underwritten Offering, pursuant to an Underwritten Demand, in good faith, advises or advise the Company, the Demanding Holders, the Requesting Holders and other persons or entities holding Registrable Securities or other equity securities of the Company that were requested to be included in such Underwritten Offering, taken together with all other shares of Common Stock or other securities which the Company desires to sell and the shares of Common Stock or other securities, if any, as to which registration has been requested pursuant to written contractual piggyback registration rights held by other equity holders of the Company who desire to sell (if any) that the dollar amount or number of Registrable Securities or other equity securities of the Company requested to be included in such Underwritten Offering exceeds the maximum dollar amount or maximum number of equity securities of the Company that can be sold in the Underwritten Offering without adversely affecting the proposed offering price, the timing, the distribution method or the probability of success of such offering (such maximum dollar amount or maximum number of such securities, as applicable, the "Maximum Number of Securities"), then the Company shall include in such Underwritten Offering, as follows: (i) first, the Registrable Securities of the Demanding Holders (pro rata based on the respective number of Registrable Securities that each Demanding Holder has requested be included in such Underwritten Offering, regardless of the number of shares held by each such person and the aggregate number of Registrable Securities that the Demanding Holders have requested be included in such Underwritten Offering (such proportion is referred to herein as "Pro Rata")) that can be sold without exceeding the Maximum Number of Securities; (ii) second, to the extent that the Maximum Number of Securities has not been reached under the foregoing clause (i), the Registrable Securities of the Requesting Holders, Pro Rata, which can be sold without exceeding the Maximum Number of Securities; (iii) third, to the extent that the Maximum Number of Securities has not been reached under the foregoing clauses (i) and (ii), the shares of Common Stock or other equity securities of the Company that the Company desires to sell and that can be sold without exceeding the Maximum Number of Securities; and (iv) fourth, to the extent that the Maximum Number of Securities has not been reached under the foregoing clauses (i), (ii) and (iii), the shares of Common Stock or other equity securities of the Company held by other persons or entities that the Company is obligated to include pursuant to separate written contractual arrangements with such persons or entities and that can be sold without exceeding the Maximum Number of Securities.

Withdrawal. Prior to the filing of the applicable "red herring" prospectus or prospectus supplement used for marketing such 2.1.7Underwritten Offering, a majority-in-interest of the Demanding Holders initiating an Underwritten Offering shall have the right to withdraw from such Underwritten Offering for any or no reason whatsoever upon written notification (a "Withdrawal Notice") to the Company and the Underwritter or Underwritters (if any) of their intention to withdraw from such Underwritten Offering, prior to the public announcement of the Underwritten Offering by the Company; provided that a Holder may elect to have the Company continue an Underwritten Offering if the Minimum Underwritten Offering Threshold would still be satisfied by the Registrable Securities proposed to be sold in the Underwritten Offering the Holders or any of their respective Permitted Transferees, as applicable. If withdrawn, a demand for an Underwritten Offering shall constitute a demand for an Underwritten Offering by the withdrawing Demanding Holder for purposes of subsection 2.1.6, unless either (i) such Demanding Holder has not previously withdrawn any Underwritten Offering or (ii) such Demanding Holder reimburses the Company for all Registration Expenses with respect to such Underwritten Offering (or, if there is more than one Demanding Holder, a pro rata portion of such Registration Expenses based on the respective number of Registrable Securities that each Demanding Holder has requested be included in such Underwritten Offering); provided that, if a Holder elects to continue an Underwritten Offering pursuant to the proviso in the immediately preceding sentence, such Underwritten Offering shall instead count as an Underwritten Offering demanded by the such Holder for purposes of subsection 2.1.6. Following the receipt of any Withdrawal Notice, the Company shall promptly forward such Withdrawal Notice to any other Holders that had elected to participate in such Underwritten Offering. Notwithstanding anything to the contrary in this Agreement, the Company shall be responsible for the Registration Expenses incurred in connection with an Underwritten Offering prior to its withdrawal under this subsection 2.1.7, other than if a Demanding Holder elects to pay such Registration Expenses pursuant to clause (ii) of the second sentence of this subsection 2.1.7.

2.2 Piggyback Registration.

Piggyback Rights. Subject to the provisions of subsection 2.2.2 and Section 2.5 hereof, if, at any time on or after the date the 2.2.1Company consummates the Merger, the Company proposes to consummate an Underwritten Offering for its own account or for the account of stockholders of the Company other than holders of any securities subject to that certain Registration Rights Agreement, dated as of November 3, 2022, by and among the Company and certain institutional investors party thereto, as amended on January 5, 2023 (the "Existing Registration Rights Agreement"), then the Company shall give written notice of such proposed action to all of the Holders as soon as practicable, which notice shall (i) describe the amount and type of securities to be included, the intended method(s) of distribution and the name of the proposed managing Underwriter or Underwriters, if any, and (ii) offer to all of the Holders the opportunity to include such number of Registrable Securities as such Holders may request in writing within two (2) business days (unless such offering is an overnight or bought Underwritten Offering, then one (1) business day), in each case after receipt of such written notice (such Registration a "Piggyback Registration"). The Company shall, in good faith, cause such Registrable Securities to be included in such Piggyback Registration and shall use its commercially reasonable efforts to cause the managing Underwriter or Underwriters of a proposed Underwritten Offering to permit the Registrable Securities requested by the Holders pursuant to this subsection 2.2.1 to be included in a Piggyback Registration on the same terms and conditions as any similar securities of the Company included in such Piggyback Registration and to permit the resale or other disposition of such Registrable Securities in accordance with the intended method(s) of distribution thereof. All such Holders proposing to include Registrable Securities in an Underwritten Offering under this subsection 2.2.1 shall enter into an underwriting agreement in customary form with the Underwriter(s) selected for such Underwritten Offering by the Company.

2.2.2 **Reduction of Piggyback Registration.** If the managing Underwriter or Underwriters in an Underwritten Offering that is to be a Piggyback Registration, in good faith, advises the Company and the Holders of Registrable Securities participating in the Piggyback Registration in writing that the dollar amount or number of shares of equity securities of the Company that the Company desires to sell, taken together with (i) the shares of equity securities of the Company, if any, as to which the Underwritten Offering has been demanded pursuant to separate written contractual arrangements with persons or entities other than the Holders of Registrable Securities hereunder, (ii) the Registrable Securities as to which a Piggyback Registration has been requested pursuant to <u>Section 2.2</u> of this Agreement and (iii) the shares of equity securities of the Company, if any, as to which inclusion in the Underwritten Offering has been requested pursuant to separate written contractual piggyback registration rights of other stockholders of the Company, exceeds the Maximum Number of Securities, then:

(a) If the Underwritten Offering is undertaken for the Company's account, the Company shall include in any such Underwritten Offering (A) first, the shares of Common Stock or other equity securities of the Company that the Company desires to sell, which can be sold without exceeding the Maximum Number of Securities; (B) second, to the extent that the Maximum Number of Securities has not been reached under the foregoing clause (A), the Registrable Securities of Holders requesting a Piggyback Registration pursuant to subsection 2.2.1 of this Agreement, Pro Rata, which can be sold without exceeding the Maximum Number of Securities; and (C) third, to the extent that the Maximum Number of Securities has not been reached under the foregoing clauses (A) and (B), the shares of Common Stock or other equity securities of the Company, if any, as to which inclusion in the Underwritten Offering has been requested pursuant to written contractual piggyback registration rights of other stockholders of the Company, which can be sold without exceeding the Maximum Number of Securities;

(b) If the Underwritten Offering is pursuant to a request by persons or entities other than the Holders of Registrable Securities, then the Company shall include in any such Underwritten Offering (A) first, the shares of Common Stock or other equity securities of the Company, if any, of such requesting persons or entities, other than the Holders, which can be sold without exceeding the Maximum Number of Securities has not been reached under the foregoing <u>clause (A)</u>, the Registrable Securities of Holders requesting a Piggyback Registration pursuant to <u>subsection 2.2.1</u> of this Agreement, Pro Rata, which can be sold without exceeding the Maximum Number of Securities; (C) third, to the extent that the Maximum Number of Securities has not been reached under the foregoing <u>clauses (A)</u> and (B), the shares of Common Stock or other equity securities of the Company that the Company desires to sell, which can be sold without exceeding the Maximum Number of Securities; and (D) fourth, to the extent that the Maximum Number of Securities has not been reached under the foregoing <u>clauses (A)</u>, (B), and (C), the shares of Common Stock or other equity securities of the Company for the account of other persons or entities that the Company is obligated to register pursuant to separate written contractual arrangements with such persons or entities, which can be sold without exceeding the Maximum Number of Securities; or

(c) If the Underwritten Offering is pursuant to a request by Holder(s) of Registrable Securities pursuant to <u>Section 2.1</u> hereof, then the Company shall include in any such Registration or registered offering securities in the priority set forth in <u>subsection 2.1.6</u>.

2.2.3 Piggyback Registration Withdrawal. Any Holder of Registrable Securities (other than a Demanding Holder, whose right to withdraw from an Underwritten Offering, and related obligations, shall be governed by <u>subsection 2.1.7</u>) shall have the right to withdraw from a Piggyback Registration upon written notification to the Company and the Underwriter or Underwriters (if any) of his, her or its intention to withdraw from such Piggyback Registration prior to the filing of the applicable "red herring" prospectus or prospectus supplement used for marketing the Underwritten Offering. Notwithstanding anything to the contrary in this Agreement, the Company shall be responsible for the Registration Expenses incurred in connection with the Piggyback Registration prior to its withdrawal under this <u>subsection 2.2.3</u>. The Company (whether on its own good faith determination or as a result of a withdrawal by persons making a demand pursuant to written contractual obligations) may withdraw an Underwritten Offering undertaken for the Company's account at any time prior to the effectiveness of such Registration Statement.

2.2.4 <u>Unlimited Piggyback Registration Rights</u>. For purposes of clarity, subject to <u>subsection 2.1.7</u>, any Piggyback Registration or Underwritten Offering effected pursuant to <u>Section 2.2</u> of this Agreement shall not be counted as an Underwritten Offering pursuant to an Underwritten Demand effected under <u>Section 2.1</u> of this Agreement.

2.3 <u>Market Stand-off</u>. In connection with any Underwritten Offering of equity securities of the Company (other than a Block Trade or Other Coordinated Offering), if requested by the managing Underwriters, each Holder of Registrable Securities in excess of five percent (5%) of the outstanding Common Stock that participates and sells Registrable Securities in such Underwritten Offering (and for which it is customary for such a Holder to agree to a lock-up) agrees that it shall not Transfer any shares of Common Stock or other equity securities of the Company (other than those included in such offering pursuant to this Agreement), without the prior written consent of the Company, during the ninety (90)-day period (or such shorter time agreed to by the managing Underwriters) beginning on the date of pricing of such offering, except as expressly permitted by such lock-up agreement or in the event the managing Underwriters otherwise agree by written consent. Each such Holder that participates and sells Registrable Securities in such Underwritters to such effect (in each case on substantially the same terms and conditions as all such Holders that execute a lock-up agreement).

2.4 Block Trades and Other Coordinated Offerings.

2.4.1 Notwithstanding any other provision of this <u>Article II</u>, but subject to <u>Section 3.4</u>, at any time and from time to time when an effective Registration Statement is on file with the Commission, if a Demanding Holder wishes to engage in (a) an underwritten registered offering not involving a "roadshow," an offer commonly known as a "block trade" (a "<u>Block Trade</u>") or (b) an "at the market" or similar registered offering through a broker, sales agent or distribution agent, whether as agent or principal, (an "<u>Other Coordinated Offering</u>"), in each case, with a total offering price reasonably expected to exceed, in the aggregate, either (x) \$25 million or (y) all remaining Registrable Securities held by the Demanding Holder, then if such Demanding Holder requires any assistance from the Company pursuant to this <u>Section 2.4</u>, such Holder shall notify the Company of the Block Trade or Other Coordinated Offering; <u>provided</u> that the Demanding Holders representing a majority of the Registrable Securities wishing to engage in the Block Trade or Other Coordinated Offering shall use commercially reasonable efforts to work with the Company and any Underwriters or brokers, sales agents or placement agents (each, a "<u>Financial Counterparty</u>") (including by disclosing the maximum number of Registrable Securities proposed to be the subject of such Block Trade or Other Coordinated Offering documentation related to the Block Trade or Other Coordinated Offering documentation related to the Block Trade or Other Coordinated Offering block Trade or Other Coordinated to

2.4.2 Prior to the filing of the applicable "red herring" prospectus or prospectus supplement used in connection with a Block Trade or Other Coordinated Offering, a majority-in interest of the Demanding Holders initiating such Block Trade or Other Coordinated Offering shall have the right to submit a Withdrawal Notice to the Company, the Underwriter or Underwriters (if any) and Financial Counterparty (if any) of their intention to withdraw from such Block Trade or Other Coordinated Offering. Notwithstanding anything to the contrary in this Agreement, the Company shall be responsible for the Registration Expenses incurred in connection with a Block Trade or Other Coordinated Offering prior to its withdrawal under this subsection 2.4.2.

2.4.3 Notwithstanding anything to the contrary in this Agreement, <u>Section 2.2</u> shall not apply to a Block Trade or Other Coordinated Offering initiated by a Demanding Holder pursuant to <u>Section 2.4</u> of this Agreement.

2.4.4 The Demanding Holder in a Block Trade or Other Coordinated Offering shall have the right to select the Underwriters and Financial Counterparty (if any) for such Block Trade or Other Coordinated Offering (in each case, which shall consist of one or more reputable nationally recognized investment banks).

2.4.5 A Holder in the aggregate may demand no more than two (2) Block Trades or Other Coordinated Offerings pursuant to this <u>Section 2.4</u> in any twelve (12) month period. For the avoidance of doubt, any Block Trade or Other Coordinated Offering effected pursuant to this <u>Section 2.4</u> shall not be counted as a demand for an Underwritten Offering pursuant to <u>subsection 2.1.5</u> hereof.

2.5 **Restrictions on Registration Rights**. If (a) the Holders have requested an Underwritten Registration and the Company and the Holders are unable to obtain the commitment of underwriters to firmly underwrite the offer; (b) in the good faith judgment of the Board such Underwritten Offering would be detrimental to the Company and the Board concludes as a result that it is necessary or appropriate to defer the undertaking of such Underwritten Offering at such time (including, without limitation, because such Underwritten Offering would be detrimental to another transaction that the Company is contemplating at that time or because such Underwritten Offering) or (c) the Underwritten Offering would require inclusion in the Registration Statement or any prospectus of financial statements that are then unavailable to the Company, then in each case, as applicable, the Company shall furnish to such Holders a certificate signed by the Chairman of the Board stating the applicable reason(s) set forth in <u>clauses (a)</u> through (c) above underlying the Company's decision to defer the undertaking of such Underwritten Offering. In such event, the Company shall have the right to defer such offering for a period of not more than seventy-five (75) days; provided, however, that the Company shall not defer its obligations in this manner more than twice, or for more than one hundred and twenty (120) days in the aggregate, in any twelve (12) month period.

2.6 Holder Right to Withdraw from Registration Statement Under Certain Circumstances. In no event shall any Holder be identified as a statutory underwriter in any Registration Statement without its prior written consent; provided, that if the Securities and Exchange Commission requires that a Holder be identified as a statutory underwriter in a Registration Statement, such Holder will have the option, in its sole and absolute discretion, to either (i) have the opportunity to withdraw from such Registration Statement upon its prompt written request to the Company or (ii) be included as such in the Registration Statement.

ARTICLE III COMPANY PROCEDURES

3.1 <u>General Procedures.</u> In connection with effecting any Underwritten Offering, Block Trade, and/or Other Coordinated Offering, subject to applicable law and any regulations promulgated by any securities exchange on which the Company's equity securities are then listed, each as interpreted by the Company with the advice of its counsel, the Company shall use its commercially reasonable efforts to effect such Registration or Underwritten Offering to permit the resale or other disposition of such Registrable Securities in accordance with the intended plan of distribution thereof (and including all manners of distribution in such Registration Statement as Holders may reasonably request in connection with the filing of such Registration Statement and as permitted by law, including distribution of Registrable Securities to a Holder's members, security holders or partners), and pursuant thereto the Company shall, as expeditiously as possible and to the extent applicable:

3.1.1 prepare and file with the Commission after the consummation of the Merger a Registration Statement with respect to such Registrable Securities and use its reasonable best efforts to cause such Registration Statement to become effective in accordance with <u>Section 2.1</u>, including filing a replacement Registration Statement, if necessary, and remain effective until all Registrable Securities covered by such Registration Statement have been sold or are no longer outstanding (such period, the "<u>Effectiveness Period</u>");

3.1.2 prepare and file with the Commission such amendments and post-effective amendments to the Registration Statement, and such supplements to the Prospectus, (a) as may be reasonably requested by any Holder that holds at least five-percent (5%) of the Registrable Securities registered on such Registration Statement or any Underwriter, or (b) as may be required by the rules, regulations or instructions applicable to the registration form used by the Company or by the Securities Act or rules and regulations thereunder to keep the Registration Statement effective until all Registrable Securities covered by such Registration Statement are sold in accordance with the plan of distribution provided by the Holders and as set forth in such Registration Statement or supplement to the Prospectus or are no longer outstanding;

3.1.3 prior to filing a Registration Statement or Prospectus, or any amendment or supplement thereto, furnish without charge to the Underwriters, if any, and the Holders of Registrable Securities included in such Registration or Underwritten Offering, and such Holders' legal counsel, copies of such Registration Statement as proposed to be filed, each amendment and supplement to such Registration Statement (in each case including all exhibits thereto and documents incorporated by reference therein), the Prospectus (including each preliminary Prospectus) and such other documents as the Underwriters and the Holders of Registrable Securities included in such Registration or Underwritten Offering or the legal counsel for any such Holders may reasonably request in order to facilitate the disposition of the Registrable Securities owned by such Holders; <u>provided</u> that the Company will not have any obligation to provide any document pursuant to this <u>subsection 3.1.3</u> that is available on the Commission's EDGAR system;

3.1.4 prior to any Underwritten Offering of Registrable Securities, use its commercially reasonable efforts to (i) register or qualify the Registrable Securities covered by the Registration Statement under such securities or "blue sky" laws of such jurisdictions in the United States as the Holders of Registrable Securities included in such Registration Statement (in light of their intended plan of distribution) may request (or provide evidence satisfactory to such Holders that the Registrable Securities are exempt from such registration or qualification) and (ii) take such action necessary to cause such Registrable Securities covered by the Registration Statement to be registered with or approved by such other governmental authorities as may be necessary by virtue of the business and operations of the Company and do any and all other acts and things that may be necessary or advisable to enable the Holders of Registrable Securities included in such Registration Statement to consummate the disposition of such Registrable Securities in such jurisdictions; provided, however, that the Company shall not be required to qualify generally to do business in any jurisdiction where it would not otherwise be required to qualify or take any action to which it would be subject to general service of process or taxation in any such jurisdiction where it is not then otherwise so subject;

3.1.5 cause all such Registrable Securities to be listed on each national securities exchange or automated quotation system on which similar securities issued by the Company are then listed;

3.1.6 provide a transfer agent or warrant agent, as applicable, and registrar for all such Registrable Securities no later than the effective date of such Registration Statement or Underwritten Offering;

3.1.7 advise each seller of such Registrable Securities, promptly after it shall receive notice or obtain knowledge thereof, of the issuance of any stop order by the Commission suspending the effectiveness of such Registration Statement or the initiation or threatening of any proceeding for such purpose and promptly use its commercially reasonable efforts to prevent the issuance of any stop order or to obtain its withdrawal if such stop order should be issued;

3.1.8 during the Effectiveness Period, furnish a conformed copy of each filing of any Registration Statement or Prospectus or any amendment or supplement to such Registration Statement or Prospectus or any document that is to be incorporated by reference into such Registration Statement or Prospectus, promptly after such filing of such documents with the Commission to each seller of such Registrable Securities or its counsel; provided that the Company will not have any obligation to provide any document pursuant to this subsection 3.1.8 that is available on the Commission's EDGAR system;

3.1.9 notify the Holders at any time when a Prospectus relating to such Registration Statement is required to be delivered under the Securities Act, of the happening of any event as a result of which the Prospectus included in such Registration Statement, as then in effect, includes a Misstatement, and then to correct such Misstatement as set forth in Section 3.4 of this Agreement;

3.1.10 in the event of an Underwritten Offering, a Block Trade, an Other Coordinated Offering, or sale by a Financial Counterparty pursuant to such Registration, permit a representative of the Holders (such representative to be selected by a majority of the Holders), the Underwriters, or other financial institutions facilitating such Underwritten Offering, Block Trade, Other Coordinated Offering or other sale pursuant to such Registration, if any, and any attorney, consultant or accountant retained by such Holders or Underwriter to participate, at each such person's or entity's own expense, in the preparation of the Registration Statement or the Prospectus, and cause the Company's officers, directors and employees to supply all information reasonably requested by any such representatives or Underwriters or financial institutions agree to confidentiality arrangements in form and substance reasonably satisfactory to the Company, prior to the release or disclosure of any such information;

3.1.11 obtain a comfort letter from the Company's independent registered public accountants in the event of an Underwritten Offering, a Block Trade or sale by a Financial Counterparty pursuant to such Registration (subject to such Financial Counterparty providing such certifications and/or representations reasonably requested by the Company's independent registered public accountants and the Company's counsel), in customary form and covering such matters of the type customarily covered by comfort letters as the managing Underwriter or other similar type of sales agent or placement agent may reasonably request;

3.1.12 in the event of an Underwritten Offering, a Block Trade, an Other Coordinated Offering or sale by an Underwriter pursuant to such Registration, on the date the Registrable Securities are delivered for sale pursuant to such Registration, obtain an opinion and negative assurance letter, dated such date, of counsel representing the Company for the purposes of such Registration, addressed to the participating Underwriter covering such legal matters with respect to the Registration in respect of which such opinion and negative assurance letter is being given as the participating Underwriter may reasonably request and as are customarily included in such opinions and negative assurance letters, and reasonably satisfactory to such participating Underwriter;

3.1.13 in the event of an Underwritten Offering or a Block Trade, or an Other Coordinated Offering or sale by an Underwriter pursuant to such Registration to which the Company has consented, to the extent reasonably requested by such Underwriter in order to engage in such offering, allow the Underwriter to conduct customary "underwriter's due diligence" with respect to the Company;

3.1.14 in the event of any Underwritten Offering, a Block Trade, an Other Coordinated Offering or sale by a Financial Counterparty pursuant to such Registration, enter into and perform its obligations under an underwriting or other purchase or sales agreement, in usual and customary form, with the managing Underwriter or the Financial Counterparty of such offering or sale;

3.1.15 make available to its security holders, as soon as reasonably practicable, an earnings statement covering the period of at least twelve (12) months beginning with the first day of the Company's first full calendar quarter after the effective date of the Registration Statement which satisfies the provisions of Section 11(a) of the Securities Act and Rule 158 thereunder (or any successor rule promulgated thereafter by the Commission);

3.1.16 with respect to an Underwritten Offering pursuant to <u>subsection 2.1.5</u> use its commercially reasonable efforts to make available senior executives of the Company to participate in customary "road show" presentations that may be reasonably requested by the Underwriter in any Underwritten Offering; and

3.1.17 otherwise, in good faith, cooperate reasonably with, and take such customary actions as may reasonably be requested by the participating Holders, consistent with the terms of this Agreement, in connection with such Registration. Notwithstanding the foregoing, the Company shall not be required to provide any documents or information to an Underwriter or Financial Counterparty if such Underwriter or Financial Counterparty has not then been named with respect to the applicable Underwritten Offering or other offering involving a registration as an Underwriter or Financial Counterparty, as applicable.

3.2 **<u>Registration Expenses.</u>** The Registration Expenses in respect of all Registrations shall be borne by the Company. It is acknowledged by the Holders that the Holders shall bear all incremental selling expenses relating to the sale of Registrable Securities, such as Underwriters' commissions and discounts, brokerage fees, Underwriter marketing costs and, other than as set forth in the definition of "Registration Expenses," all fees and expenses of any legal counsel representing the Holders, in each case pro rata based on the number of Registrable Securities that such Holders have sold in such Registration.

3.3 Requirements for Participation in Underwritten Offerings. Notwithstanding anything in this Agreement to the contrary, if any Holder does not provide the Company with its requested Holder Information within a reasonable amount of time after such request (and a minimum of five (5) business days), the Company may exclude such Holder's Registrable Securities from the applicable Registration Statement or Prospectus if the Company determines, based on the advice of counsel, that such information is necessary to effect the registration and such Holder continues thereafter to withhold such information. No person or entity may participate in any Underwritten Offering for equity securities of the Company pursuant to a Registration initiated by the Company hereunder unless such person or entity (i) agrees to sell such person's or entity's securities on the basis provided in any underwriting arrangements approved by the Company in the case of an Underwritten Offering initiated by the Company, and approved by the Demanding Holders in the case of an Underwritten Offering initiated by the Demanding Holders and (ii) completes and executes all customary questionnaires, powers of attorney, indemnities, lock-up agreements, underwriting agreements and other customary documents as may be reasonably required under the terms of such underwriting arrangements. Subject to the minimum thresholds set forth in Section 2.1.5 and 2.4, the exclusion of a Holder's Registrable Securities as a result of this Section 3.3 shall not affect the registration of the other Registrable Securities to be included in such Registration. The Company will use its commercially reasonable efforts to ensure that the underwriting agreement related to such Registration shall provide that any liability of a Holder to any Underwriter or other person pursuant to such underwriting agreement shall be limited to liability (i) arising from a breach of such Holder's representations and warranties thereto, (ii) will be several, and not joint and several, and (iii) will be limited to the net proceeds (after deducting discounts and commission, but not expenses) received by such Holder from the sale of such Holder's Registrable Securities pursuant to such underwriting agreement.

3.4 Suspension of Sales; Adverse Disclosure; Restrictions on Registration Rights.

3.4.1 Upon receipt of written notice from the Company that a Registration Statement or Prospectus contains or includes a Misstatement, each of the Holders shall forthwith discontinue disposition of Registrable Securities until he, she or it has received copies of a supplemented or amended Registration Statement or Prospectus correcting the Misstatement (it being understood that the Company hereby covenants to prepare and file such supplement or amendment as soon as practicable after the time of such notice), or until he, she or it is advised in writing by the Company that the use of the Registration Statement or Prospectus may be resumed.

3.4.2 Subject to <u>subsection 3.4.4</u>, if the filing, initial effectiveness or continued use of a Registration Statement in respect of any Registration or Underwritten Offering at any time would (a) require the Company to make an Adverse Disclosure, (b) require the inclusion in such Registration Statement of financial statements that are unavailable to the Company for reasons beyond the Company's control, or (c) in the good faith judgment of the majority of the Board, be detrimental to the Company and the majority of the Board concludes as a result that it is necessary or appropriate to defer such filing, initial effectiveness or continued use at such time, the Company may, upon giving prompt written notice of such action to the Holders, delay the filing or initial effectiveness of, or suspend use of, such Registration Statement for the shortest period of time determined in good faith by the Company to be necessary for such purpose. Notwithstanding the foregoing, the Company may delay or suspend continued use of a Registration Statement or Prospectus in respect of a Registration or Underwritten Offering in order to file and make effective a post-effective amendment to such Registration Statement in connection with the filing of the Company's Annual Report on Form 10-K, and such suspension shall not be subject to the provisions of <u>subsection 3.4.4</u>. In the event the Company exercises its rights under the preceding sentences in this <u>Section 3.4</u>, the Holders agree to suspend, immediately upon their receipt of the notices referred to in this <u>Section 3.4</u>, their use of the Registration Statement or Prospectus in connection with any resale or other disposition of Registrable Securities. The Company shall immediately notify the Holders of the expiration of any period during which it exercised its rights under this <u>Section 3.4</u>.

3.4.3 Subject to <u>subsection 3.4.4</u>, (a) during the period starting with the date thirty (30) days prior to the Company's good faith estimate of the date of the filing of, and ending on a date sixty (60) days after the effective date of, a Company-initiated Registration and <u>provided</u> that the Company continues to actively employ, in good faith, all reasonable efforts to maintain the effectiveness of the applicable Registration Statement, or (b) if, pursuant to <u>subsection 2.1.5</u>, Holders have requested an Underwritten Offering and the Company and Holders are unable to obtain the commitment of underwriters to firmly underwrite such offering, the Company may, upon giving prompt written notice of such action to the Holders, delay any other registered offering pursuant to <u>subsection 2.1.5</u> or <u>Section 2.4</u>.

3.4.4 The right to delay or suspend any filing, initial effectiveness or continued use of a Registration Statement pursuant to <u>subsection</u> <u>3.4.2</u> or a registered offering pursuant to <u>subsection 3.4.3</u> shall be exercised by the Company on not more than two (2) occasions and, in the aggregate, for not more than sixty (60) consecutive calendar days or more than one hundred-twenty (120) total calendar days in each case, during any twelve (12)-month period.

3.5 **Reporting Obligations**. As long as any Holder shall own Registrable Securities, the Company, at all times while it shall be a reporting company under the Exchange Act, covenants to file all reports required to be filed by the Company after the date hereof pursuant to Sections 13(a) or 15(d) of the Exchange Act. The Company further covenants that it shall take such further action as any Holder may reasonably request, all to the extent required from time to time to enable such Holder to resell or otherwise dispose of shares of Registrable Securities held by such Holder without registration under the Securities Act within the limitation of the exemptions provided by Rule 144 promulgated under the Securities Act (or any successor rule promulgated thereafter by the Commission), including providing any customary legal opinions, to the extent such rule is available. Upon the request of any Holder, the Company shall deliver to such Holder a written certification of a duly authorized officer as to whether it has complied with such requirements.

ARTICLE IV INDEMNIFICATION AND CONTRIBUTION

4.1 Indemnification.

4.1.1 The Company agrees to indemnify, to the extent permitted by law, each Holder of Registrable Securities, its officers, directors, employees, advisors, agents, representatives, members and each person who controls such Holder (within the meaning of the Securities Act) (collectively, the "Holder Indemnified Persons") against all losses, claims, damages, liabilities and expenses (including reasonable attorneys' fees and inclusive of all reasonable attorneys' fees arising out of the enforcement of each such persons' rights under this <u>Section 4.1</u>) as incurred arising out of or resulting from any Misstatement or alleged Misstatement, except insofar as the same are caused by or contained or included in any information furnished in writing to the Company by or on behalf of such Holder Indemnified Person specifically for use therein.

4.1.2 In connection with any Registration Statement in which a Holder of Registrable Securities is participating, such Holder shall furnish to the Company in writing such information and affidavits as the Company reasonably requests for use in connection with any such Registration Statement or Prospectus (the "Holder Information") and, to the extent permitted by law, shall indemnify and hold harmless the Company, its officers, directors, employees, advisors, agents, representatives and each person who controls the Company (within the meaning of the Securities Act) against all losses, claims, damages, liabilities and expenses (including reasonable attorneys' fees and inclusive of all reasonable attorneys' fees arising out of the enforcement of each such persons' rights under this <u>Section 4.1</u>) resulting from any Misstatement or alleged Misstatement, but only to the extent that the same are made in reliance on and in conformity with information relating to the Holder so furnished in writing to the Company by or on behalf of such Holder specifically for use therein; <u>provided</u>, <u>however</u>, that the obligation to indemnify shall be several, not joint and several, among such Holders of Registrable Securities, and in no event shall the liability of any selling Holder hereunder be greater in amount than the net proceeds received by such Holder from the sale of Registrable Securities pursuant to such Registration Statement giving rise to such indemnification obligation.

Any person or entity entitled to indemnification herein shall (i) give prompt written notice to the indemnifying party of any 413 claim with respect to which it seeks indemnification (provided that the failure to give prompt notice shall not impair any person's right to indemnification hereunder to the extent such failure has not materially prejudiced the indemnifying party) and (ii) unless in such indemnified party's reasonable judgment a conflict of interest between such indemnified and indemnifying parties may exist with respect to such claim or there may be reasonable defenses available to the indemnified party that are different from or additional to those available to the indemnifying party, permit such indemnifying party to assume the defense of such claim with counsel reasonably satisfactory to the indemnified party. If such defense is assumed, the indemnifying party shall not be subject to any liability for any settlement made by the indemnified party without its consent (but such consent shall not be unreasonably withheld). An indemnifying party who is not entitled to, or elects not to, assume the defense of a claim shall not be obligated to pay the fees and expenses of more than one counsel (in addition to any necessary local counsel) for all parties indemnified by such indemnifying party with respect to such claim, unless in the reasonable judgment of any indemnified party a conflict of interest may exist between such indemnified party and any other of such indemnified parties with respect to such claim. No indemnifying party shall, without the consent of the indemnified party, not to be unreasonably withheld or delayed, consent to the entry of any judgment or enter into any settlement which cannot be settled in all respects by the payment of money (and such money is so paid by the indemnifying party pursuant to the terms of such settlement) or which settlement includes a statement or admission of fault and culpability on the part of such indemnified party or which settlement does not include as an unconditional term thereof the giving by the claimant or plaintiff to such indemnified party of a release from all liability in respect to such claim or litigation.

4.1.4 The indemnification provided for under this Agreement shall remain in full force and effect regardless of any investigation made by or on behalf of the indemnified party or any officer, director, employee, advisor, agent, representative, member or controlling person or entity of such indemnified party and shall survive the transfer of securities. The Company and each Holder of Registrable Securities participating in an offering also agrees to make such provisions as are reasonably requested by any indemnified party for contribution to such party in the event the Company's or such Holder's indemnification is unavailable for any reason.

4.1.5 If the indemnification provided under Section 4.1 of this Agreement is held by a court of competent jurisdiction to be unavailable to an indemnified party in respect of any losses, claims, damages, liabilities and expenses referred to herein, then the indemnifying party, in lieu of indemnifying the indemnified party, shall to the extent permitted by law contribute to the amount paid or payable by the indemnified party as a result of such losses, claims, damages, liabilities and expenses in such proportion as is appropriate to reflect the relative fault of the indemnifying party and the indemnified party, as well as any other relevant equitable considerations. The relative fault of the indemnifying party and indemnified party shall be determined by a court of law by reference to, among other things, whether the Misstatement or alleged Misstatement relates to information supplied by such indemnifying party or such indemnified party and the indemnifying party's and indemnified party's relative intent, knowledge, access to information and opportunity to correct or prevent such action; provided, however, that the liability of any Holder under this subsection 4.1.5 shall be limited to the amount of the net proceeds received by such Holder in such offering giving rise to such liability. The amount paid or payable by a party as a result of the losses or other liabilities referred to above shall be deemed to include, subject to the limitations set forth in subsections 4.1.1, 4.1.2 and 4.1.3 of this Agreement, any reasonable legal or other fees, charges or expenses reasonably incurred by such party in connection with any investigation or proceeding. The parties hereto agree that it would not be just and equitable if contribution pursuant to this subsection 4.1.5 were determined by pro rata allocation or by any other method of allocation, which does not take account of the equitable considerations referred to in this subsection 4.1.5. No person guilty of fraudulent misrepresentation (within the meaning of Section 11(f) of the Securities Act) shall be entitled to contribution pursuant to this subsection 4.1.5 from any person who was not guilty of such fraudulent misrepresentation.

ARTICLE V MISCELLANEOUS

5.1 Notices. Any notice or communication under this Agreement must be in writing and given by (i) deposit in the United States mail, addressed to the party to be notified, postage prepaid and registered or certified with return receipt requested, (ii) delivery in person or by courier service or sent by overnight mail via a reputable overnight carrier, in each case providing evidence of delivery or (iii) transmission by facsimile or email. Each notice or communication that is mailed, delivered or transmitted in the manner described above shall be deemed sufficiently given, served, sent, and received, in the case of mailed notices, on the third (3rd) business day following the date on which it is mailed, in the case of notices delivered by courier service, hand delivery, electronic mail or facsimile at such time as it is delivered to the addressee (with the delivery receipt or the affidavit of messenger) or at such time as delivery is refused by the addressee upon presentation, and in the case of notices delivered by facsimile or email, at such time as it is successfully transmitted to the addressee. Any notice or communication under this Agreement must be addressed, if to the Company, to Verde Bioresins, Inc., 1431 Orangethorpe Ave, Fullerton, CA 92833, Attention: Brian Gordon, Email: brian@verdebioresins.com, or if to any Holder, to the address of such Holder as it appears in the applicable register for the Registrable Securities or such other address as may be designated in writing by such Holder (including on the signature pages hereto). Any party may change its address for notice at any time and from time to time by written notice to the other parties hereto, and such change of address shall become effective thirty (30) days after delivery of such notice as provided in this <u>Section 5.1</u>.

5.2 Assignment; No Third Party Beneficiaries.

5.2.1 This Agreement and the rights, duties and obligations of the Company and the Holders of Registrable Securities, as the case may be, hereunder may not be assigned or delegated by the Company or the Holders of such Registrable Securities, as the case may be, in whole or in part, except in connection with a Transfer of Registrable Securities by such Holder to a Permitted Transferee but only if such Permitted Transferee agrees to become bound by the restrictions set forth in this Agreement.

5.2.2 This Agreement and the provisions hereof shall be binding upon and shall inure to the benefit of each of the parties and its successors.

5.2.3 This Agreement shall not confer any rights or benefits on any persons that are not parties hereto, other than as expressly set forth in this Agreement and Section 5.2 of this Agreement.

5.2.4 No assignment by any party hereto of such party's rights, duties and obligations hereunder shall be binding upon or obligate the Company unless and until the Company shall have received (i) written notice of such assignment as provided in <u>Section 5.1</u> of this Agreement and (ii) the written agreement of the assignee, in a form reasonably satisfactory to the Company, to be bound by the terms and provisions of this Agreement (which may be accomplished by an addendum or certificate of joinder to this Agreement). Any transfer or assignment made other than as provided in this <u>Section 5.2</u> shall be null and void.

5.3 <u>Counterparts</u>. This Agreement may be executed in multiple counterparts (including facsimile or PDF counterparts), each of which shall be deemed an original, and all of which together shall constitute the same instrument, but only one of which need be produced.

5.4 <u>Adjustments.</u> If there are any changes in the Common Stock as a result of share split, share dividend, combination or reclassification, or through merger, consolidation, recapitalization or other similar event, appropriate adjustment shall be made in the provisions of this Agreement, as may be required, so that the rights, privileges, duties and obligations under this Agreement shall continue with respect to the Common Stock as so changed.

5.5 <u>Governing Law; Venue</u>. NOTWITHSTANDING THE PLACE WHERE THIS AGREEMENT MAY BE EXECUTED BY ANY OF THE PARTIES HERETO, THE PARTIES EXPRESSLY AGREE THAT THIS AGREEMENT SHALL BE GOVERNED BY AND CONSTRUED UNDER THE LAWS OF THE STATE OF NEW YORK AS APPLIED TO AGREEMENTS AMONG NEW YORK RESIDENTS ENTERED INTO AND TO BE PERFORMED ENTIRELY WITHIN NEW YORK, WITHOUT REGARD TO THE CONFLICT OF LAW PROVISIONS OF SUCH JURISDICTION.

5.6 **Trial by Jury**. EACH PARTY HERETO ACKNOWLEDGES AND AGREES THAT ANY CONTROVERSY WHICH MAY ARISE UNDER THIS AGREEMENT IS LIKELY TO INVOLVE COMPLICATED AND DIFFICULT ISSUES, AND, THEREFORE, EACH SUCH PARTY HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVES TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, ANY RIGHT SUCH PARTY MAY HAVE TO A TRIAL BY JURY IN RESPECT TO ANY ACTION DIRECTLY OR INDIRECTLY ARISING OUT OF, UNDER OR IN CONNECTION WITH OR RELATING TO THIS AGREEMENT OR THE TRANSACTIONS CONTEMPLATED BY THIS AGREEMENT.

5.7 <u>Amendments and Modifications</u>. Upon the written consent of (i) the Company and (ii) the Holders of at least a majority in interest of the Registrable Securities held by the Holders at the time in question, compliance with any of the provisions, covenants and conditions set forth in this Agreement may be waived, or any of such provisions, covenants or conditions may be amended or modified; <u>provided</u>, <u>however</u>, that notwithstanding the foregoing, any amendment hereto or waiver hereof that adversely affects any Holder, solely in his, her or its capacity as a holder of the shares of capital stock of the Company, in a manner that is materially different from the other Holders (in such capacity) shall require the consent of each such Holder so affected. No course of dealing between any Holder or the Company and any other party hereto or any failure or delay on the part of a Holder or the Company in exercising any rights or remedies under this Agreement shall operate as a waiver of any rights or remedies of any other rights or remedies under this Agreement by a party shall operate as a waiver or preclude the exercise of any other rights or remedies hereunder or thereunder by such party.

5.8 <u>Other Registration Rights.</u> The Company represents and warrants that, except for the Existing Registration Rights Agreement, no person has any right to require the Company to register any securities of the Company for sale or to include such securities of the Company in any Registration filed by the Company for the sale of securities for its own account or for the account of any other person. Further, the Company represents and warrants that this Agreement supersedes any other registration rights agreement or other agreement with similar terms and conditions entered into by and among the parties to this Agreement, and in the event of a conflict between any such agreement or agreements and this Agreement, the terms of this Agreement shall prevail.

5.9 <u>Term</u>. This Agreement shall terminate upon the earlier of (i) the tenth (10th) anniversary of the date of this Agreement and (ii) with respect to any Holder, the date as of which such Holder ceases to hold any Registrable Securities. The provisions of <u>Article IV</u> shall survive any termination.

5.10 **Holder Information**. Each Holder agrees, if requested in writing, to represent to the Company the total number of Registrable Securities held by such Holder in order for the Company to make determinations hereunder.

5.11 Additional Holders; Joinder. In addition to persons or entities who may become Holders pursuant to <u>Section 5.2</u> hereof, subject to the prior written consent of each Holder (in each case, so long as such Holder and its affiliates hold, in the aggregate, at least five percent (5%) of the outstanding shares of Common Stock of the Company), the Company may make any person or entity who acquires Common Stock or rights to acquire Common Stock after the date hereof a party to this Agreement (each such person or entity, an "<u>Additional Holder</u>") by obtaining an executed joinder to this Agreement from such Additional Holder in the form of <u>Exhibit A</u> attached hereto (a "Joinder"). Such Joinder shall specify the rights and obligations of the applicable Additional Holder under this Agreement. Upon the execution and delivery and subject to the terms of a Joinder by such Additional Holder, the Common Stock of the Company then owned, or underlying any rights then owned, by such Additional Holder (the "<u>Additional Holder</u>") shall be Registrable Securities to the extent provided herein and therein and such Additional Holder shall be a Holder under this Agreement with respect to such Additional Holder Common Stock.

5.12 <u>Severability</u>. It is the desire and intent of the parties that the provisions of this Agreement be enforced to the fullest extent permissible under the laws and public policies applied in each jurisdiction in which enforcement is sought. Accordingly, if any particular provision of this Agreement shall be adjudicated by a court of competent jurisdiction to be invalid, prohibited or unenforceable for any reason, such provision, as to such jurisdiction, shall be ineffective, without invalidating the remaining provisions of this Agreement or affecting the validity or enforceability of such provision in any other jurisdiction. Notwithstanding the foregoing, if such provision could be more narrowly drawn so as not to be invalid, prohibited or unenforceable in such jurisdiction, it shall, as to such jurisdiction, be so narrowly drawn, without invalidating the remaining provisions of this Agreement or affecting the validity or enforceable in such jurisdiction in such jurisdiction.

5.13 **Entire Agreement.** This Agreement constitutes the full and entire agreement and understanding between the parties with respect to the subject matter hereof and supersedes all prior agreements and understandings relating to such subject matter.

[Signature pages follow.]

IN WITNESS WHEREOF, the undersigned have caused this Agreement to be executed as of the date first written above.

COMPANY:

[Verde Bioresins, Corp.,] a Delaware corporation

By:

Name: Brian Gordon Title: President

[Signature Page to Registration Rights Agreement]

HOLDER:

[•]

By: Name: Title:

[Signature Page to Registration Rights Agreement]

EXHIBIT A REGISTRATION RIGHTS AGREEMENT JOINDER

The undersigned is executing and delivering this joinder (this "*Joinder*") pursuant to the Registration Rights Agreement, dated as of October [\bullet], 2024 (as the same may hereafter be amended, the "*Registration Rights Agreement*"), among [Verde Bioresins, Corp] (f/k/a NXU, Inc.), a Delaware corporation (the "*Company*"), and the other persons or entities named as parties therein. Capitalized terms used but not otherwise defined herein shall have the meanings provided in the Registration Rights Agreement.

By executing and delivering this Joinder to the Company, and upon acceptance hereof by the Company upon the execution of a counterpart hereof, the undersigned hereby agrees to become a party to, to be bound by, and to comply with the Registration Rights Agreement as a Holder of Registrable Securities in the same manner as if the undersigned were an original signatory to the Registration Rights Agreement, and the undersigned's shares of Common Stock shall be included as Registrable Securities under the Registration Rights Agreement to the extent provided therein; provided, however, that the undersigned and its permitted assigns (if any) shall not have any rights as Holders, and the undersigned's (and its transferees') shares of Common Stock shall not be included as Registrable Securities, for purposes of the Excluded Sections.

For purposes of this Joinder, "Excluded Sections" shall mean _____.

Accordingly, the undersigned has executed and delivered this Joinder as of the ______ day of _____.

Signature of Stockholder

Print Name of Stockholder

Its:

Address:

Agreed and Accepted as of _____, 20____

Verde Bioresins, Inc.

By: Name: Its:

COMPANY LOCK-UP AGREEMENT

THIS COMPANY LOCK-UP AGREEMENT (this "<u>Agreement</u>") is made and entered into as of October $[\bullet]$, 2024 between $[\bullet]$, a $[\bullet]$ (the "<u>Stockholder</u>") and [Verde Bioresins, Corp.], a Delaware corporation (the "<u>Company</u>"). The Stockholder and the Company are sometimes referred to herein individually as a "<u>Party</u>" and collectively as the "<u>Parties</u>". Capitalized terms used but not otherwise defined herein shall have the meanings ascribed to such terms in the Merger Agreement (as defined below).

WHEREAS, the Company, Verde Bioresins, Inc., a Delaware corporation ("<u>Verde</u>"), Nxu Merger Sub, Inc., a Delaware corporation and Nxu Merger Sub, LLC, a Delaware limited liability company, entered into that certain Agreement and Plan of Merger, dated as of October 23, 2024 (as it may be amended, restated or otherwise modified from time to time in accordance with its terms, the "<u>Merger Agreement</u>");

WHEREAS, the Merger Agreement contemplates that the Stockholder will receive a number of shares of Acquiror Common Stock at Closing (such shares, together with any securities paid as dividends or distributions with respect to such securities or into which such securities are exchanged or converted, the "Lock-Up Shares"); and

WHEREAS, the Merger Agreement contemplates that the Parties will enter into this Agreement, pursuant to which the Lock-Up Shares shall become subject to limitations on disposition as set forth herein.

NOW, THEREFORE, in consideration of the premises set forth above, which are incorporated in this Agreement as if fully set forth below, and intending to be legally bound hereby, the Parties hereby agree as follows:

1. For purposes of this Agreement:

- a. the term "Closing" has the meaning given to it in the Merger Agreement;
- b. the term "Initial Lock-Up Period" means the period beginning on the Closing Date and ending 180 days thereafter;
- c. the term "Lock-Up Period" means, jointly, the Initial Lock-Up Period and the Second Lock-Up Period;

d. the term "<u>Permitted Transferees</u>" means any Person to whom the Stockholder is permitted to transfer Lock-Up Shares prior to the expiration of the Lock-Up Period pursuant to <u>Section 2(a)</u>;

e. the term "Second Lock-Up Period" means the twenty-four (24) month period following the Initial Lock-Up Period; and

D-1

f. the term "<u>Transfer</u>" shall mean the (A) sale or assignment of, offer to sell, contract or agreement to sell, hypothecate, pledge, grant of any option to purchase or otherwise dispose of or agreement to dispose of, directly or indirectly, or establishment or increase of a put equivalent position or liquidation with respect to or decrease of a call equivalent position within the meaning of Section 16 of the Securities Exchange Act of 1934, as amended, and the rules and regulations of the Securities and Exchange Commission promulgated thereunder with respect to, any security, (B) entry into any swap or other arrangement that transfers to another, in whole or in part, any of the economic consequences of ownership of any security, whether any such transaction is to be settled by delivery of such securities, in cash or otherwise, or (C) public announcement of any intention to effect any transaction specified in clause (A) or (B).

2. Lock-Up Provisions.

a. Notwithstanding the provisions set forth in <u>Section 2(b)</u> and <u>Section 2(c)</u>, below, the Stockholder or its Permitted Transferees may Transfer the Lock-Up Shares:

i. during the Lock-Up Period (i) to its officers or directors, (ii) to any Affiliates of the Stockholder; (iii) in respect of (i) or (ii), in the case of an individual, by bona fide gift to a member of such individual's immediate family or to a trust, the beneficiary of which is the Stockholder, a member of such individual's immediate family, an Affiliate of such individual or to a charitable organization; (iv) in respect of (i), (ii) or (iii), in the case of an individual, by virtue of laws of descent and distribution upon death of such individual; or (v) in the case an individual, pursuant to a domestic order or a negotiated divorce settlement, in each case, subject to any such transferee signing a joinder hereto agreeing to be bound by all provisions hereof to the same extent as the Stockholder; and

ii. following such time that the Company completes any liquidation, merger, capital stock exchange, reorganization or other similar transaction that results in all stockholders of the Surviving Corporation having the right to exchange their shares for cash, securities or other property.

b. The Stockholder hereby agrees that, during the Initial Lock-Up Period, other than as permitted in Section 2(a) hereof, it shall not, and shall cause any of its Permitted Transferees not to, Transfer any Lock-Up Shares.

c. The Stockholder hereby agrees that, during the Second Lock-Up Period, other than as permitted in Section 2(a) hereof, it shall not, and shall cause any of its Permitted Transferees not to, Transfer any Lock-Up Shares representing more than five percent (5%) of the aggregate Lock-Up Shares in any calendar month.

d. If any Transfer is made or attempted contrary to the provisions of this Agreement, such Transfer shall be null and void ab initio, and the Company shall refuse to recognize any such transferee of the Lock-Up Shares as one of its equity holders for any purpose. In order to enforce this <u>Section 2</u>, the Company may impose stop-transfer instructions with respect to the Lock-Up Shares (and any Permitted Transferees and assigns thereof) until the end of the Lock-Up Period.

D-2

e. During the Lock-Up Period, each certificate (if any are issued) evidencing any Lock-Up Shares shall be stamped or otherwise imprinted with a legend in substantially the following form, in addition to any other applicable legends:

"THE SECURITIES REPRESENTED BY THIS CERTIFICATE ARE SUBJECT TO RESTRICTIONS ON TRANSFER SET FORTH IN A LOCK-UP AGREEMENT, DATED AS OF OCTOBER [•], 2024, BY AND AMONG THE ISSUER OF SUCH SECURITIES (THE "*ISSUER*") AND THE ISSUER'S SECURITY HOLDER NAMED THEREIN, AS AMENDED. A COPY OF SUCH LOCK-UP AGREEMENT WILL BE FURNISHED WITHOUT CHARGE BY THE ISSUER TO THE HOLDER HEREOF UPON WRITTEN REQUEST."

For the avoidance of any doubt, the Stockholder shall retain all of its rights as a stockholder of the Company with respect to the Lock-Up Shares during the Lock-Up Period, including the right to vote any Lock-Up Shares.

3. Miscellaneous.

a. <u>Effective Date</u>. <u>Section 1</u> of this Agreement shall become effective at the Effective Time.

b. <u>Termination of the Merger Agreement</u>. Notwithstanding anything to the contrary contained herein, in the event that the Merger Agreement is terminated in accordance with its terms prior to the Effective Time, this Agreement and all rights and obligations of the Parties hereunder shall automatically terminate and be of no further force or effect.

c. Notices. All notices, consents, waivers and other communications under this Agreement must be in writing and will be deemed to have been duly given (a) if personally delivered, on the date of delivery; (b) if delivered by express courier service of national standing for next day delivery (with charges prepaid), on the Business Day (as defined in the Merger Agreement) following the date of delivery to such courier service; (c) if delivered by electronic mail, on the date of transmission if on a Business Day before 5:00 p.m. local time of the business address of the recipient party (otherwise on the next succeeding Business Day); and (d) if deposited in the United States mail, first-class postage prepaid, on the date of delivery, in each case to the appropriate addresses or electronic mail addresses set forth below (or to such other addresses or electronic mail addresses as a party may designate by notice to the other parties in accordance with this Section 3(c)):

| [•] | With a copy to (which shall not constitute notice): Wilmer Cutler Pickering Hale and Dorr LLP 2600 El Camino Real, Suite 400 Palo Alto, CA 94306 Attn: Daniel Zimmermann E-mail: Daniel.Zimmermann@wilmerhale.com |
|----------------------------|--|
| If to the Stockholder, to: | With a copy to (which shall not constitute notice): |
| [•] | [•] |



d. <u>Incorporation by Reference</u>. Sections 1.02 (Construction) 11.03 (Assignment), 11.06 (Governing Law), 11.07 (Captions; Counterparts), 11.09 (Entire Agreement), 11.10 (Amendments), 11.11 (Severability), 11.12 (Jurisdiction; Waiver of Trial by Jury) and 11.13 (Enforcement) of the Merger Agreement are incorporated herein by reference and shall apply to this Agreement *mutatis mutandis*.

e. <u>Representations and Warranties</u>. Each of the parties hereto, by their respective execution and delivery of this Agreement, hereby represents and warrants to the other that (i) such party has the full right, capacity and authority to enter into, deliver and perform its respective obligations under this Agreement, (ii) this Agreement has been duly executed and delivered by such party and is a binding and enforceable obligation of such party and, enforceable against such party in accordance with the terms of this Agreement, and (iii) the execution, delivery and performance of such party's obligations under this Agreement will not conflict with or breach the terms of any other agreement, contract, commitment or understanding to which such party is a party or to which the assets or securities of such party are bound. The Stockholder has independently evaluated the merits of his/her/its decision to enter into and deliver this Agreement, and such Stockholder confirms that he/she/it has not relied on the advice of Company, Company's legal counsel, or any other person.

[Remainder of Page Intentionally Left Blank; Signature Pages Follow]

D-4

IN WITNESS WHEREOF, the parties have executed this Agreement as of the date first written above.

STOCKHOLDER:

| [●] | | |
|--------------------------|--|--|
| By: | | |
| Name: | | |
| Title: | | |
| THE COMPANY: | | |
| [VERDE BIORESINS, CORP.] | | |
| [•] | | |
| By: | | |
| Name: | | |
| Title: | | |

[Signature Page to Company Lock-Up Agreement]

EXHIBIT E Form of Amendment to Certificate of Incorporation

RESTATED CERTIFICATE OF INCORPORATION OF

NXU, INC.

(originally incorporated on March 10, 2023, under the name Nxu, Inc.)

ARTICLE I NAME

The name of the corporation is "Verde Bioresins, Corp." (hereinafter called the "Corporation").

ARTICLE II REGISTERED OFFICE AND AGENT

The address of the Corporation's registered office in the State of Delaware is c/o Corporation Service Company, 251 Little Falls Drive, Wilmington New Castle County, Delaware 19808. The name of its registered agent at such address is Corporation Service Company.

ARTICLE III PURPOSE

The purpose of the Corporation is to engage in any lawful act or activity for which corporations may be organized under the General Corporation Law of the State of Delaware or any applicable successor act thereto, as the same may be amended from time to time (the "DGCL").

ARTICLE IV

CAPITAL STOCK

The total number of shares of all classes of capital stock which the Corporation shall have authority to issue is [___] shares, consisting of (x) [___] authorized shares of common stock (the "Common Stock") and (y) [____] authorized shares of preferred stock, par value \$0.0001 per share ("Preferred Stock"). The number of authorized shares of Common Stock or Preferred Stock may be increased or decreased (but not below the number of shares thereof then outstanding) by the affirmative vote of the holders of capital stock representing a majority of the voting power of all the then-outstanding shares of capital stock of the Corporation entitled to vote thereon irrespective of the provisions of Section 242(b)(2) of the DGCL.

Upon the effectiveness under the DGCL of the Restated Certificate of Incorporation first inserting this paragraph, each [__] shares of the Corporation's Class A Common Stock, par value \$0.0001 per share, issued and outstanding immediately prior to such effectiveness shall be reclassified and combined into one validly issued, fully paid and non-assessable share of the Corporation's Common Stock, par value \$0.0001 per share, automatically and without any further action by the Corporation or the holder thereof. No fractional shares of Common Stock shall be issued as a result of this paragraph. Until surrendered, each certificate that, immediately prior to the effectiveness of this paragraph, represented shares of Class A Common Stock shall only represent the number of whole shares of Common Stock into which the shares of Class A Common Stock formerly represented by such certificates were reclassified and combined into as a result of this paragraph. Fractional interests will be disposed of in any manner determined by the Corporation and in accordance with the DGCL.

Each share of Class B Common Stock issued and outstanding immediately before the effectiveness of the Restated Certificate of Incorporation first inserting this sentence is hereby cancelled for no consideration.

The following is a statement of the designations and the powers, preferences, privileges and rights, and the qualifications, limitations or restrictions thereof in respect of each class of capital stock of the Corporation.

A. COMMON STOCK.

Unless otherwise indicated, references to "Sections" or "Subsections" in this Part A of this Article IV refer to sections and subsections of Part A of this Article IV.

- 1. *Equal Status; General.* Except as otherwise provided in this Certificate of Incorporation (as amended and/or restated from time to time, including pursuant to any Preferred Stock Designation (as defined below), this "Certificate of Incorporation") or required by applicable law, shares of Common Stock shall have the same rights, privileges and powers, rank equally, share ratably and be identical in all respects and as to all matters. The voting, dividend, liquidation and other rights, powers and preferences of the holders of Common Stock are subject to and qualified by the rights, powers and preferences of the Preferred Stock of any series as may be designated by the Board of Directors of the Corporation (the "Board") upon any issuance of the Preferred Stock of any series.
- 2. Voting. Except as otherwise required by applicable law, at all meetings of stockholders and on all matters submitted to a vote of stockholders of the Corporation generally, each holder of Common Stock, as such, shall have the right to one (1) vote per share of Common Stock held of record by such holder. Except as otherwise required by applicable law or provided in this Certificate of Incorporation, the holders of shares of Common Stock, as such, shall (a) at all times vote together as a single class on all matters (including the election of directors) submitted to a vote of the stockholders of the Corporation generally, (b) be entitled to notice of any stockholders' meeting in accordance with the Bylaws of the Corporation, as the same may be amended and/or restated from time to time (the "Bylaws"), and (c) be entitled to vote upon such matters and in such manner as may be provided by applicable law; provided, however, that, except as otherwise required by applicable law, holders of Common Stock, as such, shall not be entitled to vote on any amendment to this Certificate of Incorporation (including any Preferred Stock Designation) that relates solely to the terms of one or more outstanding series of Preferred Stock if the holders of such affected series of Preferred Stock are exclusively entitled, either separately or together with the holders of one or more other such series of Preferred Stock, to vote thereon pursuant to this Certificate of Incorporation or applicable law. There shall be no cumulative voting.

- 3. *Dividend and Distribution Rights.* Subject to the prior rights and preferences, if any, applicable to shares of Preferred Stock or any series thereof, the holders of shares of Common Stock shall be entitled to receive, ratably in proportion to the number of shares of Common Stock held by them, with respect to any dividends or distributions as may be declared and paid from time to time by the Board out of any assets of the Corporation legally available therefor.
- 4. *Distributions on Dissolution, etc.* Upon the dissolution, liquidation or winding up of the Corporation, whether voluntary or involuntary, holders of Common Stock, as such, will be entitled to receive all assets of the Corporation available for distribution to its stockholders, subject to any preferential liquidation or other rights or preferences of any then outstanding shares of Preferred Stock.
- 5. *Definitions*. For purposes of this Certificate of Incorporation:

"Person" means any individual, corporation, limited liability company, limited or general partnership, joint venture, association, joint-stock company, trust, unincorporated organization or other entity, whether domestic or foreign.

"Requisite Stockholder Consent" means the action at a meeting or by written consent (to the extent permitted under this Certificate of Incorporation) of the holders of at least two-thirds (2/3rds) of the voting power of the shares of capital stock of the Corporation that would then be entitled to vote in the election of directors at an annual meeting of stockholders.

B. PREFERRED STOCK

Preferred Stock may be issued from time to time in one or more series, each of such series to have such terms as stated or expressed herein and in the resolution or resolutions providing for the issue of such series adopted by the Board as hereinafter provided. Any shares of Preferred Stock that may be redeemed, purchased or acquired by the Corporation may be reissued except as otherwise provided by law.

Authority is hereby expressly granted to the Board from time to time to issue the Preferred Stock in one or more series, and in connection with the creation of any such series, by adopting a resolution or resolutions providing for the issuance of the shares thereof and by filing a certificate of designations relating thereto in accordance with the DGCL (a "Preferred Stock Designation"), to determine and fix the number of shares of such series and such voting powers, full or limited, or no voting powers, and such designations, preferences and relative participating, optional or other special rights, and qualifications, limitations or restrictions thereof, including without limitation thereof, dividend rights, conversion rights, redemption privileges and liquidation preferences, as shall be stated and expressed in such resolutions, all to the full extent now or hereafter permitted by the DGCL. Without limiting the generality of the foregoing, the resolutions providing for issuance of any series of Preferred Stock may provide that such series shall be superior or rank equally or be junior to any other series of Preferred Stock to the extent permitted by law.

ARTICLE V AMENDMENT OF THE CERTIFICATE OF INCORPORATION

The Corporation reserves the right to amend, alter, change, adopt or repeal any provision contained in this Certificate of Incorporation, in the manner now or hereafter prescribed by statute, and all rights conferred upon stockholders herein are granted subject to this reservation; provided, however, that, subject to the final sentence of this Article, notwithstanding any other provision of this Certificate of Incorporation (except the final sentence of this Article) or any provision of law that might otherwise permit a lesser vote or no vote, but in addition to any vote of the holders of shares of any class or series of capital stock of the Corporation required by law or by this Certificate of Incorporation, the affirmative vote of the holders of a majority of the voting power of the then-outstanding shares of capital stock of the Corporation entitled to vote thereon, voting together as a single class, shall be required to amend or repeal, or adopt any provision of this Certificate of Incorporation inconsistent with Articles IV, V, VI, VII, VIII, IX, X, XI and XII; provided further, so long as any shares of Common Stock remain outstanding, the Corporation shall not, without the prior affirmative vote of the holders of a majority of the outstanding shares of Common Stock, voting as a separate class, in addition to any other vote required by applicable law or this Certificate of Incorporation, directly or indirectly, whether by amendment, or through merger, recapitalization, consolidation or otherwise amend, alter, change, repeal or adopt any provision of this Certificate of Incorporation in a manner that is inconsistent with, or that otherwise alters or changes the powers, preferences, or special rights of the shares of Common Stock so as to affect them adversely. For the avoidance of doubt, (i) nothing in the immediately preceding provisos shall limit the rights of the Board as specified in Article IV, Section B or Article VI of this Certificate of Incorporation, and (ii) notwithstanding anything in this Article V to the contrary, any amendment to a provision that contemplates a specific approval requirement by the stockholders (or any class of capital stock of the Corporation) in this Certificate of Incorporation (including the definition of Requisite Stockholder Consent) shall require the greater of (x) the specific approval requirement by the stockholders (or any class of capital stock of the Corporation) contemplated in such provision, and (y) the approval requirements contemplated by this Article V. Notwithstanding the foregoing, the Corporation hereby expressly elects to be governed by Section 242(d) of the Delaware General Corporation Law.

ARTICLE VI AMENDMENT OF THE BYLAWS

In furtherance and not in limitation of the powers conferred upon it by the DGCL, and subject to the terms of any series of Preferred Stock, the Board shall have the power to adopt, amend, alter or repeal the Bylaws of the Corporation by the affirmative vote of a majority of the directors present at any regular or special meeting of the Board at which a quorum is present in any manner not inconsistent with the laws of the State of Delaware or this Certificate of Incorporation. The stockholders may not adopt, amend, alter or repeal the Bylaws of the Corporation, or adopt any provision inconsistent therewith, unless such action is approved, in addition to any other vote required by this Certificate of Incorporation, by the Requisite Stockholder Consent. Notwithstanding any other provisions of law, this Certificate of Incorporation or the Bylaws of the Corporation that may otherwise permit a lesser vote, but in addition to any vote of the holders of shares of any class or series of capital stock of the Corporation required by law or this Certificate of Incorporation, the affirmative vote of the Requisite Stockholder Consent shall be required to amend or repeal, or to adopt any provision inconsistent with, this Article.

ARTICLE VII INDEMNIFICATION AND ADVANCEMENT

- (A) The Corporation shall provide indemnification and advancement of expenses as follows:
 - 1. Actions, Suits and Proceedings Other than by or in the Right of the Corporation. The Corporation shall indemnify each person who was or is a party or threatened to be made a party to any threatened, pending or completed action, suit or proceeding, whether civil, criminal, administrative or investigative (other than an action by or in the right of the Corporation) by reason of the fact that he or she is or was, or has agreed to become, a director or officer of the Corporation or, while a director or officer of the Corporation, is or was serving, or has agreed to serve, at the request of the Corporation, as a director, officer, partner, employee or trustee of, or in a similar capacity with, another corporation, partnership, joint venture, trust or other enterprise (including any employee benefit plan) (all such persons being referred to hereafter as an "Indemnitee"), or by reason of any action alleged to have been taken or omitted by an Indemnitee in such capacity, against all expenses (including attorneys' fees), liabilities, losses, judgments, fines (including excise taxes and penalties arising under the Employee Retirement Income Security Act of 1974), and amounts paid in settlement actually and reasonably incurred by or on behalf of Indemnitee reasonably believed to be in, or not opposed to, the best interests of the Corporation, and, with respect to any criminal action or proceeding, had no reasonable cause to believe his or her conduct was unlawful. The termination of any action, suit or proceeding by judgment, order, settlement, conviction or upon a plea of nolo contendere or its equivalent, shall not, of itself, create a presumption that Indemnitee did not act in good faith and in a manner which Indemnitee reasonably believed to be in, or not opposed to, the best interests of the Corporation, and, with respect to any criminal action or proceeding, had reasonable cause to believe that his or her conduct was unlawful.
 - 2. Actions or Suits by or in the Right of the Corporation. The Corporation shall indemnify any Indemnitee who was or is a party to or threatened to be made a party to any threatened, pending or completed action or suit by or in the right of the Corporation to procure a judgment in its favor by reason of the fact that Indemnitee is or was, or has agreed to become, a director or officer of the Corporation or, while a director or officer of the Corporation, is or was serving, or has agreed to serve, at the request of the Corporation, as a director, officer, partner, employee or trustee of, or in a similar capacity with, another corporation, partnership, joint venture, trust or other enterprise (including any employee benefit plan), or by reason of any action alleged to have been taken or omitted by an Indemnitee in such capacity, against all expenses (including attorneys' fees) and, to the extent permitted by law, amounts paid in settlement actually and reasonably incurred by or on behalf of Indemnitee reasonably believed to be in, or not opposed to, the best interests of the Corporation, except that no indemnification shall be made under this Section 2 in respect of any claim, issue or matter as to which Indemnitee shall have been adjudged to be liable to the Corporation, unless, and only to the extent, that the Court of Chancery of Delaware or the court in which such action or suit was brought shall determine upon application that, despite the adjudication of such liability but in view of all the circumstances of the case, Indemnitee is fairly and reasonably entitled to indemnity for such expenses (including attorneys' fees) which the Court of Chancery of Delaware or such other court shall deem proper.

- 3. Indemnification for Expenses of Successful Party. Notwithstanding any other provisions of this Article, to the extent that an Indemnitee has been successful on the merits or otherwise in defense of any action, suit or proceeding referred to in Sections 1 and 2 of this Article, or in defense of any claim, issue or matter therein, or on appeal from any such action, suit or proceeding, Indemnitee shall be indemnified against all expenses (including attorneys' fees) actually and reasonably incurred by or on behalf of Indemnitee in connection therewith. Without limiting the foregoing, if any action, suit or proceeding is disposed of on the merits or otherwise (including a disposition without prejudice) without (i) the disposition being adverse to Indemnitee, (ii) an adjudication that Indemnitee was liable to the Corporation, (iii) a plea of guilty or nolo contendere by Indemnitee, (iv) an adjudication that Indemnitee did not act in good faith and in a manner he or she reasonably believed to be in or not opposed to the best interests of the Corporation, and (v) with respect to any criminal proceeding, an adjudication that Indemnitee shall be considered for the purposes hereof to have been wholly successful with respect thereto.
- 4. Notification and Defense of Claim. As a condition precedent to an Indemnitee's right to be indemnified under this Article, such Indemnitee must notify the Corporation in writing as soon as practicable of any action, suit, proceeding or investigation involving such Indemnitee for which indemnity will or could be sought. With respect to any action, suit, proceeding or investigation of which the Corporation is so notified, the Corporation will be entitled to participate therein at its own expense and/or to assume the defense thereof at its own expense, with legal counsel reasonably acceptable to Indemnitee. After notice from the Corporation to Indemnitee of its election so to assume such defense, the Corporation shall not be liable to Indemnitee for any legal or other expenses subsequently incurred by Indemnitee in connection with such action, suit, proceeding or investigation, other than as provided below in this Section 4. Indemnitee shall have the right to employ his or her own counsel in connection with such action, suit, proceeding or investigation, but the fees and expenses of such counsel incurred after notice from the Corporation of its assumption of the defense thereof shall be at the expense of Indemnitee unless (i) the employment of counsel by Indemnitee has been authorized by the Corporation, (ii) counsel to Indemnitee shall have reasonably concluded that there may be a conflict of interest or position on any significant issue between the Corporation and Indemnitee in the conduct of the defense of such action, suit, proceeding or investigation or (iii) the Corporation shall not in fact have employed counsel to assume the defense of such action, suit, proceeding or investigation, in each of which cases the fees and expenses of counsel for Indemnitee shall be at the expense of the Corporation, except as otherwise expressly provided by this Article. The Corporation shall not be entitled, without the consent of Indemnitee, to assume the defense of any claim brought by or in the right of the Corporation or as to which counsel for Indemnitee shall have reasonably made the conclusion provided for in clause (ii) above. The Corporation shall not be required to indemnify Indemnitee under this Article for any amounts paid in settlement of any action, suit, proceeding or investigation effected without its written consent. The Corporation shall not settle any action, suit, proceeding or investigation in any manner which would impose any penalty or limitation on Indemnitee without Indemnitee's written consent. Neither the Corporation nor Indemnitee will unreasonably withhold or delay its consent to any proposed settlement.

- 5. Advancement of Expenses. Subject to the provisions of Section 6 of this Article, in the event of any threatened or pending action, suit, proceeding or investigation of which the Corporation receives notice under this Article, any expenses (including attorneys' fees) incurred by or on behalf of Indemnitee in defending an action, suit, proceeding or investigation or any appeal therefrom shall be paid by the Corporation in advance of the final disposition of such matter; provided, however, that the payment of such expenses incurred by or on behalf of Indemnitee in advance of the final disposition of such matter shall be made only upon receipt of an undertaking by or on behalf of Indemnitee to repay all amounts so advanced in the event that it shall ultimately be determined by final judicial decision from which there is no further right to appeal that Indemnitee is not entitled to be indemnified by the Corporation as authorized in this Article; and provided further that no such advancement of expenses shall be made under this Article if it is determined (in the manner described in Section 6) that (i) Indemnitee did not act in good faith and in a manner he or she reasonably believed to be in, or not opposed to, the best interests of the Corporation, or (ii) with respect to any criminal action or proceeding, Indemnitee had reasonable cause to believe his or her conduct was unlawful. Such undertaking shall be accepted without reference to the financial ability of Indemnitee to make such repayment.
- 6. Procedure for Indemnification and Advancement of Expenses. In order to obtain indemnification or advancement of expenses pursuant to Section 1, 2, 3 or 5 of this Article, an Indemnitee shall submit to the Corporation a written request therefor. Any such advancement of expenses shall be made promptly, and in any event within 60 days after receipt by the Corporation of the written request of Indemnitee therefor, unless (i) the Corporation has assumed the defense pursuant to Section 4 of this Article (and none of the circumstances described in Section 4 of this Article that would nonetheless entitle the Indemnitee to indemnification for the fees and expenses of separate counsel have occurred) or (ii) the Corporation determines within such 60 day period that Indemnitee did not meet the applicable standard of conduct set forth in Section 1, 2 or 5 of this Article, as the case may be. Any such indemnification, unless ordered by a court, shall be made with respect to requests under Section 1 or 2 only as authorized in the specific case upon a determination by the Corporation that the indemnification of Indemnitee is proper because Indemnitee has met the applicable standard of conduct set forth in Section 1 or 2, as the case may be. Such determination shall be made in each instance (a) by a majority vote of the directors of the Corporation consisting of persons who are not at that time parties to the action, suit or proceeding in question ("disinterested directors"), whether or not a quorum, (b) by a committee of disinterested directors so direct, by independent legal counsel (who may, to the extent permitted by law, be regular legal counsel to the Corporation) in a written opinion, or (d) by the stockholders of the Corporation.

- 7. Remedies. Subject to Article XIII, the right to indemnification or advancement of expenses as granted by this Article shall be enforceable by Indemnitee in any court of competent jurisdiction. Neither the failure of the Corporation to have made a determination prior to the commencement of such action that indemnification or advancement of expenses is proper in the circumstances because Indemnitee has met the applicable standard of conduct, nor an actual determination by the Corporation pursuant to Section 6 of this Article that Indemnitee has not met such applicable standard of conduct, shall be a defense to the action or create a presumption that Indemnitee has not met the applicable standard of conduct. In any suit brought by Indemnitee to enforce a right to indemnification or advancement of expenses, or brought by the Corporation to recover an advancement of expenses pursuant to the terms of an undertaking, the Corporation shall have the burden of proving that Indemnitee is not entitled to be indemnified, or to such advancement of expenses, under this Article. Indemnitee's expenses (including attorneys' fees) reasonably incurred in connection with successfully establishing Indemnitee's right to indemnification or advancement of expenses, in whole or in part, in any such proceeding shall also be indemnified by the Corporation to the fullest extent permitted by applicable law. Notwithstanding the foregoing, in any suit brought by Indemnitee to enforce a right to indemnification or advancement of expenses hereunder it shall be a defense that the Indemnitee has not met any applicable standard for indemnification set forth in the DGCL or in this Article.
- 8. Limitations. Notwithstanding anything to the contrary in this Article, except as set forth in Section 7 of this Article, the Corporation shall not indemnify, or advance expenses to, an Indemnitee pursuant to this Article in connection with a proceeding (or part thereof) initiated by such Indemnitee unless the initiation thereof was approved by the Board of Directors. Notwithstanding anything to the contrary in this Article, the Corporation shall not indemnify or advance expenses to an Indemnitee to the extent such Indemnitee is reimbursed from the proceeds of insurance, and in the event the Corporation makes any indemnification or advancement payments to an Indemnitee is subsequently reimbursed from the proceeds of insurance, such Indemnitee shall promptly refund indemnification or advancement payments to the Corporation to the extent of such insurance reimbursement.

- 9. Subsequent Amendment. No amendment, termination or repeal of this Article or of the relevant provisions of the DGCL or any other applicable laws shall adversely affect or diminish in any way the rights of any Indemnitee to indemnification or advancement of expenses under the provisions hereof with respect to any action, suit, proceeding or investigation arising out of or relating to any actions, transactions or facts occurring prior to the final adoption of such amendment, termination or repeal.
- 10. Other Rights. The indemnification and advancement of expenses provided by this Article shall not be deemed exclusive of any other rights to which an Indemnitee seeking indemnification or advancement of expenses may be entitled under any law (common or statutory), agreement or vote of stockholders or disinterested directors or otherwise, both as to action in Indemnitee's official capacity and as to action in any other capacity while holding office for the Corporation, and shall continue as to an Indemnitee who has ceased to be a director or officer, and shall inure to the benefit of the estate, heirs, executors and administrators of Indemnitee. Nothing contained in this Article shall be deemed to prohibit, and the Corporation is specifically authorized to enter into, agreements with officers and directors providing indemnification and expense advancement rights and procedures different from those set forth in this Article. In addition, the Corporation may, to the extent authorized from time to time by its Board of Directors, grant indemnification and expense advancement rights to other employees or agents of the Corporation or other persons serving the Corporation and such rights may be equivalent to, or greater or less than, those set forth in this Article.
- 11. *Partial Indemnification.* If an Indemnitee is entitled under any provision of this Article to indemnification by the Corporation for some or a portion of the expenses (including attorneys' fees), liabilities, losses, judgments, fines (including excise taxes and penalties arising under the Employee Retirement Income Security Act of 1974) or amounts paid in settlement actually and reasonably incurred by or on behalf of Indemnitee in connection with any action, suit, proceeding or investigation and any appeal therefrom but not, however, for the total amount thereof, the Corporation shall nevertheless indemnify Indemnitee for the portion of such expenses (including attorneys' fees), liabilities, losses, judgments, fines (including excise taxes and penalties arising under the Employee Retirement Income Security Act of 1974) or amounts paid in settlement to which Indemnitee is entitled.
- (B) Insurance. The Corporation may purchase and maintain insurance, at its expense, to protect itself and any director, officer, employee or agent of the Corporation or another corporation, partnership, joint venture, trust or other enterprise (including any employee benefit plan) against any expense, liability or loss incurred by him or her in any such capacity, or arising out of his or her status as such, whether or not the Corporation would have the power to indemnify such person against such expense, liability or loss under the General Corporation Law of the State of Delaware. Such insurance may include insurance provided directly or indirectly by or through a captive insurance company to the extent permitted by Section 145(g) of the General Corporation Law of the State of Delaware.

- (C) Savings Clause. If this Article VII or any portion hereof shall be invalidated on any ground by any court of competent jurisdiction, then the Corporation shall nevertheless indemnify each Indemnitee as to any expenses (including attorneys' fees), liabilities, losses, judgments, fines (including excise taxes and penalties arising under the Employee Retirement Income Security Act of 1974) and amounts paid in settlement in connection with any action, suit, proceeding or investigation, whether civil, criminal or administrative, including an action by or in the right of the Corporation, to the fullest extent permitted by any applicable portion of this Article that shall not have been invalidated and to the fullest extent permitted by applicable law.
- (D) *Definitions.* Terms used in this Article VII and defined in Section 145(h) and Section 145(i) of the General Corporation Law of the State of Delaware shall have the respective meanings assigned to such terms in such Section 145(h) and Section 145(i).

ARTICLE VIII BOARD OF DIRECTORS

This Article VIII is inserted for the management of the business and for the conduct of the affairs of the Corporation, and for further definition, limitation and regulation of the powers of the Corporation and of its directors and stockholders.

- (A) General Powers. The business and affairs of the Corporation shall be managed by or under the direction of the Board, except as otherwise provided by law.
- (B) Quorum; Action at Meeting. The greater of (a) a majority of the directors at any time in office and (b) one third of the number of directors fixed pursuant to paragraph (C) of this Article shall constitute a quorum of the Board of Directors. If at any meeting of the Board of Directors there shall be less than such a quorum, a majority of the directors present may adjourn the meeting from time to time without further notice other than announcement at the meeting, until a quorum shall be present. Every act or decision done or made by a majority of the directors present at a meeting duly held at which a quorum is present shall be regarded as the act of the Board of Directors unless a greater number is required by law or by this Certificate of Incorporation.
- (C) Number of Directors; Election of Directors. Subject to the rights of holders of any series of Preferred Stock to elect directors, the number of directors of the Corporation shall be established from time to time by the Board of Directors; provided, further, that unless otherwise approved by the Requisite Stockholder Consent, the number of directors shall not exceed nine (9). Election of directors need not be by written ballot, except as and to the extent provided in the Bylaws of the Corporation.
- (D) Classes of Directors. Subject to the rights of holders of any series of Preferred Stock to elect directors, the Board of Directors shall be and is divided into three classes, designated Class I, Class II and Class III. Each class shall consist, as nearly as may be possible, of one third of the total number of directors constituting the entire Board of Directors. The Board of Directors is authorized to assign members of the Board of Directors already in office to Class I, Class III at the time such classification becomes effective.

- (E) Terms of Office. Subject to the rights of holders of any series of Preferred Stock to elect directors, each director shall serve for a term ending on the date of the third annual meeting of stockholders following the annual meeting of stockholders at which such director was elected; provided that each director initially assigned to Class I shall serve for a term expiring at the Corporation's first annual meeting of stockholders held after the time at which the initial classification of the Board of Directors becomes effective; each director initially assigned to Class II shall serve for a term expiring at the Corporation's second annual meeting of stockholders held after the time at which the initial classification of the Board of Directors becomes effective; and each director initially assigned to Class III shall serve for a term expiring at the Corporation's third annual meeting of stockholders held after the time at which the initial classification of the Board of Directors becomes effective; provided further, that the term of each director shall continue until the election and qualification of his or her successor and be subject to his or her earlier death, resignation or removal. Any director may resign at any time upon notice to the Corporation given in writing by any electronic transmission permitted in the Corporation's Bylaws or in accordance with applicable law.
- (F) Vacancies. Subject to the rights of holders of any series of Preferred Stock, any vacancies or newly-created directorships on the Board of Directors, however occurring, shall be filled only by vote of a majority of the directors then in office, although less than a quorum, or by a sole remaining director and shall not be filled by the stockholders. A director elected to fill a vacancy or to fill a position resulting from a newlycreated directorship shall hold office until the next election of the class for which such director shall have been chosen, subject to the election and qualification of a successor and to such director's earlier death, resignation or removal.
- (G) Removal. Subject to the rights of the holders of any series of Preferred Stock expressly set forth in a Preferred Stock Designation adopted in compliance with this Certificate of Incorporation, any director or the entire Board may be removed from office only for cause by the affirmative vote of the Requisite Stockholder Consent.
- (H) Committees. Pursuant to the Bylaws of the Corporation, the Board may establish one or more committees to which may be delegated any or all of the powers and duties of the Board to the full extent permitted by law.
- (I) Stockholder Nominations and Introduction of Business. Advance notice of stockholder nominations for election of directors and other business to be brought by stockholders before a meeting of stockholders shall be given in the manner provided by the Bylaws.

- (J) Preferred Stock Directors. During any period when the holders of any series of Preferred Stock have the right to elect additional directors as provided for or fixed pursuant to and in accordance with the provisions of Article IV hereof or any Preferred Stock Designation, then upon commencement and for the duration of the period during which such right continues: (i) the then otherwise total number of authorized directors of the Corporation shall automatically be increased by such specified number of directors, and the holders of such Preferred Stock shall be entitled to elect the additional directors so provided for or fixed pursuant to said provisions, and (ii) each such additional director shall serve until such director's successor shall have been duly elected and qualified, or until such director's right to hold such office terminates pursuant to said provisions, whichever occurs earlier, subject to his earlier death, disqualification, resignation or removal. Except as otherwise provided for or fixed pursuant to and in accordance with the provisions of Article IV hereof or any Preferred Stock Designation, whenever the holders of any series of Preferred Stock having such right to elect additional directors are divested of such right pursuant to the provisions of such stock, all such additional directors elected by the holders of such stock, or elected or appointed to fill any vacancies resulting from the death, resignation, disqualification or removal of such additional directors shall automatically cease to be qualified as directors, the terms of office of all such directors shall forthwith terminate and the total authorized number of directors of the Corporation shall be reduced accordingly.
- (K) Amendments to Article. Notwithstanding any other provisions of law, this Certificate of Incorporation or the Bylaws of the Corporation that may otherwise permit a lesser vote, but in addition to any vote of the holders of shares of any class or series of capital stock of the Corporation required by law or this Certificate of Incorporation, the affirmative vote of the Requisite Stockholder Consent shall be required to amend or repeal, or to adopt any provision inconsistent with, this Article.

ARTICLE IX ELECTION OF DIRECTORS

Unless and except to the extent that the Bylaws shall so require, the election of directors of the Corporation need not be by written ballot. The vote required for election of a director by the stockholders at a meeting of stockholders shall, except in a contested election, be the affirmative vote of a majority of the votes cast in favor or against the election of a nominee at a meeting of stockholders. In a contested election, (i) the directors shall be elected by a plurality of the votes cast at a meeting of stockholders of stock entitled to vote in such election, and (ii) stockholders shall not be permitted to vote against a nominee. An election shall be considered contested if, as the tenth (10th) preceding the date the Corporation first mails its notice of meeting for such meeting to the stockholders of the Corporation, there are more nominees for election than directorships on the Board to be filled by election at the meeting.

ARTICLE X LIMITATION OF DIRECTOR AND OFFICER LIABILITY

To the fullest extent permitted by the DGCL as the same exists or as may hereafter be amended, each director and officer of the Corporation shall not be personally liable to the Corporation or its stockholders for monetary damages for breach of fiduciary duty as a director or officer; provided, however, that nothing contained in this Article shall eliminate or limit the liability: (i) of a director or officer for any breach of the director's or officer's duty of loyalty to the Corporation or its stockholders, (ii) a director or officer for acts or omissions not in good faith or which involve intentional misconduct or a knowing violation of law, (iii) a director pursuant to the provisions of Section 174 of the DGCL, (iv) a director or officer for any transaction from which the director or officer derived an improper personal benefit, or (v) an officer in any action by or in the right of the Corporation. No repeal or modification of this Article shall apply to or have any adverse effect on any right or protection of, or any limitation of the liability of, a director or officer of the Corporation existing at the time of such repeal or modification with respect to acts or omissions occurring prior to such repeal or modification. For purposes of this paragraph, "officer" has the meaning determined in accordance with Section 102(b)(7) of the DGCL.

ARTICLE XI CONSENT OF STOCKHOLDERS IN LIEU OF MEETING

Subject to the terms of any series of Preferred Stock, any action required or permitted to be taken by the stockholders of the Corporation must be effected at an annual or special meeting of the stockholders and may not be effected by written consent in lieu of a meeting. Notwithstanding any other provisions of law, this Certificate of Incorporation or the Bylaws of the Corporation that may otherwise permit a lesser vote, but in addition to any vote of the holders of shares of any class or series of capital stock of the Corporation required by law or this Certificate of Incorporation, the affirmative vote of the Requisite Stockholder Consent shall be required to amend or repeal, or to adopt any provision inconsistent with, this Article.

ARTICLE XII SPECIAL MEETING OF STOCKHOLDERS

Special meetings of stockholders for any purpose or purposes may be called at any time by the Board, the Chairperson of the Board or the Chief Executive Officer of the Corporation, and may not be called by another other Person or Persons; provided that special meetings of stockholders for any purpose or purposes may also be called by or at the request of stockholders of the Corporation collectively holding of record shares of capital stock of the Corporation with voting power sufficient, if such shares and only such shares were voted at such special meeting, to provide the Requisite Stockholder Consent. Business transacted at any special meeting of stockholders shall be limited to matters relating to the purpose or purposes stated in the notice of meeting. Notwithstanding any other provisions of law, this Certificate of Incorporation or the Bylaws of the Corporation that may otherwise permit a lesser vote, but in addition to any vote of the holders of shares of any class or series of capital stock of the Corporation required by law or this Certificate of Incorporation, the affirmative vote of the Requisite Stockholder Consent shall be required to amend or repeal, or to adopt any provision inconsistent with, this Article.

ARTICLE XIII FORUM SELECTION

Unless the Corporation consents in writing to the selection of an alternative forum, (i) the Court of Chancery (the "Chancery Court") of the State of Delaware (or, in the event that the Chancery Court does not have jurisdiction, the federal district court for the District of Delaware or other state courts of the State of Delaware) shall, to the fullest extent permitted by law, be the sole and exclusive forum for (1) any derivative action or proceeding brought on behalf of the Corporation, (2) any action asserting a claim of breach of a fiduciary duty owed by, or any other wrongdoing by, any current or former director, officer, other employee or stockholder of the Corporation, (3) any action asserting a claim against the Corporation arising pursuant to any provision of the DGCL, this Certificate of Incorporation or the Bylaws or as to which the DGCL confers jurisdiction on the Court of Chancery. (4) any action to interpret, apply, enforce or determine the validity of any provisions of this Certificate of Incorporation or the Bylaws, or (5) any other action asserting a claim governed by the internal affairs doctrine and (ii) notwithstanding anything to the contrary herein, but subject to the foregoing provisions of this Article XIV, the federal district courts of the United States shall be the exclusive forum for the resolution of any action, suit or proceeding asserting a cause of action arising under the Securities Act of 1933, as amended. If any action the subject matter of which is within the scope of the preceding sentence is filed in a court other than the applicable courts specified in the immediately preceding sentence (a "Foreign Action") in the name of any stockholder, such stockholder shall, to the fullest extent permitted by applicable law, be deemed to have consented to (a) the personal jurisdiction of the state and federal courts located within the State of Delaware in connection with any action brought in any such court to enforce the preceding sentence and (b) having service of process made upon such stockholder in any such action by service upon such stockholder's counsel in the Foreign Action as agent for such stockholder. This provision will not apply to claims arising under the Securities Exchange Act of 1934, as amended, or other federal securities laws for which there is exclusive federal jurisdiction. Any Person or entity purchasing or otherwise acquiring or holding any interest in shares of capital stock of the Corporation shall be deemed to have notice of and consented to the provisions of this Article.

ARTICLE XIV

MISCELLANEOUS

(A) Severability. If any provision or provisions of this Certificate of Incorporation shall be held to be invalid, illegal or unenforceable as applied to any circumstance for any reason whatsoever: (i) the validity, legality and enforceability of such provisions in any other circumstance and of the remaining provisions of this Certificate of Incorporation (including, without limitation, each portion of any paragraph of this Certificate of Incorporation containing any such provision held to be invalid, illegal or unenforceable that is not itself held to be invalid, illegal or unenforceable) shall not in any way be affected or impaired thereby and (ii) to the fullest extent possible and without limiting any other provisions of this Certificate of Incorporation (or any other provision of the Bylaws or any agreement entered into by the Corporation, the provisions of this Certificate of Incorporation (including, without limitation, each such portion of any paragraph of this Certificate of Incorporation (including, without limitation, each such portion of any paragraph of this Certificate of Incorporation containing any such provision of the Bylaws or any agreement entered into by the Corporation containing any such provision held to be invalid, illegal or unenforceable) shall be construed so as to permit the Corporation to protect its directors, officers, employees and agents from personal liability in respect of their good faith service to, or for the benefit of, the Corporation to the fullest extent permitted by law.

- (B) Notice of Provisions. To the fullest extent permitted by law, each and every Person purchasing or otherwise acquiring any interest (of any nature whatsoever) in any shares of the capital stock of the Corporation shall be deemed, by reason of and from and after the time of such purchase or other acquisition, to have notice of and to have consented to all of the provisions of (a) this Certificate of Incorporation, (b) the Bylaws and (c) any amendment to this Certificate of Incorporation or the Bylaws enacted or adopted in accordance with this Certificate of Incorporation, the Bylaws and applicable law.
- (C) *Opt-Out of Section 203 of the DGCL*. The Corporation shall not be governed by Section 203 of the DGCL.

[Remainder of Page Intentionally Left Blank]

In witness whereof, this Restated Certificate of Incorporation, which restates and integrates and further amends the provisions of the Certificate of Incorporation of this Corporation, has been duly adopted in accordance with Sections 242 and 245 of the Delaware General Corporation Law and has been executed by its duly authorized officer on the date set forth below.

| Nxu, | Inc. |
|------|------|
|------|------|

| BW | • |
|----|---|
| | • |

[Name] [Title]

VERDE BIORESINS, INC.

STATEMENT UNDER TREASURY REGULATIONS SECTIONS 1.897-2(h) AND 1.1445-2(c)(3) RELATING TO STATUS OF INTERESTS AS "UNITED STATES REAL PROPERTY INTERESTS"

Reference is made to the Agreement and Plan of Merger, dated as of October $[\bullet]$, 2024 (the "<u>Agreement</u>"), by and among Nxu, Inc., a Delaware corporation ("<u>Acquiror</u>"), Nxu Merger Sub, Inc., a Delaware corporation ("<u>Merger Sub I</u>"), Nxu Merger Sub, LLC, a Delaware limited liability company ("<u>Merger Sub II</u>", and together with Merger Sub I, the "<u>Merger Subs</u>"), and Verde Bioresins, Inc., a Delaware corporation (the "<u>Company</u>").

Section 1445 of the Internal Revenue Code of 1986, as amended (the "<u>Code</u>") provides that a transferee of a United States real property interest (as defined in Section 897(c) of the Code) must withhold tax if the transferor is a foreign person. This certificate is being provided pursuant to Treasury Regulations Sections 1.897-2(h) and 1.1445-2(c)(3) and Section 8.03(d) of the Agreement. To inform Acquiror and Merger Subs that withholding of tax is not required under Section 1445 of the Code in connection with the transactions contemplated under the Agreement, the undersigned hereby certifies the following on behalf of the Company:

- 1. The Company is not, and has not been, a "United States real property holding corporation" within the meaning of Section 897(c)(2) of the Code and the applicable Treasury Regulations during the applicable period specified in Section 897(c)(1)(A)(ii) of the Code.
- 2. Interests in the Company are not "United States real property interests" within the meaning of Section 897(c)(1) of the Code.
- 3. The Company's registered office is at 1431 E. Orangethorpe Ave. Fullerton, CA 92831.
- 4. The Company's U.S. employer identification number is [•].
- 5. The Company understands that this certification may be disclosed to the U.S. Internal Revenue Service by Acquiror and Merger Subs and that any false statement contained herein could be punished by fine, imprisonment or both.

Under penalties of perjury, the undersigned, a responsible officer of the Company, declares (i) that the undersigned has examined this certification and, to the best of the undersigned's knowledge and belief, it is true, correct, and complete and (ii) that the undersigned has the authority to sign this document on behalf of the Company.

VERDE BIORESINS, INC.

By:

| -) • | |
|--------|--------------|
| Name: | Brian Gordon |
| Title: | President |
| Date: | [•], 2024 |

F-1

[Date]

CERTIFIED MAIL & RETURN RECEIPT REQUESTED Internal Revenue Service Ogden Service Midpoint P.O. Box 409101 Ogden, UT 84409

Re: NOTICE OF STATEMENT UNDER TREASURY REGULATIONS SECTIONS 1.897-2(h) AND 1.1445-2(c)(3)

Dear Sir or Madam:

At the request of Nxu, Inc., a Delaware corporation ("<u>Acquiror</u>"), Nxu Merger Sub, Inc., a Delaware corporation ("<u>Merger Sub I</u>"), Nxu Merger Sub, LLC, a Delaware limited liability company ("<u>Merger Sub II</u>", and together with Merger Sub I, the "<u>Merger Subs</u>"), in connection with the merger contemplated under the Agreement and Plan of Merger dated as of October [\bullet], 2024 (the "<u>Agreement</u>") by and among Acquiror, Merger Subs, and Verde Bioresins, Inc. a Delaware corporation (the "<u>Company</u>"), the Company provided the attached statement (the "<u>Statement</u>") to Acquiror and Merger Subs on October [\bullet], 2024. The undersigned, being a duly authorized officer of the Company, hereby affirms the following:

- (i) This notice is being provided to the Internal Revenue Service pursuant to Treasury Regulations Sections 1.897-2(h) and 1.1445-2(c)(3);
- (ii) The following information relates to the Company providing this notice:

Name: Verde Bioresins, Inc. Address: 1431 E. Orangethorpe Ave. Fullerton, CA 92831 U.S. Employer Identification Number: [•]

(iii) The attached statement was not requested by a foreign interest holder. The attached statement was voluntarily provided by the Company in response to a request from Acquiror and Merger Subs, in accordance with Treasury Regulation Section 1.1445-2(c)(3)(i). The following information relates to Acquiror and Merger Subs respectively:

Name: [•] Address: [•] U.S. Employer Identification Number: [•]

Name: Verde Merger Sub I Address: [•] U.S. Employer Identification Number: [•]

F-2

Name: Verde Merger Sub II Address: [•] U.S. Employer Identification Number: [•]

(iv) The Company has provided to Acquiror and Merger Subs the attached Statement, certifying that: (A) the Company is not and has not been a "United States real property holding corporation" within the meaning of Section 897(c)(2) of the Internal Revenue Code of 1986, as amended (the "Code") and the applicable Treasury Regulations during the applicable period specified in Section 897(c)(1)(A)(ii) of the Code and (B) interests in the Company are not "United States real property interests" within the meaning of Section 897(c)(1)(A)(ii) of the Code and (B) interests in the

Under penalties of perjury, the undersigned, a responsible officer of the Company declares (i) that the undersigned has examined this affidavit and the attached Statement and, to the best of the undersigned's knowledge and belief, this affidavit and the attached Statement are true, correct and complete and (ii) that the undersigned has the authority to execute this document on behalf of the Company.

VERDE BIORESINS, INC.

By: Name: Brian Gordon Title: President Date:

F-3

Exhibit G Post-Closing Directors and Officers

- 1. Brian Gordon
- 2. Jessica Billingsley
- 3. Jefferey Brown
- 4. Michael Sherman
- 5. Mark Adler
- 6. Cuong Do
- 7. One director to be identified by the Company post-signing.

G-1

ACQUIROR SUPPORT AGREEMENT

This Acquiror Support Agreement (this "<u>Support Agreement</u>") is dated as of October 23, 2024, by and among, Mark Hanchett ("<u>Hanchett</u>"), Annie Pratt ("<u>Pratt</u>", and together with Hanchett, the "<u>Stockholders</u>"), NXU, Inc., a Delaware corporation ("<u>NXU</u>"), and Verde Bioresins, Inc., a Delaware corporation (the "<u>Company</u>"). Capitalized terms used but not defined herein shall have the meanings ascribed to such terms in the Merger Agreement.

RECITALS

WHEREAS, as of the date hereof, each Stockholder is the holder of record and/or the "beneficial owner" (within the meaning of Rule 13d-3 under the Exchange Act) of that number of (a) shares of Class A Common Stock, par value \$0.0001, of NXU (the "<u>Class A Common Stock</u>"), (b) Class B Common Stock, par value \$0.0001, of NXU (the "<u>Class B Common Stock</u>", and together with the Class A Common Stock, the "<u>Common Stock</u>") and (c) options set forth on the signature page of such Stockholder (all such securities held by a Stockholder, the "<u>Stockholder Securities</u>");

WHEREAS, contemporaneously with the execution and delivery of this Support Agreement, NXU, Nxu Merger Sub, Inc., a Delaware corporation and a direct wholly owned subsidiary of NXU ("<u>Merger Sub I</u>"), Nxu Merger Sub, LLC, a Delaware limited liability company ("<u>Merger Sub II</u>", and together with Merger Sub I, the "<u>Merger Subs</u>"), and the Company are entering into an Agreement and Plan of Merger (as amended or modified from time to time, the "<u>Merger Agreement</u>"), dated as of the date hereof, pursuant to which, among other transactions, Merger Sub I is to merge with and into the Company, with the Company continuing on as the surviving company and a wholly owned subsidiary of NXU and then merging with and into Merger Sub II, with Merger Sub II becoming the surviving entity, on the terms and conditions set forth therein (the "<u>Merger</u>"); and

WHEREAS, as an inducement to NXU and the Company to enter into the Merger Agreement and to consummate the transactions contemplated therein, the parties hereto desire to agree to certain matters as set forth herein.

AGREEMENT

NOW, THEREFORE, in consideration of the foregoing and the mutual agreements contained herein, and intending to be legally bound hereby, the parties hereto hereby agree as follows:

Article I SUPPORT AGREEMENT; COVENANTS; RELEASE

I.1 <u>Binding Effect of Merger Agreement</u>. Each Stockholder hereby acknowledges that he or she has read the Merger Agreement and this Support Agreement and has had the opportunity to consult with his or her tax and legal advisors. During the period commencing on the date hereof and ending at the Expiration Time (as defined below), each Stockholder shall be bound by and comply with Sections 7.08 (*Exclusivity*) and 8.04 (*Confidentiality; Publicity*) of the Merger Agreement (and any relevant definitions contained in any such Sections) as if (a) such Stockholder was an original signatory to the Merger Agreement with respect to such provisions, and (b) each reference to the "Acquiror" contained in Sections 7.08 and 8.04 also referred to such Stockholder.

No Transfer. During the period commencing on the date hereof and ending on the earliest to occur of (a) the Effective Time (and, in such 12 case, for the avoidance of doubt, not following the Effective Time) and (b) such date and time as the Merger Agreement shall be terminated in accordance with its terms (the earlier of clauses (a) and (b), the "Expiration Time"), neither Stockholder shall Transfer any Stockholder Securities held by such Stockholder as of the date of this Support Agreement; provided, however, either Stockholder may Transfer shares of Stockholder Securities in connection with bona fide estate planning purposes, either during such Stockholder's lifetime or on death by will or intestacy to such Stockholder's spouse, child (natural or adopted), or any other direct lineal descendant, or any custodian or trustee of any trust, partnership, limited liability company or other corporate entity for the benefit of, or the ownership interests of which are owned wholly by such Stockholder or any such family members, provided that such permitted transferees execute and deliver to NXU and the Company a joinder to this Support Agreement in substantially the form attached hereto as Annex A; provided, further, that any Transfer permitted under this Section 1.2 shall not relieve such Stockholder of his or her obligations under this Support Agreement. Any Transfer in violation of this Section 1.2 shall be null and void. "Transfer" shall mean the (i) sale or assignment of, offer to sell, contract or agreement to sell, hypothecate, pledge, grant of any option to purchase or otherwise dispose of or agreement to dispose of, directly or indirectly, or establishment or increase of a put equivalent position or liquidation with respect to or decrease of a call equivalent position within the meaning of Section 16 of the Securities Exchange Act of 1934, as amended, and the rules and regulations of the Securities and Exchange Commission promulgated thereunder with respect to, any security, (ii) entry into any swap or other arrangement that transfers to another, in whole or in part, any of the economic consequences of ownership of any security, whether any such transaction is to be settled by delivery of such securities, in cash or otherwise, or (iii) public announcement of any intention to effect any transaction specified in clause (i) or (ii).

I.3 <u>New Shares</u>. In the event that after the date hereof but prior to the Expiration Time (a) either Stockholder acquires any shares of Common Stock or other equity securities of NXU or any rights to acquire such shares of Common Stock or other equity securities of NXU pursuant to any stock dividend, stock split, recapitalization, reclassification, combination or exchange of shares of Common Stock, on or affecting the shares of Common Stock or other equity securities of NXU, or (b) either Stockholder purchases or otherwise acquires beneficial ownership of any shares of Common Stock or other equity securities of NXU, or (c) either Stockholder acquires the right to vote or share in the voting of any shares of Common Stock or other equity securities of NXU, collectively, the "<u>New Securities</u>"), then such New Securities acquired or purchased by either such Stockholder shall be subject to the terms of this Support Agreement to the same extent as if they constituted the shares of Common Stock owned by such Stockholder as of the date hereof.

I.4 Voting and Support Agreement.

a. Each Stockholder irrevocably and unconditionally agrees that, prior to the Expiration Time, at any meeting of the stockholders of NXU, however called, or at any adjournment thereof, or in any other circumstance in which the vote, consent or other approval of the stockholders of NXU is sought, each Stockholder shall (i) appear at each such meeting or otherwise cause all of his or her Common Stock to be counted as present thereat for purposes of calculating a quorum and (ii) vote (or cause to be voted), or execute and deliver a written consent (or cause a written consent to be executed and delivered) covering, all of his or her Common Stock:

i. in favor of the approval and adoption of the Proposals;

ii. against any Competing Proposal;

iii. against any change in the business, management or Board of Directors of NXU (other than in connection with the Transaction Proposals) that would or would reasonably be expected to adversely affect the ability of NXU to consummate the transactions contemplated by the Merger Agreement, including the Merger; and

iv. against any proposal, action or agreement that would (A) impede, frustrate, prevent or nullify any provision of this Support Agreement, the Merger Agreement or the Transactions, including the Merger, (B) result in a breach in any respect of any covenant, representation, warranty or any other obligation or agreement of NXU or Merger Sub under the Merger Agreement, (C) result in any of the conditions set forth in Article IX of the Merger Agreement not being fulfilled or (D) change in any manner the dividend policy or capitalization of, including the voting rights of any class of capital stock of, NXU.

Each Stockholder hereby agrees that he or she shall not commit or agree to take any action inconsistent with the foregoing.

b. During the period commencing on the date hereof and ending at the Expiration Time, such Stockholder shall not modify or amend any agreement set forth on <u>Schedule I</u>.

I.5 <u>Further Assurances</u>. Each Stockholder shall take, or cause to be taken, all such further actions and do, or cause to be done, all things reasonably necessary (including under applicable Law) to consummate the Merger and the other transactions contemplated by the Merger Agreement and this Support Agreement, in each case, on the terms and subject to the conditions set forth therein and herein, as applicable.

I.6 <u>No Inconsistent Agreement</u>. Each Stockholder hereby represents and covenants that such Stockholder has not entered into, and shall not enter into, any agreement that would restrict, limit or interfere with the performance of his or her obligations hereunder.

I.7 <u>Waiver and Release of Claims</u>. Each Stockholder hereby covenants and agrees as follows:

(a) Subject to and conditioned upon the Effective Time, effective as of the Effective Time (and subject to the limitations set forth in paragraph (c) below), such Stockholder, on behalf of himself, herself and his or her Affiliates and his or her and their respective successors, assigns, representatives, administrators, executors and agents, and any other person or entity claiming by, through, or under any of the foregoing (each a "<u>Releasing Party</u>" and, collectively, the "<u>Releasing Parties</u>," provided, for the avoidance of doubt, that NXU, shall not be deemed a Releasing Party hereunder), does hereby unconditionally and irrevocably release, waive and forever discharge the NXU, the Company, and each of their past and present directors, officers, employees, agents, predecessors, successors, assigns, and Subsidiaries, from any and all past or present claims, demands, damages, judgments, causes of action and liabilities of any nature whatsoever, whether or not known, suspected or claimed, arising directly or indirectly from any act, omission, event or transaction occurring (or any circumstances existing) at or prior to the Effective Time (each a "<u>Claim</u>" and, collectively, the "<u>Claims</u>").

(b) Such Stockholder acknowledges that he or she may hereafter discover facts in addition to or different from those which he or she now knows or believes to be true with respect to the subject matter of this Agreement, and that he or she may hereafter come to have a different understanding of the law that may apply to potential claims which it is releasing hereunder, but he or she affirms that, except as is otherwise specifically provided herein, it is his or her intention to fully, finally and forever settle and release any and all Claims. In furtherance of this intention, such Stockholder acknowledges that the releases contained herein shall be and remain in effect as full and complete general releases notwithstanding the discovery or existence of any such additional facts or different understandings of law. Such Stockholder knowingly and voluntarily waives and releases any and all rights and benefits he or she may now have, or in the future may have, under Section 1542 of the California Civil Code (or any analogous law of any other state), which reads as follows:

"A GENERAL RELEASE DOES NOT EXTEND TO CLAIMS THAT THE CREDITOR OR RELEASING PARTY DOES NOT KNOW OR SUSPECT TO EXIST IN HIS OR HER FAVOR AT THE TIME OF EXECUTING THE RELEASE AND THAT, IF KNOWN BY HIM OR HER, WOULD HAVE MATERIALLY AFFECTED HIS OR HER SETTLEMENT WITH THE DEBTOR OR RELEASED PARTY."

Such Stockholder understands that Section 1542, or a comparable statute, rule, regulation or order of another jurisdiction, gives such Stockholder the right not to release existing Claims of which such Stockholder is not aware, unless such Stockholder voluntarily chooses to waive this right. Having been so apprised, such Stockholder nevertheless hereby voluntarily elects to and does waive any rights he or she may have that are described in Section 1542, or such other comparable statute, rule, regulation or order, and elects to assume all risks for Claims that exist, existed or may hereafter exist in his or her favor, known or unknown, suspected or unsuspected, arising out of or related to claims or other matters purported to be released pursuant to this Section 1.7, in each case, effective as of the Effective Time. Such Stockholder acknowledges and agrees that the foregoing waiver is an essential and material term of the release provided pursuant to this Section 1.7 and that, without such waiver, NXU and the Company would not have agreed to the terms of this Agreement.

(c) Notwithstanding the foregoing provisions of this Section 1.7 or anything to the contrary set forth herein, the Releasing Parties do not release or discharge, and each Releasing Party expressly does not release or discharge any Claims (i) that arise under or are based upon the terms of this Agreement, the Merger Agreement, any of the Ancillary Agreements or any other document, certificate or Contract executed or delivered in connection with the Merger Agreement or any similar agreement in respect of the transactions contemplated by the Merger Agreement to which a Releasing Party may be a party and that (in each case) is expressly disclosed in the Schedules of NXU, as each such agreement or instrument described above may be amended in accordance with its terms and the terms set forth in (A) the Merger Agreement or (B) this Agreement or the other Ancillary Agreements (if and to the extent applicable), (ii) any rights with respect to the Common Stock owned by such Releasing Party, or (iii) any Claims for indemnification, contribution, set-off, reimbursement or similar rights pursuant to any certificate of incorporation or bylaws of NXU or any of its Subsidiaries or any indemnity or similar agreements by the NXU with or for the benefit of a Releasing Party solely to the extent (in each case) set forth in the Schedules of Acquiror or as contemplated by Section 7.01 of the Merger Agreement.

(d) Notwithstanding the foregoing provisions of this Section 1.7, nothing contained in this Agreement shall be construed as an admission by any party hereto of any liability of any kind to any other party hereto.

Article II <u>REPRESENTATIONS AND WARRANTIES</u>

II.1 <u>No Inconsistent Agreement</u>. Except as contemplated or permitted under the Merger Agreement, each Stockholder hereby represents and covenants that such Stockholder has not entered into, and shall not enter into, any agreement that would restrict, limit or interfere with the performance of such Stockholder's obligations hereunder.

II.2 <u>Representations and Warranties of the Stockholders</u>. Each Stockholder hereby represents and warrants (severally as to such Stockholder and not jointly) as of the date hereof to NXU and the Company as follows:

a. <u>Organization; Due Authorization</u>. This Agreement has been duly executed and delivered by such Stockholder, and, assuming due authorization, execution and delivery by the other parties to this Agreement, this Agreement constitutes a legally valid and binding obligation of such Stockholder, enforceable against such Stockholder in accordance with the terms hereof (except as enforceability may be limited by bankruptcy laws, other similar laws affecting creditors' rights and general principles of equity affecting the availability of specific performance and other equitable remedies).

b. <u>Ownership</u>. Such Stockholder is the record and/or beneficial owner (within the meaning of Rule 13d-3 under the Exchange Act) of, and has good title to, all of his or her Stockholder Securities, and there exist no Liens or any other limitation or restriction (including any restriction on the right to vote, sell or otherwise dispose) affecting any such Stockholder Securities, other than pursuant to (i) this Agreement, (ii) the organizational documents of NXU, (iii) the Merger Agreement or (iv) any applicable securities laws. Other than the Stockholder Securities described on the signature page of such Stockholder hereto, such Stockholder does not hold or own any rights to acquire (directly or indirectly) any equity securities of NXU or outstanding options, warrants, rights, convertible or exchangeable securities, "phantom" stock rights, stock appreciation rights, stock-based performance units or commitments for shares of NXU.

c. <u>No Conflicts</u>. The execution and delivery of this Agreement by such Stockholder does not, and the performance by such Stockholder of his or her obligations hereunder will not, require any consent or approval that has not been given or other action that has not been taken by any Person (including under any contract binding upon such Stockholder or the Stockholder Shares), in each case to the extent such consent, approval or other action would prevent, enjoin or materially delay the performance by such Stockholder of his or her obligations under this Agreement. Such Stockholder has full right and power to enter into this Agreement.

d. Litigation. There are no Actions pending against such Stockholder, or to the knowledge of such Stockholder threatened against such Stockholder, before (or, in the case of threatened Legal Proceedings, that would be before) any Governmental Entity, that in any manner challenges or seeks to prevent, enjoin or materially delay the performance by such Stockholder of his or her obligations under this Agreement. Such Stockholder (i) is not subject to or a respondent in any legal action for any injunction, cease and desist order or order or stipulation to desist or refrain from any act or practice relating to the offering of securities in any jurisdiction, (ii) has never been convicted of, or pleaded guilty to, any crime involving fraud relating to any financial transaction or handling of funds of another person, or pertaining to any dealings in any securities and (iii) is not currently a defendant in any such criminal proceeding.

e. <u>Acknowledgment</u>. Such Stockholder understands and acknowledges that the Company is entering into the Merger Agreement in reliance upon such Stockholder's execution, delivery and performance of this Support Agreement.

Article III <u>MISCELLANEOUS</u>

III.1 <u>Termination</u>. This Support Agreement and all of its provisions shall automatically terminate and be of no further force and effect upon the earliest to occur of (i) the Effective Time, and (ii) such date and time as the Merger Agreement shall be terminated in accordance with its terms. Upon such termination of this Support Agreement, all obligations of the parties under this Support Agreement will terminate, without any liability or other obligation on the part of any party hereto to any Person in respect hereof or the transactions contemplated hereby, and no party hereto shall have any claim against another (and no person shall have any rights against such party), whether under contract, tort or otherwise, with respect to the subject matter hereof; <u>provided</u>, <u>however</u>, that the termination of this Support Agreement shall not relieve any party hereto from liability arising in respect of any breach of this Support Agreement prior to such termination. This ARTICLE III shall survive the termination of this Support Agreement.

III.2 <u>Governing Law</u>. Sections 11.06 and 11.12 of the Merger Agreement are incorporated herein by reference, *mutatis mutandis*.

III.3 <u>Assignment</u>. This Support Agreement and all of the provisions hereof will be binding upon and inure to the benefit of the parties hereto and their respective heirs, successors and permitted assigns. Neither this Support Agreement nor any of the rights, interests or obligations hereunder will be assigned (including by operation of law) without the prior written consent of all of the other parties hereto.

III.4 <u>Specific Performance</u>. The parties hereto agree that irreparable damage may occur in the event that any of the provisions of this Support Agreement were not performed in accordance with their specific terms or were otherwise breached. It is accordingly agreed that the parties hereto shall be entitled to seek an injunction or injunctions to prevent breaches of this Support Agreement and to enforce specifically the terms and provisions of this Support Agreement in the Court of Chancery of the State of Delaware (or, to the extent such court does not have subject matter jurisdiction, the Superior Court of the State of Delaware or the United States District Court for the District of Delaware), this being in addition to any other remedy to which such party is entitled at law or in equity. In the event that any Action shall be brought in equity to enforce the provisions of this Support Agreement, no party shall allege, and each party hereby waives the defense, that there is an adequate remedy at law, and each party agrees to waive any requirement for the securing or posting of any bond in connection therewith.

III.5 <u>Amendment; Waiver</u>. This Support Agreement or any provision hereof may not be amended, changed, supplemented, waived or otherwise modified or terminated, except upon the execution and delivery of a written agreement executed by NXU, the Company and each Stockholder.

III.6 <u>Severability</u>. If any provision of this Support Agreement is held invalid or unenforceable by any court of competent jurisdiction, the other provisions of this Support Agreement will remain in full force and effect. Any provision of this Support Agreement held invalid or unenforceable only in part or degree will remain in full force and effect to the extent not held invalid or unenforceable.

III.7 Notices. All notices, consents, waivers and other communications under this Agreement must be in writing and will be deemed to have been duly given (a) if personally delivered, on the date of delivery; (b) if delivered by express courier service of national standing for next day delivery (with charges prepaid), on the Business Day following the date of delivery to such courier service; (c) if delivered by electronic mail, on the date of transmission if on a Business Day before 5:00 p.m. local time of the business address of the recipient party (otherwise on the next succeeding Business Day); and (d) if deposited in the United States mail, first-class postage prepaid, on the date of delivery, in each case to the appropriate addresses or electronic mail addresses as a party may designate by notice to the other parties in accordance with this Section 3.7):

If to NXU:

Attn: Mark Hanchett E-mail: mark@nxuenergy.com

with a copy (which shall not constitute notice) to:

Attn: Mike Donahey 1 E Washington Street, Suite 2700 Phoenix, AZ 85004 E-mail: mdonahey@swlaw.com

If to the Company:

Verde Bioresins, Inc. 1431 E. Orangethorpe Ave. Fullerton, CA 92831 Attn: Brian Gordon E-mail: brian@verdebioresins.com

with a copy (which shall not constitute notice) to:

Wilmer Cutler Pickering Hale and Dorr LLP 2600 El Camino Real, Suite 400 Palo Alto, CA 94306 Attn: Daniel Zimmermann E-mail: Daniel.Zimmerman@wilmerhale.com

III.8 <u>Counterparts</u>. This Support Agreement may be executed in two or more counterparts (any of which may be delivered by electronic transmission), each of which shall constitute an original, and all of which taken together shall constitute one and the same instrument.

III.9 <u>Entire Agreement</u>. This Support Agreement and the agreements referenced herein constitute the entire agreement and understanding of the parties hereto in respect of the subject matter hereof and supersede all prior understandings, agreements or representations by or among the parties hereto to the extent they relate in any way to the subject matter hereof.

[THE REMAINDER OF THIS PAGE IS INTENTIONALLY BLANK]

NXU, Inc.

By: /s/ Mark Hanchett

Name: Mark Hanchett Title: Chief Executive Officer

Verde Bioresins, Inc.

By: /s/ Brian Gordon

Name: Brian Gordon Title: President

/s/ Mark Hanchett Mark Hanchett

Notice Address:

Stockholder Securities: 196,692 Options (underwater, none exercised) 3,826.3 shares of Class A Common Stock 26,503,675 shares of Class B Common Stock

/s/ Annie Pratt Annie Pratt

Notice Address:

Stockholder Securities: 76,812 Options (underwater, none exercised) 3,186 shares of Class A Common Stock 10,021,695 shares of Class B Common Stock

<u>Annex A</u>

Form of Joinder Agreement

This Joinder Agreement (this "Joinder Agreement") is made as of the date written below by the undersigned (the "Joining Party") in accordance with the Acquiror Support Agreement, dated as of October 23, 2024 (as amended, supplemented or otherwise modified from time to time, the "Support Agreement"), by and among Mark Hanchett, Annie Pratt, Nxu, Inc., a Delaware corporation, and Verde Bioresins, Inc., a Delaware corporation. Capitalized terms used herein and not otherwise defined shall have the meaning ascribed to them in the Support Agreement.

The Joining Party hereby acknowledges, agrees and confirms that, by its execution of this Joinder Agreement, the Joining Party shall be deemed to be a party to, and a "Stockholder" under, the Support Agreement as of the date hereof and shall have all of the rights and obligations of the Stockholders as if it had executed the Support Agreement. The Joining Party hereby ratifies, as of the date hereof, and agrees to be bound by, all of the terms, provisions and conditions contained in the Support Agreement.

IN WITNESS WHEREOF, the undersigned has duly executed this Joinder Agreement as of the date written below.

Name:

By: Title:

Date:

Notice Address:

COMPANY SUPPORT AGREEMENT

This Company Support Agreement (this "<u>Support Agreement</u>") is dated as of October 23, 2024, by and among Humanitario Capital LLC, a Puerto Rico limited liability company (the "<u>Stockholder</u>"), NXU, Inc., a Delaware corporation prior ("<u>Acquiror</u>"), and Verde Bioresins, Inc., a Delaware corporation (the "<u>Company</u>"). Capitalized terms used but not defined herein shall have the meanings ascribed to such terms in the Merger Agreement (as defined below).

RECITALS

WHEREAS, as of the date hereof, the Stockholder is the holder of record and/or the "beneficial owner" (within the meaning of Rule 13d-3 under the Exchange Act) of (i) 2,648,364,984 shares of the Company's Common Stock (the "<u>Stockholder Shares</u>"), (ii) (a) that certain Common Stock Purchase Warrant, dated as of February 17, 2023, (b) that certain Common Stock Purchase Warrant, dated as of February 8, 2021, and (c) that certain Common Stock Purchase Warrant, dated as of June 15, 2023 (together, the "<u>Stockholder Warrants</u>"), and (iii) (a) that certain Secured Convertible Promissory Note issued by the Company to Stockholder, dated as of February 17, 2023, (b) that certain Secured Convertible Promissory Note issued by the Company to Stockholder, dated June 15, 2023, (c) that certain Senior Secured Convertible Promissory Note issued by the Company to Stockholder, dated March 6, 2024, (d) that certain Senior Secured Convertible Promissory Note issued by the Company to Stockholder, dated March 6, secured Convertible Promissory Note issued by the Company to Stockholder, dated June 5, 2024, (f) that certain Senior Secured Convertible Promissory Note issued by the Company to Stockholder, dated July 26, 2024 and (g) that certain Senior Secured Convertible Promissory Note issued by the Company to Stockholder, dated August 30, 2024 (together, the "<u>Stockholder Notes</u>");

WHEREAS, prior to the execution and delivery of this Support Agreement, Acquiror, Nxu Merger Sub, Inc., a Delaware corporation and a direct wholly owned subsidiary of Acquiror ("<u>Merger Sub I</u>"), Nxu Merger Sub, LLC a Delaware limited liability company and a wholly owned subsidiary of Acquiror ("<u>Merger Sub II</u>") and the Company have entered into an Agreement and Plan of Merger (as amended or modified from time to time, the "<u>Merger Agreement</u>"), dated as of October [•], 2024, pursuant to which, among other transactions, Merger Sub I is to merge with and into the Company, with the Company continuing on as the surviving company and a wholly owned subsidiary of Acquiror (the "<u>First Merger</u>") and, promptly thereafter, such surviving company shall merge with and into Merger Sub II with Merger Sub II continuing as the surviving entity of such merger as a wholly owned subsidiary of Acquiror, on the terms and conditions set forth therein (the "<u>Second Merger</u>" and together with the First Merger, the "<u>Merger</u>");

WHEREAS, the Merger Agreement contemplates that the Stockholder will receive, pursuant to the terms of the Merger Agreement, certain shares of Acquiror Common Stock at the Closing (such shares, together with any securities paid as dividends or distributions with respect to such securities or into which such securities are exchanged or converted, the "<u>Closing Shares</u>"); and

WHEREAS, as an inducement to Acquiror and the Company to enter into the Merger Agreement and to consummate the transactions contemplated therein, the parties hereto desire to agree to certain matters as set forth herein.

AGREEMENT

NOW, THEREFORE, in consideration of the foregoing and the mutual agreements contained herein, and intending to be legally bound hereby, the parties hereto hereby agree as follows:

ARTICLE I <u>SUPPORT AGREEMENT; COVENANTS; RELEASE; VOTING TRUST AGREEMENT</u>

1.1 <u>Binding Effect of Merger Agreement</u>. The Stockholder hereby acknowledges that it has read the Merger Agreement and this Support Agreement and has had the opportunity to consult with its tax and legal advisors. During the term commencing on the date hereof and ending at the Expiration Time, the Stockholder shall be bound by and comply with Section 8.04 (*Confidentiality; Publicity*) of the Merger Agreement (and any relevant definitions contained in any such Sections) as if (a) the Stockholder was an original signatory to the Merger Agreement with respect to such provision, and (b) each reference to the "Stockholder" contained in Section 8.04 of the Merger Agreement also referred to the Stockholder.

No Transfer. During the period commencing on the date hereof and ending on the earliest to occur of (a) the Effective Time and (b) such 1.2 date and time as the Merger Agreement shall be terminated in accordance with its terms (the earlier of clauses (a) and (b), the "Expiration Time"), the Stockholder shall not, except in each case pursuant to the Merger Agreement, Transfer any of the Stockholder Shares, the Stockholder Warrants or the Stockholder Notes; provided, however, that the foregoing shall not prohibit Transfers between the Stockholder and any Affiliate of the Stockholder, so long as, prior to and as a condition to the effectiveness of any such Transfer, such Affiliate executes and delivers to Acquiror and the Company a joinder to this Support Agreement in substantially the form attached hereto as Annex A; provided, further, that any Transfer permitted under this Section 1.2 shall not relieve such Stockholder of its obligations under this Support Agreement. Any Transfer in violation of this Section 1.2 shall be null and void. Notwithstanding anything to the contrary above, the Stockholder shall not be prohibited from exercising or converting the Stockholder Warrants or the Stockholder Notes into Company Common Stock, and such Company Common Stock issued upon conversion of the Stockholder Warrants or the Stockholder Notes shall be deemed to be "Stockholder Shares" for purposes of this Support Agreement. "Transfer" shall mean the (a) sale or assignment of, offer to sell, contract or agreement to sell, hypothecate, pledge, grant of any option to purchase or otherwise dispose of or agreement to dispose of, directly or indirectly, or establishment or increase of a put equivalent position or liquidation with respect to or decrease of a call equivalent position within the meaning of Section 16 of the Securities Exchange Act of 1934, as amended, and the rules and regulations of the Securities and Exchange Commission promulgated thereunder with respect to, any security, (b) entry into any swap or other arrangement that transfers to another, in whole or in part, any of the economic consequences of ownership of any security, whether any such transaction is to be settled by delivery of such securities, in cash or otherwise, or (c) public announcement of any intention to effect any transaction specified in clause (a) or (b).

1.3 <u>New Shares</u>. In the event that after the date hereof (a) any shares of Company Common Stock are issued to the Stockholder pursuant to any stock dividend, stock split, recapitalization, reclassification, combination or exchange of the Stockholder Shares, on or affecting the Stockholder Shares owned by the Stockholder or otherwise, (b) the Stockholder purchases or otherwise acquires beneficial ownership of any Company Common Stock or other equity securities of the Company (such Company Common Stock or other equity securities of the Company (such Company Common Stock or other equity securities of the Company (such Company Common Stock or other equity securities of the Company (such Company Common Stock or other equity securities of the Stockholder acquired or purchased by the Stockholder shall be subject to the terms of this Support Agreement to the same extent as if they constituted the Stockholder Shares owned by the Stockholder as of the date hereof.

1.4 <u>Closing Date Deliverables</u>. Upon the Closing, the Stockholder shall deliver to Acquiror and the Company a duly executed copy of the Company Lock-Up Agreement, by and among Acquiror and the Stockholder.

1.5 <u>Conversion of Stockholder Notes</u>. In connection with the Closing and immediately prior to the Effective Time, the Stockholder and the Company each shall take all actions for the Stockholder Notes to be converted into shares of the Company Common Stock pursuant to the terms of the applicable Stockholder Note.

1.6 <u>Conversion of Stockholder Warrants</u>. In connection with the Closing and immediately prior to the Effective Time, the Stockholder and the Company each shall take all actions for the Stockholder Warrants to be converted into shares of the Company Common Stock pursuant to the terms of that certain Convertible Note and Warrant Purchase Agreement, dated as of June 15, 2023, between the Company and the Stockholder.

1.7 Voting and Support Agreement.

(a) The Stockholder irrevocably and unconditionally agrees that, prior to the Expiration Time, at any meeting of the stockholders of the Company, and in any action by written consent of the stockholders of the Company distributed by the Board of Directors of the Company or otherwise undertaken in respect of or as contemplated by the Merger Agreement or the transactions contemplated thereby in a form reasonably acceptable to Acquiror, the Stockholder shall, if a meeting is held, appear at the meeting, in person or by proxy, or otherwise cause its Stockholder Shares (to the extent such Stockholder Shares are entitled to vote on or provide consent with respect to such matter):

(i) to approve and adopt the Merger Agreement and the transactions contemplated thereby, including the Merger;

(ii) in any other circumstances upon which a consent, waiver or other approval is required under the Company Organizational Documents or under any agreements between the Company and its stockholders or otherwise sought with respect to the Merger Agreement or the transactions contemplated thereby, to vote, consent, waive or approve (or cause to be voted, consented, waived or approved) all of the Stockholder's Stockholder Shares held at such time in favor thereof (to the extent such Stockholder Shares are entitled to vote on or provide consent, waiver or approval with respect to such matter);

(iii) against any merger agreement or merger, consolidation, combination, sale of substantial assets, reorganization, recapitalization, dissolution, liquidation or winding up of or by the Company (other than the Merger Agreement and the transactions contemplated thereby, including the Merger);

(iv) against any change in the business, management or Board of Directors of the Company that would or would reasonably be expected to adversely affect the ability of the Company to consummate the transactions contemplated by the Merger Agreement, including the Merger; and

(v) against any proposal, action or agreement that would (A) impede, frustrate, prevent or nullify any provision of this Support Agreement, the Merger Agreement or the transactions contemplated thereby, including the Merger, (B) result in a breach in any respect of any covenant, representation, warranty or any other obligation or agreement of the Company under the Merger Agreement, (C) result in any of the conditions set forth in Article IX of the Merger Agreement not being fulfilled or (D) change in any manner the dividend policy or capitalization of, including the voting rights of any class of capital stock or securities convertible into capital stock of, the Company; and

(vi) the Stockholder hereby agrees that it shall not commit or agree to take any action inconsistent with the foregoing.

1.8 <u>Further Assurances</u>. The Stockholder shall take, or cause to be taken, all such further actions and do, or cause to be done, all things reasonably necessary (including under applicable Law) to consummate the Merger and the other transactions contemplated by the Merger Agreement and this Support Agreement, in each case, on the terms and subject to the conditions set forth therein and herein, as applicable.

1.9 <u>No Inconsistent Agreement</u>. The Stockholder hereby represents and covenants that such Stockholder has not entered into, and shall not enter into, any agreement that would restrict, limit or interfere with the performance of the Stockholder's obligations hereunder.

1.10 <u>Voting Trust Agreement</u>. The Stockholder hereby covenants and agrees that, within ten (10) Business Days after the Registration Statement is declared effective under the Securities Act, the Stockholder will enter into a voting trust agreement, on terms and conditions set forth in <u>Exhibit A</u> hereto and otherwise on terms and conditions to the reasonable satisfaction of the Stockholder, the Company and the Acquiror (the "<u>Voting Trust</u> <u>Agreement</u>"), with an independent third-party trustee appointed by the Stockholder (which trustee shall not be an Affiliate of the Stockholder or a member of the immediate family of any Affiliate of the Stockholder) (the "<u>Voting Trustee</u>"), pursuant to which the Stockholder will transfer and assign to the Voting Trustee all of the Closing Shares received by the Stockholder pursuant to the terms of the Merger Agreement, effective as of the Effective Time.

1.11 <u>Waiver and Release of Claims</u>. The Stockholder hereby covenants and agrees as follows:

(a) Subject to and conditioned upon the Effective Time, effective as of the Effective Time (and subject to the limitations set forth in paragraph (c) below), the Stockholder, on behalf of itself and its Affiliates and its and their respective successors, assigns, representatives, administrators, executors and agents, and any other person or entity claiming by, through, or under any of the foregoing (each a "<u>Releasing Party</u>" and, collectively, the "<u>Releasing Parties</u>," provided, for the avoidance of doubt, that the Acquiror, shall not be deemed a Releasing Party hereunder), does hereby unconditionally and irrevocably release, waive and forever discharge the Acquiror, the Company, and each of their past and present directors, officers, employees, agents, predecessors, successors, assigns, and Subsidiaries, from any and all past or present claims, demands, damages, judgments, causes of action and liabilities of any nature whatsoever, whether or not known, suspected or claimed, arising directly or indirectly from any act, omission, event or transaction occurring (or any circumstances existing) at or prior to the Effective Time (each a "<u>Claim</u>" and, collectively, the "<u>Claims</u>").

(b) The Stockholder acknowledges that it may hereafter discover facts in addition to or different from those which it now knows or believes to be true with respect to the subject matter of this Support Agreement, and that it may hereafter come to have a different understanding of the law that may apply to potential claims which it is releasing hereunder, but it affirms that, except as is otherwise specifically provided herein, it is its intention to fully, finally and forever settle and release any and all Claims. In furtherance of this intention, the Stockholder acknowledges that the releases contained herein shall be and remain in effect as full and complete general releases notwithstanding the discovery or existence of any such additional facts or different understandings of law. The Stockholder knowingly and voluntarily waives and releases any and all rights and benefits it may now have, or in the future may have, under Section 1542 of the California Civil Code (or any analogous law of any other state), which reads as follows:

"A GENERAL RELEASE DOES NOT EXTEND TO CLAIMS THAT THE CREDITOR OR RELEASING PARTY DOES NOT KNOW OR SUSPECT TO EXIST IN HIS OR HER FAVOR AT THE TIME OF EXECUTING THE RELEASE AND THAT, IF KNOWN BY HIM OR HER, WOULD HAVE MATERIALLY AFFECTED HIS OR HER SETTLEMENT WITH THE DEBTOR OR RELEASED PARTY."

The Stockholder understands that Section 1542, or a comparable statute, rule, regulation or order of another jurisdiction, gives the Stockholder the right not to release existing Claims of which the Stockholder is not aware, unless the Stockholder voluntarily chooses to waive this right. Having been so apprised, the Stockholder nevertheless hereby voluntarily elects to and does waive any rights it may have that are described in Section 1542, or such other comparable statute, rule, regulation or order, and elects to assume all risks for Claims that exist, existed or may hereafter exist in its favor, known or unknown, suspected or unsuspected, arising out of or related to claims or other matters purported to be released pursuant to this <u>Section 1.11</u>, in each case, effective as of the Effective Time. The Stockholder acknowledges and agrees that the foregoing waiver is an essential and material term of the release provided pursuant to this <u>Section 1.11</u> and that, without such waiver, the Acquiror and the Company would not have agreed to the terms of this Support Agreement.

(c) Notwithstanding the foregoing provisions of this Section 1.11 or anything to the contrary set forth herein, the Releasing Parties do not release or discharge, and each Releasing Party expressly does not release or discharge: (i) any Claims that arise under or are based upon the terms of (A) this Support Agreement, the Merger Agreement, any of the Ancillary Agreements or any other document, certificate or Contract executed or delivered in connection with the Merger Agreement; (B) the Registration Rights Agreement; or (C) any underwriting agreement, business combination marketing agreement, financial advisory agreement, engagement letter or any similar agreement in respect of the transactions contemplated by the Merger Agreement or instrument described in this clause (i) may be amended in accordance with its terms and the terms set forth in (x) the Merger Agreement or (y) this Support Agreement or the other Ancillary Agreements (if and to the extent applicable), (ii) any rights with respect to the capital stock or warrants of the Company owned by such Releasing Party or the promissory notes issued by the Company to such Releasing Party, or (iii) any Claims for indemnification, contribution, set-off, reimbursement or similar rights pursuant to any certificate of incorporation or bylaws of the Acquiror or any of its Subsidiaries or any of its Subsidiaries with or for the benefit of a Releasing Party solely to the extent (in each case) set forth in the Schedules of the Company or as contemplated by Section 7.01 of the Merger Agreement.

(d) Notwithstanding the foregoing provisions of this <u>Section 1.11</u>, nothing contained in this Support Agreement shall be construed as an admission by any party hereto of any liability of any kind to any other party hereto.

REPRESENTATIONS AND WARRANTIES

2.1 <u>No Inconsistent Agreement</u>. The Stockholder hereby represents and covenants that the Stockholder has not entered into, and shall not enter into, any agreement that would restrict, limit or interfere with the performance of the Stockholder's obligations hereunder.

2.2 <u>Representations and Warranties of the Stockholder</u>. The Stockholder hereby represents and warrants as of the date hereof to Acquiror and the Company as follows:

(a) <u>Organization; Due Authorization</u>. The Stockholder is duly organized, validly existing and in good standing under the laws of the jurisdiction in which it is incorporated, formed, organized or constituted, and the execution, delivery and performance of this Support Agreement and the consummation of the transactions contemplated hereby are within the Stockholder's corporate, limited liability company or organizational powers and have been duly authorized by all necessary corporate, limited liability company or organizational actions on the part of the Stockholder. This Support Agreement has been duly executed and delivered by the Stockholder, and, assuming due authorization, execution and delivery by the other parties to this Support Agreement, this Support Agreement constitutes a legally valid and binding obligation of the Stockholder, enforceable against the Stockholder in accordance with the terms hereof (except as enforceability may be limited by bankruptcy laws, other similar laws affecting creditors' rights and general principles of equity affecting the availability of specific performance and other equitable remedies).

(b) <u>Ownership</u>. The Stockholder is the record and/or beneficial owner (within the meaning of Rule 13d-3 under the Exchange Act) of, and have good title to, all of the Stockholder Shares, which represent 81.66% of the voting power of the outstanding Company Stock entitled to vote, voting together as a single class, and there exist no Liens or any other limitation or restriction (including any restriction on the right to vote, sell or otherwise dispose) affecting any Stockholder Shares, other than pursuant to (i) this Support Agreement, (ii) the Company Organizational Documents, or (iii) any applicable securities laws. Other than the Stockholder Shares, the Stockholder Warrants and the Stockholder Notes, the Stockholder does not hold or own any rights to acquire (directly or indirectly) any equity securities of the Company or outstanding options, warrants, rights, convertible or exchangeable securities, "phantom" stock rights, stock appreciation rights, stock-based performance units or commitments for shares of the Company.

(c) <u>No Conflicts</u>. The execution and delivery of this Support Agreement by the Stockholder does not, and the performance by the Stockholder of its obligations hereunder will not, (i) conflict with or result in a violation of the organizational documents of the Stockholder or (ii) require any consent or approval that has not been given or other action that has not been taken by any Person (including under any contract binding upon the Stockholder or the Stockholder of its obligations under this Support Agreement. The Stockholder has full right and power to enter into this Support Agreement. For the avoidance of doubt, pursuant to the Company Organizational Documents, other than the consent of the Stockholder, no other consent of any other stockholder of the Company is required to obtain the Company Stockholder Approvals.

(d) <u>Litigation</u>. Except as set forth in Schedule 2.2(d) attached hereto, (i) there are no Actions pending against the Stockholder or, to the knowledge of the Stockholder, threatened against the Stockholder, before (or, in the case of any threatened Actions, that would be before) any Governmental Entity, that in any manner challenge or seek to prevent, enjoin or materially delay the performance by the Stockholder of its obligations under this Support Agreement, (ii) the Stockholder has not been suspended or expelled from membership in any securities or commodities exchange or association or had a securities or commodities license or registration denied, suspended or revoked and (iii) the Stockholder (A) is not subject to or a respondent in any legal action for any injunction, cease and desist order or order or stipulation to desist or refrain from any act or practice relating to the offering of securities in any jurisdiction, (B) has not been convicted of, or pleaded guilty to, any crime involving fraud relating to any financial transaction or handling of funds of another person, or pertaining to any dealings in any securities and (C) is not currently a defendant in any such criminal proceeding.

(e) <u>Acknowledgment</u>. The Stockholder understands and acknowledges that each of the Company and the Acquiror is entering into the Merger Agreement in reliance upon the Stockholder's execution, delivery and performance of this Support Agreement.

ARTICLE III MISCELLANEOUS

3.1 Termination. This Support Agreement and all of its provisions shall automatically terminate and be of no further force and effect upon the earliest to occur of (i) the Effective Time, and (ii) such date and time as the Merger Agreement shall be terminated in accordance with its terms. Upon such termination of this Support Agreement, all obligations of the parties under this Support Agreement will terminate, without any liability or other obligation on the part of any party hereto to any Person in respect hereof or the transactions contemplated hereby, and no party hereto shall have any claim against another (and no person shall have any rights against such party), whether under contract, tort or otherwise, with respect to the subject matter hereof; provided, however, that the termination of this Support Agreement shall not relieve any party hereto from liability arising in respect of any breach of this Support Agreement prior to such termination. This <u>ARTICLE III</u> shall survive the termination of this Support Agreement.

3.2 <u>Governing Law</u>. Sections 11.06 and 11.12 of the Merger Agreement are incorporated herein by reference, *mutatis mutandis*.

3.3 <u>Assignment</u>. This Support Agreement and all of the provisions hereof will be binding upon and inure to the benefit of the parties hereto and their respective heirs, successors and permitted assigns. Neither this Support Agreement nor any of the rights, interests or obligations hereunder will be assigned (including by operation of law) without the prior written consent of all of the other parties hereto.

3.4 <u>Specific Performance</u>. The parties hereto agree that irreparable damage may occur in the event that any of the provisions of this Support Agreement were not performed in accordance with their specific terms or were otherwise breached. It is accordingly agreed that the parties hereto shall be entitled to seek an injunction or injunctions to prevent breaches of this Support Agreement and to enforce specifically the terms and provisions of this Support Agreement in the Court of Chancery of the State of Delaware (or, to the extent such court does not have subject matter jurisdiction, the Superior Court of the State of Delaware or the United States District Court for the District of Delaware), this being in addition to any other remedy to which such party is entitled at law or in equity. In the event that any Action shall be brought in equity to enforce the provisions of this Support Agreement, no party shall allege, and each party hereby waives the defense, that there is an adequate remedy at law, and each party agrees to waive any requirement for the securing or posting of any bond in connection therewith.

3.5 <u>Amendment; Waiver</u>. This Support Agreement or any provision hereof may not be amended, changed, supplemented, waived or otherwise modified or terminated, except upon the execution and delivery of a written agreement executed by Acquiror, the Company and the Stockholder.

3.6 <u>Severability</u>. If any provision of this Support Agreement is held invalid or unenforceable by any court of competent jurisdiction, the other provisions of this Support Agreement will remain in full force and effect. Any provision of this Support Agreement held invalid or unenforceable only in part or degree will remain in full force and effect to the extent not held invalid or unenforceable.

3.7 <u>Notices</u>. All notices, consents, waivers and other communications under this Support Agreement must be in writing and will be deemed to have been duly given (a) if personally delivered, on the date of delivery; (b) if delivered by express courier service of national standing for next day delivery (with charges prepaid), on the Business Day following the date of delivery to such courier service; (c) if delivered by electronic mail, on the date of transmission if on a Business Day before 5:00 p.m. local time of the business address of the recipient party (otherwise on the next succeeding Business Day); and (d) if deposited in the United States mail, first-class postage prepaid, on the date of delivery, in each case to the appropriate addresses or electronic mail addresses as a party may designate by notice to the other parties in accordance with this <u>Section 3.7</u>):

If to Acquiror (before the Effective Time):

Attn: Mark Hanchett

E-mail: mark@nxuenergy.com

with a copy (which shall not constitute notice) to:

Snell & Wilmer L.L.P. Attn: Mike Donahey 1 E Washington Street, Suite 2700 Phoenix, AZ 85004 E-mail: mdonahey@swlaw.com

If to the Company:

Verde Bioresins, Inc. 1431 Orangethorpe Ave. Fullerton, CA 92831 Attn: Brian Gordon E-mail: brian@verdebioresins.com

with a copy (which shall not constitute notice) to:

Wilmer Cutler Pickering Hale and Dorr LLP 2600 El Camino Real, Suite 400 Palo Alto, CA 94306 Attn: Daniel Zimmermann E-mail: Daniel.Zimmermann@wilmerhale.com

If to the Stockholder:

Humanitario Capital LLC Attn: Terren S. Peizer E-mail: terren@acuitasgh.com

3.8 <u>Counterparts</u>. This Support Agreement may be executed in two or more counterparts (any of which may be delivered by electronic transmission), each of which shall constitute an original, and all of which taken together shall constitute one and the same instrument.

3.9 <u>Entire Agreement</u>. This Support Agreement and the agreements referenced herein constitute the entire agreement and understanding of the parties hereto in respect of the subject matter hereof and supersede all prior understandings, agreements or representations by or among the parties hereto to the extent they relate in any way to the subject matter hereof.

3.10 <u>Rights of Third Parties</u>. Nothing expressed or implied in this Support Agreement is intended or shall be construed to confer upon or give any Person, other than the parties hereto, any right or remedies under or by reason of this Support Agreement.

3.11 <u>Stockholder Transaction Expenses</u>. If the Closing occurs, at the Closing, Acquiror shall pay, or cause to be paid, to the Stockholder (or as directed by the Stockholder), by wire transfer of immediately available funds, the Stockholder Transaction Expenses (as defined below). If the Merger Agreement is terminated in accordance with its terms, promptly upon such termination, the Company shall reimburse the Stockholder for all Stockholder Transaction Expenses. As used herein, "<u>Stockholder Transaction Expenses</u>" means all reasonable and documented fees and disbursements of the Stockholder and its Affiliates payable to unaffiliated third parties (including, without limitation, outside counsel, agents, advisors, consultants, experts and financial advisors employed by or on behalf of the Stockholder or any of its Affiliates) incurred in connection with the Transactions, in each case, to the extent payable in cash; provided that such fees and disbursements of the Stockholder and its Affiliates shall not exceed \$200,000.00 in the aggregate.

[THE REMAINDER OF THIS PAGE IS INTENTIONALLY BLANK]

IN WITNESS WHEREOF, the Stockholder, Acquiror and the Company have each caused this Support Agreement to be duly executed as of the date first written above.

NXU, Inc.

By: /s/Mark Hanchett Name: Mark Hanchett Title: Chief Executive Officer

[Signature Page to Company Support Agreement]

IN WITNESS WHEREOF, the Stockholder, Acquiror and the Company have each caused this Support Agreement to be duly executed as of the date first written above.

Humanitario Capital LLC

By: <u>/s/Terren S. Peizer</u> Name: Terren S. Peizer Title: Authorized Signatory

[Signature Page to Company Support Agreement]

IN WITNESS WHEREOF, the Stockholder, Acquiror and the Company have each caused this Support Agreement to be duly executed as of the date first written above.

Verde Bioresins, Inc.

By: /s/Brian Gordon

Name: Brian Gordon Title: President

[Signature Page to Company Support Agreement]

Annex A

Form of Joinder Agreement

This Joinder Agreement (this "Joinder Agreement") is made as of the date written below by the undersigned (the "Joining Party") in accordance with the Company Support Agreement, dated as of October 23, 2024 (as amended, supplemented or otherwise modified from time to time, the "Support Agreement"), by and among Humanitario Capital LLC, a Puerto Rico limited liability company (the "Stockholder"), NXU, Inc., a Delaware corporation prior ("Acquiror"), and Verde Bioresins, Inc., a Delaware corporation (the "Company"). Capitalized terms used herein and not otherwise defined shall have the meaning ascribed to them in the Support Agreement.

The Joining Party hereby acknowledges, agrees and confirms that, by its execution of this Joinder Agreement, the Joining Party shall be deemed to be a party to, and a "Stockholder" under, the Support Agreement as of the date hereof and shall have all of the rights and obligations of the Stockholder as if it had executed the Support Agreement. The Joining Party hereby ratifies, as of the date hereof, and agrees to be bound by, all of the terms, provisions and conditions contained in the Support Agreement.

IN WITNESS WHEREOF, the undersigned has duly executed this Joinder Agreement as of the date written below.

Name:

By:______ Title: Date:

Notice Address:

[•]

[•]

[•]

<u>Exhibit A</u>

Terms of Voting Trust Agreement

In the event that, at any time during the term of the Voting Trust Agreement, (a) any shares of Acquiror Common Stock are issued to the Stockholder pursuant to any stock dividend, stock split, recapitalization, reclassification, combination or exchange of any of the Closing Shares, on or affecting any of the Closing Shares owned by the Stockholder or otherwise, (b) the Stockholder purchases or otherwise acquires beneficial ownership of any shares of Acquiror Common Stock or (c) the Stockholder acquires the right to vote or share in the voting of any shares of Acquiror Common Stock, then all such shares of Acquiror Common Stock shall be subject to the Voting Trust Agreement.

During the term of the Voting Trust Agreement, the Voting Trustee shall vote all shares of Acquiror Common Stock subject to the Voting Trust Agreement in favor of any vote, consent or other action in the same proportion as voted by public stockholders of the Company that are not Affiliates of the Company with respect to such matter.

The Voting Trust Agreement shall remain in effect until the earliest to occur of the following: (a) Mr. Peizer is sentenced to a non-custodial sentence, or finishes service of any custodial part of a sentence, on the charges described in Schedule 2.2(d) attached to the Support Agreement, or the conviction on such charges is set aside or reversed, (b) the fifth anniversary of the Closing Date, and (c) such time as the Stockholder and its Affiliates collectively beneficially own shares of Acquiror Common Stock representing in the aggregate less than 10% of the total voting power of all shares of Acquiror Common Stock then outstanding.

FORM OF COMPANY LOCK-UP AGREEMENT

THIS COMPANY LOCK-UP AGREEMENT (this "<u>Agreement</u>") is made and entered into as of October $[\bullet]$, 2024 between $[\bullet]$, a $[\bullet]$ (the "<u>Stockholder</u>") and [Verde Bioresins, Corp.], a Delaware corporation (the "<u>Company</u>"). The Stockholder and the Company are sometimes referred to herein individually as a "<u>Party</u>" and collectively as the "<u>Parties</u>". Capitalized terms used but not otherwise defined herein shall have the meanings ascribed to such terms in the Merger Agreement (as defined below).

WHEREAS, the Company, Verde Bioresins, Inc., a Delaware corporation ("<u>Verde</u>"), Nxu Merger Sub, Inc., a Delaware corporation and Nxu Merger Sub, LLC, a Delaware limited liability company, entered into that certain Agreement and Plan of Merger, dated as of October 23, 2024 (as it may be amended, restated or otherwise modified from time to time in accordance with its terms, the "<u>Merger Agreement</u>");

WHEREAS, the Merger Agreement contemplates that the Stockholder will receive a number of shares of Acquiror Common Stock at Closing (such shares, together with any securities paid as dividends or distributions with respect to such securities or into which such securities are exchanged or converted, the "Lock-Up Shares"); and

WHEREAS, the Merger Agreement contemplates that the Parties will enter into this Agreement, pursuant to which the Lock-Up Shares shall become subject to limitations on disposition as set forth herein.

NOW, THEREFORE, in consideration of the premises set forth above, which are incorporated in this Agreement as if fully set forth below, and intending to be legally bound hereby, the Parties hereby agree as follows:

1. For purposes of this Agreement:

- a. the term "Closing" has the meaning given to it in the Merger Agreement;
- b. the term "Initial Lock-Up Period" means the period beginning on the Closing Date and ending 180 days thereafter;
- c. the term "Lock-Up Period" means, jointly, the Initial Lock-Up Period and the Second Lock-Up Period;

d. the term "<u>Permitted Transferees</u>" means any Person to whom the Stockholder is permitted to transfer Lock-Up Shares prior to the expiration of the Lock-Up Period pursuant to <u>Section 2(a)</u>;

e. the term "Second Lock-Up Period" means the twenty-four (24) month period following the Initial Lock-Up Period; and

f. the term "<u>Transfer</u>" shall mean the (A) sale or assignment of, offer to sell, contract or agreement to sell, hypothecate, pledge, grant of any option to purchase or otherwise dispose of or agreement to dispose of, directly or indirectly, or establishment or increase of a put equivalent position or liquidation with respect to or decrease of a call equivalent position within the meaning of Section 16 of the Securities Exchange Act of 1934, as amended, and the rules and regulations of the Securities and Exchange Commission promulgated thereunder with respect to, any security, (B) entry into any swap or other arrangement that transfers to another, in whole or in part, any of the economic consequences of ownership of any security, whether any such transaction is to be settled by delivery of such securities, in cash or otherwise, or (C) public announcement of any intention to effect any transaction specified in clause (A) or (B).

2. Lock-Up Provisions.

a. Notwithstanding the provisions set forth in <u>Section 2(b)</u> and <u>Section 2(c)</u>, below, the Stockholder or its Permitted Transferees may Transfer the Lock-Up Shares:

i. during the Lock-Up Period (i) to its officers or directors, (ii) to any Affiliates of the Stockholder; (iii) in respect of (i) or (ii), in the case of an individual, by bona fide gift to a member of such individual's immediate family or to a trust, the beneficiary of which is the Stockholder, a member of such individual's immediate family, an Affiliate of such individual or to a charitable organization; (iv) in respect of (i), (ii) or (iii), in the case of an individual, by virtue of laws of descent and distribution upon death of such individual; or (v) in the case an individual, pursuant to a domestic order or a negotiated divorce settlement, in each case, subject to any such transferee signing a joinder hereto agreeing to be bound by all provisions hereof to the same extent as the Stockholder; and

ii. following such time that the Company completes any liquidation, merger, capital stock exchange, reorganization or other similar transaction that results in all stockholders of the Surviving Corporation having the right to exchange their shares for cash, securities or other property.

b. The Stockholder hereby agrees that, during the Initial Lock-Up Period, other than as permitted in Section 2(a) hereof, it shall not, and shall cause any of its Permitted Transferees not to, Transfer any Lock-Up Shares.

c. The Stockholder hereby agrees that, during the Second Lock-Up Period, other than as permitted in Section 2(a) hereof, it shall not, and shall cause any of its Permitted Transferees not to, Transfer any Lock-Up Shares representing more than five percent (5%) of the aggregate Lock-Up Shares in any calendar month.

d. If any Transfer is made or attempted contrary to the provisions of this Agreement, such Transfer shall be null and void ab initio, and the Company shall refuse to recognize any such transferee of the Lock-Up Shares as one of its equity holders for any purpose. In order to enforce this <u>Section 2</u>, the Company may impose stop-transfer instructions with respect to the Lock-Up Shares (and any Permitted Transferees and assigns thereof) until the end of the Lock-Up Period.

e. During the Lock-Up Period, each certificate (if any are issued) evidencing any Lock-Up Shares shall be stamped or otherwise imprinted with a legend in substantially the following form, in addition to any other applicable legends:

"THE SECURITIES REPRESENTED BY THIS CERTIFICATE ARE SUBJECT TO RESTRICTIONS ON TRANSFER SET FORTH IN A LOCK-UP AGREEMENT, DATED AS OF OCTOBER [•], 2024, BY AND AMONG THE ISSUER OF SUCH SECURITIES (THE "*ISSUER*") AND THE ISSUER'S SECURITY HOLDER NAMED THEREIN, AS AMENDED. A COPY OF SUCH LOCK-UP AGREEMENT WILL BE FURNISHED WITHOUT CHARGE BY THE ISSUER TO THE HOLDER HEREOF UPON WRITTEN REQUEST."

For the avoidance of any doubt, the Stockholder shall retain all of its rights as a stockholder of the Company with respect to the Lock-Up Shares during the Lock-Up Period, including the right to vote any Lock-Up Shares.

3. Miscellaneous.

a. <u>Effective Date</u>. <u>Section 1</u> of this Agreement shall become effective at the Effective Time.

b. <u>Termination of the Merger Agreement</u>. Notwithstanding anything to the contrary contained herein, in the event that the Merger Agreement is terminated in accordance with its terms prior to the Effective Time, this Agreement and all rights and obligations of the Parties hereunder shall automatically terminate and be of no further force or effect.

c. Notices. All notices, consents, waivers and other communications under this Agreement must be in writing and will be deemed to have been duly given (a) if personally delivered, on the date of delivery; (b) if delivered by express courier service of national standing for next day delivery (with charges prepaid), on the Business Day (as defined in the Merger Agreement) following the date of delivery to such courier service; (c) if delivered by electronic mail, on the date of transmission if on a Business Day before 5:00 p.m. local time of the business address of the recipient party (otherwise on the next succeeding Business Day); and (d) if deposited in the United States mail, first-class postage prepaid, on the date of delivery, in each case to the appropriate addresses or electronic mail addresses set forth below (or to such other addresses or electronic mail addresses as a party may designate by notice to the other parties in accordance with this Section 3(c)):

| [•] | With a copy to (which shall not constitute notice): Wilmer Cutler Pickering Hale and Dorr LLP 2600 El Camino Real, Suite 400 Palo Alto, CA 94306 Attn: Daniel Zimmermann E-mail: Daniel.Zimmermann@wilmerhale.com |
|-----------------------------------|--|
| If to the Stockholder, to: [•] | <i>With a copy to (which shall not constitute notice):</i> [•] |

d. <u>Incorporation by Reference</u>. Sections 1.02 (Construction) 11.03 (Assignment), 11.06 (Governing Law), 11.07 (Captions; Counterparts), 11.09 (Entire Agreement), 11.10 (Amendments), 11.11 (Severability), 11.12 (Jurisdiction; Waiver of Trial by Jury) and 11.13 (Enforcement) of the Merger Agreement are incorporated herein by reference and shall apply to this Agreement *mutatis mutandis*.

e. <u>Representations and Warranties</u>. Each of the parties hereto, by their respective execution and delivery of this Agreement, hereby represents and warrants to the other that (i) such party has the full right, capacity and authority to enter into, deliver and perform its respective obligations under this Agreement, (ii) this Agreement has been duly executed and delivered by such party and is a binding and enforceable obligation of such party and, enforceable against such party in accordance with the terms of this Agreement, and (iii) the execution, delivery and performance of such party's obligations under this Agreement will not conflict with or breach the terms of any other agreement, contract, commitment or understanding to which such party is a party or to which the assets or securities of such party are bound. The Stockholder has independently evaluated the merits of his/her/its decision to enter into and deliver this Agreement, and such Stockholder confirms that he/she/it has not relied on the advice of Company, Company's legal counsel, or any other person.

[Remainder of Page Intentionally Left Blank; Signature Pages Follow]

IN WITNESS WHEREOF, the parties have executed this Agreement as of the date first written above.

STOCKHOLDER:

| By: | | | | |
|---------------|-----------|------------|--|--|
| Name: | | | | |
| Title: | | | | |
| THE CO | OMPANY: | | | |
| | | | | |
| | | | | |
| | | | | |
| [VERDF | E BIORESI | NS, CORP.] | | |
| [VERDF [•] | E BIORESI | NS, CORP.] | | |
| | E BIORESI | NS, CORP.] | | |
| | E BIORESI | NS, CORP.] | | |

[Signature Page to Company Lock-Up Agreement]

VOTING AGREEMENT AND IRREVOCABLE PROXY

Dated October 23, 2024

This Voting Agreement and Irrevocable Proxy (this "<u>Agreement</u>") is entered into as of the date set forth above by and between the Stockholder set forth on the signature pages hereof (the "**Holder**") and Nxu, Inc., a Delaware corporation (the "<u>Corporation</u>").

Whereas, the Corporation has issued, or will issue, a share of Series B Preferred Stock to the Stockholder, and in exchange therefor, the Stockholder has agreed to the covenants and proxy set forth herein.

Now, therefore, in consideration of the mutual covenants and agreements contained herein and for good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, and intending to be legally bound, the parties hereto hereby agree as follows:

Section 1. Voting Covenants.

- (a) For purposes of this Agreement, "Special Action" means each proposal, act and transaction (including without limitation any amendment or restatement of the Certificate of Incorporation of the Corporation, but excluding the election of directors) submitted for stockholder approval at or in connection with any special meeting of stockholders of the Corporation (including any adjournment, postponement or rescheduling thereof) called for the purpose of (among other things) approving a transaction among the Corporation, Nxu Merger Sub, Inc., Mxu Merger Sub, LLC and Verde Bioresins, Inc.
- (b) The Holder agrees that such Holder shall cast all of the votes to which he or she is entitled by reason of holding the Series B Preferred Stock:
 - "For" a Special Action if the votes cast "for" such Special Action by the holders of Class A Common Stock and Class B Common Stock exceed the votes cast "against" such Special Action by the holders of Class A Common Stock and Class B Common Stock; and
 - "Against" a Special Action if the votes cast "against" such Special Action by the holders of Class A Common Stock and Class B Common Stock exceed the votes cast "for" such Special Action by the holders of Class A Common Stock and Class B Common Stock.

For the avoidance of doubt, the following shall not constitute votes cast "for" or "against": (x) abstentions, (y) broker non-votes and (z) shares that are not present in person or by proxy at a stockholder meeting.

- (c) With respect to the election of directors, the Holder may cast 100 of the votes to which he or she is entitled by reason of holding the Series B Preferred Stock in his or her sole discretion.
- (d) With respect to the balance of the votes on an election of directors and all other matters submitted for action by the stockholders of the Corporation that are not Special Actions, the Holder agrees that such Holder shall cast all of the votes to which he or she is entitled by reason of holding the Series B Preferred Stock in a manner that is proportionate to the manner in which all shares of Class A Common Stock and Class B Common Stock of the Corporation are voted with respect to such matters and nominees, such that, for any matter or nominee, the votes to which the Holder is entitled by reason of holding the Series B Preferred Stock shall reflect voting results with respect to "shares voted for" "shared voted against" "shares abstained" "shares withheld" "broker non-votes" and "shares not present at the meeting" proportionate to such aggregate voting results for shares of Class A Common Stock and Class B Common Stock.

Section 2. *Irrevocable Proxy.* The Holder hereby irrevocably grants to, and appoints, each director of the Corporation (each of whom may act alone) as such Holder's proxy and attorney-in-fact (with full power of substitution), for and in the name, place and stead of the Holder to vote the Series B Preferred Stock as required herein, if the Holder purports to fail at any time to vote as required herein. The Holder hereby acknowledges that the proxy granted herein is coupled with an interest and is irrevocable.

Section 3. *Term.* The voting covenants and irrevocable proxy set forth in this Agreement shall be effective until the redemption of the Series B Preferred Stock, even if such covenants and proxy remain effective for more than three years.

Section 4. *Other shares owned by Holder.* For the avoidance of doubt, the provisions of this Agreement shall apply solely with respect to the Series B Preferred Stock and the voting power associated therewith and shall not apply to any other class or series of capital stock of the Corporation held by the Holder.

Section 5. No Certificate. The Series B Preferred Stock shall be uncertificated.

Section 6. *Ownership and Transfer.* The Holder may not directly or indirectly transfer the Series B Preferred Stock to any other person or entity without the prior approval of the Board of Directors (acting in its sole discretion). Any attempted transfer in violation of this paragraph shall be void and of no force or effect.

Section 7. *Governing Law.* This Agreement shall be governed by and construed and interpreted in accordance with the laws of the State of Delaware, regardless of any laws or legal principles that might otherwise govern under the applicable principles of conflicts of law thereof.

Section 8. *Severability.* The provisions of this Agreement are severable, so that the invalidity or unenforceability of any provision of this Agreement shall not affect the validity or enforceability of any other provision of this Agreement.

Section 9. *Forum.* The Holder irrevocably consents to the personal jurisdiction of the Delaware Court of Chancery with respect to any dispute arising out of, relating to, or in connection with this Agreement and HEREBY WAIVES ANY RIGHT TO TRIAL BY JURY WITH RESPECT TO ANY SUCH DISPUTE. The Holder agrees to mailing of process or other papers to the Holder's address as it appears on the records of the Corporation.

Section 10. Amendment. Amendments to this Agreement shall be made only upon written approval by the Holder and the Corporation.

[Signature Pages Follow]

In witness whereof, each party hereto has caused this Agreement to be executed as of the date first written above.

NXU, INC., a Delaware corporation

By: /s/ Annie Pratt

Name: Annie Pratt Title: President

STOCKHOLDER:

MARK HANCHETT, an individual

/s/ Mark Hanchett

[Signature Page to Series B Preferred Stock Voting Agreement]

FORM OF AMENDMENT TO EMPLOYMENT AGREEMENT FOR [____]

This First Amendment to the Employment Agreement by and between [____] ("Executive") and NXU, Inc. (the "Company") is effective as of October 23, 2024 (the "Amendment Date"). Executive and the Company are parties to an Employment Agreement effective as of May 12, 2023. By this instrument, the Company and the Executive desire to amend the Employment Agreement as set forth below.

1. The last sentence of Section 1(b) (<u>Retention and Duties</u>) is hereby amended and restated in its entirety to read as follows:

This position will be based remotely and in Mesa, AZ, and Executive may be required to travel in fulfillment of Executive's duties and responsibilities hereunder.

2. Section 1(c)(iii) (<u>Retention and Duties</u>) is hereby amended and restated in its entirety to read as follows:

(iii) hold no other employment, unless such employment does not interfere or conflict with the Executive's duties hereunder; provided that the foregoing shall not prevent Executive from continuing to participate in such charitable, civic, educational, professional, community or industry affairs as long as such activities do not, either individually or in the aggregate, interfere or conflict with Executive's duties hereunder.

3. A new Section 3(e) shall be inserted as follows:

Financing Bonus. The Company intends to enter into an Agreement and Plan of Merger, by and among the Company, (e) Verde Bioresins, Inc., NXU Merger Sub, Inc. and NXU Merger Sub, LLC (the "Merger Agreement"). Subject to and following the closing of the transactions contemplated by the Merger Agreement (the "Transaction") and before December 31, 2025, the Company intends to attempt to obtain financing through the sale, in one or more transactions, of Company equity securities to one or more third parties that are unaffiliated with both the Company and Verde Bioresins, Inc. (each such transaction, a "Post-Closing Financing"). Provided Executive was employed by the Company on the date of the closing of the Transaction, within thirty (30) days of the date on which the Company raises at least \$5 million as a result of one or more Post-Closing Financings, the Company shall pay to Executive a lump sum payment in an amount equal to (a) a fraction (i) the numerator of which is the amount raised by the Company in such Post-Closing Financings (provided that such amount shall be deemed not to exceed \$10 million) and (ii) the denominator of which is \$10 million, times (b) six months of Executive's base salary as of the date of the closing of the Transaction (the "Financing Bonus"). In the event that the Company subsequently closes any additional Post-Closing Financings prior to December 31, 2025, the Financing Bonus shall be increased by an amount equal to the excess of (A) an amount equal to (1) a fraction (x) the numerator of which is the cumulative amount raised by the Company in all Post-Closing Financings (provided that such amount shall be deemed not to exceed \$10 million) and (y) the denominator of which is \$10 million times (2) six months of Executive's base salary as of the date of the closing of the Transaction over (B) the aggregate Financing Bonus that has been previously paid to Executive. For the avoidance of doubt, (i) Executive is not required to remain employed following the closing of the Transaction to receive payment of the Financing Bonus; (ii) eligibility for the Financing Bonus is subject to Executive having timely executed and not revoked the general release substantially in the form attached hereto as Exhibit A in connection with any termination of employment; (iii) the Financing Bonus shall be made, if at all, during the 2025 calendar year; (iv) if the Company is unable to raise at least \$5 million through one or more Post-Closing Financings in 2025, the Financing Bonus shall not be paid and Executive shall have no further rights with respect thereto and (v) in no event will the aggregate Financing Bonus payable to the Executive exceed an amount equal to six months of Executive's base salary as of the date of the closing of the Transaction.



4. Section 5(c) (<u>Termination of Employment – Benefits upon Termination</u>) is hereby amended and restated in its entirety to read as follows:

(c) **Benefits upon Termination**. If Executive's employment with the Company is terminated during the Period of Employment for any reason by the Company or by Executive, the Company shall have no further obligation to make or provide to Executive, and Executive shall have no further right to receive or obtain from the Company, any payments or benefits except as follows:

(i) The Company shall pay Executive (or, in the event of Executive's death, Executive's estate) any Accrued Obligations (as defined below) within the thirty (30) day period following the date Executive's employment terminates (the "Separation Date"), or such earlier date as may be required by applicable law, and Executive shall receive any vested accrued benefits for which Executive remains eligible under the Company's employee welfare benefit and retirement plans, payable according to the terms of such plans.

Only following a termination of the Merger Agreement in accordance with its terms which results in the abandonment of (ii) the transactions contemplated thereby, if, during the Period of Employment, Executive's employment with the Company ends as a result of an involuntary termination by the Company without Cause, or Executive's resignation for Good Reason, then, in addition to the amounts payable under Section 5(c)(i), subject to Executive's timely execution and non-revocation of the general release described in Section 5(d) (the "General Release") and the other conditions and limitations herein, Executive shall be eligible to receive (A) a lump sum equal to six (6) months of Base Salary, subject to all applicable taxes and withholdings, payable as of the first payroll date following the expiration of the applicable revocation period for the General Release; (B) Equity Compensation continuation by which all Equity Compensation due Executive, as outlined in Exhibit C, shall continue to vest according to the original vesting schedule during the Severance Period, subject to Executive's compliance with the terms of this Agreement and the applicable award agreements; and (C) in the event that Executive and Executive's dependents are eligible for and timely elect medical and dental continuation coverage under the Consolidated Omnibus Budget Reconciliation Act of 1985, as amended ("COBRA"), a monthly cash payment equal to the monthly premium costs for medical and dental COBRA continuation coverage at the rate of the Company's normal contribution for active employees at the Executive's coverage level as in effect immediately prior to the Executive's termination for the period starting on Executive's Separation Date and ending on the earliest of (1) the last day of the twelve (12) month period following the Separation Date (the "Severance Period"), (2) Executive (or Executive's dependents) becoming eligible for medical or dental (as applicable) benefits from a subsequent employer or (3) Executive (or Executive's dependents) otherwise becoming ineligible for COBRA continuation coverage; provided that such COBRA payments do not result in a violation of applicable law by, or in the imposition of penalties, fines, or excise taxes to, the Company ((A) through (C) herein, the "Severance Payments"); provided, that no installment or portion of the Severance Payments shall be payable or paid until the expiration of the applicable revocation period for the General Release. For the avoidance of doubt, Executive shall have no rights to any Severance Payments while the Merger Agreement is in effect or from and after the consummation of the Transaction.

(iii) Notwithstanding anything herein to the contrary, if any Severance Payment is "nonqualified deferred compensation" within the meaning of Section 409A of the Code (as defined below) and the period to consider the General Release and, if applicable, revoke the General Release *plus* the first regular payroll date thereafter spans two calendar years, then no portion of the Severance Payments shall be paid until the Company's first payroll payment date in the year following the year in which the Separation Date occurs, and any amount that is not paid prior to such date due to such restriction shall be paid in a lump sum along with the installment scheduled to be paid on that date.

5. Section 5(d)(i) (<u>Termination of Employment – Release; Exclusive Remedy; Cooperation Upon Termination</u>) is hereby amended to delete "and (iii)" in the second sentence and last sentence of the section.

6. A new sentence shall be added to the end of Section 5(f)(v) (Termination of Employment – Certain Defined Terms; "Good Reason") as follows:

Notwithstanding the foregoing, none of (x) the execution of the First Amendment to the Employment Agreement, (y) any change to any of Executive's Base Salary, Executive's duties, authority or responsibilities or Executive's principal place of employment while the Merger Agreement is in effect, or (z) any of the transactions contemplated by the Merger Agreement, including the closing of the Transaction itself or Executive's resignation from her position as an officer of the Company or a member of the Board, shall be an event that constitutes Good Reason for purposes of this Agreement.

- 7. Exhibit A (Form of Agreement and General Release) to the Employment Agreement is hereby amended as set forth below:
 - a. Section 2 is hereby amended to replace the words "Employer, its respective" with "Employer, Verde Bioresins Inc., it or their respective"
 - b. Section 4 is hereby amended to begin the first two sentences with "To the extent permitted by applicable law and Sections 7(e) and (f) and 13,"
 - c. Section 5 is hereby amended and restated in its entirety to say "Reserved."
 - d. Section 9 is hereby amended and restated in its entirety to read as follows:

Executive acknowledges that if Executive breaches any of Executive's obligations under Section 6 of the Employment Agreement during the relevant restricted period, then, and not in any way in limitation of any right or remedy otherwise available to the Company, Executive will forfeit, and the Company will cease payment of, any portion of the Financing Bonus set forth in Section 3(e) that has not yet been paid, equal to up to ninety percent (90%) of the economic value of the Financing Bonus, as determined by the Board (or its delegate) in its sole discretion. If the Financing Bonus has already been paid, Executive acknowledges that the Board may request Executive pay back to the Company an amount in cash as determined by the Board in its sole discretion, up to ninety percent (90%) of the economic value of the Financing Bonus.

- e. The last sentence of Section 10 is hereby amended by deleting the words "or Caryn Nightengale at caryn.nightengale@gmail.com."
- f. Section 14 is hereby amended to replace the words "whether or not the Executive is entitled to severance" with "subject to Sections 7(e) and (f) and 13 and whether or not the Executive is entitled to the Financing Bonus or severance"
- 8. Exhibit C (Equity Compensation Addendum) to the Employment Agreement is hereby amended and restated as Appendix 1 hereto.

| EXECUTIVE: |
|------------|
|------------|

| |] | | | |
|--|--------------------|--|--|--|
| Da | te: | | | |
| | MPANY: (U, Inc. | | | |
| Na | me Printed: | | | |
| Tit | le: | | | |
| Da | te: | | | |
| [Signature Page to Amendment to [] Employment Agreement] | | | | |
| | 5 | | | |

<u>Appendix 1</u>

EXHIBIT C

EQUITY COMPENSATION ADDENDUM

For the avoidance of doubt, all Restricted Stock Units described in this Exhibit C shall be subject to adjustment upon a Change in Capitalization as defined in the Amended 2023 Omnibus Plan (as defined below) and shall, except as described below, remain subject to the terms and conditions thereof.

- On May 12, 2023, Executive was granted Base equity compensation ("Base Equity") of \$75,000 in Restricted Stock Units ("RSUs") each quarter. Of this original Base Equity grant, Executive has received RSUs with an aggregate vesting date value of \$150,000, representing the Base Equity that vested on June 30, 2023 and September 30, 2023. The Board and the Executive previously agreed to extend the vesting dates for all of the Executive's RSUs, including the Base Equity RSUs that would otherwise have vested on and between October 31, 2023 and December 31, 2024 until January 31, 2025 (the "Delayed Vesting"). The Delayed Vesting was the result of (1) the prohibitively high transaction fees associated with sell-to-cover ("STC") tax treatment required when the RSUs vest, which continues to jeopardize the Company's ability to continue as a going concern and (2) the Company failing to have sufficient shares available for issuance pursuant to the NXU, Inc. 2023 Incentive Omnibus Plan (the "2023 Omnibus Plan"). Effective as of August 14, 2024, the Company's shareholders approved an amendment and restatement of the 2023 Omnibus Plan (the "Amended 2023 Omnibus Plan") to, among other things, authorize additional shares for current and future equity awards. The Company and Executive hereby affirm that Executive is entitled to \$75,000 in RSUs for each complete calendar guarter between October 1, 2023 and September 30, 2024, totaling \$300,000. The Company and Executive hereby desire to set the number of the RSUs due under the Base Equity award through the September 30, 2024 original vesting date. The number of RSUs shall be calculated by dividing \$300,000 by the closing share price August 14, 2024 (\$0.3399) (the "Calculation Date"), resulting in an award of 882,612 RSUs. The Executive shall not be entitled to any RSUs for the vesting periods after September 30, 2024. The Company has determined that the execution of the Merger Agreement will enable it to deliver a portion of the RSUs subject to Delayed Vesting without jeopardizing its ability to continue as a going concern. Following the execution of the Merger Agreement, the Company will deliver the RSUs to the Participant in installments in such amounts as the Company determines may be delivered to the Executive without jeopardizing the Company's ability to continue as a going concern. Such RSUs shall be delivered until the earlier of (A) the date all the RSUs subject to Delayed Vesting have been delivered to the Participant or (B) the date that is no later than 5 business days prior to the closing of the Transaction (such earlier date, the "Cut-Off Date"). Any RSUs that have not been delivered to the Participant as of the Cut-Off Date shall be forfeited for no consideration.
- Executive has been eligible to receive a short term equity incentive plan bonuses ("<u>STIP Bonus</u>") of up to \$60,000 in RSUs each quarter if certain pre-established short term goals were met. The Executive has received STIP Bonuses equal to \$135,000. To date, \$12,000 of the STIP Bonus has been delivered. As with the RSUs described above, the Company and the Executive agreed to delay the delivery of the remaining STIP Bonus payments due to it jeopardizing the Company's ability to continue as a going concern. The Company and Executive hereby agree that Executive is entitled to 361,871 RSUs under the previously earned STIP Bonuses (\$123,000/share price on the Calculation Date). As with the RSUs described above, following the execution of the Merger Agreement, the Company will deliver the RSUs to the Participant in installments in such amounts as the Company determines may be delivered to the Executive without jeopardizing the Company's ability to continue as a going concern. Such RSUs shall be delivered until the Cut-Off Date. Any RSUs that have not been delivered to the Participant as of the Cut-Off Date shall be forfeited for no consideration. Executive is not entitled to any additional STIP Bonus for any calendar quartering beginning after June 30, 2024.

C-1

- Long term equity incentive plan payment: None.
- Effective as of the Amendment Date, Executive holds stock options under the Stock Option Plan of Atlis Motor Vehicles, Inc. (the "Atlis Plan"). All of the stock options have a per share exercise price in excess of the current fair market value of the Company's stock as of the Amendment Date. Executive hereby agrees to forfeit all outstanding stock options, whether vested or unvested, for no consideration immediately prior to a Change in Control, provided the per share exercise price, as adjusted to reflect any changes in the Company's capitalization between the Amendment Date and the date on which a Change in Control occurs, remains in excess of the Company's closing stock price on the trading day immediately preceding the date on which the Change in Control occurs.
- On May 12, 2023, Executive received an Executive High Impact Award ("High Impact Award") of \$1,450,000, vesting over thirty-six (36) months, to reflect Executive's considerable efforts and sacrifices in growing the Company from its infancy and facilitating the Company's public listing on Nasdaq. Executive has received RSUs with an aggregate vesting date value of \$161,111, representing the High Impact Award that vested between June 30, 2023 through September 30, 2023. As with the RSUs described above, the Board and Executive previously agreed to amend the vesting dates for all the RSUs as the delivery of the RSUs would jeopardize the Company's ability to continue as a going concern. The Company and Executive hereby desire to set the number of the RSUs due under the High Impact Award through the September 30, 2024 vesting date. The number of RSUs shall be calculated by dividing \$483,333.36 (the value of the High Impact Award that would have vested on and between October 1, 2023 and September 30, 2024) by the closing share price on the Calculation Date, resulting in an award of 1,421,986 RSUs. The Executive shall not be entitled to any additional RSUs under the High Impact Award for the vesting periods after September 30, 2024. Following the execution of the Merger Agreement, the Company will deliver the RSUs to the Participant in installments in such amounts as the Company determines may be delivered to the Executive without jeopardizing the Company's ability to continue as a going concern. Such RSUs shall be delivered until the Cut-Off Date. Any RSUs that have not been delivered to the Participant as of the Cut-Off Date shall be forfeited for no consideration.

Notwithstanding anything to the contrary in this Exhibit C, in no event shall any RSUs (whether pursuant to a Base Equity award, an STIP Bonus, a High Impact Award or otherwise) be settled pursuant to this Exhibit C unless the Company has provided to counsel to Verde Bioresins, Inc. ("Counsel"), on the business day immediately preceding the date of proposed settlement, an updated analysis under Section 280G of the Code in substantially the form provided to Counsel immediately prior to the date of the Merger Agreement, which reflects the approximate effect of the settlement of the RSUs to be settled on such date, and which revised analysis Counsel has confirmed is acceptable (which confirmation shall not be unreasonably withheld, conditioned or delayed). For the avoidance of doubt, any updated analysis under Section 280G of the Code provided to Counsel shall treat settled RSUs as compensatory payments that are contingent on a change of control in the ownership or effective control of the Company within the meaning of Section 280G of the Code, shall assume that the Financing Bonus shall be paid in full, and shall not reduce the value of any compensatory payments payable to the Executive by a valuation of services rendered or to be rendered by the Executive. It is understood and agreed that no RSUs shall be settled to the extent such settlement would cause the Executive to be subject to the excise taxes imposed by Section 4999 based on the analysis described in this paragraph; provided, however, that at the Company and Executive's request, Verde Bio resins, Inc. may consider permitting such RSUs to be settled upon receipt of satisfactory evidence, which shall be determined in Counsel's sole discretion, that the Executive has remitted to an escrow account established by the Company for this purpose, the amount(s) necessary to cover the Company's tax withholding obligations under Section 4999 upon the closing of the Transaction.

C-2

The amendment to this Exhibit C supersedes and replaces any award agreement between the Company and Executive to the extent the terms of such award agreement are inconsistent with this Exhibit C.



October 23, 2024

Re: Employment Offer Letter

Dear Ms. Wyant,

You ("you" or "Executive") have been Acting Chief Financial Officer for Nxu, Inc ("Nxu" or the "Company") since December 5, 2023. After reviewing your performance, we would like to remove the "Acting" designation and make you the Company's Chief Financial Officer. You are hereby offered the position of Chief Financial Officer of Nxu. Please accept this communication as an offer letter and an agreement on the basic terms of your employment relationship ("Offer Letter"). This Offer Letter is subject to Nxu's employment policies and procedures as they relate to all of Nxu's employees, as well as any other documents you must execute before your employment, as listed herein. You will continue to manage the Finance and Accounting teams and will report directly to Nxu's Chief Executive Officer, Mark Hanchett ("CEO").

1. **Duties**. Although subject to change based on management and decisions at the discretion of the CEO, your initial tasks will include, but are not limited to:

Planning:

- Monitor and direct the implementation of strategic business plans, specifically related to the Company's pursuit of a business combination
- · Support tactical initiatives to reduce liabilities, collect receivables and sell Company assets
- · Manage, implement, and oversee the Company's cash position and forecasts of projected cash inflows and outflows

Operations:

- Meaningfully and reasonably participate in key decisions as a member of Company's executive team, as directed
- Maintain quality and professional relationships with all members of the management team and teams you supervise
- · Manage the accounting, finance, and tax functions of the company, as assigned
- · Oversee Company's transaction processing systems
- · Identify and implement operational best practices within the finance and accounting functional areas
- · Assist with acquisitions and related due diligence and negotiations, as directed

Financial Information:

- · Oversee the issuance of required financial information and reporting
- Personally review, amend, and approve all Form 8-K, 10-K, and 10-Q filings with the Securities and Exchange Commission

1

· Report financial results to the Board of Directors

1828 N Higley Rd. Ste. 116 Mesa, AZ 85205

nxuenergy.com

Risk Management:

- Understand and help to mitigate key elements of Company's risk profile
- · Construct and monitor reliable control systems
- · Maintain appropriate insurance coverage
- · Ensure that record keeping meets the requirements of auditors and government agencies
- · Report risk issues to the audit committee of the Board of Directors
- · Maintain relations with external auditors and lead the investigation of their findings and recommendations Funding:
- · Monitor cash balances and cash forecasts
- Invest funds

Third Parties:

· Maintain banking relationships

You will also be responsible for compliance with existing Company policies and procedures. All Company policies and procedures are subject to change and you will be notified of any approved changes in writing. Once changes are approved and communicated to you, you will be responsible for your compliance with all such policies and procedures, as amended.

2. Compensation and Reimbursement.

(a) **Compensation**. Effective as of the date of this Agreement, your initial compensation package includes base annual salary of \$300,000, payable in accordance with the Company's regular payroll practices and subject to all legal withholdings and deductions.

(b) Equity Compensation. The equity compensation to which you are entitled is set forth on Exhibit A (collectively "Equity Compensation").

(c) **Reimbursement of Business Expenses**. You are authorized to incur reasonable expenses in carrying out your duties hereunder and shall, upon receipt by the Company of proper documentation with respect thereto (setting forth the amount, business purpose and establishing payment) be reimbursed for all such business expenses, subject to the Company's written expense reimbursement policies, employee travel policies and any written pre-approval policies in effect from time to time.

(d) **Clawback Policy**. To the extent required by applicable law or regulation, any applicable stock exchange listing standards or any clawback policy, adopted by the Company pursuant to any such law, regulation or stock exchange listing standards, or to comport with good corporate governance practices, any incentive compensation granted to you (whether pursuant to this Agreement or otherwise) shall be subject to the provisions of any applicable clawback policies or procedures, which may provide for forfeiture and/or recoupment of such amounts paid or payable under this Agreement or otherwise, including incentive equity awards granted to you.

1828 N Higley Rd. Ste. 116 Mesa, AZ 85205

nxuenergy.com





Financing Bonus. The Company intends to enter into an Agreement and Plan of Merger, by and among the Company, Verde Bioresins, Inc., (e) NXU Merger Sub, Inc. and NXU Merger Sub, LLC (the "Merger Agreement"). Subject to and following the closing of the transactions contemplated by the Merger Agreement (the "Transaction") and before December 31, 2025, the Company intends to attempt to obtain financing through the sale, in one or more transactions, of Company equity securities to one or more third parties that are unaffiliated with both the Company and Verde Bioresins, Inc. (each such transaction, a "Post-Closing Financing"). Provided you were employed by the Company on the date of the closing of the Transaction, within thirty (30) days of the date on which the Company raises at least \$5 million as a result of one or more Post-Closing Financings, the Company shall pay to Executive a lump sum payment in an amount equal to (a) a fraction (i) the numerator of which is the amount raised by the Company in such Post-Closing Financings (provided that such amount shall be deemed not to exceed \$10 million) and (ii) the denominator of which is \$10 million, times (b) six months of Executive's base salary as of the date of the closing of the Transaction (the "Financing Bonus"). In the event that the Company subsequently closes any additional Post-Closing Financings prior to December 31, 2025, the Financing Bonus shall be increased by an amount equal to the excess of (A) an amount equal to (1) a fraction (x) the numerator of which is the cumulative amount raised by the Company in all Post-Closing Financings (provided that such amount shall be deemed not to exceed \$10 million) and (y) the denominator of which is \$10 million times (2) six months of Executive's base salary as of the date of the closing of the Transaction over (B) the aggregate Financing Bonus that has been previously paid to Executive. For the avoidance of doubt, (i) Executive is not required to remain employed following the closing of the Transaction to receive payment of the Financing Bonus; (ii) eligibility for the Financing Bonus is subject to Executive having timely executed and not revoked the general release substantially in the form attached hereto as Exhibit B in connection with any termination of employment; (iii) the Financing Bonus shall be made, if at all, during the 2025 calendar year; (iv) if the Company is unable to raise at least \$5 million through one or more Post-Closing Financings in 2025, the Financing Bonus shall not be paid and Executive shall have no further rights with respect thereto and (v) in no event will the aggregate Financing Bonus payable to the Executive exceed an amount equal to six months of Executive's base salary as of the date of the closing of the Transaction.

3. Executive Benefits.

You shall be provided the opportunity to participate in all standard employee benefit programs made available by the Company to the Company's employees generally, in accordance with the eligibility and participation provisions of such plans and as such plans or programs may be in effect from time to time. The Company reserves the right to amend any employee benefit plan, policy, program or arrangement from time to time, or to terminate such plan, policy, program or arrangement, consistent with the terms thereof at any time and for any reason without providing you with notice.

You shall be eligible to receive paid vacation in accordance with and subject to the Company's vacation policies in effect from time to time.

4. Termination of Employment.

<u>4.1</u> <u>Termination Due to Death</u>. If Executive shall die during the Term, this Agreement shall terminate, except for payment of Base Salary, Equity Compensation, Bonus Compensation and other amounts as otherwise provided in Section 2 of this Agreement. No severance benefits shall be due to Executive's estate in connection with such termination.

4.2 <u>Termination at the Option of Executive</u>. Executive may resign and terminate her employment without further liability upon thirty (30) days prior written notice to the Company.

4.3 Termination by Executive for Good Reason. For the purposes of this Section 4.3, termination of this Agreement by Executive for "Good Reason" means the termination of this Agreement initiated by Executive resulting from any of the following events without Executive's express written consent which is not cured by the Company within thirty (30) days of the date of the written notice thereof from Executive to the Company, provided that such written notice must be provided to the Company by the Executive within thirty (30) days of the event occurring and Executive must terminate employment within 30 days of the end of the Company's cure period: (i) a material diminution in Executive's base salary (except for any across-the-board reduction impacting substantially all executives of the Company of not more than 10%); (ii) the assignment to Executive of any duties that would result in Executive's violation of applicable law or regulation, including the AICPA Code of Conduct; (iii) any action by Company that results in a material diminishment in Executive's authority, duties or responsibilities; or (iv) the material breach by the Company of this Agreement. Notwithstanding the foregoing, neither (y) any change to any of Executive's base salary or Executive's duties, authority or responsibilities or Executive's principal place of employment while the Merger Agreement is in effect, nor (z) any of the transactions contemplated by the Merger Agreement, including the closing of the Transaction itself or Executive's resignation from her position as an officer of the Company or a member of the Board, shall be an event that constitutes Good Reason for purposes of this Agreement.

1828 N Higley Rd. Ste. 116 Mesa, AZ 85205

nxuenergy.com

D 🞯 f Y in



Ø

G

Following a termination of the Merger Agreement in accordance with its terms which results in the abandonment of the transactions contemplated thereby, if this Agreement is terminated by Executive for Good Reason, in addition to all other amounts payable to Executive hereunder, Executive shall receive severance payments provided in Section 4.5 of this Agreement in the same manner as if the Company had terminated Executive's employment without Cause.

<u>4.4</u> <u>Termination by the Company for Cause</u>. The Company may terminate Executive's employment hereunder upon the occurrence of one or more of the following events listed in subsection (a) through (e) below ("<u>Cause</u>") and no payments, severance or other benefits shall be due to Executive in connection with such termination for Cause other than accrued and unpaid Base Salary, Equity Compensation, and Bonus Compensation due, if any, up to the Date of Termination (as defined below):

(a) Executive commits fraud, embezzlement, or an act of material willful misconduct;

(b) Executive is convicted of a felony or of any crime or offense involving moral turpitude or any property of the Company or its Affiliates (as defined below) or any other crime that reflects adversely on the Company;

(c) Executive commits an act of gross negligence in carrying out her duties under this Agreement (for purposes of this Agreement, "gross negligence" means a higher level of negligence representing an extreme departure from the ordinary standard of care);

(d) Executive breaches any material provision of this Agreement or any other material provision of any other agreement between Executive and the Company (or its Affiliates) or acts other than in accordance with the reasonable directions of the Board (which are consistent with this Agreement and Executive's position with the Company) and, if curable, Executive fails to cure the breach or failure within thirty (30) days of the Company's notice thereof to Executive; or

(e) Executive is unable to perform Executive's job responsibilities in all material respects as a result of a physical or mental illness, injury or condition that persists for more than six (6) months (whether or not consecutive) ("<u>Disability</u>").

4.5 Termination by the Company Without Cause. Only following a termination of the Merger Agreement in accordance with its terms which results in the abandonment of the transactions contemplated thereby, if the Company terminates Executive's employment hereunder (other than as a result of Executive's death, Disability or for Cause) (i) Executive shall be entitled to receive, as severance, any amount accrued and payable for services provided under normal employment as provided in Section 2, (ii) a lump sum of \$150,000, less applicable taxes and withholding, payable in one lump sum on the next regularly scheduled payroll date following the expiration of the release revocation period described in Section 4.6, and (iii) Equity Compensation as outlined on Exhibit A shall continue to be paid to Executive in accordance with Exhibit A. For the avoidance of doubt, Executive shall have no rights to any severance benefits under this Section 4 while the Merger Agreement is in effect or from and after the consummation of the Transaction.

1828 N Higley Rd. Ste. 116 Mesa, AZ 85205

nxuenergy.com





<u>4.6</u> Release. As a condition of Executive's receiving the severance in Section 4.3 or 4.5 or the Financing Bonus in Section 2(e), the Executive shall provide the Company with a valid, executed General Release in the form substantially attached hereto as Exhibit B, and not revoke such General Release prior to the expiration of any revocation rights afforded to Executive by applicable law. The Company shall provide Executive with the General Release on or within ten (10) days following the Separation Date, and Executive must deliver the executed General Release to the Company within twenty-one (21) days (or, if greater, the minimum period required by applicable law for a fully valid release) after the later of the Separation Date or the receipt of the release in substantially final form. If Executive does not return the General Release within this time period, Executive shall not be entitled to the benefits provided under Sections 2(e), 4.3 or 4.5 of this Agreement.

5. Acknowledgement.

You acknowledge that the compensation, including, but not limited to the offer, the specialized training, and the Confidential Information provided to you pursuant to your employment with Nxu give rise to Nxu's interest in restraining you from violating any restrictive covenants by which you are bound, and that said covenants are designed to enforce such consideration and that any limitations as to time, geographic scope and scope of activity to be restrained as defined herein are reasonable and do not impose a greater restraint than is necessary to protect the goodwill or other business interest of Nxu. Nothing in this offer supersedes or nullifies any portion or provision in Nxu's Employee Covenants Agreement and/or Nxu's policies and procedures. The terms of this offer shall be read in cooperation and conjunction with any such policies and procedures.

In accepting this offer, you also authorize the Company to share with the prospective buyer such personal employment data as would reasonably allow the buyer to evaluate your contribution to the business and future role.

6. Confidentiality.

You agree that during your employment, you shall not, and shall cause your affiliates and representatives not to, disclose to any individual or natural person, partnership (including a limited liability partnership), corporation, limited liability company, association, joint stock company, trust, joint venture, unincorporated organization or governmental authority (each, a "Person"), except (i) to the your legal, financial, tax or accounting advisors, or (ii) as compelled by law, any Confidential Information for any reason or purpose whatsoever, and you shall not, and shall cause your affiliates or representatives not to, make use of any of the Confidential Information for their own purposes or for the benefit of any Person except the Company or its affiliates or subsidiaries (the "Company Group" and each a "Company Group Member"). In the event that you or any of your affiliates or representatives are compelled by law to disclose any Confidential Information, you shall promptly provide written notice to the Company of the request or requirement so that the Company may seek (at the Company's sole cost and expense) an appropriate protective order or waive compliance with the provisions of this section. If, in the absence of a protective order or the receipt of a waiver hereunder, you or any of your affiliates or representatives are compelled by law to disclose any tribunal, you or your affiliates, as applicable, may disclose the Confidential Information to the tribunal; provided, that you or your affiliates, as applicable, shall use commercially reasonable efforts to obtain, at the request and sole expense of the Company, an order or other assurance that confidential treatment shall be accorded to such portion of the Confidential Information required to be disclosed as the Company shall designate.

1828 N Higley Rd. Ste. 116 Mesa, AZ 85205

nxuenergy.com

□ ◎ f y in

5



ര

For purposes of this Agreement, "<u>Confidential Information</u>" shall mean any information of a confidential or proprietary nature concerning the business or affairs of the Company; provided however, "<u>Confidential Information</u>" shall not include information which (i) is or becomes generally available to the public other than as a result of disclosure by you to the receiving party in violation of this Agreement or (ii) becomes available to you or your affiliates on a non-confidential basis, so long as such source is not known by you to be bound by a confidentiality agreement with the Company prohibiting such disclosure.

Nothing in this Agreement or otherwise shall be construed to limit Executive's ability to file a charge or complaint with the EEOC, the National Labor Relations Board, the Occupational Safety and Health Administration, the Securities and Exchange Commission or any other federal, state or local governmental agency or commission ("Government Agencies" or limit Executive's ability to communicate with any Government Agencies or otherwise participate in any investigation or proceeding that may be conducted by any Government Agency, including providing documents or other information, without notice to the Company. Any non-disclosure provision in this Agreement or other policy or agreement from the Company does not prohibit or restrict Executive (or Executive's attorney) from responding to any inquiry about this Agreement by the Financial Industry Regulatory Authority, any other self-regulatory organizations or Government Agencies, or making other disclosures that are protected under the whistleblower provisions of federal law or regulation. Executive understands and acknowledges that Executive does not need the prior authorization of the Company to make any such reports or disclosures and that Executive is not required to notify the Company that Executive has made such reports or disclosures. Further, notwithstanding Executive's confidentiality and nondisclosure obligations, Executive is hereby advised as follows pursuant to the Defend Trade Secrets Act: "An individual shall not be held criminally or civilly liable under any Federal or State trade secret law for the disclosure of a trade secret that (A) is made (i) in confidence to a Federal, State, or local government official, either directly or indirectly, or to an attorney; and (ii) solely for the purpose of reporting or investigating a suspected violation of law; or (B) is made in a complaint or other document filed in a lawsuit or other proceeding, if such filing is made under seal. An individual who files a lawsuit for retaliation by an employer for reporting a suspected violation of law may disclose the trade secret to the attorney of the individual and use the trade secret information in the court proceeding, if the individual (A) files any document containing the trade secret under seal; and (B) does not disclose the trade secret, except pursuant to court order."

7. Assignment.

This Agreement cannot be assigned by the Company without your prior written consent, other than (a) an assignment to an affiliate of the Company made as part of an internal reorganization applicable to employees of the Company generally or (b) to any successor or assign (whether direct or indirect, by purchase, merger, consolidation or otherwise) to all or substantially all of the business or assets of the Company, and shall be binding and inure to the benefit of the Company, its successors and any permitted assigns. No right, obligation or duty or duty of this Agreement may be assigned by you without the prior written consent of the Company.

8. <u>Counterparts</u>.

This offer may be executed in any number of counterparts, each of which when so executed and delivered shall be deemed to be an original, and all of which counterparts of this offer, when taken together shall constitute one and the same instrument.

9. No Third-Party Beneficiaries.

Nothing in this offer, express or implied, is intended or will be construed to confer on any person, other than the parties to this offer, any right, remedy, or claim under or with respect to this offer.

1828 N Higley Rd. Ste. 116 Mesa, AZ 85205

nxuenergy.com





10. Notices.

All notices and other communications under this offer must be in writing and will be deemed to have been given if delivered personally, mailed by certified mail, or delivered by an overnight delivery service (with confirmation), to the parties at the addresses as a party may designate from time-to-time. Any notice or other communication will be deemed given: (a) on the date of personal delivery; (b) at the expiration of the third day after the date of deposit in the United States mail; or (c) on the date of confirmation of delivery by overnight delivery service.

11. Waiver of Jury Trial.

Each of the parties hereto hereby irrevocably waives all right to trial by jury in any action, proceeding or counterclaim arising out of or relating to this Agreement.

12. Entire Agreement; Amendment.

This Agreement embodies the entire agreement of the parties hereto respecting the matters within its scope and supersedes all prior agreements (including, without limitation, any offer letters, term sheets, consulting agreements and correspondence relating thereto), whether written or oral, that directly or indirectly bear upon the subject matter hereof, excluding any restrictive covenant agreement by and between the Parties. This Agreement may not be amended, modified or changed (in whole or in part), except by written agreement executed by both of the parties hereto.

13. Further Assurances.

Each party agrees: (a) to execute and deliver such other documents; and (b) to do and perform such other acts and things, as any other party may reasonably request, to carry out the intent and accomplish the purposes of this offer.

14. <u>Waiver</u>.

Any provision or condition of this offer may be waived at any time, in writing, by the party entitled to the benefit of such provision or condition. Waiver of any breach of any provision will not be a waiver of any succeeding breach of the provision or a waiver of the provision itself or any other provision.

15. Governing Law.

This offer will be governed by and construed in accordance with the laws of the State of Arizona, without regard to conflict of law provisions.

16. Severability.

It is the desire and intent of the parties hereto that the provisions of this Agreement be enforced to the fullest extent permissible under the laws and public policies applied in each jurisdiction in which enforcement is sought. Accordingly, if any particular provision of this Agreement shall be adjudicated by a court of competent jurisdiction to be invalid, prohibited or unenforceable under applicable law, such provision, as to such jurisdiction, shall be ineffective without invalidating the remaining provisions of this Agreement or affecting the validity or enforceability of such provision in any other jurisdiction.

1828 N Higley Rd. Ste. 116 Mesa, AZ 85205

nxuenergy.com



7



17. Headings.

Section headings in this offer are for convenience and reference only and shall not govern the interpretation of any of the provisions of this offer.

18. Attorneys' Fees and Costs.

If attorneys' fees or other costs are incurred to secure performance of any obligations hereunder, or to establish damages for the breach thereof or to obtain any other appropriate relief, whether by way of prosecution or defense, the prevailing party will be entitled to recover reasonable attorneys' fees and costs incurred in connection therewith.

Sincerely,

/s/ Mark Hanchett

Mark Hanchett, Chairman and CEO of Nxu

Employee Signature: /s/ Sarah Wyant

Date: October 23, 2024

1828 N Higley Rd. Ste. 116 Mesa, AZ 85205

nxuenergy.com

□ ◎ f y in

8

<u>EXHIBIT A</u>

EQUITY COMPENSATION ADDENDUM

For the avoidance of doubt, all Restricted Stock Units described in this Exhibit A shall be subject to adjustment upon a Change in Capitalization as defined in the Amended 2023 Omnibus Plan (as defined below) and shall, except as described below, remain subject to the terms and conditions thereof.

- On August 21, 2023, Executive received a grant of \$190,000 in Restricted Stock Units ("RSUs"). Pursuant to the RSU Exchange Program, on February 23, 2024, \$47,500 of Executive's RSUs were exchanged for nonqualified stock options. Executive has \$142,500 in RSUs outstanding, of which \$11,875 were vested as of September 30, 2024. The Company and Executive hereby desire to set the number of the RSUs due under the Base Equity award through the September 30, 2024 original vesting date. The number of RSUs shall be calculated by dividing \$11,875 by the closing share price August 14, 2024 (\$0.3399) (the "Calculation Date"), resulting in an award of 34,936 RSUs. The Executive shall not be entitled to any RSUs for the vesting periods after September 30, 2024. The Company has determined that the execution of the Merger Agreement will enable it to deliver a portion of the RSUs subject to Delayed Vesting without jeopardizing its ability to continue as a going concern. Following the execution of the Merger Agreement, the Company will deliver the RSUs to the Participant in installments in such amounts as the Company determines may be delivered to the Executive without jeopardizing the Company's ability to continue as a going concern. Such RSUs shall be delivered until the earlier of (A) the date all the RSUs subject to Delayed Vesting have been delivered to the Participant or (B) the date that is no later than 5 business days prior to the closing of the Transaction (such earlier date, the "Cut-Off Date"). Any RSUs that have not been delivered to the Participant as of the Cut-Off Date shall be forfeited for no consideration.
- Effective as of the Amendment Date, Executive holds stock options under the Amended 2023 Omnibus Plan. All of the stock options have a per share exercise price in excess of the current fair market value of the Company's stock as of the Amendment Date. Executive hereby agrees to forfeit all outstanding stock options, whether vested or unvested, for no consideration immediately prior to a Change in Control, provided the per share exercise price, as adjusted to reflect any changes in the Company's capitalization between the Amendment Date and the date on which a Change in Control occurs, remains in excess of the Company's closing stock price on the trading day immediately preceding the date on which the Change in Control occurs.



Notwithstanding anything to the contrary in this Exhibit A, in no event shall any RSUs (whether pursuant to a Base Equity award, an STIP Bonus, a High Impact Award or otherwise) be settled pursuant to this Exhibit A unless the Company has provided to counsel to Verde Bioresins, Inc. ("Counsel"), on the business day immediately preceding the date of proposed settlement, an updated analysis under Section 280G of the Code in substantially the form provided to Counsel immediately prior to the date of the Merger Agreement, which reflects the approximate effect of the settlement of the RSUs to be settled on such date, and which revised analysis Counsel has confirmed is acceptable (which confirmation shall not be unreasonably withheld, conditioned or delayed). For the avoidance of doubt, any updated analysis under Section 280G of the Code provided to Counsel shall treat settled RSUs as compensatory payments that are contingent on a change of control in the ownership or effective control of the Company within the meaning of Section 280G of the Code, shall assume that the Financing Bonus shall be paid in full, and shall not reduce the value of any compensatory payments payable to the Executive by a valuation of services rendered or to be rendered by the Executive. It is understood and agreed that no RSUs shall be settled to the extent such settlement would cause the Executive to be subject to the excise taxes imposed by Section 4999 based on the analysis described in this paragraph; provided, however, that at the Company and Executive's request, Verde Bioresins, Inc. may consider permitting such RSUs to be settled upon receipt of satisfactory evidence, which shall be determined in Counsel's sole discretion, that the Executive has remitted to an escrow account established by the Company for this purpose, the amount(s) necessary to cover the Company's tax withholding obligations under Section 4999 upon the closing of the Transaction.

The amendment to this Exhibit A supersedes and replaces any award agreement between the Company and Executive to the extent the terms of such award agreement are inconsistent with this Exhibit A.

A-2

<u>EXHIBIT B</u>

FORM OF AGREEMENT AND GENERAL RELEASE

THIS AGREEMENT AND GENERAL RELEASE (this "Agreement and General Release") is made and entered into on , 20 by and between Sarah Wyart ("Executive") and NXU, Inc. ("Employer").

WHEREAS, Executive has been employed by Employer and the parties wish to resolve all outstanding claims and disputes between them relating to such employment;

NOW, THEREFORE, in consideration of the mutual promises, covenants and agreements set forth in this Agreement and General Release, the sufficiency of which the parties acknowledge, it is agreed as follows:

- 1. In consideration for Executive's promises, covenants and agreements in this Agreement and General Release, Employer agrees to make the payments to and on behalf of Executive provided under Section 2 or 4 of the employment agreement between Executive and Employer dated October 23, 2024 (the "Employment Agreement"), in accordance with the terms and subject to the conditions of such Employment Agreement. Executive would not otherwise be entitled to such payments but for Executive's promises, covenants and agreements in this Agreement and General Release.
- 2. The parties agree that the payments described in Section 1 of this Agreement and General Release are in full, final and complete settlement of all claims Executive may have against Employer, Verde Bioresins Inc., it or their respective past and present subsidiaries and affiliates, and their respective officers, directors, owners, members, shareholders, employees, agents, advisors, consultants, insurers, attorneys, successors and/or assigns of each of the foregoing (collectively, the "Releasees") arising out of or in any way connected with Executive's employment with Employer or any of its affiliates or the termination of such employment. For the avoidance of doubt, this Agreement and General Release provides for the sole and exclusive benefits for which Executive is eligible as a result of Executive's termination of employment, and Executive shall not be eligible for any benefits under Employer's severance plan, if any, or any other agreement or arrangement providing for benefits upon a separation from service other than the Employment Agreement.
- 3. Nothing in this Agreement and General Release shall be construed as an admission of liability by Employer or any other Releasee, and Employer specifically disclaims liability to or wrongful treatment of Executive on the part of itself and all other Releasees.
- 4. To the extent permitted by applicable law and Sections 6(e) and (f) and 11, Executive agrees that Executive will not encourage or assist any person to litigate claims or file administrative charges against Employer or any other Releasee, unless required to provide testimony or documents pursuant to a lawful subpoena or other compulsory legal process, in which case Executive agrees to notify Employer immediately of Executive's receipt of such subpoena so that Employer has the opportunity to contest the same. To the extent permitted by applicable law and Sections 6(e) and (f) and 11, if any court has or assumes jurisdiction of any action against Employer or any of its affiliates on behalf of Executive, Executive will request the court withdraw from or dismiss the matter with prejudice. Executive further represents that Executive has reported to Employer in writing any and all work-related injuries that Executive has suffered or sustained during Executive's employment with Employer or its affiliates.



- 5. Executive covenants not to sue, and fully and forever releases and discharges Employer and all other Releasees from any and all legally waivable claims, liabilities, damages, demands, and causes of action or liabilities of any nature or kind, whether now known or unknown, arising out of or in any way connected with Executive's employment with Employer or any of its affiliates or the termination of such employment. This release includes but is not limited to claims arising under federal, state or local laws concerning employment discrimination, termination, retaliation and equal opportunity, including but not limited to Title VII of the Civil Rights Act of 1964, as amended, the Age Discrimination in Employment Act of 1967, as amended (the "ADEA"), the Equal Pay Act of 1963, the Americans with Disabilities Act of 1990, as amended, the Worker Adjustment and Retraining Notification Act of 1988, as amended, the Employee Retirement Income Security Act of 1974, as amended (including but not limited to fiduciary claims), claims for attorneys' fees or costs, any and all statutory or common law provisions relating to or affecting Executive's employment with Employer or its affiliates, and any and all claims in contract, tort, or premised on any other legal theory. Executive acknowledges that Executive is releasing claims based on age, race, color, sex, sexual orientation or preference, marital status, religion, national origin, citizenship, veteran status, disability and other legally protected categories. This provision is intended to constitute a general release of all of Executive's presently existing covered claims against the Releasees arising out of or in any way connected with Executive's employment with Employer or any of its affiliates or the termination of such employment, to the maximum extent permitted by law.
- Nothing in this Agreement and General Release shall be construed to: (a) waive any rights or claims of Executive that arise after Executive 6. signs this Agreement and General Release; (b) waive any rights or claims of Executive to enforce the terms of this Agreement and General Release; (c) waive any claim for worker's compensation or unemployment benefits; (d) waive any rights or claims for the provision of accrued benefits conferred to Executive or Executive's beneficiaries under the terms of Employer's medical, dental, life insurance or defined contribution retirement benefit plans; (e) limit Executive's ability to file a charge or complaint with the EEOC, the National Labor Relations Board, the Occupational Safety and Health Administration, the Securities and Exchange Commission or any other federal, state or local governmental agency or commission ("Government Agencies"); (f) limit Executive's ability to communicate with any Government Agencies or otherwise participate in any investigation or proceeding that may be conducted by any Government Agency, including providing documents or other information, without notice to the Company; (g) release claims challenging the validity of this Agreement under the ADEA; (h) release the Releasees or any of them from any claim that by law cannot be waived or released; (i) release any existing rights that Executive may have, if any, to indemnification pursuant to the Employer's or an affiliate's governing documents and/or any directors' and officers' insurance policy of the Employer for acts committed during the course of Executive's employment; or (j) waive any rights of Executive with respect to vested equity. Executive expressly waives and agrees to waive (y) any right to recover monetary damages for personal injuries in any charge, complaint or lawsuit filed by Executive or anyone else on behalf of Executive for any released claims and (z) any right to receive any monetary relief from any action brought by Executive or on Executive's behalf pursuant to the Federal False Claims Act, 31 U.S. Code § 3729, any federal law with a gui tam recovery provision, or any similar state law. This Agreement and General Release does not limit Executive's right to receive an award for information provided to any Government Agencies.
- 7. Executive acknowledges that all Confidential Information (as defined in the Employment Agreement) regarding Employer's or any Company Group Member's (as defined in the Employment Agreement) business compiled, created or obtained by, or furnished to, Executive during the course of or in connection with Executive's employment with Employer or any Company Group member is the exclusive property of Employer or such Company Group member. Upon or before execution of this Agreement and General Release, Executive will return to Employer all originals and copies of any material containing Confidential Information, and Executive further agrees that Executive will not, directly or indirectly, use or disclose such information in violation of the Employment Agreement. Executive will also return to Employer upon or before execution of this Agreement and General Release any other items in Executive's possession, custody or control that are the property of Employer, including, but not limited to, Executive's files, credit cards, identification card, data storage devices, passwords and office keys.

B-2

- 8. Executive acknowledges that (a) Executive has been given at least twenty-one (21)¹ calendar days to consider this Agreement and General Release and that modifications hereof which are mutually agreed upon by the parties hereto, whether material or immaterial, do not restart the twenty-one day period; (b) Executive has seven (7) calendar days from the date Executive executes this Agreement and General Release in which to revoke it; and (c) this Agreement and General Release will not be effective or enforceable nor the amounts set forth in Section 1 paid unless the seven- day revocation period ends without revocation by Executive. Revocation can be made by delivery and receipt of a written notice of revocation to Britt Ide at britt@ideenergy.com by midnight on or before the seventh calendar day after Executive signs this Agreement and General Release.
- 9. Executive acknowledges that Executive has been advised to consult with an attorney of Executive's choice with regard to this Agreement and General Release. Executive hereby acknowledges that Executive understands the significance of this Agreement and General Release, and represents that the terms of this Agreement and General Release are fully understood and voluntarily accepted by Executive.
- Executive agrees that Executive will treat the existence and terms of this Agreement and General Release as confidential and will not discuss this Agreement and General Release, its terms or the circumstances surrounding Executive's separation from service with Employer or its affiliates with anyone other than: (a) Executive's counsel or tax or financial advisor as necessary to secure their professional advice, (b) Executive's spouse or (c) as may be required by law.
- 11. Any non-disclosure provision in this Agreement and General Release does not prohibit or restrict Executive (or Executive's attorney) from responding to any inquiry about this Agreement and General Release or its underlying facts and circumstances by the Securities and Exchange Commission, the Financial Industry Regulatory Authority, any other self-regulatory organizations or Government Agencies, or making other disclosures that are protected under the whistleblower provisions of federal law or regulation. Executive understands and acknowledges that Executive does not need the prior authorization of the Employer to make any such reports or disclosures and that Executive is not required to notify the Employer that Executive has made such reports or disclosures.
- 12. Following the termination of Executive's employment, subject to Sections 6(e) and (f) and 11 and whether or not the Executive is entitled to the Financing Bonus or severance, Executive shall not make any negative statements or communications about any Company Group Member or any of their respective direct or indirect equity holders, directors, managers, officers or employees. Notwithstanding the foregoing, nothing in this Agreement is intended to require any person to make any untruthful statement or to violate any law.
- 13. In the event of any lawsuit against Employer or any of its affiliates that relates to alleged acts or omissions by Executive during Executive's employment with Employer or any affiliate, to the extent permitted by applicable law Executive agrees to cooperate with Employer and its affiliates by voluntarily providing truthful and full information as reasonably necessary for Employer and its affiliates to defend against such lawsuit, provided that the Employer shall reimburse Executive's reasonable expenses incurred in providing such assistance (including, without limitation, attorneys' fees and lodging and meals) subject to Executive's delivery of written notice to the Employer prior to the time such expenses are incurred.

¹ Change to forty-five (45) days in the case of a group termination under the ADEA.



- 14. This Agreement and General Release shall be binding on Employer and Executive and upon their respective heirs, representatives, successors and assigns, and shall run to the benefit of the Releasees and each of them and to their respective heirs, representatives, successors and assigns.
- 15. This Agreement and General Release (and, to the extent explicitly provided herein, the Employment Agreement) set forth the entire agreement between Executive and Employer, and fully supersede any and all prior agreements or understandings between them regarding their subject matter; provided, however, that nothing in this Agreement and General Release is intended to or shall be construed to limit, impair or terminate any obligation of Executive pursuant to any non- competition, non-solicitation, confidentiality or intellectual property agreements that have been signed by Executive where such agreements by their terms continue after Executive's employment with Employer terminates, including, but not limited to, the provisions of Section 6 of the Employment Agreement. This Agreement and General Release may only be modified by written agreement signed by both parties.
- 16. The Employer and Executive agree that in the event any provision of this Agreement and General Release is deemed to be invalid or unenforceable by any court or administrative agency of competent jurisdiction, or in the event that any provision cannot be modified so as to be valid and enforceable, then that provision shall be deemed severed from this Agreement and General Release and the remainder of this Agreement and General Release shall remain in full force and effect.
- 17. This Agreement and General Release will be governed by and construed in accordance with the laws of the State of Delaware without giving effect to any choice or conflict of law provision or rule (whether of the State of Delaware or any other jurisdiction) that would cause the application of the laws of any jurisdiction other than the State of Delaware.
- 18. All judicial proceedings brought against any party arising out of or relating to this Agreement and General Release, or any obligations or liabilities hereunder, shall be brought and maintained in the courts of the State of Delaware or the federal courts located in the State of Delaware. By executing this Agreement and General Release, each party irrevocably: (a) accepts generally and unconditionally the exclusive jurisdiction and venue of such courts; (b) waives, to the fullest extent permitted by applicable law any objection which they may now or hereafter have to the laying of venue of any such dispute brought in such court or any defense of inconvenient forum for the maintenance of such dispute; and (c) agrees that the provisions of this Section 18 relating to jurisdiction and venue shall be binding and enforceable to the fullest extent permissible under applicable law. Notwithstanding the foregoing, the Employer may seek injunctive or equitable relief to enforce the terms of this Agreement and General Release in any court of competent jurisdiction.
- 19. Each of the parties hereto hereby irrevocably waives all right to trial by jury in any action, proceeding or counterclaim arising out of or relating to this Agreement and General Release.
- 20. The language of all parts of this Agreement and General Release in all cases shall be construed as a whole, according to its fair meaning, and not strictly for or against any of the parties.

[Signature Page Follows]

B-4

PLEASE READ CAREFULLY. THIS AGREEMENT AND GENERAL RELEASE INCLUDES A RELEASE OF ALL KNOWN AND UNKNOWN CLAIMS.

"EMPLOYER"

| | NXU, Inc. |
|---|-------------|
| | By: |
| | Name: |
| | Title: |
| | "EXECUTIVE" |
| | Sarah Wyant |
| | Date: |
|] | B-5 |

FORM OF BOARD OF DIRECTORS AGREEMENT

THIS BOARD OF DIRECTORS AGREEMENT ("Agreement") is made and entered into as of October 23, 2024 (the "Effective Date"), by and between NXU, Inc., a Delaware corporation (the "Company") with its principal place of business located at 1828 N. Higley Rd # 116, Mesa, AZ 85205 and [____], an individual ("Director").

1. Term

This Agreement shall continue for a period of up to twelve (12) months from the Effective Date, or earlier in the event of an annual shareholders meeting, and shall continue thereafter for as long as Director is elected as a member of the Board of Directors by the shareholders of the Company on a yearly basis. If re-elected, this Agreement may be renewed for successive one year terms on the same or different terms. If Director is not duly re-appointed by shareholder vote, Director's term will end ten (10) calendar days following the annual shareholders meeting.

2. Position and Responsibilities

- (a) <u>Position</u>. The Board of Directors hereby appoints Director to serve as a Board Member until the 2025 annual shareholders' meeting or Director's earlier resignation, removal or death. Director shall perform such duties and responsibilities as are customarily related to such position in accordance with Company's bylaws and applicable law, including, but not limited to, those services described on <u>Exhibit A</u> attached hereto (the "Services"). Director hereby agrees to use his/her best efforts to provide the Services. Director shall not allow any other person or entity to perform any of the Services for or instead of Director. Director shall comply with the statutes, rules, regulations and orders of any governmental or quasi-governmental authority, which are applicable to the Company and the performance of the Services, and Company's rules, regulations, and practices as they may from time-to-time be adopted or modified.
- (b) Other Activities. Director may be employed by another company, may serve on other boards of directors or advisory boards, and may engage in any other business activity (whether or not pursued for pecuniary advantage), as long as such outside activities do not violate Director's obligations under this Agreement or Director's fiduciary obligations to the Company's shareholders. The ownership of less than a 5% interest in an entity, by itself, shall not constitute a violation of this duty. Director represents that Director has no outstanding agreement or obligation that is in conflict with any of the provisions of this Agreement, and Director agrees to use her best efforts to avoid or minimize any such conflict and agrees not to enter into any agreement or obligation that could create such a conflict without the approval of a majority of the Board of Directors. If, at any time, Director is required to make any disclosure or take any action that may conflict with any of the provisions of this Agreement, Director will promptly notify the Board of such obligation, prior to making such disclosure or taking such action.

(c) <u>No Conflict</u>. Director will not engage in any activity that creates an actual or perceived conflict of interest with Company, regardless of whether such activity is prohibited by Company's conflict of interest guidelines or this Agreement, and Director agrees to notify the Board of Directors before engaging in any activity that could reasonably be assumed to create a potential conflict of interest with Company. Notwithstanding the provisions of Section 2(b) hereof, Director shall not engage in any activity that is in direct competition with the Company or serve in any capacity (including, but not limited to, as an employee, consultant, advisor or director) in any company or entity that competes directly or indirectly with the Company, as reasonably determined by a majority of Company's disinterested board members, without the approval of the Board of Directors.

3. Compensation and Benefits

- (a) Equity Compensation. In consideration of the services rendered under this Agreement, on October 22, 2024, the Company shall grant Director 591,715 restricted stock units ("RSUs") in respect of her service to the Board of Directors from the Company's 2024 Annual Meeting until the Company's next annual shareholders' meeting (the "2024 Annual Grant") under the Nxu, Inc. Amended and Restated 2023 Omnibus Incentive Plan ("Plan"). The 2024 Annual Grant shall vest as follows: (1) 147,928 RSUs are consideration for services rendered for the quarter ending September 30, 2024 and shall be immediately vested (the "Fully Vested RSUs"); (2) 147,929 RSUs shall vest on December 31, 2024, provided Director performs continuous service as a director until such date; (3) 147,929 RSUs shall vest on March 31, 2025, provided Director performs continuous service as a director until such date; (3) 147,929 RSUs shall vest on June 30, 2025, provided Director performs continuous service as a director until such date, and (4) 147,929 RSUs shall vest on June 30, 2025, provided Director performs continuous service as a director until such date. The Fully Vested RSUs shall be delivered within thirty (30) days of the Grant Date. The remaining RSUs shall be delivered within thirty (30) days following the date on which they vest. The Company intends to enter into an Agreement and Plan of Merger, by and among the Company, Verde Bioresins, Inc., NXU Merger Sub, Inc. and NXU Merger Sub, LLC (the "Merger Agreement"). Notwithstanding anything to the contrary in this paragraph, the vesting of the 2024 Annual Grant shall not accelerate, in whole or in part, in connection with the closing of the transactions contemplated by the Merger Agreement, shall not accelerate, in whole or in part, in connection with the closing of the closing of the transactions contemplated by the Merger Agreement shall be forfeited for no consideration at such time. In addition, any vested RSUs that are not settled prior to the Cut-Off Date (as defined below) shall also
- (b) Delayed Equity Compensation for Prior Service. Director also is entitled to RSUs for her prior service as a director of the Company. The Company and Director previously agreed to extend the vesting dates for all of Director's RSUs that would otherwise have vested on and between January 1, 2024 and the Company's 2024 Annual Meeting (the "<u>Delayed Vesting</u>"). The Delayed Vesting for Directors was the result of the Company failing to have sufficient shares available for issuance pursuant to the Plan. Effective as of August 14, 2024, the Company's shareholders approved an amendment and restatement of the 2023 Omnibus Plan to, among other things, authorize additional shares for current and future equity awards. The Company and Director hereby affirm that Director is entitled to \$35,000 in RSUs for each complete calendar quarter between January 1, 2024 and August 14, 2024, totaling \$70,000. The Company and Director hereby desire to set the number of the RSUs due under Director equity compensation award through the August 14, 2024 vesting date. The number of RSUs shall be calculated by dividing \$70,000 by the closing share price August 14, 2024 (\$0.3399) (the "<u>Calculation Date</u>"), resulting in an award of 205,943 RSUs. Following the execution of the Merger Agreement, the Company will deliver the RSUs to Director in installments in such amounts as the Company determines may be delivered to Director without jeopardizing the Company's ability to continue as a going concern. Such RSUs shall be delivered until the earlier of (A) the date all the RSUs subject to Delayed Vesting have been delivered to Director or (B) the date that is no later than 5 business days prior to the closing of the transaction contemplated by the Merger Agreement described above (such earlier date, the "<u>Cut-Off Date</u>"). Any RSUs that have not been delivered to Director as of the Cut-Off Date shall be forfeited for no consideration.

- (c) <u>Stipend.</u> In addition to the 2024 Annual Grant, Director shall be entitled to receive a \$15,000 quarterly cash stipend following the close of each financial quarter beginning with the calendar quarter beginning July 1, 2024 and ending with the calendar quarter beginning April 1, 2025, provided that Director has been in continuous service to the Company as a director through the end of the applicable financial quarter. The stipend payment for the calendar quarter beginning July 1, 2024, shall be paid within fifteen (15) calendar days following the date hereof and stipend payments for subsequent calendar quarters shall be paid within fifteen (15) calendar days of the applicable financial quarter closing date (i.e, December 31, 2024, March 31, 2025, June 30, 2025) until the 2025 annual shareholder meeting. Notwithstanding the foregoing, quarterly stipend payments shall cease upon the closing of the transactions contemplated by the Merger Agreement, whether or not Director continues to service on the Company's Board of Directors.
- (d) <u>Taxes</u>. Director is and will remain an independent contractor. The Company shall not be responsible for withholding taxes on any consideration provided to Director pursuant to this Agreement.
- (e) <u>Expenses</u>. The Company shall reimburse Director for all reasonable business expenses incurred in the performance of the Services in accordance with Company's expense reimbursement guidelines.
- (f) Indemnification. Company will indemnify and defend Director against any liability incurred in the performance of the Services to the fullest extent authorized in Company's Articles of Incorporation, as amended, bylaws, as amended and applicable law. Company agrees to maintain a Director's and Officer's (D&O) insurance policy to further indemnify Director against any liability incurred in the performance of her duties as a director. Company agrees to notify Director of any changes in the D&O policy within 30 days of such change being effectuated. Company agrees to furnish Director with a current copy of each D&O policy.
- (g) <u>Records</u>. So long as Director shall serve as a member of the Company's Board of Directors Director shall have full access to books and records of Company and access to management of the Company.

4. Termination

- (a) <u>Right to Terminate</u>. At any time, Director may be removed as Board Member as provided in Company's Articles of Incorporation, as amended, bylaws, as amended, and applicable law. Director may resign as Board Member or Director as provided in Company's Articles of Incorporation, as amended, bylaws, as amended, and applicable law. Notwithstanding anything to the contrary contained in or arising from this Agreement or any statements, policies, or practices of Company, neither Director nor Company shall be required to provide any advance notice or any reason or cause for termination of Director's status as Board Member, except as provided in Company's Articles of Incorporation, as amended, and applicable law.
- (b) <u>Effect of Termination as Director</u>. Upon Director's termination this Agreement will terminate; Company shall pay to Director all compensation and expenses to which Director is entitled up through the date of termination; and Director shall be entitled to her rights under any other applicable law. Thereafter, all of Company's obligations under this Agreement shall cease.

5. Termination Obligations

- (a) Director agrees that all property, including, without limitation, all equipment, tangible proprietary information, documents, records, notes, contracts, and computer-generated materials provided to or prepared by Director incident to the Services and her membership on the Company's Board of Directors or any committee therefore the sole and exclusive property of the Company and shall be promptly returned to the Company at such time as Director is no longer a member of the Company's Board of Directors.
- (b) Upon termination of this Agreement, Director shall be deemed to have resigned from all offices then held with Company by virtue of her position as Board Member. Director agrees that following any termination of this Agreement, he shall cooperate with Company in the winding up or transferring to other directors of any pending work and shall also cooperate with Company (to the extent allowed by law, and at Company's expense) in the defense of any action brought by any third party against Company that relates to the Services.

6. Nondisclosure Obligations

Director shall maintain in confidence and shall not, directly or indirectly, disclose or use, either during or after the term of this Agreement, any Confidential Information (as defined below) or trade secrets belonging to Company, whether or not it is in written or permanent form, except to the extent necessary to perform the Services, as required by a lawful government order or subpoena, or as authorized in writing by Company. These nondisclosure obligations also apply to Proprietary Information belonging to customers and suppliers of Company, and other third parties, learned by Director as a result of performing the Services. "**Confidential Information**" means all information pertaining in any manner to the business of Company, unless (i) the information is or becomes publicly known through lawful means; (ii) the information was part of Director's general knowledge prior to her relationship with Company; or (iii) the information is disclosed to Director without restriction by a third party who rightfully possesses the information and did not learn of it from Company. Notwithstanding this provision, nothing herein prohibits or restricts Director from communicating with government or regulatory entities, legislative agencies, or self-regulatory organizations without notice to or approval from the Company about possible violations of law.

7. Dispute Resolution

- (a) <u>Mediation.</u> In the event of a dispute between Director and Company, the parties agree to exercise good faith efforts to negotiate a resolution. If the parties are unable to resolve the dispute themselves, they agree to employ a mutually agreeable third party mediator. Parties agree to participate in the mediation in good faith in an attempt to resolve the conflict in advance of initiating litigation.
- (b) <u>Jurisdiction and Venue</u>. The parties agree that any suit, action, or proceeding between Director and Company (and its affiliates, shareholders, directors, officers, employees, members, agents, successors, attorneys, and assigns) relating to this Agreement shall be brought in either the United States District Court for the State of Arizona or in an Arizona state court and that the parties shall submit to the jurisdiction of such court. The parties irrevocably waive, to the fullest extent permitted by law, any objection the party may have to the laying of venue for any such suit, action or proceeding brought in such court. If any one or more provisions of this Section shall for any reason be held invalid or unenforceable, it is the specific intent of the parties that such provisions shall be modified to the minimum extent necessary to make it or its application valid and enforceable.
- (c) <u>Attorneys' Fees</u>. Should any litigation, arbitration or other proceeding be commenced between the parties concerning the rights or obligations of the parties under this Agreement, the party prevailing in such proceeding shall be entitled, in addition to such other relief as may be granted, to a reasonable sum as and for its attorneys' fees in such proceeding. This amount shall be determined by the court in such proceeding or in a separate action brought for that purpose. In addition to any amount received as attorneys' fees, the prevailing party also shall be entitled to receive from the party held to be liable, an amount equal to the attorneys' fees and costs incurred in enforcing any judgment against such party. This Section is severable from the other provisions of this Agreement and survives any judgment and is not deemed merged into any judgment.

8. Entire Agreement

This Agreement constitutes the entire understanding between the parties hereto superseding all prior and contemporaneous agreements or understandings among the parties hereto concerning the Agreement.

9. Amendments; Waivers

This Agreement may be amended, modified, superseded or canceled, and any of the terms, covenants, representations, warranties or conditions hereof may be waived, only by a written instrument executed by the parties or, in the case of a waiver, by the party to be charged. Any amendment or waiver by the Company must be approved by the Company's Board of Directors and executed on behalf of the Company by its Chief Executive Officer. If Director shall also serve as Chief Executive Officer, such amendment or waiver must be executed on behalf of the Company by an officer designed by the Company's Board of Directors.

10. Severability

If any provision of this Agreement shall be held by a court to be invalid, unenforceable, or void, such provision shall be enforced to fullest extent permitted by law, and the remainder of this Agreement shall remain in full force and effect. In the event that the time period or scope of any provision is declared by a court of competent jurisdiction to exceed the maximum time period or scope that such court deems enforceable, then such court shall reduce the time period or scope to the maximum time period or scope permitted by law.

11. Assignment

This Agreement shall not be assignable by either party.

12. Governing Law

This Agreement shall be governed by and construed in accordance with the laws of the State of Arizona and the relevant provisions of the Delaware General Corporation Law.

13. Interpretation

This Agreement shall be construed as a whole, according to its fair meaning, and not in favor of or against any party. Captions are used for reference purposes only and should be ignored in the interpretation of the Agreement.

14. Binding Agreement

Each party represents and warrants to the other that the person(s) signing this Agreement below has authority to bind the party to this Agreement and that this Agreement will legally bind both Company and Director. To the extent that the practices, policies, or procedures of Company, now or in the future, are inconsistent with the terms of this Agreement, the provisions of this Agreement shall control. Any subsequent change in Director's duties or compensation as Board Member will not affect the validity or scope of the remainder of this Agreement.

15. Mutual Non-Disparagement

Director and the Company mutually agree to forbear from making, causing to be made, publishing, ratifying, or endorsing any and all disparaging remarks, derogatory statements or comments made to any party with respect to either of them. Further, the Parties agree to forbear from making any public or non-confidential statement with respect to any claim or complaint against either party without the mutual consent of each party, to be given in advance of any such statement.

16. Director Acknowledgment

Director acknowledges Director has had the opportunity to consult legal counsel concerning this Agreement, that Director has read and understands the Agreement, that Director is fully aware of its legal effect, and that Director has entered into it freely based on her own judgment and not on any representations or promises other than those contained in this Agreement.

17. Cooperation

In the event of any claim or litigation against the Company and/or Director based upon any alleged conduct, acts or omissions of Director during the tenure of Director as an officer of the Company, whether known or unknown, threatened or not as of the time of this writing, the Company will cooperate with Director and provide to Director such information and documents as are necessary and reasonably requested by Director or her counsel, subject to restrictions imposed by federal or state securities laws or court order or injunction. The Company shall cooperate in all respects to ensure that Director has access all available insurance coverage and shall do nothing to damage Director's status as an insured, and shall provide all necessary information for Director to make or tender any claim under applicable coverage.

18. Counterparts

This Agreement may be executed in any number of counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument.

19. Date of Agreement

The parties have duly signed this Agreement as of October 23, 2024 with an effective date of ______

20. Signatures

NXU, INC. a Delaware Corporation

Name Title Mark Hanchett Chairman Name Title

Director

[_____ Individual

EXHIBIT A

DESCRIPTION OF SERVICES

<u>Responsibilities as Director</u>. Director shall have all responsibilities of a Director of the Company imposed by Delaware, Arizona, or other applicable law, the Articles of Incorporation, as amended, and Bylaws, as amended, of Company. These responsibilities shall include, but shall not be limited to, the following:

- 1. <u>Attendance</u>. Use best efforts to attend regularly scheduled and special meetings of Company's Board of Directors, along with Committee Meetings. Company will conduct four (4) in-person quarterly board meetings and Committee Meetings as regularly scheduled, but may call special board meetings as necessary. Special board meetings shall be conducted on an as-needed basis via remote teleconference.
- 2. <u>Executive Sessions</u>. From time to time, and especially during the term of this agreement, the Chairman of the Board may call Executive Sessions or emergency board meetings. Director will be expected to participate in such meetings, as necessary.
- 3. <u>Consult.</u> Meet with the Company upon request to discuss any matter involving the Company or its Subsidiaries, which may involve issues of which Director has knowledge or expertise. Director acknowledges and agrees that the Company may rely upon Director's expertise in any business discipline where Director has knowledge and/or experience that may contribute to the facilitation of Company's operations. Director acknowledges that such requests may involve substantial time and efforts.
- 4. Act as a Fiduciary. Represent the shareholders and the interests of Company as a fiduciary; and
- 5. <u>Participation</u>. Participate as a full voting member of Company's Board of Directors in setting overall objectives, approving plans and programs of operation, formulating general policies, offering advice and counsel, serving on Board Committees, and reviewing management performance.
- 6. <u>Commitment.</u> Based upon the foregoing responsibilities, Director shall be expected to commit an average of at least 10 hours per month to Company business. Director shall not be expected to dedicate more than 20 hours of time to Company business per month.

AMENDMENT TO RESTRICTED STOCK UNIT AWARD AGREEMENT

This Amendment by and between Caryn Nightengale ("Director") and NXU, Inc. (the "Company") is effective as of October 23, 2024 (the "Amendment Date"). Director and the Company are parties to the Nxu Inc. Restricted Stock Unit Award Agreement that granted Director restricted stock units ("RSUs") equal to \$35,000 per quarter beginning with the second calendar quarter of 2023 until the 2024 annual shareholder meeting, subject to the terms and conditions set forth in the Restricted Stock Unit Award Agreement (the "RSU Award Agreement"). By this instrument, the Company and the Director desire to amend the RSU Award Agreement as set forth below.

1. Effective as of the Amendment Date, the fifth and sixth paragraphs of the Introduction to the RSU Award Agreement are hereby deleted.

2. Section 1(1) (<u>Award and Vesting of Restricted Stock Units – Award</u>) of the Restricted Stock Unit Award Agreement is hereby amended and restated in its entirety to read as follows:

Award. On May 12, 2023, Director was granted equity compensation ("Director Equity") of \$35,000 in Restricted Stock 1 Units ("<u>RSUs</u>") each quarter. Of this original Director Equity grant, Director has received RSUs with an aggregate vesting date value of \$105,000, representing the Director Equity that vested on June 30, 2023, September 30, 2023, and December 31, 2023. The Board and the Director previously agreed to extend the vesting dates for all of the Director Equity RSUs that would otherwise have vested on and between January 1, 2024 and August 14, 2024, the date on which Director ceased serving as a Director of the Company (the "Delayed Vesting"). The Delayed Vesting was the result of the Company failing to have sufficient shares available for issuance pursuant to the NXU, Inc. 2023 Incentive Omnibus Plan (the "2023 Omnibus Plan"). Effective as of August 14, 2024, the Company's shareholders approved an amendment and restatement of the 2023 Omnibus Plan (the "Amended 2023 Omnibus Plan") to, among other things, authorize additional shares for current and future equity awards. The Company and Director hereby affirm that Director is entitled to \$35,000 in RSUs for each complete calendar quarter between January 1, 2024 and August 14, 2024, totaling \$70,000. The Company and Director hereby desire to set the number of the RSUs due to Director. The number of RSUs shall be calculated by dividing \$70,000 by the closing share price on August 14, 2024 (\$0.3399) (the "Calculation Date"), resulting in an award of 205,943 RSUs. The Director shall not be entitled to any RSUs for the vesting periods after her termination of service as a Director. The Company intends to enter into an Agreement and Plan of Merger, by and among the Company, Verde Bioresins, Inc., NXU Merger Sub, Inc. and NXU Merger Sub, LLC (the "Merger Agreement"). The Company has determined that the execution of the Merger Agreement will enable it to deliver a portion of the RSUs subject to Delayed Vesting without jeopardizing its ability to continue as a going concern. Following the execution of the Merger Agreement, the Company will deliver the RSUs to the Director in installments in such amounts as the Company determines may be delivered to the Director without jeopardizing the Company's ability to continue as a going concern. Such RSUs shall be delivered until the earlier of (A) the date all the RSUs subject to Delayed Vesting have been delivered to the Director or (B) the date that is no later than 5 business days prior to the closing of the Transaction (such earlier date, the "Cut-Off Date"). Any RSUs that have not been delivered to the Director as of the Cut-Off Date shall be forfeited for no consideration.

DIRECTOR:

/s/ Caryn Nightengale

Date: October 23, 2024

COMPANY: NXU, Inc.

/s/ Mark Hanchett

Name Printed: Mark Hanchett

Title: Chief Executive Officer

Date: October 23, 2024



Press Release

FOR RELEASE ON: October 23, 2024

Nxu and Verde Bioresins Announce Proposed Merger

Combination intended to create significant shareholder value in pursuit of a sustainable future An Economically Feasible and Scalable Alternative Bioresin to Traditional Petroleum-based Plastics

MESA, Ariz., October 23, 2024 – Nxu, Inc., (NASDAQ: NXU) ("Nxu", "the Company"), a domestic technology company focused on energy storage and charging solutions for the infrastructure we need to power our electrified future, and Verde Bioresins, Inc. ("Verde"), a leader in sustainable product innovation and full-service bioplastics production, have entered into a merger agreement pursuant to which Nxu will acquire all of the issued and outstanding common shares of Verde in an all-stock transaction. Upon closing of the merger, the combined company is expected to be renamed "Verde Bioresins, Corp." and list its common stock on Nasdaq under the new ticker symbol "VRDE." The members of the boards of directors of both companies unanimously approved, and will recommend that shareholders of each company approve, the proposed transaction.

Verde pioneered PolyEarthyleneTM, an innovative and proprietary bioresin that has the potential to replace traditional petroleum-based plastics and disrupt the plastics industry. It is an economically feasible alternative that is intended to be dropped into existing plastics manufacturing processes, and Verde is one of the first scalable full-service bio-resin market solutions intended to meet most of the environmental, application, manufacturing, and cost requirements of the industry. Verde's current polyethylene and polypropylene replacement applications are designed to address nearly half of the approximate \$600 billion total addressable global plastics market.

"We are thrilled to merge with Nxu, a company at the forefront of sustainable solutions in transportation," said Brian Gordon, Co-founder and President of Verde. "In addition to selling our biobased, biodegradable and recyclable PolyEarthylene resins to global brands and large plastics converters, we look forward to developing ways to incorporate PolyEarthylene resins into Nxu's EV charging systems with the hopes of creating an even more environmentally friendly offering."

"Nxu has remained steadfast in its pursuit of a sustainable future through clean technology, and a combination with Verde is a creative path to that future that has the potential for long-term value creation for our shareholders." said Nxu Founder, Chairman and CEO Mark Hanchett. "Verde's PolyEarthylene resins are making waves in the plastics market, and I'm incredibly excited about the potential for disruption."

Proposed Transaction Details

Upon the closing of the Merger, assuming Nxu's aggregate enterprise value is approximately \$16.2 million, pre-Merger Verde stockholders will own approximately 95% of the combined company and pre-Merger Nxu stockholders will own approximately 5% of the combined company, in each case, on a fully-diluted and as-converted basis. For purposes of determining the exchange ratio in the transaction, Verde has been ascribed an aggregate enterprise value of approximately \$306.9 million, and Nxu's aggregate enterprise value will be an amount equal to approximately \$16.2 million less an amount equal to the excess of certain lease payments remaining unpaid at closing over Nxu's cash balance at closing.

The transaction is subject to certain closing conditions, including approval by Nxu and Verde shareholders. Upon closing of the Merger, the board of directors of the combined company will consist of seven members, six of whom will be appointed by Verde and one of whom will be appointed by Nxu; and the Verde management team will manage the business of the combined company.

Verde's controlling shareholder Humanitario Capital, LLC has signed a support agreement to vote all of its shares of Verde common stock in favor of the transaction. Similarly, Mark Hanchett Chairman, CEO, & Founder and Annie Pratt, President, holding an aggregate of approximately 19% of the total voting power of Nxu's common stock, have entered into support agreements to vote all of their shares of Nxu common stock in favor of the Merger. Humanitario Capital, LLC has also entered into a Lock-Up Agreement with an initial lock-up period of 180 days, after which the transfer of any shares of Nxu common stock representing more than 5% of the aggregate shares in any calendar month is prohibited for a period of 24 months.

The proposed transaction between Verde and Nxu is the culmination of a formal process initiated by Nxu's Board of Directors to explore a range of possible strategic alternatives for optimizing Nxu's assets and generating sustained shareholder value while still managing potential risks. Since April 2024, the Strategic Planning Committee of the Nxu Board, with the assistance of its advisors and management, evaluated many former potential merger and acquisition candidates as well as new ones, including Verde. Nxu's Strategic Planning Committee made a unanimous, formal recommendation to the Nxu Board to pursue a combination with Verde and to proceed with the proposed transaction.

Britt Ide, co-chair of the Strategic Planning Committee, commented, "Nxu's Strategic Committee and senior management team have conducted thorough due diligence on Verde, and we have been tremendously impressed with their technology, management team, business, and potential for growth." Co-chair Jessica Billingsley added, "Verde has the opportunity to be a major player in the plastics industry, and we believe that by leveraging the public markets they will be able to grow their market share quickly. The proposed merger provides Nxu shareholders with the exciting opportunity to participate in the creation of a more sustainable future via Verde's growth."

Advisors

Lake Street Capital Markets, LLC has provided a fairness opinion to the Nxu Board in connection with the proposed transaction. Snell & Wilmer L.L.P. is serving as legal counsel to Nxu in connection with the proposed transaction.

Roth Capital Partners, LLC is serving as financial advisor to Verde. Wilmer Cutler Pickering Hale and Dorr LLP is serving as legal counsel to Verde in connection with the proposed transaction.

Proxy Solicitor Information and Shareholder Questions

Nxu's shareholders who have questions should contact info@nxuenergy.com

It is anticipated that the Nxu special shareholder meeting will take place in the first quarter of 2025. Nxu's proxy solicitation will be handled by D.F. King & Co.

About Verde

Verde is addressing the Growing Global Plastic Pollution Crisis: 85% of all plastics in the United States end up in a landfill, taking hundreds of years or more to decompose. Through its sustainable product innovation and proprietary technology, Verde provides comprehensive design and development solutions for companies seeking alternatives to conventional plastics. Verde targets food service, consumer goods, skincare and beauty, automotive, agricultural, and other packaging end markets with resins for blown film, thermoform, injection molding and blow molding applications. For additional information, please visit verdebioresins.com

About Nxu, Inc.

Nxu, Inc. is a domestic technology company leveraging its intellectual property and innovations to support e-Mobility and energy storage solutions. For more information, visit www.nxuenergy.com.

Forward-Looking Statements

Certain statements in this press release may constitute "forward-looking statements" within the meaning of the Private Securities Litigation Reform Act of 1995. Forward-looking statements can typically be identified by such words as "aim", "anticipate," "believe," "continue," "could," "estimate," "evolve," "expect," "forecast," "intend," "looking ahead," "may," "opinion," "plan," "possible," "potential," "project," "should," "will," and variations of such words and other similar expressions.

These forward-looking statements are only predictions based on current expectations and assumptions and are inherently subject to risks and uncertainties, some of which cannot be predicted or quantified. There are important factors that could cause actual results to differ materially from the results expressed or implied by forward-looking statements, including the risk factors set forth under "Risk Factors" in Part I, Item 1A of Nxu's Annual Report on Form 10-K for the fiscal year ended December 31, 2023, set forth in Nxu's subsequent Quarterly Reports on Form 10-Q or set forth in Nxu's other filings with the Securities and Exchange Commission ("SEC"), and any of the following: (i) the expected timing to consummate the Merger and/or the transactions contemplated by the Merger Agreement; (ii) the satisfaction (or waiver) of closing conditions to the consummation of the Merger and/or the transactions contemplated by the Merger Agreement, including obtaining stockholder approvals and any required regulatory clearances; (iii) potential litigation relating to the Merger and/or the transactions contemplated by the Merger Agreement that could be instituted against Nxu, Verde or their respective directors; (iv) risks related to Nxu's continued listing on Nasdaq until closing of the Merger; (v) risks related to (a) the combined company's satisfaction of the Nasdaq initial listing application qualifications for the listing on the Nasdaq Capital Market and (b) Nasdaq's approval of the listing of the shares of Nxu common stock to be issued in connection with the Merger; (vi) risks associated with the possible failure to realize certain anticipated benefits of the Merger, including with respect to future financial and operating results; (vii) uncertainties regarding unexpected costs, liabilities or delays in connection with the consummation of the Merger and/or the transactions contemplated by the Merger Agreement, and the impact such costs, liabilities or delays would have on the anticipated cash resources of the combined company upon closing; (viii) operating costs, customer loss and business disruptions arising from the Merger and/or the transactions contemplated by the Merger Agreement and the pendency or consummation thereof (including, without limitation, difficulties in maintaining relationships with employees, customers, clients or suppliers), which may be greater than expected; (ix) the ability of Nxu and Verde to protect their respective intellectual property rights; (x) competitive responses to the proposed Merger and changes in expected or existing competition; (xi) Nxu's, Verde's, and the combined company's limited operating history; (xii) the combined company's ability to manage growth; (xiii) changes in capital requirements' (xiv) the combined company's ability to obtain additional financing; (xv) other adverse economic, business, and/or competitive factors; and (xvi) other risks to the consummation of the Merger and the other transactions contemplated by the Merger Agreement, including circumstances that could give rise to the termination of the Merger Agreement and the risk that the transactions contemplated thereby will not be consummated within the expected time period, without undue delay, cost or expense, or at all.

All forward-looking statements are qualified by, and should be considered in conjunction with, these cautionary statements. Readers are cautioned not to place undue reliance on any forward-looking statements, which speak only as of the date on which such statements are made. Except as required by applicable law, neither Nxu nor Verde undertakes any obligation to update forward-looking statements to reflect events or circumstances arising after such date.

Important Additional Information About the Proposed Transaction and Where to Find It

In connection with the proposed transaction, Nxu intends to file relevant material with the SEC. including a Registration Statement on Form S-4 (as amended, the "Registration Statement"), that will contain a proxy statement of Nxu and that will constitute a prospectus with respect to shares of Nxu's common stock to be issued in connection with the proposed transaction (the "Proxy Statement/Prospectus"). NXU URGES ITS INVESTORS AND STOCKHOLDERS TO READ THE REGISTRATION STATEMENT, PROXY STATEMENT/PROSPECTUS AND ANY OTHER RELEVANT DOCUMENTS THAT MAY BE FILED WITH THE SEC, AS WELL AS ANY AMENDMENTS OR SUPPLEMENTS TO THESE DOCUMENTS, CAREFULLY AND IN THEIR ENTIRETY IF AND WHEN THEY BECOME AVAILABLE BECAUSE THEY WILL CONTAIN IMPORTANT INFORMATION ABOUT NXU, VERDE, THE PROPOSED TRANSACTION AND RELATED MATTERS. Investors and stockholders will be able to obtain free copies of the Registration Statement, Proxy Statement/Prospectus and other documents filed by Nxu with the SEC (when they become available) through the website maintained by the SEC at www.sec.gov. In addition, investors and stockholders should note that Nxu communicates with investors and the public using its website (www.nxuenergy.com) and the investor relations website (investors.nxuenergy.com), where anyone will be able to obtain free copies of the Registration Statement, Proxy Statement/Prospectus and other documents filed by Nxu with the SEC and investors and stockholders should note that Nxu communicates with investors are urged to read the Registration Statement/Prospectus and other documents filed by Nxu with the SEC and investors and stockholders are urged to read the Registration Statement/Prospectus and the other relevant materials when they become available before making any voting or investment decision with respect to the proposed transaction.

No Offer or Solicitation

This press release is not intended to and does not constitute (i) a solicitation of a proxy, consent or approval with respect to any securities or in respect of the proposed transaction or (ii) an offer to sell or the solicitation of an offer to subscribe for or buy or any invitation to purchase or subscribe for any securities pursuant to the proposed transaction or otherwise, nor shall there be any sale, issuance or transfer of securities in any jurisdiction in contravention of applicable law. No offer of securities shall be made except by means of a prospectus meeting the requirements of the Securities Act or an exemption therefrom. Subject to certain exceptions to be approved by the relevant regulators or certain facts to be ascertained, the public offer will not be made directly or indirectly, in or into any jurisdiction where to do so would constitute a violation of the laws of such jurisdiction, or by use of the mails or by any means or instrumentality (including without limitation, facsimile transmission, telephone and the internet) of interstate or foreign commerce, or any facility of a national securities exchange, of any such jurisdiction.

NEITHER THE SEC NOR ANY STATE SECURITIES COMMISSION HAS APPROVED OR DISAPPROVED OF THE SECURITIES OR DETERMINED IF THIS PRESS RELEASE IS TRUTHFUL OR COMPLETE.

Participants in the Solicitation

Nxu, Verde and their respective directors, executive officers and other members of management may be considered participants in the solicitation of proxies in connection with the proposed transactions. Information about Nxu's directors and executive officers, consisting of Mark Hanchett, Annie Pratt, Britt Ide, Jessica Billingsley and Sarah Wyant, including a description of their direct or indirect interest, by security holdings or otherwise, can be found under the captions, "Security Ownership of Certain Beneficial Owners and Management," "Executive Compensation," and "Director Compensation" contained in the definitive proxy statement on Schedule 14A for Nxu's 2024 annual meeting of stockholders, filed with the SEC on May 2, 2024 (the "2024 Nxu Proxy Statement"). To the extent that Nxu's directors and executive officers and their respective affiliates have acquired or disposed of security holdings since the applicable "as of" date disclosed in the 2024 Nxu Proxy Statement, such transactions have been or will be reflected on Statements of Change in Beneficial Ownership on Form 4 filed with the SEC. Additional information regarding the persons who may be deemed participants in the solicitation of proxies, including the information about the directors and executive officers of Verde, will be included in the Proxy Statement/Prospectus and other relevant materials relating to the proposed transaction when it is filed with the SEC. Investors should read the Registration Statement, Proxy Statement/Prospectus and the other relevant materials when they become available before making any voting or investment decision with respect to the proposed transaction. These documents can be obtained free of charge from the sources indicated above.

CONTACT: Nxu, Inc. info@nxuenergy.com